Generating Video Narratives to Support Learning

Abrar Mohammed

Submitted in accordance with the requirements for the degree of Doctorate of Philosophy



School of Computing University of Leeds

2024

Publications

Some of the work in this thesis has been presented prior to thesis submission:

The semantic tagging algorithm designed as part of characterising videos segments to support learning which is part of the VISC-L framework presented in **Chapter 4** and **Chapter 6** is discussed in :

Mohammed, A. and Dimitrova, V., 2020. Characterising video segments to support learning. In Proceedings of the 28th International Conference on Computers in Education.

The implementation of VISC-L framework in the Giving Pitch Presentations domain that has been presented in **Chapter 4** and **Chapter 6** is discussed in:

Mohammed, A. and Dimitrova, V., 2022, September. Video Segmentation and Characterisation to Support Learning. In European Conference on Technology Enhanced Learning (pp. 229-242). Cham: Springer International Publishing.

An overview of the approach has been discussed in :

Mohammed, A., 2022, July. Generating Narratives of Video Segments to Support Learning. In International Conference on Artificial Intelligence in Education (pp. 22-28). Cham: Springer International Publishing.

To my Parents

Acknowledgements

With the grace and will of Allah, I managed to accomplish my PhD research and present this work which I wish to be beneficial and inspire anyone who reads it or uses it.

Throughout my PhD journey, I thought I was setting goals for me to achieve but, in reality, the goals have been set for me by Allah! These goals were represented by how to accept fate (the loss of the meaning of my life- my parents), and deal with hard feelings and translate these feelings to success. I am thankful and grateful for having my Mum and Dad, who I express as angels sent to me from heaven, for bringing me up, teaching me and showing me how to live in the right way. Now, I am proud to say, yes Mum and Dad, I have finished my PhD and it is the beginning of more success to come- this is my promise. Thank you for being so compassionate, patient, supportive and for filling my life with happiness. One day we will meet, in the other life, and I will tell you everything since you left me and we will celebrate this success together as we wished to do in this life. You are in my heart to the last beat and after.

Sincere thanks and gratitude goes to my supervisor, Vania Dimitrova, who believed in me, encouraged and guided me. Your support throughout all the difficulties I have passed and all the challenges I have faced, your words and support were just what I needed. I am lucky to have an amazing and successful person like you in my life. You deserve all the best!

A huge thanks go to my second supervisor David Hogg for his support, feedback and encouragement.

Thank you very much to Professor Tanja Mitrovic, Lydia Lau, the PhD researchers in the University of Leeds, the PhD researchers in the University of Canterbury and to the researchers in the InAdvance project for their support throughout my research. I wish you all the success!

To Karam and Ahmed (my children), I am gifted to have you in my life. Many times you have made me feel so enthusiastic, and my eyes would go full of tears from pride to see you grow up, be so intellectual, successful, and compliant. No words can express my love to you, all my gratitude sent to Allah for his grace to me. Love you!

I believe that real siblings who will stand side by side with you do not only share feelings with you, but also make you feel that you are altogether one soul living in different bodies. For me those are: Mohammed, Maan, Rathwan and Wiqar- my brothers and sister. Thank you very much for being real siblings and may Allah protect you!

I kept my half to the end as we together share this message. To my husband, Khalid, your support, love, interest and encouragement is what I live with. I am glad that you are my half and I do not mind if you take my heart as well- you deserve it.

Abstract

Videos have become popular learning resources which are available through various online platforms, e.g. YouTube, Khan Academy and MOOCS. However, using videos for learning can be challenging, especially when learners are new to the domain. Learners may not know what key points to notice, can lose concentration, and become frustrated. Some of the effective ways to address these challenges are to provide video annotations, summarise the main points, and link different video segments. Existing approaches for annotation and summarisation are either manual and do not scale, or automatic but do not explicitly relate to main domain concepts. There are no approaches which provide automated ways to segment and characterise videos by relating to domain concepts and linking segments from different videos to support learning key domain areas.

The thesis embarks on the challenge of addressing this shortcoming by proposing an automated way to generate video narratives that group segments from different videos to support learning key domain concepts. Our approach is knowledge driven and presumes the domain is computationally represented via a taxonomy. Our first contribution is by developing the Video Segmentation and Characterisation for Learning (VISC-L) framework which uses video transcripts and a domain taxonomy to characterise video segments. Characterising video segments with a domain related topic(s) and concept(s) generated by combining both the Semantic Tagging and Topic Classifier characterisation is found to be an interesting way of linking video content to its domain as recognised by the Technology Enhance Learning community. Our second contribution is the designing of the Videos Narratives for Learning (VIN-L) framework which links different video segments from different videos following a pedagogical theory called 'Subsumption Theory for meaningful learning.' Four types of video narratives, as defined in the Subsumption Theory, are generated–Combinational, Correlative, Derivative, and Super-Ordinate. These narrative types represent how the information is linked in the learner's cognitive structure, for full details about the narrative types, see Section 3.3.3. VIN-L is a novel and generic framework which contributes to the area of using Artificial Intelligence in Education by linking domain related parts from different videos using a pedagogical theory and this framework is new and the idea of this type of narrative has not been done before. Both VISC-L and VIN-L are implemented in two domains for learning soft skills: Giving Pitch Presentations and Health Related Quality of Life Awareness. Soft skill domains, (See Section 3.7), are an important domain but are not well structured and the main learning resources are available as online videos like: tutorials, interviews, stories, etc. The learners of these domains need to find the learning materials among the huge amount of available online videos. Additionally, learners need to be able to identify and remember different domain areas in these videos which could cause learning barriers if the learners are new to the domain and may not able to learn these domains systematically by themselves.

The resultant video narratives are evaluated with domain experts (11 experts for the first domain and 6 experts for the second domain) and advanced learners (in the presentation skills domain)—we could not evaluate VIN-L with learners in the second domain because of the time and budget constraints which are out of the scope of this thesis and will be considered as future work (see Section 7.3). With the first domain (Giving Pitch Presentations), we have evaluated VIN-L with advanced learners because we used workers from Amazon Mechanical Turk to participate in the evaluation study. The number of workers was 217 and they had different knowledge of the domain

(80% had domain related experience) and this is why we refer to them as advanced learners. The Quality and the Perceived Usefulness of the video narratives was assessed. The evaluation outcomes from the evaluation studies are presented as average values in (Sections 5.4 and 5.5–the experts and learners outcomes in the first domain respectively, Section 6.6–the experts outcome in the second domain). For the actual numbers used in these studies, see (Appendices L for the first domain and O for the second domain). All the evaluation studies conducted in this thesis are saved in our GitHub account ¹. The findings of the responses of all video narrative types show that the overall Quality of the narratives is good, as well as indicate some areas for improving the supporting text that clarifies the domain connections. All video narrative types have been proved to have potential to be useful if used to support learning, considering Perceived Usefulness, Learning Effect and Cognitive Work Load evaluation results.

¹https://github.com/Generating-Video-Narratives/Generating-Video-Narratives. git[Accessed: June 2023]

Abbreviations

- **VISC-L** Video Segmentation and Characterisation for Learning
- **VIN-L** Video Narratives for Learning
- CO Combinational Narratives
- **CR** Correlative Narratives
- **D** Derivative Narratives
- **S** Super Ordinate Narratives

Contents

1 Introduction

2	Rel	ated Work	30
	2.1	Introduction	30
	2.2	Background: Video Based Learning (VBL)	32
		2.2.1 Importance of VBL	32
		2.2.2 Challenges faced by teachers and learners	32
		2.2.3 Possible approaches to help learners	33
	2.3	Video Segmentation and Characterisation	37
		2.3.1 Non-learning Contexts	38
		2.3.2 Learning-related Contexts	40
		2.3.3 Video Segmentation and Characterisation in this Thesis	46
	2.4	Video narratives for learning	48
		2.4.1 Narratives using Storytelling Design	48
		2.4.2 Narratives using Hypermedia	50
		2.4.3 Narratives with Concept Maps	51
		2.4.4 Video narratives in this thesis	52
	2.5	Summary	53
3	Res	search Methodology	55
	3.1	Introduction	55
	3.2	Overview of the Research Methodology	56
	3.3	Theoretical underpinning	58
		3.3.1 Text Tiling (Hearst, 1993) \ldots	58
		3.3.2 Thematic Progression (TP) Theory (Danes, 1974)	62

 $\mathbf{22}$

CONTENTS

		3.3.3	Subsumption Theory For Meaningful Learning (Ausubel	
			et al., 1968) \ldots	64
	3.4	Prelim	inaries	66
	3.5	Video	Segmentation and Characterisation for Learning (VISC-L)	
		Frame	work	71
		3.5.1	Input	71
		3.5.2	Initial Segments	73
		3.5.3	Segment Characterisation	76
		3.5.4	Segment Aggregation	80
		3.5.5	Output	83
	3.6	Video	Narratives for Learning (VIN-L) Framework	83
		3.6.1	Input	84
		3.6.2	Subsumption Linking Algorithms	85
		3.6.3	Output	101
	3.7	Domai	ins Used in this Thesis	102
		3.7.1	Giving Pitch Presentations	103
		3.7.2	Health Related Quality of Life Awareness (HRQLA)	106
	3.8	Summ	ary	108
4	Trace	lonoon	tation and Validation of VISC L in the Dresentation	
4	ւար ՏԻ։լ	le Don	tation and valuation of vISC-L in the Presentation	110
		Introd		110
	4.1	Drogon		110
	4.2	Data (Collection	111
	4.5	Data (montotion of VISC I	112
	4.4			110
		4.4.1	Input	110
		4.4.2	Segmentation	117
		4.4.5	Segment Characterisation	117
		4.4.4	Segment Aggregation	120
	4 5	4.4.3		129
	4.0	$0 \text{ ser } \mathbf{E}$	Studie Satur	101
		4.5.1	Degultar Quality of the Segment Characteristics	13Z
		4.3.2	Results: Quality of the Segment Characterisations	139

CONTENTS

		4.5.3	Results: Usefulness of the Segment Characterisations	137
		4.5.4	Discussion	143
	4.6	Summ	nary	144
5	Imp	olemen	ntation and Validation of VIN-L in the Presentation	
	Skil	ls Don	nain 1	46
	5.1	Introd	luction	146
	5.2	Imple	mentation of VIN-L	147
	5.3	Exper	imental Setup	155
	5.4	Findir	ngs from Study-1: Experts	163
		5.4.1	Quality of the Video Narratives	163
		5.4.2	Perceived Usefulness of the generated Video Narratives	166
		5.4.3	Qualitative feedback from experts	167
	5.5	Findir	ngs from Study-2: Potential learners	172
		5.5.1	Quality of the Video Narratives	172
		5.5.2	Perceived Usefulness of the generated video narratives	177
		5.5.3	Qualitative feedback from learners	182
	5.6	Discus	ssion	186
		5.6.1	RQ3: Quality of the video narratives	187
		5.6.2	RQ4: Usefulness of the generated video narratives	189
		5.6.3	Strengths	190
		5.6.4	Areas for video narratives' improvement	190
		5.6.5	Limitations of the studies	191
	5.7	Summ	nary	193
6	Imp	olemen	ntation and Validation of VISC-L and VIN-L in the	
	Hea	lth Do	omain 1	95
	6.1	Introd	luction	195
	6.2	Health	n Domain Taxonomy	197
		6.2.1	Background	197
		6.2.2	Taxonomy Design	199
	6.3	Data	Collection	204
		6.3.1	Patient Stories.	205
		6.3.2	Tutorial Videos	207

CONTENTS

	6.4	Impler	nentation of VISC-L	2	10
		6.4.1	Input	2	10
		6.4.2	Initial Segmentation	2	11
		6.4.3	Segment Characterisation	2	11
		6.4.4	Segment Aggregation	2	17
		6.4.5	Output	2	17
	6.5	Impler	mentation of VIN-L	22	20
		6.5.1	Input	22	21
		6.5.2	Subsumption Linking Algorithms	22	22
		6.5.3	Output	22	26
	6.6	Exper	t Evaluation	22	28
		6.6.1	Study Setup	22	29
		6.6.2	Results: Quality of Video Narratives	23	32
		6.6.3	Results: Perceived Usefulness of the generated Video		
			Narratives	23	36
		6.6.4	Discussion	24	42
	6.7	Summ	ary	24	46
7	Con	clusio	n and Future Work	2 4	17
	7.1	Synop	sis	24	47
	7.2	Contri	butions	24	1 8
	7.3	Limita	ations	2	51
	7.4	Future	e Work	2	52
		7.4.1	Immediately implementable	2	53
		7.4.2	Long term	25	53
R	efere	nces		27	′ 5
A	ppen	dix A	Algorithm 1 tracing (Initial Segments)	27	' 6
A	ppen	dix B	Algorithm 2 tracing (Segment Characterisation)	28	33
A	ppen	dix C	Algorithm 3 tracing (Segment Aggregation)	28	37
A	ppen	dix D	Derivative Algorithm tracing	29)1

Appendix E	Super Ordinate Algorithm tracing	29 4
Appendix F	Correlative Algorithm tracing	297
Appendix G	Combinational Algorithm tracing	300
Appendix H Algorithm	Expert's Evaluation Results of the Semantic Tagging (First Domain)	g 303
Appendix I	Google Forms for VISC-L Evaluation with expert	S 919
(First Dom	lain)	313
Appendix J	Google Form for VIN-L Evaluation (First Domain)	3 81
Appendix K	Amazon Mechanical Turk Study Design	431
Appendix L	Evaluation Results of VIN-L (First Domain)	438
Appendix M Algorithm	Expert's Evaluation Results of the Semantic Tagging (Second Domain)	g 443
Appendix N	Google Form for VIN-L Evaluation (Second Domain)) <mark>465</mark>
Appendix O	Evaluation Results of VIN-L (Second Domain)	534

List of Figures

Overall Research Methodology.	57
(a) Simple Linear (b) Constant Theme (c) Derived Themes (Danes,	
1974)	63
Domain taxonomy. The top classes T represent the main domain	
topics. The hierarchy of a concept (class or individual) represent	
its path to the main topic, e.g. $c_{i1_1} \in C_{i1}$ and the latter is \subseteq of the	
topic T_i (which is one of the top classes in the taxonomy). Hence,	
the path of the concept c_{i1_1} is: $c_{i1_1} \to C_{i1} \to T_i$	67
Video segmentation and characterisation framework. Notice that	
FT means a set of focus topics and FC means a set of focus	
concepts within the focus topics, e.g. T_1C_1 means there is a set of	
focus concepts FC_1 in the focus topic FT_1	72
Video Segment Characterisation Pipeline	77
VIN-L Framework. The taxonomy is a list of topics which are top	
level in the taxonomy structure (black nodes), and a list of concepts	
(classes or individuals) C (white nodes)	83
	Overall Research Methodology

3.7	The domain taxonomy consists of: top concepts T that represent	
	the main domain topics. The hierarchy of a concept (class or	
	an individuals) represent its path to the main topic, e.g. the	
	concept $c_{i1_1} \in C_{i1}$ and the latter is \subseteq of the topic T_i . Hence, the	
	path of the concept c_{i1_1} is: $c_{i1_1} \to C_{i1} \to T_i$. The Subsumption	
	links are illustrated using the domain taxonomy hierarchy. The	
	Combinational linking used to learn concepts related to different	
	topics (T_{i-1}, T_i) . The Correlative linking used to learn concepts	
	related to the same topic (e.g c_{i1_1} and c_{i1_2}). The Derivative linking	
	used to learn the specific concept c_{22_1} by first learning its generic	
	concept C_{22} . The Super-Ordinate linking is used to learn the generic	
	concept C_{12} by start learning its specific concept c_{12_1} or/and c_{12_2} .	87
3.8	The template used for generating the augmented Derivative Video	
	Narrative.	90
3.9	The template used for generating the augmented Super-Ordinate	
	Video Narrative.	94
3.10	The template used for generating the augmented Correlative Video	
	Narrative.	98
3.11	The template used for generating the augmented Combinational	
	Video Narrative.	101
3.12	AVW-Personal Space (Mohammadhassan <i>et al.</i> , 2022)	105
3.13	AVW- Social Space (Mohammadhassan <i>et al.</i> , 2022).	105
4.1	Giving Pitch Presentations main topics based on the PreSOn ontology.	112
4.2	Part of the PreSOn ontology representing the hierarchy of the topic	
	DELIVERY and its concepts.	113
4.3	The collected domain related videos have been filtered using the	
	scoping review approach	114

4.4	Example of aggregating adjacent segments by interpolating the gap	
	segments. For example, we have aggregated some of the segments	
	in one of the collected videos which has the YouTube video ID=	
	Hp7Id3Yb9XQ–this is taken from the segment characterisation	
	which saved the segment Id that is created using the video Id as	
	described in Algorithm 1. The aggregation process is as follows:	
	the segment with duration $[3.06-3.13]$ is with duration < 30 seconds	
	so it is aggregated with the previous segments. The same process	
	is followed to aggregate the segments with duration [3.20-3.27],	
	[4.19-4.28], [4.42-4.48] and $[4.49-4.57] $ with the previous segments.	130
4.5	VISC-L output using the videos with views $\geq 100K$ after restoring	
	the punctuation of their transcript $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	131
4.6	Evaluation results of the Perceived Usefulness of the segment	
	characterisation using the VISC-L and Google characterisations.	137
4.7	Example 1 of characterising a video segment using VISC-L and	
	Google. The segment has been taken from one of the collected	
	domain related videos with ID= dEDcc0aCjaA, and starts at minute	
	2.09 and ends at minute 4.20 from the video footage	138
4.8	Example 2 of characterising a video segment using VISC-L and	
	Google. The segment has been taken from one of the collected	
	domain related videos with ID= ADJAcyTq1us, and starts at	
	minute 16.55 and ends at minute 17.55 from the video footage	139
4.9	Example 3 of characterising a video segment using VISC-L and	
	Google. The segment has been taken from one of the collected	
	domain related videos with ID= Hp7Id3Yb9XQ, and starts at	
	minute 2.45 and ends at minute 5.04 from the video footage	139
4.10	Mental Load estimation of the segment characterisation using	

VISC-L and Google. The score range is 1-Low to 20-High. \hdots 142

LIST OF FIGURES

5.1	The output after applying the four Subsumption Linking
	Algorithms. Four types of video narratives have been generated:
	Combinational, Correlative, Derivative, Super Ordinate. The
	number of narratives per type with their duration Average and
	Standard Deviation (STDV) is reported
5.2	Example of a video narrative augmented (Section 3.4–Definition
	12) using the template we designed in (Section 3.6.2)
5.3	Example of the Derivative Narrative to support learning showing
	that to learn the specific concept 'GRAPH', its generic (parent)
	concept 'FIGURE' must be introduced first before introducing the
	specific concept. Both concepts belong to the topic 'VISUAL AIDS'. 151
5.4	Example of the Super Ordinate Narrative to support learning
	showing that to learn the generic concept 'POSTURE', its specific
	concept 'SHOULDER' must be introduced first before introducing
	the generic concept. Both concepts belong to the topic 'DELIVERY'.152
5.5	Example of the Correlative Narrative to support learning showing
	that to learn the specific concept 'CONNECTION', its similar
	concepts 'PAUSE' and 'FOCUS' must be introduced first before
	introducing the specific concept. All concepts belong to the topic
	'DELIVERY'
5.6	Example of the Combinational Narrative to support learning the
	topics 'STRUCTURE' and 'PRESENTATION ATTRIBUTE' by
	introducing video segments that focus on different concepts belong
	to these topics. $\dots \dots \dots$
5.7	Implementation of VIN-L in the Giving Pitch Presentations domain. 155
5.8	Experts' profiles
5.9	Learners' profiles which include: Age group, Gender, English native
	speakers and Job sector
5.10	Learners' profile including data about: how often they watch
	YouTube videos, how often they watch YouTube for learning, the
	type of presentations they have given, the amount and type of
	training they had about presentation skills

LIST OF FIGURES

5.11	Quality of the generated narratives of the Giving Pitch	
	Presentations Domain as evaluated by experts. We have used the	
	quality average value of ≥ 4 as filtering criteria to select from the	
	video narratives which their quality match this criteria to be used	
	for the evaluation study with learners. \ldots \ldots \ldots \ldots \ldots	162
5.12	Quality of the generated narratives of the Giving Pitch Presentation	
	Domain as evaluated by the experts	165
5.13	Perceived Usefulness of the generated narratives of the Giving Pitch	
	Presentations domain evaluated by experts	169
5.14	Quality of the generated narratives of the Giving Pitch	
	Presentations domain as evaluated by learners. $\hfill\hfil$	175
5.15	Perceived Usefulness of the generated narratives of Giving Pitch	
	Presentation Domain as evaluated by the learners. \ldots	179
5.16	Cognitive Work Load estimation of the generated video narratives	
	to support learning the Giving Pitch Presentations domain based	
	on the evaluation study with the learners. The score range is 1-Low	
	to 20-High	182
6.1	Iterative Taxonomy Development Methodology inspired by the	
	ontology building steps (Fernández-López, 1999; Noy et al., 2001).	199
6.2	Main Topics of the in the Health Related Quality of Life Awareness	
	taxonomy	203
6.3	The concepts that belong to the topics ENVIRONMENT, LEVEL	
	Of INDEPENDENCE, PERSONAL VALUES AND BELIEFS and	
	SOCIAL RELATIONSHIP in the Health Related Quality of Life	
	Awareness taxonomy.	203
6.4	The concepts that belong to the topics PHYSICAL HEALTH and	
	PSYCHOLOGICAL HEALTH in the Health Related Quality of	
	Life Awareness taxonomy.	204
6.5	Filtering the collected domain related patient story videos using	
	the scoping review approach	206
6.6	Filtering the collected domain related tutorial videos using the	
	•	

6.7	Example of aggregating adjacent segments with interpolating the	
	gap segment.	218
6.8	Example 1 of a segment characterisation by VISC-L. The segment	
	has been taken from one of the patient story videos with YouTube	
	video $ID = hQQ5xQGZxdA$ -this is taken from the segment	
	characterisation which saved the segment Id that is created using	
	the video Id as described in Algorithm 1. It starts at $04:41$ with	
	duration of 128 seconds. \ldots	219
6.9	Example 2 of a segment characterisation by VISC-L. The	
	segment has been taken from one of the patient story videos with	
	YouTube video ID= wwiwjZw7d6o-this is taken from the segment	
	characterisation which saved the segment Id that is created using	
	the video Id as described in Algorithm 1. It starts at $01:14$ with	
	duration of 118 seconds. \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots	220
6.10	Example 3 of a segment characterisation by VISC-L. The segment	
	has been taken from one of the patient story videos with YouTube	
	video ID= GV3Rew061FA-this is taken from the segment	
	characterisation which saved the segment Id that is created using	
	the video Id as described in Algorithm 1. It starts at $01:31$ with	
	a duration of 216 seconds	221
6.11	Example of the Derivative Narrative to support learning showing	
	that to learn the concept 'EXERCISE', its generic (parent) concept	
	'REHABILITATION' must be introduced first before introducing	
	the specific concept. All of the concepts belong to the topic	
	'ENVIRONMENT'	223
6.12	Example of the Super Ordinate Narrative to support learning	
	showing that to learn the generic (parent) concept 'ENERGY',	
	its specific concept 'STRENGTH' must be introduced first before	
	introducing the parent concept. All of the concepts belong to the	
	topic 'PHYSICAL HEALTH'	224

6.13 An example of one of the generated Correlative Narratives to	
support learning shows that to learn the concept 'BELIEF', a similar	
concept to it must be introduced first which is 'MEDITATION' and	
both concepts belong to the topic 'PERSONAL VALUES AND	
BELIEFS'	25
6.14 An example of a Combinational Narrative that supports the learning	
the topics 'SOCIAL RELATIONSHIP' and 'PSYCHOLOGICAL	
HEALTH' by introducing video segments that focus on different	
concepts belonging to these topics. $\ldots \ldots \ldots \ldots \ldots \ldots \ldots 22$	27
6.15 The output after implementing VIN-L following the research	
methodology. \ldots 22	28
6.16 Evaluation results of the quality of the video narratives 23 $$	35
6.17 Evaluation results of the Perceived Usefulness of the video narratives. 23	39
6.18 Cognitive workload estimation of the generated video narratives.	
The scores range is from 1-Low to 20-High. \ldots \ldots \ldots 24	42

Chapter 1

Introduction

Videos for learning have showed their effect to support teachers when embedded in lessons. In recent years, they have been seen as the main method for learning for both teachers and learners. Videos have been widely used in various learning settings to facilitate independent learning and are becoming a key platform for digital learning (Hsin & Cigas, 2013; June *et al.*, 2014). They provide learners with audio and visual learning channels that can meet vast learning needs (Haridakis & Hanson, 2009; Khan, 2017). Teachers, when using videos for learning, have a responsibility to select the most suitable videos (Lange & Costley, 2020; Yacobson & Alexandron, 2023), provide suitable feedback (Cohen & Cohen, 2023; Majumdar *et al.*, 2023; Villagrán *et al.*, 2022) and supply the learning platform with digital tools to help learners meet their learning objectives. Learners, who are familiar with Video Based Learning (VBL), are able to develop self-regulated skills, learn at their own pace, set goals and plan their learning independently (Tammets *et al.*, 2022).

However, there are major challenges that affect user engagement with videos. Learners prefer watching short videos as their concentration span is reduced over time (Meseguer-Martinez *et al.*, 2017; Risko *et al.*, 2012). Video content complexity could affect a user's engagement with the video and hence cause confusion or boredom (Mongkhonvanit *et al.*, 2019). Consequently, learners may have to watch videos many times and may not be able to identify the most relevant key points. Furthermore, there are learners who are either new to the domain or have a lack of prior knowledge. In such cases, they do not know the key points

to look out for in the videos which causes frustration and reduces engagement. As a result, they will not be able to meet their learning goal(s). This calls for finding new ways to identify the main points in a video and to direct learners to the corresponding parts in the video that elaborate on specific key points. Consequently, a teacher's support is required by providing a suitable technique(s) (Esterl *et al.*, 2022; Richtberg & Girwidz, 2019; Rohloff & Meinel, 2018); feedback, quizzes, setting goals, annotation tools, summarisation, and providing a forum for group discussions.

This research proposes a novel automatic approach that supports video annotation and summarisation. Summarising videos for learning attracts different researchers to enhance the use of videos for learning. This includes segmenting and characterising videos to help in identifying key domain areas to direct a student's focus. Teachers can manually identify relevant areas in the videos where the domain key points are elaborated and highlight these areas with key words (Castro & Tumibay, 2021; Tseng, 2021). Meanwhile, learners can segment and characterise the videos via note-taking (Mirriahi *et al.*, 2021). However, the aforementioned challenges are experienced at a large scale with the increase of both the amount of video footage available and the number of learners who use videos for learning. Therefore, manual solutions would not scale up: using experts to analyse the video segments and identify how they will be used by potential learners can be costly and ineffective. Hence, there is a need for computational means to automate the characterisation of video segments in order to identify what concepts are covered and whether learners will grasp these concepts or not.

Automatic approaches have tried to address the aforementioned challenges and scale up by utilising different AI tools and the support of knowledge models (e.g, DPBedia, knowledge graphs, etc.). This included: using the learners' interaction with the videos to highlight high attention areas (Bywater *et al.*, 2021; Mitrovic *et al.*, 2016), using visual and textual video content to characterise video segments (Liu & Lauw, 2022; Schlotterbeck *et al.*, 2021), segmenting the videos based on the change in the topic recognised in the video's textual data with the aid of Wikipedia pages (Das & Das, 2019; Sikos, 2017). Other researchers attempted to enrich VBL through a storytelling style (Bianchi & Ragonese, 2023) to arrange learning resources and support for learners (Aqqal *et al.*, 2007; Lee *et al.*, 2021;

McQuiggan *et al.*, 2008). Likewise, other researchers found that concept maps help in presenting videos and learning resources as narratives (Hao *et al.*, 2021; Pinandito *et al.*, 2021; Tang *et al.*, 2021).

Most of the existing approaches have notable limitations; they rely on visual content when segmenting and characterising videos (Liu & Lauw, 2022; Schlotterbeck *et al.*, 2021), they use general knowledge models when characterising the videos (Das & Das, 2019; Sikos, 2017) where domain specific concepts within the main topics are missed, they link the whole videos together but without specifying or locating the key domain areas within the videos, and they do not link different video segments to other related segments from other videos (Hao *et al.*, 2021; Pinandito *et al.*, 2021; Tang *et al.*, 2021).

The aim of this thesis is to support video based learning by designing, implementing and evaluating the generic and novel frameworks we designed for Video Segmentation and Characterisation for Learning (VISC-L), and Video Narratives for Learning (VIN-L). VISC-L helps in segmenting and characterising videos by identifying the domain related knowledge presented in the video transcript. The **segmentation** of a video transcript is to divide it into smaller segments. This aims to reduce watching time and enhance concentration as cognitive load and attention span play a significant role in learning. The characterisation (summarisation) using the video transcript is applied in this thesis by combining the Semantic Tagging and Topic Classifier characterisations. The Topic Classifier helps in generating the contextual characterisation of the video segments' transcript and the Semantic Tagging characterisation identifies the domain-terms' full path to their main domain topics using the taxonomy hierarchy as explained in detail in (Section 3.5). Using the video segment characterisation with the domain taxonomy helps in implementing VIN-L to generate video narratives by linking different segments from different videos which have the same Focus Topic and Concept under the guidance of the Subsumption Theory. VIN-L aims to provide learners with a cohesive learning experience. The Subsumption Theory helps in identifying the type of learning while trying different linking types, e.g., the learning type from a generic domain concept to a specific domain concept can be followed by implementing the Derivative linking from the Subsumption Theory. The generic and the specific

concepts are identified by the characterisation and the relationship between them is presented in the hierarchy of the domain taxonomy, (see Section 3.6). Generating video narratives by linking video segments helps to minimise watching time, keep learners engaged and helps them remember important parts and retain the critical information from videos.

We have two main assumptions in our work. The first assumption is that the domains tackled should be represented by videos which are associated with their transcript. This is crucial for the domains we are tackling- soft-skills domainswhere people share their experiences using videos or podcasts. Although the visual content can give useful visual cues (demonstrations, hints, etc), it does not necessarily represent the domain and will narrow our approach to specific types of videos (e.g., lecture videos with slides). Hence, the video transcript contains the useful domain related information for learners to use and improve their self-regulated skills. Most of the videos freely available online are designed and associated with a transcript, e.g., the videos available on YouTube. Using video transcripts enables us to handle a wide range of video types. The second assumption is that the concepts of the domain are represented via a taxonomy or ontology. To generate video narratives from video segments, we have adopted the Ausubel Subsumption Theory of Meaningful Learning (Ausubel et al., 1968) where four types of subsumption links were identified to link learning materials. We implemented these links using the characterisation of the video segments and the domain taxonomy to generate four types of video narratives which have been presented to learners as learning materials (see Chapters 5 and 6).

Our hypothesis is as follows: using video narratives designed based on a pedagogical theory utilising video segment characterisation and domain taxonomy will help in providing useful learning materials. Learners, who will use these learning materials, will be able to focus on key domain areas, recall and recognise the link between them. For that purpose, segmenting and characterising videos is needed to identify key domain parts and should be evaluated by users to assess their benefit to help learners in identifying such parts. Then, domain related key parts (characterised video segments) from different videos should be linked to generate the video narratives. These narratives should be evaluated by experts

and learners to assess their benefits of guiding learning while watching videos and help in focusing on key domain areas.

Following this hypothesis, we address the following research questions where the quality of the generated video segment characterisation (RQ1 and RQ2) is crucial to help in generating good quality video narratives to support domain learning (RQ3 and RQ4):

- **RQ1:** How to automatically generate good quality characterisations of video segments that can be used for learning?
- **RQ2:** Does the characterisation of video segments help in identifying and focusing on the key domain concepts?
- **RQ3:** How to generate good quality narratives that can be used to support perceiving domain learning?
- **RQ4:** Are the generated narratives useful to support perceiving domain learning?

We present two novel domain taxonomy-underpinned frameworks, VISC-L and VIN-L, to address our research questions.

To address **RQ1**, the first framework, Video Segmentation and Characterisation for Learning (VISC-L) uses a domain taxonomy and video transcripts to segment, characterise and link videos to the domain knowledge covered in the segments. This framework is presented and described in (Chapter 3) and is implemented in (Chapters 4 and 6) in two different domains.

To address **RQ2**, our work has a key difference from previous approaches. While none of them assesses the effect on learning, we provide an evaluation study with learners to examine the Quality, Perceived Usefulness and Cognitive Work Load of the segmentation and characterisation in a learning context (Chapter 4). We also compare the VISC-L characterisation with a state-of-the-art video segmentation and characterisation interface that is available for YouTube videos which is Offered by Google and produced by Google Video AI¹.

To address **RQ3**, we designed a second framework, Video Narratives for Learning (VIN-L), which represents the main contribution of this thesis. VIN-L generates video narratives using different characterised video segments that have a domain topic(s) and concept(s) in common, and link them following the Ausubel Subsumption Theory. We generated four types of video narratives that are used to learn a domain from different perspectives based on each narrative's definition. The video narratives we have generated are augmented with text representing the segment characterisation and the type of linking used. The Quality of the generated video narratives has been assessed via an evaluation study with domain experts whose domain experience varies based on the domain, e.g. teaching, delivering (as presented in Section 5.3), practicing, or researching (as presented in Section 6.6).

To address **RQ4**, VIN-L has been implemented in two different domains and the generated narratives have been evaluated with domain experts (Chapter 5 and Chapter 6). The generated video narratives of the first domain (Giving Pitch Presentations) have also been evaluated with advanced learners from Amazon Mechanical Turk in which it is proven that the video narratives have potential to support domain learning (presented in Section 5.5).

This thesis contributes to the following areas: (a) segmenting and characterising videos for learning, (b) using narratives for learning and (c) video based learning environments. A number of original contributions have been made to the following research communities:

• Video segmentation and characterisation-this thesis presents automatic methods by designing algorithms to: (a) segment video transcripts based on the Text Tiling approach, and (b) characterise video segments using a semantic tagging algorithm and the BERT model. Then, adjacent segments

¹https://cloud.google.com/video-intelligence/docs/analyze-labels[Accessed:June 2022]

are aggregated based on the Thematic Progression Theory as described in (Chapter 4 and 6). In-case the domain taxonomy is not available, we can create this taxonomy either manually, which makes our method semi-automatic (Chapter 6), or we create the taxonomy automatically to have a fully automatic way of characterising video segments. The automatic method for taxonomy creation is considered as a limitation in this thesis (see Section 7.3).

- Video narratives for learning- generated by developing a novel and generic framework to link different segments from different videos following the Ausubel Subsumption Theory of meaningful learning. We have designed four algorithms for this purpose as described in (Chapter 5 and 6).
- Video based learning environments-where: (a) we have presented a semi-automatic way to search and collect domain related videos from online video platforms (e.g, YouTube) as described in (Section 3.5.1 and implemented in Sections 4.3 and 6.3.1). This includes designing a search schema that is based on a domain taxonomy and implementing it using Python libraries (the full code is available on our GitHub account), (b) we have validated the frameworks (VISC-L and VIN-L) in two different domains with experts and validated the pedagogical value of the frameworks within one domain (Giving Pitch Presentations). The average value of the responses from the expert and learner studies showed that the Quality and the Usefulness of the generated narratives is good and the video narratives have a potential to support learning when used by domain learners. However, a future study is needed to assess the usefulness of using the video narratives with beginner learners as described in (Sections 5.6.5 and 7.4.2).

Thesis structure. The thesis is structured into seven chapters. In this chapter we have introduced the context of this thesis, the research questions and the thesis' contribution.

Chapter 2 provides related work and background research where the context of this thesis is defined and the key issues the research aims to address are highlighted .

Chapter 3 describes and discusses the overall research methodology to address the research questions detailed in Chapter 1. The algorithms and the theories used to design them are described and the domains where these algorithms have been implemented are also identified.

Chapter 4 describes the implementation of the first framework, VISC-L, in our first domain (Giving Pitch Presentations). The evaluation with domain experts to assess the Quality and Usefulness of the VISC-L results compared to the video characterised segments generated by Google is also presented in this chapter.

Chapter 5 describes the implementation of VIN-L in our first domain (Giving Pitch Presentation). The evaluation of the generated video narratives with experts and advanced domain learners is presented.

Chapter 6 describes the implementation of VISC-L and VIN-L in the second domain (Health related Quality of Life Awareness). The evaluation of the frameworks has been conducted exclusively by domain experts.

Finally, Chapter 7 summarises the main aspects of our research, discusses the main contributions, and highlights the future directions of this research.

Our full work in this thesis including code, taxonomies and experimental data can be accessed using the link to our GitHub account ¹.

¹https://github.com/Generating-Video-Narratives/Generating-Video-Narratives. git[Accessed: April 2023]

Chapter 2

Related Work

2.1 Introduction

Video Based Learning (VBL) has become widely spread in education as it is used to enrich the learning experience, promote engagement, and empower both learners and educators. However, there are still challenges associated with using VBL which are represented by: selecting suitable videos from the vast amount of available online learning platforms (Haridakis & Hanson, 2009; Khan, 2017), locating the key points in these videos and remembering them (Bywater *et al.*, 2021; Schlotterbeck *et al.*, 2021), and not all of the learners know the key domain points to identify in the videos. These challenges cause frustration and introduce a lack of interactivity with the videos among learners. As a result, educators need to tackle these issues which is a costly process as it requires time and effort.

This motivates studies to use manual and automatic approaches (Dodson *et al.*, 2019; Dutta & Zisserman, 2019) to identify key parts in the videos by segmenting and characterising them. The manual approaches did not scale, and were time consuming and subjective (Chiu *et al.*, 2018; Colasante & Douglas, 2016). The automatic approaches succeed to scale and be objective by relying on the visual content for segmenting and characterising the videos (Bywater *et al.*, 2021; Liu & Lauw, 2022). Additionally, these approaches try to enrich the characterisation of the video segments using general knowledge models (Das & Das, 2019; Sikos, 2017). However, this limits the use of these automatic approaches to tutorial videos, which are designed using slides, and will not be applicable to other type of

videos, e.g., interviews, podcasts, patient stories, etc. Moreover, the use of general knowledge models will not capture domain specific topics and concepts.

On the other hand, many researchers advocate using Artificial Intelligence (AI) tools to enhance video based learning by: helping to set goals (McQuiggan *et al.*, 2008), linking to additional resources (Recke & Perna, 2021), including quizzes, and improving the interaction by generating nudges utilising learner comments or by adding a tutor agent by asking knowledge-building questions (Serrano-Iglesias *et al.*, 2021).

Along this line is the research on facilitating video based learning by supporting the linking of learning materials by exploring the use of concept maps (Pinandito *et al.*, 2021; Shao *et al.*, 2020; Spicer *et al.*, 2020; Tang *et al.*, 2021). However, these approaches are computationally expensive and are usually domain-dependent. Crucially, while the visual representation of complex knowledge structures can be helpful for navigating through videos, it may be overwhelming for learners not familiar with the domain.

The aim of the work in this thesis is to help learners interact with the videos by developing an automatic way to link different video segments from different videos together following a pedagogical theory. Using this kind of linkage between video segments will enable learners to identify key areas in the domain and the relationship between them.

In this chapter, we position our work in the relevant literature which allows us to contextualise our research, identify the key challenges and show the difference of our research in comparison to other researchers as presented in (Section 2.2). In (Section 2.3), we review the approaches that focus on segmenting and characterising videos to support learning as they help in understanding the main videos' content which is related to the first part of our approach-video segmentation and characterisation. Additionally, we describe how our segmentation and characterisation is different from the aforementioned approaches in (Section 2.3.3). To be able to pedagogically link the key domain areas presented in different domain related videos to generate video narratives, in (Section 2.4), we show some of the approaches used by other researchers to adopt the idea of narratives in learning and highlight how our approach is different. In (Section 2.5), we summarise the researches we reviewed, the identified gaps and identify our own way to tackle these gaps.

2.2 Background: Video Based Learning (VBL)

In this section, we review the importance of video based learning and highlight the challenges that teachers and learners face while using this learning method.

2.2.1 Importance of VBL

The use of video based learning has increased rapidly. It offers the flexibility of having visual and auditory channels that make it easier for learners to gather information and to support their learning (Haridakis & Hanson, 2009; Khan, 2017). This is noticed when instructional videos have been used successfully in flipped learning by enhancing the learning achievement of students and helping in supporting self-regulated learning (Kim *et al.*, 2023; van Alten *et al.*, 2020). Additionally, students highly ranked e-lectures in terms of their technology acceptance (Sprenger & Schwaninger, 2021) as they allow them to learn at their own pace and revise for their tests accordingly. Furthermore, videos can be used before lessons for preparation (Förster *et al.*, 2022) to help them strengthen their knowledge retention and recognise any knowledge gaps. The design and content of the videos affects their impact on learning as students engage more when watching short animated videos as opposed to using videos with slides (Köhler *et al.*, 2023). Consequently, video based learning is not free of challenges for both teachers and students which, in turn, will affect the learning process if not resolved.

2.2.2 Challenges faced by teachers and learners

There are some challenges faced by teachers when adopting videos in their teaching which require support. An example of these challenges is that some teachers (Mhamdi, 2017) are affected with the profession's workload and struggle to find time to search for appropriate videos and define suitable video parts to be used by their students, and thus, struggle to find suitable videos for learning. Some teachers rely on using specific criteria while searching for educational videos: a

curriculum, learning objectives, the reliability of the resources which is decided by recommendations from peers, or the credentials of the content designer (Yacobson & Alexandron, 2023). Hence, it is beneficial for teachers to know: video designers, downloaders and the associated video comments. Finding the videos is not the only challenge faced by teachers; providing suitable usage-support techniques to the videos to help their learners is another. Some of these techniques are: reducing watching time, enhancing video quality and intelligibility (Lange & Costley, 2020). To help learners while watching videos, teachers provide them with tools to segment the videos into short clips and to decide how many segments to watch (too many segments may slow down the understanding of the information). Video quality can be improved when the media diversity is used properly, e.g, when the visual content complements the audio content. Meanwhile, video intelligibility can be improved by adding media controls for learners e.g, volume and brightness controls.

Nevertheless, these solutions are not applicable for many teachers as some teachers have a lack of professional training in digital technology; consequently, training workshops are required to equip the teachers with the necessary digital knowledge (Michel & Pierrot, 2022).

The aforementioned challenges faced by teachers also affect learners who face other challenges while using VBL; watching videos takes up time and, for many learners with a lack of domain knowledge, it can be hard to identify key points in the videos and link these points to the study domain (Bywater *et al.*, 2021; Schlotterbeck *et al.*, 2021). Consequently, support is needed to help learners interact with VBL to improve their performance. The aim of this thesis is to support learners while using VBL, and accordingly, we review the possible approaches to help them.

2.2.3 Possible approaches to help learners

Learners' self regulation skills and engagement within VBL can be enhanced via: responding to the teacher's support provided to them and by following suitable learning strategies. **Teacher support to learners** while using VBL is varied and can be classified as providing: (a) feedback and (b) educational resources which can be used to serve the learning objectives.

(a) Using feedback can take different forms, e.g., Villagrán *et al.* (2022) used multimodel feedback which is based on pedagogical goals and not criticism to help learners improve their work. The type of feedback should match the types of interaction tools and resources available to learners. For instance, the learners in Villalobos et al. (2022) interact differently with a learning system that includes: topics addressed to be learnt every week, quizzes and assignments, practical exercises, reading materials, practical and theoretical solutions, watching videos, and system visualisation. Accordingly, teachers can personalise their feedback to learners based on the difficulty level of the task (Cohen & Cohen, 2023; Majumdar et al., 2023). Verbal feedback, using dialogic interactions with learners is found to be beneficial especially for learners with limited domain knowledge or learners with special needs (Mirzababaei et al., 2023). Automatic feedback can be used to improve the learners' performance and help them correct their mistakes quickly (Gabbay & Cohen, 2022). The generation of automatic feedback can be varied, e.g. Abu-Rasheed et al. (2023) includes a chat-bot within an educational web application where teachers design exercises for their learners following a pedagogical informed template. The learners who use this application and interact with the exercises will be provided with a chat-bot that provides predefined teacher feedback. Varying feedback based on each learner's performance is more beneficial than using the same feedback for all learners (no one type fits all). Enhancing online teaching with pedagogical strategies is essential for successful online learning as stated by Archambault et al. (2022) who identified the five pedagogical pillars for this purpose "the ability to: Build Relationships and Community, Incorporate Active Learning, Leverage Learner Agency, Embrace Mastery Learning, and Personalize the Learning Process." This idea has been adopted by Dever *et al.* (2022), where a pedagogical agent has been used with ITS tutoring systems through dialogue and feedback to learners which helped them be more self-regulated. Similarly, in Ausin *et al.* (2021) the ITS has been supported with a neural network model to help in predicting the type of feedback

to be presented to the learners which is either a problem to be solved or an expert step-by-step solution to a problem. Experts' feedback, designed following a pedagogical policy, shows a positive impact on the learners' performance.

In addition to feedback, learners use other supporting tools differently, e.g, learners who have been classified as high attention learners were found to use play/pause buttons often while watching videos (Rabin *et al.*, 2019). This indicates that their focus is on knowledge acquisition. Differently, learners with a low attention classification, tried quizzes and final exams more than watching educational videos which affected their knowledge gain. Hence, personalised feedback and learning strategies are required to direct learners to their best learning behaviour. Teachers also use different types of quizzes which are found to not only familiarise learners with the technology, but also to engage them more with the learning materials (Tammets *et al.*, 2022).

(b) Educational resources can be added to learning environments to help learners. For example, Esterl *et al.* (2022) shows that using websites with hyper text, playful programming platforms, and forum discussions helped online learners during the pandemic. Similarly, multimodel learning systems use different data about learners (Lee *et al.* (2023) uses visual and audio data, physical movement, and Lee *et al.* (2023) uses taste and smell sensors) to assess the learners' interactivity and guide the designing of customised feedback. Learners' engagement can be detected using computer vision techniques where their facial expressions are recognised and mapped to a lecture video timeline (Linson *et al.*, 2022). This helps teachers identify the times in the lecture where learners need support. Furthermore, the learners' noise, silence and speech times can be detected using speech recognition via supervised machine learning models (Barzola *et al.*, 2022). Based on these signals, automatic feedback can be provided to learners within a virtual learning environment. Although speech analysis can lead to more accurate results, it can raise data privacy concerns.

Learners follow learning strategies. Adding to the teacher support within VBL, learners, from their perspective, can use different learning strategies. For example, learners who use MOOC learning platforms are able to improve

their self-regulated skills via some strategies, e.g. goal setting, learning planning, self-learning evaluation, and summarisation (Richtberg & Girwidz, 2019; Rohloff & Meinel, 2018). Additionally, when the learner's behaviour while watching videos has been analysed, it was found that active learners engage more through: social interaction (discussion forums and messenger apps), information seeking (annotation tools), and environmental configuration (buttons, progress bars, pausing, moving forwards and backwards) (Yoon *et al.*, 2021). An interaction with VBL can be more informative if learning pathways and progress were combined together to guide teachers in giving support (Bourguet, 2022). Another learning strategy used by learners is manual video annotation which was found to be easy to use and helped learners boost their knowledge (Seo *et al.*, 2021). Learning via video annotation was found to be more appealing than learning via annotating textual learning resources (Lee & List, 2019). However, this type of annotation is subjective and prone to errors. For that reason, teachers annotate their videos by highlighting main points and adding different types of questions to enhance their students' engagement and information retention (Tseng, 2021).

A learner's interactivity with the learning materials can produce different visual or textual data which can be used to train a classifier to help in predicting their knowledge gain successfully (Otto *et al.*, 2021).

The problem is that not all learners who use VBL are equipped with suitable learning strategies and teacher-support techniques to aid in using VBL. Consequently, many learners will be overwhelmed when searching and selecting videos from the massive amount of freely available online videos. Adding to that, selecting the most suitable videos for learning will take up a significant amount of their time. Furthermore, and more crucially, is the challenge of identifying key points in different videos and linking these points to the study domain (Bywater et al., 2021; Schlotterbeck et al., 2021). Accordingly, VBL needs to be enriched with interactive techniques that aid learners and help them engage more while using learning resources. Various interaction types can be provided to learners that have been identified by Palaigeorgiou et al. (2019); video annotation–annotating fixed positions within a video (video segment characterisation), adding side content, questions, reflective pauses, video summaries (to help information recall), formative and summative assessments, giving feedback, structuring the video
content, encouraging replays and tracking interactions.

Hence, relying on a video's visual content will limit the education to using only tutorial videos where the visual and textual video content is related to the learning domain. However, if we want to use videos with examples of activities (e.g, interviews, presentations and negotiations) related to our main domain (soft skills), the visual content (e.g, nodding, smiling) may not be directly related to the domain. Accordingly, our approach focuses predominantly on textual video data (video transcripts) to automatically detect the key domain areas and segment the videos correspondingly. We aim to link different domain related video parts together to generate video narratives–our ultimate goal. For that purpose, within the next two sections, we review the literature that tackled video segmentation and characterisation, and the generation of video narratives to help in motivating our research questions.

2.3 Video Segmentation and Characterisation

Segmenting and characterising videos helps in identifying the different key parts in them and can be done using their visual or textual content. Video segmentation involves dividing a video transcript into individual segments and, in this thesis, we segment the video transcript using a text-window following the Text Tiling approach, as described in (Section 3.5.2) and implemented in (Sections 4.4.2 and 6.4.2). Video segment characterisation aims to understand and describe the content of the segment to facilitate learning. In our work, characterising video segments involves identifying the focus topics and concepts using: a Deep learning model (BERT) to contextually characterise a video segment by identifying its focus topic, and using Semantic Tagging to identify the focus concept(s) and its full path to the focus topic(s) using a taxonomy hierarchy. The characterisation process is defined in (Section 3.5.3) and implemented in (Sections 4.4.3 and 6.4.3) Generally, video segmentation and characterisation can be done for different purposes, not only for learning. In this section, we review the application of video segmentation and characterisation in non-learning and in learning contexts.

2.3.1 Non-learning Contexts

Video segmentation and characterisation can be approached differently for different purposes, for example: the characterisation of archival sport videos (Xue *et al.*, 2017) is based on the collected metadata about these videos using an assembly of computer vision techniques (shoot boundary detection, a tensor based compact representation, player detection and tracking, and optical character recognition). A video characterisation (Fajtl *et al.*, 2019; Rochan & Wang, 2019) can be created using the visual content (key frames) of the videos. The key frames, which are similar in their visual features, will be grouped and the point where the scene is changed will be detected using Kernel Temporal Segmentation. The key shots (each one is a group of similar key frames) represent the video characterisation which can be used to identify key areas in the video. Similarly, characterising a video can be done using the video summarisation Cycle-SUM model (Yuan *et al.*, 2019) which takes, as an input, a sequence of video frames and produces an importance score frame-wise as an output. The frames with the highest scores will be selected as a key area in the videos.

According to our approach where we focus on video transcripts, here we are reviewing the literature that focuses on characterising other types of text, e.g, app reviews and product information. These types of text are unstructured and this is similar to the nature of a video transcript. For example, Tushev *et al.* (2022) used an automatic unsupervised approach (KeyATM) for extracting cohesive domain related topics from user reviews on mobile apps which can be considered to look at by the users of the app. Similarly, Khatter *et al.* (2022) showed that product information available on e-commerce websites works as a summary of the product which can be collected via a website designed for this purpose that utilises the web scraping methodology. Based on the collected data, the user will have an overview about a product.

Adding to that, Google has started annotating YouTube videos $^{1\ 2}$ to engage users and communicate with them while watching the actual video footage. For

¹https://www.youtube.com/watch?v=wf_77z1H-vQ[Accessed: September 2021]

²https://cloud.google.com/video-intelligence/docs/reference/rest/v1/videos/ annotate[Accessed: September 2021]

2.3. VIDEO SEGMENTATION AND CHARACTERISATION

that purpose, Google developers began by using the Freebase knowledge graph 1 where nodes correspond to real world things and have their own unique ID. Google uses three types of data sources to be able to annotate a video (text metadata, audio/visual features and video context). The text metadata is the description of the video that is provided by the up-loader which, in many cases, has many-to-many associations that cause text ambiguity. For that reason, text metadata should be consistent to allow it to be correlated with other sources of data. Google developers first extract main entities from the text and assign scores to them based on their connection with other entities in the text. First, the relationship between these entities will be extracted from the open internet. A score will be assigned to all entities for being mentioned in the text and another score will be added if they appear in the title as well. Additional scores are given for any link that the entity has with other entities. It is essential now to know the central entity to be included in the annotation. This will be processed by applying a specific threshold, e.g., keeping the entities with scores $\geq \theta$. However, in-case the text metadata is of poor quality or not available, the video's audio and visual features will be used. For that purpose, a classifier can be used but an issue of types of nuances in the features will affect its accuracy. Consequently, the context of the video (user comments, web pages linked to it, and the overall user engagement) can be used to detect what is notable about the video. Aggregating these data sources could boost the annotation quality. Video annotation should be central which means it should be: complete (completely describe the video), specific (no more specific words can replace them), and compact (cannot be removed). To support the video annotation, developers can use the structured data on Freebase to get more information about the video entities.

The aforementioned segmentation and characterisation approaches showed the benefits of summarising videos or textual resources using visual or textual cues to aid the process of sorting or recalling these resources. However, these segmentation approaches have not been tested across multiple domains.

According to the context of our work, we review in more depth the approaches for video segmentation and characterisation in a learning context.

¹Freebase knowledge graph has now been replaced by Google with Firebase-Google's App Development Platform

2.3.2 Learning-related Contexts

In a learning context, there are a number of approaches to support video segmentation and characterisation: (a) manual, (b) automatic, (c) textual and (d) ontological approaches.

(a) Manual Approaches. A widely used approach for video segmentation and characterisation is manual annotation (note-taking) which helps in identifying references of the main points mentioned within a video (Dodson *et al.*, 2019; Dutta & Zisserman, 2019). Video characterisation can be done by students, teachers or by both. In Castro & Tumibay (2021), teachers segmented videos by providing sections within the videos related to specific courses. Additionally, teachers in Tseng (2021) highlighted video content with a phrase or keyword or by adding questions in order to improve learner engagement and to help the digestion of the learning material. Teachers in Tseng (2021) characterised videos by highlighting the contents with a phrase or a keyword or by adding questions.

Characterising videos can also be done by learners. For instance, teachers have asked learners to annotate videos and test their effect on their learning (Mirriahi *et al.*, 2021). Though learner annotation can improve engagement with videos, it is dependent on the learner's engagement (e.g, high self-efficacy learners engage better) and their prior knowledge (e.g, learners may not be able to notice key points). Co-working can be done by teachers and learners (Chiu *et al.*, 2018; Colasante & Douglas, 2016) to characterise and highlight video content for learning to help find such content quickly. To speed up the video characterisation process, Benkada & Moccozet (2017) used a constrained tracker to choose which video frame a user should characterise, and also allow collaborative characterisation by a group of people. These approaches characterise the whole video considering different purposes, including learning.

While such characterisation is closely linked to the learning goals set by the teachers, it is subjective, laborious and does not scale across different educational domains. Hence, automatic characterisation approaches will be considered to improve video characterisation in terms of the speed and the applicability in

different educational domains with characterisation-quality preservation.

(b) Automatic Approaches. Recent works have developed approaches for automating the process of video segmentation and characterisation for learning. This includes using different types of data. For example, a learner's interaction while watching videos is represented by their comments (Mitrovic *et al.*, 2016); these are aggregated to identify "high attention intervals" which refer to key points noted by learners. These intervals are used to facilitate interactions with videos by offering an interactive visualisation interface. These interactions can indicate points of interest or confusion. However, this type of segmentation depends on the learner's engagement and learners may not appropriately capture the key points in the videos.

Alternative approaches focusing on the video content are proposed. Videos' different associated data can be used (Schlotterbeck *et al.*, 2021) to detect teaching practices (presenting, guiding, administration) in recorded lessons of trainee teachers. For that purpose, a multi-learning model is designed where the acoustic features are extracted from the audio using the Random Forest model to classify the three aforementioned teaching practice categories. The textual features were extracted from the transcripts of the recorded videos using the Google Speech API while a Spanish BERT Model was used to add contextual information to the data.

Enriching tutorial videos containing slides is done by matching a slide to a video snippet (Liu & Lauw, 2022) via a deck of slides (containing several slides) and its related videos. First, for each slide, (Liu & Lauw, 2022) start with a window size and accordingly, a number of slides will be combined with the first slide and a number of video snippets that match the window size. Dynamic Time Wrapping (DTW) is where the similarity between the time series of the slides and the video snippets within a window is measured. This process is repeated for different window sizes to identify probable irrelevant slides to exclude them. Within this process, several calculations have been done; the best match between the video snippets and the slides, the total distance between the video snippets and the slides, and the cosine distance between each slide and the video snippets

they are matched to. (Liu & Lauw, 2022) use all of these calculations to decide a threshold to keep relevant slides and exclude any irrelevant ones.

Recent video segmentation approaches based on video content utilise state-of-the-art tools in natural language processing and tap into the availability of knowledge models. For example, MOOC video lectures were automatically segmented in Das & Das (2019) using a neural network over adjacent sentences; the neural network was trained on Wikipedia pages. The transcript of the lecture videos was then segmented into sentences and each sentence associated with its universal encoding using the Stanford Natural Language Inference. Then, the sentence, together with its preceding sentence and its following sentence, were fed to a Long Short Term Model (LSTM) classifier to decide a topic boundary among the sentences and identify which sentence represents the segment boundary. To annotate each segment, the text on the slides of the lecture video is first detected using Stroke Width Transform where the slide text will be bounded with a box. Accordingly, the top third slide –text within a box would be extracted as an image and passed to an OCR engine to recognise the text. Main terms (words) would be extracted from the text lines and would be compared to the list of topics extracted from the course syllabus. The similarities between the terms and the list of topics were measured using Word Mover's Distance to assign the selected topic for segment characterisation.

Learners' action while using open ended educational environment (CAD) (Bywater *et al.*, 2021), are collected and segmented into different groups based on the difference in the actions. These segments can be used later to identify a student's engagement and the type of personalised feedback to help them further improve their performance. The learner's sequence of actions is collected where each sequence will be segmented using an algorithm designed for this purpose. Each segment will consist of actions which are different from the actions in the neighbouring segments. As the design behaviour among the students might be similar, a clustering step is conducted to group the similar designs and the proportion of actions within each cluster. The clustering step is done using K-means clustering and the distance measure between the proportion of actions within each segments is calculated using Euclidean distance. K-means is repeated 100 times and every time the K-means is repeated, a silhouette width average

value is calculated to measure the quality of the cluster. The clusters with the highest silhouette average value will be selected as segments of similar actions. Accordingly, personalised feedback for each segment (cluster) can be generated to help learners.

From the aforementioned approaches, we can conclude that the video transcript allows us to understand the domain. For example, when listening to patient stories, it is the video transcript that explains the patient conditions but not the visual contents. This is also noticed when listening to people talk about job interviews or their experiences with health systems. Analysing the visual contents gives us some gestures, whereas the transcript gives us the richness of the domain related information. Using video transcripts allows us to manipulate both tutorial and example videos where people share their experiences. Accordingly, we investigate the characterisation approaches that tackle educational text as well, e.g., intelligent textbooks.

(c) Intelligent Textbooks. Different AI technology generations have been utilised to improve the use of digital textbooks, some are: Semantic Web and NLP and Machine Learning. A comparison between textbooks and intelligent textbooks using AI has been conducted in Koć-Januchta et al. (2020) where it has been found that AI textbooks, if used for longer, will have a noticeable effect on learning. This is because AI textbooks offer linking between content sections. The value that AI added to the textbooks can surpass the content of an individual textbook. For instance, in Sosnovsky (2022), they enrich the textbooks with domain related external smart activities by using the index terms of textbooks to extract domain knowledge models. As a result, each index has its reference pages and the corresponding domain concepts presented in these pages, linked to sections in the knowledge model. To enrich the textbook with related external resources, the concepts from textbooks will used by a parser application to extract concepts from the smart learning resources (which includes domain related examples and problems). To decide to which extent the extracted resources are linked to the concepts in the textbooks, an expert is invited to manually annotate the smart learning activities previously annotated by the parser application to filter out the concepts that the learners should focus on.

Inspired by the importance of extracting the domain knowledge presented in the text (video transcripts, in our work), we review the literature about enriching video characterisation using domain ontologies. This guides us on how to extract the domain's hierarchical knowledge presented in the video transcript which, in turn, help in linking different video segments together.

(d) Using Domain Knowledge. Using domain related ontologies adds semantic meaning when characterising videos which will facilitate their usage for learning. A recent review of semantic video characterisation (Sikos, 2017) points to key challenges, including: the wide variety of video codecs, the lack of standardised vocabularies, the vast number of video resources, the inherent ambiguity of audio-visual content and the unstructured nature of user generated content. As a way to address these limitations, Sikos (2017) recommends semiautomatic or automatic video annotation using ontologies and Linked Data combined with semantic tagging tools. Semantic video characterisation adds meaning by linking objects from the videos to a specific domain, and has been proposed in several works.

Labelling videos with textual data (Kithmi *et al.*, 2016) can be done using shot boundaries, extracting features, and conducting automatic text annotation. These features have been used to conduct semantic matching (via an ontology) between them and the user's query to select the most related videos to answer the query. For this purpose, the work began by analysing a video using a hash algorithm to identify shots where each shot has been given a fingerprint string based on the image details. A single unique frame from each shot has been sent to a deep learning model for object classification. This model has been trained on image data sets categorised with 1000 classes to generate textual annotation. The labelled shots have been used to annotate their video and the latter has been stored in the MySQL database. The stored videos are retrieved based on a comparison between the user query and domain ontology terms, if a match is found, the related stored videos are retrieved and displayed to the user.

Different video textual modalities (Cagliero *et al.*, 2019), e.g., transcripts, slide text and hand written text on whiteboards, can be used to segment and characterise videos. A lecture video is segmented in a semi-automatic way based

on detecting slide transitions. The text in each segment is represented by the associated transcript, slide text and the notes written on the whiteboard by the teacher. The text is pre-processed and analysed using a supervised approach (Named Entity Recognition and Disambiguation), and for disambiguation, they used the decision tree classification model. Then, the tokens from the text are mapped to Wiki-data and DPBedia terms. The annotation of the segments is associated with the video where the user can browse to select a segment to start with.

Some video characterisation approaches utilise video transcripts only. For example, Schulten *et al.* (2020) used video transcripts to extract keywords using DBPedia Spotlight and ranked them using tf-idf to measure their suitability to annotate the transcript. The keywords are presented to learners as learning resources in their learning environment. Accordingly, these keywords can be used via Google Custom Search API to collect domain related resources. Additionally, the associated DBpedia URLs related to each key word can be presented to learners as learning resources to enhance their engagement.

Instead of using generic knowledge models (DBPedia), domain related knowledge graphs (Das & Das, 2020), with the aid of the BERT model, can be used for topical segmentation of lecture videos. At the beginning, the semantic similarity between the concepts mentioned in the video segments is computed. The segments with the same focus concepts are grouped together. Then, the slides of the lecture videos are analysed using inception-V3 to classify slide frames and apply frame difference techniques to extract unique slides. Each slide has a starting and ending time within the lecture and their associated transcript is extracted. They use the FLAIR-api⁻¹ to calculate the average score of the similarity between the mentioned concepts and assign the value as an edge weight in the domain knowledge graph. For each slide, a graphlet, where the concepts mentioned in the transcript are connected, and the edge weight are assigned from the knowledge graph. The frequency of the concepts is used to measure the similarity between consecutive slides which indicates that the instructor is carrying on in explaining a specific concept. Otherwise, a topic boundary will be

 $^{^{1}\}mathrm{FLAIR}$ Api

detected between these consecutive slides. The final output is a lecture video where the topics taught are highlighted.

Likewise, domain knowledge graphs (prerequisite relations) (Torre *et al.*, 2022) can be built using the video transcript. It presents an unsupervised method for prerequisite identification and graph building. The video transcript has been used to identify the intervals where a concept is intensely mentioned and to track it within the video. To identify concept intervals, a Hidden Markov Model is used, which is a statistical model that aims to describe the probabilistic relationship between sequences of observations and a sequence of hidden states. Video analysis tools are used to improve the identification of concept intervals and these tools are: video segmentation using segment similarity, using the XGBoost decision tree machine learning library (used for, e.g., classification and ranking) for lecture type identification, and Tesseract-OCR to detect concepts and definitions from slide text. To extract the prerequisite relations, Allen's algebra (Allen, 1983) is applied on the extracted concept intervals to decide if: one concept precedes the other, the two concepts meet each other, or one concept includes the other concept. Based on that, the prerequisite relations have been determined and used to build the prerequisite graph using RDF data.

2.3.3 Video Segmentation and Characterisation in this Thesis

Our main focus is to help learners to notice different perspectives of key domain concepts in video transcripts and the complementary between them. The existing approaches tried to achieve this goal using manual or automatic video segmentation and characterisation. The manual approaches helps in identifying domain related parts but they are:time consuming, prone to errors, subjective and do not scale (Dodson *et al.*, 2019; Dutta & Zisserman, 2019). The automatic approaches can scale and capture video content objectively but they: depends on the learner's engagement (Mitrovic *et al.*, 2016) and learners may not appropriately capture the key points in the videos, depends on the acoustic features which extracted from the audio recording (Schlotterbeck *et al.*, 2021) where these features are prone to noise and not focusing on the main points, rely on the videos' visual contents, e.g. video slides (Das & Das, 2019; Liu & Lauw, 2022) where not all of videos used for learning contain slides (especially the video for soft skills learning which is the main domains in this thesis) and this counts as a limitation to using videos for learning, and use general knowledge representation, e.g. DBPedia or Wikipedia to annotate videos semantically where domain specific topics and concepts could be missed in these general models Sikos (2017).

Our main contribution in the video segmentation and characterisation area is the design, implementation and evaluation of the Video Segmentation and Characterisation for learning Framework (VISC-L). VISC-L is different from the aforementioned methodologies as it relies only on video-transcript to understand the main video contents. This is based on the fact that not all videos (e.g. videos used to learn soft skills, interviews, podcast, patient stories, etc.) are designed by including informative visual contents, e.g.slides. To segment the video-transcript, VISC-L use the TextTiling approach, which is well known for segmenting documents into coherent multi paragraphs units (see Section 3.3.1) to generate initial segments from the video transcript. To create coherent segments, we need first to understand the content of the video segments: VISC-L uses a Deep learning model to identify the context of the video segments and use semantic tagging to link the video segment-content to the domain topics and concepts. The novelty of VISC-L is represented by combining between the characterisation generated by the semantic tagging and the Deep learning model to identify the domain focus topic and concept in the video segment utilising the hierarchy of the domain taxonomy. To complete the segmentation work of VISC-L which is to generate coherent segments, VISC-L aggregates adjacent video segments that focus on the same topic and concept. Given that the aim of VISC-L is to support learning using video segments, and that the video transcripts represent the way that the speaker constructs their message, we underpin the aggregation process of video segments with Thematic Progression Theory (see Section 3.3.2). This theory is based on the link between the main starting point in the speakers message (Theme–we use domain topics to refer to the Theme of a video segment) and the added information (Rheme–we use domain concepts to refer to the Rheme of a video segment). Accordingly, to complete the segmenting work, we aggregate the initial segments using the Aggregation with interpolation algorithm (see

Section 3.5) to generate coherent segments as they focus on the same domain topic and concept. VISC-L is implemented and evaluated with users in our prime domain 'Giving Pitch Presentation' with a comparison with Google Segmentation and Characterisation methodology (see Chapter 4). Additionally, we prove that VISC-L is generic by applying it in our second domain 'Health Related Quality of Life Awareness' as presented in (Chapter 6) where different challenges related to the implementation process has been discussed.

The semantic characterisation of the video segments allows us to link related segments together following the ontology hierarchy. This enables us to generate video narratives from these characterised video segments to be used for focus-learning. For this purpose, in the next section, we review the use of narratives in learning as another useful interaction technique.

2.4 Video narratives for learning

A narrative, in education, is defined as a sequence of ideas or actions that belong to or are done by humans or machines which relate to a specific event (Greenhalgh & Hurwitz, 1999). The intersection of technology and storytelling has led to fascinating developments in linking domain related areas together by focusing on: Storytelling Design, Hypermedia, and Concept Maps.

2.4.1 Narratives using Storytelling Design

A well known type of narratives in learning is storytelling. The vast production of personal videos, web pages (used for conversation blogs), and games (that use dialogues in their design) have affected the symbolic communication used during storytelling (Bianchi & Ragonese, 2023). Narrative based learning environments (McQuiggan *et al.*, 2008) have been used to teach different subjects where the domain context is delivered using storytelling following pedagogical strategies, ensuring effective and engaging learning. The idea of linking educational content in a Web Based Training (WBT) product has been supported (Aqqal *et al.*, 2007) by providing teachers with a "macro design" methodology. Hence, the teachers identified the concepts to learn and their relationship with the previous concepts that the student knows. Then, the teachers followed the Rhetorical Structure theory (Mann & Thompson, 1987) to justify their design of the WBT by stating the reason for the WBT, the knowledge to be produced within it and the form it is used in to produce this knowledge.

To help learners access the most related video parts from lecture videos to help them revise quickly, a digital library has been designed (Lee *et al.*, 2021) that includes managing educational videos. Students, through using a web browser where this library is deployed, can access and play the short videos which are part of lecture videos. These short video parts have been detected using the logs of other students who visited and spent more time on them in comparison to the other parts of the video. The students are presented with a video usage graph to show high attention sections of the videos to help them select their revision materials. Differently, Pina *et al.* (2022) designed a set of instructions based on the narrative idea for learners who are attending face to face classes or are learning remotely to study principles in engineering design.

To help students set their ultimate goals and design their learning plans, a *narrative experience design canvas* has been designed in Recke & Perna (2021) for that purpose. Many researches, as reviewed by Naul & Liu (2020), focus on using narratives in video games designed for learning purposes which showed enhancements in the students' engagement, motivation and performance. For example, Colaço *et al.* (2022) designed an interactive narrative game that aims to raise awareness of gender equality in STEM for a group of teenagers.

To automate the generation of the narratives, their computational definition is followed (Oghaz *et al.*, 2020) which includes the following: "topic mining, text summarisation, machine translation, graph visualisation". First, to generate a summary-based narrative of short text, it requires defining the main topics in the domain of the text. The topic is selected using an unsupervised model. Secondly, selecting the sentences that summarise their focus topic –the sentences that better represent the topic are selected based on word co-occurrence and temporal information. The discovered probabilities of the topics are applied over the whole document to extract the sentences that belong to the same topic. The higher the number of topic-words in a sentence, the higher this sentence is ranked. The sentences with the highest ranking are presented as summary-based narratives for that topic starting from earliest to latest in their time stamp (e.g, a summary of tweets).

It is noticeable that narratives in education became a focus of many researchers and are implemented via different tools. Researchers have tried to use different tools in conjunction to watching educational videos to enhance learning, e.g, using hypermedia tools and concept maps.

2.4.2 Narratives using Hypermedia

Hypermedia takes different shapes like graphics, audio, video, plain text and hypertext to be used in online learning. Hypertext can be used to implement the idea of narratives to learn foreign languages (Bourina & Dunaeva, 2019) by linking contextual visual and textual learning modalities following a pedagogical theory. Educational videos, which are a type of hypermedia, have been used widely in online learning, e.g, YouTube educational videos, and have been ranked highly by learners for their suitability and usefulness to learn different domains (Maziriri *et al.*, 2020). Learning based videos are one of the instructors' main instruments to help in delivering topics (Guo *et al.*, 2022) or track the students' progress in conversational videos (Dey-Plissonneau *et al.*, 2021). Other hypermedia types have been applied differently, e.g, in Serrano-Iglesias *et al.* (2021), a smart learning environment has been designed where YouTube videos, web pages and quizzes are utilised by the instructors to set goals and provide their learners with personalised resources.

Similarly, the web application designed by Fessl *et al.* (2021) has been found to be useful as mentioned in a study with instructors and learners. It could facilitate communication between instructors and students, allowing instructors to set goals and design lectures while enabling students to direct their learning and to develop self-regulation skills.

Using Hypermedia in VBL helps in varying the learning resources which serve different students' needs. However, to help students identify the key domain learning points in these resources, we require other tools. Next, we review the idea of using concept maps which, in their structure, resemble the idea of video narratives –the main contribution of our work.

2.4.3 Narratives with Concept Maps

Domain key points can be either defined clearly by storytelling, or be extracted from the learning materials. Based on that, the idea of creating concept maps is developed to enhance engagement and improve student performance. The idea of concept maps is first developed by Novak (1964), and established on the subsumption theory of meaningful learning by Ausubel (Ausubel, 1963). It should be for a specific domain or theme and have a structure (nodes and links). A node represents a topic in the domain while the link represents the relationship between these topics. The use of concept maps as an annotation tool helps in grasping the key domain areas and develop a student's critical thinking (Machado & Carvalho, 2020). These concepts maps can be created using (a) **semi automatic approaches** or (b) be generated **automatically**.

(a) Semi automatic approaches. Students (Spicer *et al.*, 2020) created their own concept maps using a tool designed for them which also includes prompts as recommended concepts. The students can use these prompts or create their own concepts to build a concept map. Manually created concept maps help students improve their critical thinking and domain concept comprehension. Semi-automatic concept map creation is also attempted in Pinandito *et al.* (2021) by providing learners with a system that has concept labels extracted from lecture videos to help them build a concept map for these videos. However, students might have a learning barrier towards the new technology of using concept maps which can be expressed as: creating concept maps is time consuming, and the students' limited domain knowledge prevents them from identifying the relationship between domain topics. Nevertheless, with the development of machine learning and the intensive work in NLP field, an automatic approach has been developed for aiding the automatic creation of concept maps.

(b) Automatic approaches. The automatic creation of concept maps has been developed in Tang *et al.* (2021) by designing ConceptGuide based on the idea of concept maps. The designing of ConceptGuide includes extracting the topics of YouTube videos (via the transcript and metadata), using Wikipedia glossaries and using NLP techniques to identify the possible semantic similarities between the topics. ConceptGuide helps students by identifying the topics of the videos to be used for learning. Likewise, in Shao *et al.* (2020) concept maps generation has been automated by extracting the domain concepts and identifying the relationships between them using textual data. This has been done by extracting the concepts from text questions and student answers and using these questions and answers to link the main concepts together. Then, implementing "Association Rules Mining between Concepts" for the concept map generation. However, they assume that each question considers one concept—they did not consider multiple concepts recognition per test. Hence, if the number of concepts increased, the generated concept map will be different from the ground truth concept map generated manually by an expert using the same dataset. Differently, Hao *et al.* (2021) links online resources as narratives and represents them via a knowledge-graph to help in filtering the relevant resources and avoiding the irrelevant ones during the search process.

2.4.4 Video narratives in this thesis

Along the segmentation and characterisation line, is the research on facilitating video-based learning by supporting the linking of learning materials (Bianchi & Ragonese, 2023; Liu & Lauw, 2022; Pina *et al.*, 2022). Building on this idea, some studies have explored the use of concept maps. Semi-automatic concept maps (Lecuyer *et al.*, 2020) rely on the learners' abilities to identify key points, and may not be effective when the learners are not familiar with the domain. Automatic concept maps provide visual representation of video content which can support learning (Tang *et al.*, 2021). However, these approaches are computationally expensive and are usually domain-dependent (Shao *et al.*, 2020). Crucially, while the visual representation of complex knowledge structures can be helpful for navigating through videos, it may be overwhelming for learners not familiar with the domain. The current automated approaches for linking videos do not explicitly utilise pedagogical models for linking knowledge structures to support learning, while adopting pedagogical theories to link videos is seen as key by teachers (Archambault *et al.*, 2022).

This thesis contributes to the stream of automating the linking between video parts by developing a novel and generic framework for generating video narratives which utilises Ausubel's subsumption theory of meaningful learning(Ausubel et al., 1968) (see Section 3.3.3 which introduces the theory and Section 3.6 which presents how the theory is utilised in our work). The theory focuses on how the learning materials are linked in the learners' cognitive structure. The cognitive structure is a hierarchical organisation of general (high inclusive) concepts from which there is a link to specific (less inclusive) sub concepts. Ausubel argued that new material would be learned by relating to relevant concepts in the existing cognitive structures. Accordingly, Ausubel defined four subsumption links to learn new concepts (Ausubel, 2012): (a) Derivative: when the new learning material is an example of an idea that exists in the cognitive structure, (b) Super-ordinate: when the new learning material is related to a generic idea in the cognitive structure, (d) Correlative: when the the new learning material is linked to a similar previously learned idea, and (c) Combinational: when the new learning material is not (super-ordinate or sub-ordinate) with any existing ideas in the cognitive structure but rather can be learnt in combination with other, more generic, less relevant ideas. These narratives aim to support learners by providing them with short learning resources (video segments) that have been connected to provide meaningful learning.

To be able to implement these subsumption links to generate the video narratives, we have designed the Video Narratives for Learning (VIN-L) framework which requires, as an input, the characterised video segments (with domain topics or concepts) and a domain taxonomy representing the key concepts and their hierarchy in the domain. VIN-L has been applied and evaluated with experts and potential learners in the prime domain 'Giving Pitch Presentation' (see Chapter 5) and implemented and evaluated with domain experts in the second domain 'Health Related Quality of Life Awareness' (see Chapter 6).

2.5 Summary

In our work, we have been motivated by the vital role that VBL plays in promoting and facilitating learning among students online. Accordingly, we have reviewed the literature that focuses on: using VBL, the challenges associated with using VBL and the possible approaches to address these challenges (video segmentation and characterisation, and generating video narratives).

Using videos available online (e.g, YouTube videos) helps teachers embed technology into their lessons to help students meet the learning objectives. Nevertheless, there are many challenges associated with using VBL for both teachers and learners. They are varied; from teacher perspectives, these challenges include: selecting the best videos for learning, identifying the related parts in the videos to learn from, engaging and supporting students while using VBL. From the learners' perspectives, these challenges include: identifying the key domain points and finding ways to recall them, watching videos without getting overwhelmed by the vast amount of available videos, and getting support while watching the videos to clarify any misconceptions.

The aim of our research is to address the above challenges by developing a methodology that can be applicable to different soft skills learning domains and scale for a vast number of videos related to these domains. The main steps included in our methodology are : semi-automating the collection of domain related videos, automating the identification of the key domain points in the videos via segmentation and characterisation approaches using domain taxonomy, and automating the linking of related parts from different videos (to generate video narratives) based on the Ausubel Subsumption Theory to support domain learning. Our methodology has been motivated in this chapter, clarified and justified in Chapter 3, and implemented and evaluated in two different domains in Chapters 4, 5, and 6.

Chapter 3

Research Methodology

3.1 Introduction

The main aim of this research is to support learning by providing video narratives that are automatically collected, segmented, characterised and linked following a learning theory. For this purpose, we designed two generic frameworks, one of which is for Video Segmentation and Characterisation to Support Learning (VISC-L). The second framework is to link these characterised segments based on a learning theory to generate Video Narratives to Support Learning (VIN-L). To ensure that these frameworks are applicable in soft skills learning domain, we have underpinned their design with theories. For video segmentation, we based our work on the Text Tiling approach (Hearst, 1993), and for video segments' aggregation we adopted the Thematic Progression theory (Danes, 1974). Finally, we adopted Ausubel's Subsumption Theory for meaningful learning (Ausubel *et al.*, 1968) to guide the linking of the characterised video segments to generate the video narratives.

The following sections will provide details about: the research methodology (Section 3.2), underpinning theories (Section 3.3), the research preliminaries (Section 3.4), the Video Segmentation and Characterisation framework (VISC-L) (Section 3.5), and the framework of video narratives for learning (VIN-L) (Section 3.6). The domain of soft skills we will use to implement and validate the frameworks will also be illustrated (Section 3.7).

3.2 Overview of the Research Methodology

In our work, we have adopted a knowledge driven approach that consists of two main frameworks: Video Segmentation and Characterisation for Learning (VISC-L) and Video Narratives for Learning (VIN-L).

In **VISC-L**, we first collect domain related videos by utilising the domain taxonomy to implement the search schema we designed for this purpose-the limitation of this step is described in (Section 3.5.1). The collected videos are filtered based on duration, transcript availability and via manual inspection to check their domain relevancy. These videos are segmented following the Text Tiling approach (described in Section 3.3.1). Then, the segments are characterised using two algorithms (Semantic Tagging and contextual characterisation using a Topic Classifier). The characterisation results from both algorithms are combined and the characterised segments are aggregated following Thematic Progression Theory to improve their quality. Thematic Progression Theory has been selected to underpin the aggregation process. This is due to the nature of the video transcript that holds the main presenter message. The video transcript is subdivided into segments which are aggregated coherently based on the common focus topic and concept between them. To prove the applicability of VISC-L, we implement its algorithms described in (Section 3.5) in the prime domain-Giving Pitch Presentations. The full implementation details is described in (Chapter 4) where the output of VISC-L is evaluated with users by assessing the Quality, Perceived Usefulness, Cognitive Work Load and the Learning effect in comparison with Google segmentation and characterisation methodology. To prove the generality of VISC-L, we apply and evaluate it in a second domain–Health Related Quality of Life Awareness, see (Section 6.4). In the second domain, the domain taxonomy was not available and we have created it manually which causes the VISC-L implementation to be semi-automatic-this limitation is described in (Section 3.5.1).

In **VIN-L** to be able to link video segments together in a way close to the domain structure which is represented by a taxonomy, Ausubel Subsumption Theory is used. Subsumption Theory focus on how the learning materials are

linked together in the learner's cognitive structure. Accordingly, VIN-L takes, as an input, the characterised video segments (VISC-L output) and utilises the domain taxonomy to implement Subsumption Theory. This theory is fully described in the next section. Four types of video linking are generated based on the definition of Subsumption Theory (Derivative, Super-Ordinate, Correlative and Combinational). The video narratives are as web pages to facilitate its usage by learners. VIN-L is applied and evaluated in the first domain–Giving Pitch Presentations. To prove the generality of VIN-L, we apply and evaluate it in the second domain–Health Related Quality of Life Awareness. The overall research methodology is presented in Figure 3.1.



Figure 3.1: Overall Research Methodology.

The two main practical stages in our methodology are: Video segmentation and Characterisation for Learning (VISC-L) and Video Narratives for Learning (VIN-L).

3.3 Theoretical underpinning

In this section we will review the main approach and theories adopted in this research.

3.3.1 Text Tiling (Hearst, 1993)

To be able to achieve our goal of generating video narratives to support learning, our work is concerned with the segmentation of videos to identify the key domain concepts. In order to carry out video segmentation, we adopt a text segmentation approach by focusing on the video transcript—Text Tiling which is one of the most elegant and commonly applied approaches for text segmentation. Although this approach is old but it is still adopted successfully by researchers to help in identifying key areas in a document. Below we explain the Text Tiling approach and review the way it is adopted in different work. Then, we described how we adopt this approach to help us in implementing our methodology described in the previous section. Additionally, we describe our work similarity and difference from the other work that adopt this approach.

Text Tiling is designed by Hearst (1993, 1997a) and is based on an assumption that a discourse is mostly consisting of paragraphs which are well structured with opening sentences introducing the main topic and ending sentences that summarise the paragraph. Consequently, sub-dividing this text into subtopics (multi-paragraph units) makes it coherent. The shift between the subtopics within a discourse is based on *patterns of lexical co-occurrence and distribution*. In her work, Hearst implemented Text Tiling in two applications: Hypertext and Information Retrieval. Hearst imposes the use of Text Tiling in hypertext on the hypothesis that it is better to segment long online documents when presenting them to users to help them recall useful information. These segments, which are passages, need to be meaningful units.

The main steps included in the Text Tiling algorithm (Hearst, 1997a) are: Tokenisation, Lexical Score Determination and Boundary Identification, Smoothing the Plot and Determining the Number of Boundaries. In the Tokenisation step, the text is split into tokens which will be grouped into blocks of size K (3 to 5 sentences long)-the value of K is based on the average paragraph length within the actual text. Then, the similarity among the token-sequence between two adjacent blocks is calculated and the gap token-sequence is also calculated to detect the boundaries to decide the segmentation of text into subtopics. Following this, she uses the initial segments of several sentences in the video transcript, and then applies segment aggregation to detect segment boundaries.

Counting the frequency of words between the blocks is not enough to decide the similarity between these blocks. Consequently, Hearst (1993) decided to use Yarowsky's statistical lexical disambiguation algorithm (Yarowsky, 1992) which helps in calculating the word sense. These values are also used when calculating the similarity of the words among adjacent blocks. Although Yarowsky's algorithm does not need training data, it requires a thesaurus. As a result, WordNet-an openly available hand-built lexical thesaurus-has been used. To better identify the boundaries between blocks (paragraphs), the depth score is calculated and plotted to present the valleys and peaks. The depth score presents the depth of the valley which is a token-sequence gap. The plotted values can be smoothed to remove small gaps by using average smoothing. The scores used in the averaging formula are the depth scores of the blocks to the left and to the right of the token-sequence gap and the score of the token sequence gap. The blocks' boundaries will be adjusted to the nearest block break. For the evaluation, the results of the Text Tiling algorithm against human judgment of segmentation were compared and showed acceptable performance but there is still room for improvement. One of the suggested improvements is to decide the boundaries based on the subtopics hierarchy. Another suggestion is to improve the word sense calculation to better decide the similarity of words among adjacent blocks. In the same vein, we use similarity between segments to define segment boundaries. However, we apply a more elaborate characterisation approach which allows us to identify the stretch of a topic, called Thematic Progression Theory.

The Text Tiling approach has been utilised by He *et al.* (2020) where the similarity values between blocks have been plotted. The resulting curve is smoothed by using a sliding window over the interval points. The average similarity values within each window are calculated and placed in the middle of the windows.

Having done so, the local variations in the curve are terminated and highlighted to show large similarity variations.

Text Tiling techniques have been adopted by other researchers, e.g. in Ghinassi (2021) the approach was used for linear text segmentation. To calculate the similarity between adjacent blocks in the text, the two techniques Text Tiling and deep sentence encoding have been used. This includes extracting sentences from a document, using the sentence encoder to calculate sentence embeddings, selecting a window size to have blocks of sentences, computing the score of each sentence in a block using cosine similarities with the embedding- the average of the sentences in the adjacent blocks- then, calculating the depth scores for each position (sentence) in the window. The number of boundaries in the document is decided by the number of the highest depth scores; otherwise, the number of boundaries is the depth scores that fall above a predefined threshold. The evaluation with other methods revealed that this methodology produced better results when compared with unsupervised methods also based on Text Tiling. Meanwhile, this methodology did not compete well with supervised approaches when enough training data was available. This proved that using deep sentence encoders helps in improving the text segmentation and annotation which showed a potential to successfully apply it on different broadcasting media types. In comparison, our work does not rely on the similarities between sentence embeddings but on the similarities in topics in the adjacent segments. To identify the focus topic in the segments, we presume that the tackled domain should be presented with a domain taxonomy. Correspondingly, we have utilised transformer-encoders as a topic classifier to identify the focus topics in the segments. Accordingly, adjacent segments with the same focus topic will be aggregated to facilitate learning key areas in the domain.

Moreover, Miculicich & Han (2023) hypothesised that accurate text segmentation leads to a better summarisation prediction by the model. Accordingly, two methods have been used. One method is unsupervised where the improved Text Tiling model is used to calculate the similarity score between adjacent blocks of tokens with the aid of BERT embeddings. The other method is the supervised model that compares adjacent sentences to detect the beginning of a new section. The combined results from the two methods have been used to improve the accuracy of an extractive summarisation model.

To summarise, Text Tiling starts with segmenting text (initial segments) based on sentences from documents then deciding the boundaries by combining segments together. To do so, the similarities between the text in adjacent segments are calculated between the words in the text (Hearst, 1993, 1997a) or via using machine learning models (Ghinassi, 2021; Miculicich & Han, 2023).

In our work, we are using a video transcript as a source of domain related information. To be able to understand which domain areas presented in the video transcript, we decided to segment the video transcript into coherent parts-which is the same goal of Text Tiling approach. Accordingly, this approach is adopted in our work to help in generating initial segments but instead of segmenting based on paragraphs or sentences, we segmented the transcript into a window of 6 text lines segments which resembles the size of a paragraph. We have been inspired by the aforementioned work that adopts Text Tiling and uses contextual characterisation to recognise segments' contents and based on that aggregate adjacent segments. In our work, to better understand segments' content, we combine contextual characterisation and semantic tagging to identify domain key areas. Instead of deciding the segment boundary by calculating the patterns of co-occurrence of words in different segments and aggregate them as in Text Tiling, we characterise the video segments with domain topics and concepts. These topics and concepts are defined by domain taxonomy and use the taxonomy hierarchy to link related segments together to generate video narratives-our ultimate goal in this thesis. The second step of Text Tiling which is to aggregate adjacent segments that has the pattern of words co-occurrence, has been adopted but in a different way. To ensure that the aggregation of the segments is implemented coherently which serves our goal to support learning while watching video segments, we underpin the aggregation process with Thematic Progression Theory (see Section 3.3.2). This theory is used to teach coherent writing where the aggregation is based on the Theme and Rheme of adjacent paragraphs (video segments in our work). We use the focus topic as the Theme and the focus concept as the Rheme of a video segment. The use of the Text Tiling theory in generating video initial segments is described in (Section 3.5.2) and its implementation in the prime domain 'Giving

Pitch Presentations' is described in (Section 4.4.2) and its implementation in the second domain is described in (Section 6.4.2).

3.3.2 Thematic Progression (TP) Theory (Danes, 1974)

Text Tiling allows us to propagate the initial segments which shows a shift in the concepts based on the similarity between adjacent segments' text. Hence, we will use it as the basis to identify adjacent segments in the video that can be aggregated and the gap segments that can be interpolated. Since in our work we are using the video transcript only and to ensure that our aggregation mechanism is generic, we need a theory or an approach to guide us on how to aggregate small segments into larger segments. For that purpose, we will follow TP theory which is often used for analysing a student's writing coherence. TP is designed by Danes (1974) that is based on the systematic functional grammar theory which shows how to write in a cohesive manner. This means how to arrange the main components of a sentence (clauses) cohesively. Clauses are considered similar if they begin with the same elements. These elements are called Theme and the remaining elements of the clause are called Rheme.

The basic terms used by this theory are 'Theme' and 'Rheme' which were defined by Halliday & Matthiessen (2014). He defines the term Theme as the element that serves as the point of departure of the message while the term Rheme is defined as the part in which the Theme is developed (Halliday & Matthiessen, 2014, pg.89). Accordingly, any sentence begins with a Theme and is supported by/ ends with a Rheme.

Based on these terms (Theme and Rheme), (Danes, 1974, pg.114) defines the TP theory as the *skeleton of the plot*. TP has been used by Danes (1974) to analyse scientific and professional text. Accordingly, three types of Thematic Progressions have been identified, see Figure 3.2:

(a) Simple Linear TP: the Rheme of one utterance (statement) becomes the Theme of the next utterance.

(b) TP with a Constant Theme: the Theme of one utterance becomes the same Theme of the next utterance.

(c) TP with Derived Themes: the utterance has multiple Themes.



Figure 3.2: (a) Simple Linear (b) Constant Theme (c) Derived Themes (Danes, 1974).

The Thematic Progression theory has been applied in many learning contexts. To help English non-native learners better understand the lectures and the learning materials that are delivered or written in the English language, both Theme and Thematic Progression Theory have been applied. Jing (2015) tried to help learners with being able to write cohesively by presenting them with an instructional package. This package is based on the Thematic Progression theory for Chinese students to improve their English writing. In Kuswoyo *et al.* (2020), to help Indonesian learners understand the Engineering lectures delivered in English, they examined the use of the Theme and Thematic progression theory. Their findings proved that if TP is followed by lecturers or writers, the learning materials will direct learners to the key point in the discourse. In the domain they tackled which is English Engineering lectures, both Constant and Derived TP were applied.

In this thesis we use TP theory when developing the aggregation mechanism to inform how to link the small segments. Considering the characterisation nature of the video segments that is identified in (Section 3.4) which consists of a set of focus topic(s) and focus concept(s). We can refer to the focus topic as Theme and the focus concept as Rheme. As the focus topic and its focus concept can be extended among adjacent segments in a video, we found the most suitable TP type for our work is Simple Linear as the focus concept (Rheme) is extended to be the focus in the adjacent segment(s). The description of how this theory is used to design VISC-L framework is in (Section3.5) and the implementation of this theory, as part of VISC-L, in the prime domain 'Giving Pitch Presentation' is described in (Section 4.4) and in the second domain 'Health Related Quality of Life Awareness' is described in (Section 6.4).

3.3.3 Subsumption Theory For Meaningful Learning (Ausubel *et al.*, 1968)

The ultimate goal of our computational approach is to automatically generate video narratives to support learning. A video narrative is an account of domain topic(s) and concept(s) elucidated via video segments that are arranged following a specific structure. To be able to arrange the video segments, the hierarchy of their focus topic(s) and concept(s) should be identified. Accordingly, our knowledge driven approach presumed that the domain we tackle should be presented via a taxonomy that specified the hierarchy of the domain topic(s) and their concept(s). Additionally, we need to follow a systematic learning theory to guide the linking of the video segments that can be implemented by utilising the domain taxonomy. Correspondingly, Subsumption Theory (Ausubel *et al.*, 1968) seemed to be the most appropriate theory to adopt in this work as it has been applied to underpin learning using a taxonomical model.

A primary process in learning has been defined in Ausubel *et al.* (1968) as a subsumption in which new material related to relevant ideas in the existing cognitive structures is derived from learning experiences. A learners' cognitive structure is a hierarchical organisation of general (high inclusive) concepts from which there will be a link to specific (less inclusive) sub concepts. Ausubel argued that new material should be integrated with previously presented information through comparisons and crossreferencing of new and familiar concepts. Hence, during the learning process the new learning material interacts with the learners' cognitive structure and will be linked to the more inclusive concepts in it. However, if the learning material cannot be subsumed (linked) to an existing concept in the cognitive structure, then it will be classified as a high inclusive concept with no further subsumed concepts yet.

Four subsumption linking types, that link learning material to each other, have been identified by Ausubel Ausubel (2012):

(a) Combinational: when the new learning material is not super ordinate or sub ordinate with any existing ideas in the cognitive structure but rather can be learnt in combination with other, more generic, less relevant ideas.

(b) Correlative: when the new learning material is an elaboration of a previously learned idea.

(c) Derivative: when the new learning material is an example of an idea that exists in the cognitive structure.

(d) Super-ordinate: when the new learning material is related to a generic idea in the cognitive structure.

Successful adoption of Subsumption theory for meaningful learning includes using concept maps (Katagall *et al.*, 2015; Liu *et al.*, 2018) that allow learners to group information in related modules, making the connections between modules more apparent. While concept maps provide domain structure, they do not allow reasoning and automation. Similarly to concept maps, taxonomies can define the main concepts and relationships in a domain. In addition, using taxonomies (or more generic ontologies) allow reasoning to automate the generation of instruction paths through the domain. Therefore, instead of concept maps, we use a taxonomy to represent the domain.

Similarly to Al-Tawil *et al.* (2020), who operationalises Subsumption Theory to generate information exploration paths through a large knowledge graph, we explore the hierarchical relationships between domain taxonomy concepts to identify how to link content. In this thesis, the **fundamental difference** is that we first need to characterise video segments by mapping them to a taxonomy which allows the subsumption processes proposed by Ausubel. To do that, we firstly identify focus topics and concepts of video segments using reasoning over the hierarchical links in the taxonomy. This enables subsumption links (Derivative, Correlative, Super-Ordinate, and Combinational subsumption) to combine video segments for meaningful learning, as described in (Section 3.6). The implementation of Ausubel Subsumption Theory in the prime domain 'Giving Pitch Presentations' is described in (Chapter 5) where the output is evaluated by domain experts and advanced learners using Amazon Mechanical Turk. Additionally, the theory is implemented in the second domain 'Health Related Quality of Life' is described in (Section 6.5) and the output is evaluated with domain experts as described in (Section 6.6).

3.4 Preliminaries

This section includes the main definitions used in the formal description of the domain context, frameworks and the algorithms throughout this thesis.

Definition 1: Video segment. A video segment VS is a video interval that has: an id which is a YouTube video ID with a number relating to its position in the video as a video can include several segments, a start time VS_s (with respect to the start of the full video from which this segment comes) and an end time VS_e . We assume that each video has a transcript (text) VS_t . Consequently, each video segment is associated with an id, start time, end time and a transcript representing the presenter utterance in the segments, $VS = \langle VS_{Id}, VS_s, VS_e, VS_t \rangle$.

Definition 2: Domain taxonomy. We assume that the domain is defined with a taxonomy $\Omega = \{C, H\}$ which consists of classes and individuals (that belong to these classes). C is a non empty set of relevant domain classes and individuals which are linked via a hierarchy H. We use $c_i \in C_j$ to denote that c_i is an individual of a class C_j meaning that c_i is directly beneath C_j in the taxonomy hierarchy. We refer to the immediate classes of the root of the hierarchy as domain topics T_i , where each T_i is a set of classes and each class is a set of its individuals, $T_i = \{C_1, ..., C_m\}, m > 0$, i.e. $C_i \in T$. Throughout this thesis, any class C_i at the very top level in the taxonomy hierarchy will be referred as T_i , see Figure 3.3. Hereafter, we are going to call any domain entity (class C_j or individual c_i) mentioned in the video transcript as a concept. This is also because in learning (which is the focus of our work), teachers refer to the domain as concepts not classes or individuals.

Using semantic tagging, we can link a video transcript to concepts (classes or individuals) in the taxonomy. The mentions of a concept c_i in a video segment transcript VS_t accumulate all the explicit mentions of c_i , i.e. the function $f(c_i, VS_t) > 0$, indicates the number of times that the concept c_i has been mentioned explicitly in the video segment transcript VS_t . For each video segment VS we identify the domain concepts mentioned at least once in the transcript VS_t as $f(C_j, VS_t) > 0$ which indicates the number of times that the concepts C_j



Figure 3.3: Domain taxonomy. The top classes T represent the main domain topics. The hierarchy of a concept (class or individual) represent its path to the main topic, e.g. $c_{i1_1} \in C_{i1}$ and the latter is \subseteq of the topic T_i (which is one of the top classes in the taxonomy). Hence, the path of the concept c_{i1_1} is: $c_{i1_1} \to C_{i1} \to T_i$.

have been mentioned in the text VS_t explicitly or implicitly-if at least one of its concepts (individuals) c_i has been mentioned explicitly. This is because these concepts C could be classes or individuals. Using the taxonomy hierarchy, we can implicitly identify the topic T of the mentioned concepts by referring them to their topic.

Definition 3: Concept Path. It represents the hierarchy of the concept (class or individual) in the domain taxonomy to reach its topic. For example, the path PC of a concept c_i (which is an individual in this example) will be: $PC(c_i) = \langle c_i, C_j, T_i \rangle$, this indicates that the concept $c_i \in (\text{class}) \ C_j$ and $C_j \subseteq (\text{of top class in the taxonomy hierarchy}) T$.

Based on the above definitions, we can define the Focus Topic and Focus Concepts presented in the video segment transcript.

Definition 4: Focus Topics. A focus topic FT in the transcript of a video segment is one of the top classes in the taxonomy hierarchy which is mentioned explicitly or implicitly if at least one of its concepts (classes or

individuals) has a notable number of mentions in the video segment transcript VS_t which we indicate with a parameter θ . The function f is to calculate the frequency that FT_i is mentioned in the video segment transcript $f(FT_i, VS_t)$ where $f(FT_i, VS_t) \geq \theta$. Depending on θ , a video segment can have several focus topics (for example, in the application presented in the (Section 3.5.3), we used $\theta \geq 1/3$ (third) of all concepts or topics mentions number). However, in case all the mentioned concepts in the video segments belong to a single domain topic then we use Ft to denote that it is a single focus topic not a set of focus topics FT.

Definition 5: Focus Concepts. Within a focus topic, we can identify corresponding focus concepts (classes or individuals). A set of Focus concepts FC within the focus topic FT as expressed in the transcript of a video segment VS_t is defined as the mentions frequency of focus concepts in the video transcript. The function f is to calculate the frequency that a FC_i is mentioned in the video segment transcript $f(FC_i, VS_t)$ where $f(FC_i, VS_t) \ge \theta$. Depending on θ , a video segment can have several focus concepts within the same focus topic. However, if none of the mentioned concepts in the segment transcript passed the condition $f(FC_i, VS_t) \ge \theta$, then these concepts will be considered as mentioned but not as a focus in the video segment. In case there is only one concept that is mentioned in the video segment and its mentions $\ge \theta$ then we use Fc to denote a single focus concept.

Definition 6: Characterised Video Segment. Any video segment VS that has assigned focus topics FT and focus concepts FC, after implementing the characterisation algorithms, is called a characterised video segment Ch. Correspondingly, a characterised video segment will have its: id, start time, end time, transcript (text) and the focus topic(s) and a focus concept assigned to it: $Ch = \langle VS_{\text{Id}}, VS_{\text{s}}, VS_{\text{e}}, VS_{\text{t}}, FT, FC \rangle$. In case a video segment has only one focus topic where only one of its concepts is a focus then the definition will be: $Ch = \langle VS_{\text{Id}}, VS_{\text{s}}, VS_{\text{e}}, VS_{\text{t}}, Ft, Fc \rangle$

Definition 7: Video Narrative (VN) is a sequence of characterised video segments which are linked based on a specific domain linking type.

VN is based on two components, the (FT or FC) to be learned and a sequence of video segments containing this (FT or FC) in their characterisation and these segments are linked following domain linking. Notice that the focus topic can be a set of focus topics FT or a single focus topic Ft which is based on the type of the domain linking followed to generate the video narrative. Similarly, focus concepts can be a set of focus concepts FC or a single focus concept Fc.

Following Ausubel's Subsumption Theory, we define four types of narratives that are based on the main subsumption links and utilise the domain taxonomy. In our work, each video narrative has a duration of ≥ 3 minutes and ≤ 6 minutes and this is based on the recommendation in (Brame, 2016) where the duration of educational videos should not exceed 6 minutes.

Definition 8: Derivative Video Narrative (D) is a type of video narratives that defines the learning to have the generic concept in the cognitive structure first before introducing one of its specific concepts to bind with it. Hence, to learn the specific concept c_i , the learning should start from the generic C_j , where $c_i \in C_j$. Consequently, the structure of the Derivative narrative will start with a group of video segments from the set of input segments which focus on the generic concept. Each video segment in this group has this definition: $Ch = \langle VS_{\text{Id}}, VS_{\text{s}}, VS_{\text{e}}, VS_{\text{t}}, Ft, FC_j \rangle$. Then, the Derivative narrative ends with a group of video segments that focus on the specific concept c_i and have this characterisation: $Ch = \langle VS_{\text{Id}}, VS_{\text{s}}, VS_{\text{e}}, VS_{\text{t}}, Ft, Fc_j \rangle$. Notice that both concepts c_i and C_j belong to one topic Ft.

Definition 9: Super-Ordinate Video Narrative (S) is a type of video narratives that defines the learning to have the specific concept in the cognitive structure first before introducing its generic concept to bind with it. Hence, to learn a generic concept C_j , the learning starts from one or more of its specific concept(s) c_i , where $c_i \in C_j$. The structure of the Super-Ordinate narrative will start with a group of video segments from the set of input segments with the following characterisation $Ch = \langle VS_{Id}, VS_s, VS_e, VS_t, Ft, Fc_i \rangle$. Accordingly, the specific concept c_i is in the cognitive structure now and ready to bind to its generic concept C_j . Therefore the last group of video segments in the Super-Ordinate narrative structure will have the following characterisation $Ch = \langle VS_{Id}, VS_s, VS_e, VS_t, Ft, FC_j \rangle$ where both the generic concept C_j and the specific concept c_i belong to the same topic Ft.

Definition 10: Correlative Video Narrative (CR) states that to learn a target concept c_i , the learning starts from one or more of its similar concepts (same level in the taxonomy hierarchy). Hence, to learn a target concept c_i , the structure of the Correlative narratives starts with video segments that focus on similar concepts to it c_{i-1} and c_{i+1} . Hence the video segments have the following characterisation: $Ch = \langle VS_{\text{Id}}, VS_{\text{s}}, VS_{\text{e}}, VS_{\text{t}}, Ft, Fc_{i-1} \rangle$ and Ch = $\langle VS_{\text{Id}}, VS_{\text{s}}, VS_{\text{e}}, VS_{\text{t}}, Ft, Fc_{i+1} \rangle$ respectively. The last group of the video segments will focus on the target concept c_i , $Ch = \langle VS_{\text{Id}}, VS_{\text{s}}, VS_{\text{e}}, VS_{\text{t}}, Ft, Fc_i \rangle$ where c_{i-1} , c_i , $c_{i+1} \in C_i$ and $C_j \subset T_i$.

Notice that all the concepts (class C_j and individuals c_{i-1} , c_i , and c_{i+1}) belong to one topic Ft.

Definition 11: Combinational Video Narrative (CO) means learning two topics together by exploring how some of their concepts are related to each other as they are mentioned in the same video segment. Hence, to learn the topics T_i and T_j , the Combinational narrative will list a group of video segments from the input set of segments that have the focus concepts c_i and c_j from the topics T_i and T_j respectively. Each selected segment to be in the Combinational narrative will have the following characterisation: $Ch = \langle VS_{\text{Id}}, VS_{\text{s}}, VS_{\text{e}}, VS_{\text{t}}, FT, FC \rangle$. Hence, FT refer to the two topics to be learnt and FC refers to the set of focus concepts (classes or individuals belong to the focus topics and mentioned together in the same video segment).

Definition 12: Augmented Video Narrative is when we augment the Video Narrative with text to make it suitable for learning as the sequence of video segments alone is not sufficient for learning.

These definitions will be used when presenting the frameworks VISC-L and VIN-L in (Section 3.5 and 3.6).

3.5 Video Segmentation and Characterisation for Learning (VISC-L) Framework

The first step of our computational approach is to define ways of segmenting and characterising videos. In order to make it generic to ensure it is applicable in different domains, contexts and independent on the area of learning, we define a framework for video segmentation and characterisation which we call VISC-L.

These algorithms are not novel but they are part of the novel framework (VISC-L) which is presented in Figure 3.4. It includes three main steps which is implemented in their order as presented in Figure 3.4 where the output of one algorithm is the input of the following algorithm. These algorithms, according to their order, are: Input, Initial Segments (Algorithm 1) where the initial segments are generated, Segment Characterisation (Algorithm 2) where the initial segments will be characterised, and Aggregation (Algorithm 3) where the characterised adjacent video segments is aggregated based on the common domain topic(s) and concept(s).

The novelty of VISC-L is represented by the combining of the Semantic Tagging characterisation and Topic Classifier characterisation which is different from other methodologies as described in (Section 2.3.3). Hence, the final video segment characterisation is represented by the: segment Id, segment start time, segment end time, segment transcript, the focus topic and the focus concept. The application of these algorithms in the selected domains is described in (Chapter 4 and 6). This kind of characterisation is essential to link different video segments from different videos as described in (Chapter 5 and 6).

3.5.1 Input

VISC-L is based on two assumptions. Firstly, it is assumed that the *video transcript* relates to the domain which will be learned. Hence, the text in the video transcript is taken as an input. The second assumption is that there is a *domain taxonomy*, refer to Definition 2. The taxonomy of the available ontologies, the Linked Open

3.5. VIDEO SEGMENTATION AND CHARACTERISATION FOR LEARNING (VISC-L) FRAMEWORK



Figure 3.4: Video segmentation and characterisation framework. Notice that FT means a set of focus topics and FC means a set of focus concepts within the focus topics, e.g. T_1C_1 means there is a set of focus concepts FC_1 in the focus topic FT_1 .

Data Cloud ¹ can be used or the domain taxonomy can be developed with domain experts. The latter is used in this work. With the first domain, we used the available domain ontology (Chapter 4), while with the second domain we built the domain taxonomy manually as described in (Section 6.2). The implementation of the VISC-L framework requires a domain taxonomy. If a domain taxonomy is not available, it can be generated either manually or automatically (Avron *et al.*, 2022; Liu *et al.*, 2012; Robin *et al.*, 2017). The need for a domain taxonomy is further discussed in the limitations, see Section 7.3).

If there is no pre-selected set of videos related to the domain, the domain taxonomy can be used to automatically collect videos from available social platforms (like YouTube). Following the concept hierarchy, a search schema can be developed, similarly to Pritoni *et al.* (2021). For example, using a combination of $< Domain, t_i, c_i$, "tutorial" > as search terms, videos with tutorials related to the concept c_i (class or individual) from topic t_i in the domain can be collected. The word "tutorial" can be replaced with other words [lectures/tutorials/conversations, patient stories, etc.] based on the required videos that relate to the domain to be learned. Although the collection of the videos from YouTube has been done

¹https://lod-cloud.net/[Accessed: March 2021]
automatically, the checking of the videos, to select domain related videos only, has been done manually. Hence, the method of video collection is semi-automatic.

In order to identify the main topics in the video transcripts, as part of the characterisation step, *training data* with domain topics as labels is needed. This can either be created with expert annotators, collected from past user interactions or automatically created (the latter two options are followed in the application of VISC-L in the prime and second domain, respectively). The collected video transcripts are passed as an input to the VISC-L framework.

3.5.2 Initial Segments

Following the Text Tiling approach we start with initial segments. Then, these segments are aggregated following TP theory to get larger coherent units (e.g. paragraphs). Hence, we include an initial segmentation step where the video transcripts are cut into small segments and are aggregated based on their characterisation. The size of the video segment can be decided using a window of text lines that has a specific size. For example, in Hearst (1997a), as described in Section 3.3.1, the tokens within a text are grouped into blocks of size K(3 to)5 sentences long) each and K represent the average paragraph length within the actual text. Video transcript doesn't consist of paragraphs but text lines, that represent the utterance, where most of the text lines are shorter than a normal sentence in a document. For that reason, in our work, the size of the video segments is determined by two conditions: it should include at least 6 text lines from the utterance, and the last text line should include one of the end of sentence punctuation symbols, [.,?,!]. This punctuation determines a segment boundary. Adding to that, these are initial segments, where the final segment is an aggregation of the adjacent segments that has the same focus topic and concept where the size of the aggregated segments can be increase. Hence, our segmentation Algorithm 1, presented next, describes the initial segmentation step of VISC-L where it divides video transcript text lines into segment. The tracing of this algorithm is in (Appendix A). The output of this algorithm is used as an input to the next step of VISC-L. This algorithm is not new or original but it is

part of the original framework VISC-L and we run it on each video we have in our input data.

Alg	gorithm 1: VISC-L–Initial segments		
D	Data: V_{id} , V_t and V_s /* V_{id} is a YouTube video id. V_t is a list of video		
	transcript's text lines where each text line is a list of characters. V_s		
	is a list of the start time of each text line in the video. Both V_t and		
	V_s have the same length*/		
R	esult: List of video segments $L /*$ List of lists where each list includes:		
	segment Id, start time, end time, transcript of video segment*/		
1 pu	inctuationList = [".", "?", "!"] /* Used to smoothen the segment		
b	ooundary */		
2 te	$xtCounter = 0 /* Used to go through V_t */$		
3 se	gmentStartTime=0; segmentEndTime=0; n=0 /* used to number the		
s	eegments */; $segmentId=` '/* It is a string which is a combination of$		
I	V_{id} and $n */$; $textLines = []; L=[]; \theta = 6 /* Starting with 6 text lines */$		
$4 \mathbf{W}$	hile $textCounter + \theta < length(V_t)$ and $textCounter < length(V_t)$ do		
5	/* length (V_t): returns the number of text lines in the video transcript.		
	The loop should stop if the number of the remaining text lines in the		
	transcript $+\theta \ge length(V_t)$. The loop also should stop if there are no		
	remaining lines where $textCounter \ge length (V_t)*/$		
6	if $textLines = //$ then		
7	$segmentStartTime=V_s [textCounter];$		
8	$n = n + 1; segmentId = string(V_{id} + n);$		
9	$\mathbf{for} \ i = 0 \ to \ \theta \ \mathbf{do}$		
10	$textLines.append(V_t[textCounter]);$		
11			
12	/* Check the segment boundary */		
13	if intersection(punctuationList, V_t [textCounter-1]) $\neq \phi$ then		
14	$segmentEndTime = V_e \ [textCounter-1]; \ L.append([segmentId, v]) \ L.append([segmentId, v]) \ L.append(v) \ L.$		
	$segmentStartTime, \ segmentEndTime, \ textLines]);$		
15	textLines = [] /* To start new video segment */		
16	else		
17	$textLine.append(V_t[textCounter]);$		

19 if $textCounter + \theta \ge length(V_t)$ then /* Either create a new segment to include the remaining text lines or 20 add the remaining text lines to the current segment */ if textLines = // then $\mathbf{21}$ $segmentStartTime = V_s$ [textCounter]; $\mathbf{22}$ n = n + 1;23 $segmentId = string(V_{id} + n);$ $\mathbf{24}$ for i = textCounter - 1 to length (V_t) do $\mathbf{25}$ $textLines.append(V_t[i]);$ $\mathbf{26}$ textCounter = textCounter+1;27 $segmentEndTime = V_e [textCounter-1];$ $\mathbf{28}$ L.append([segmentId, segmentStartTime, segmentEndTime, textLines]); 29 30 Return L

3.5.3 Segment Characterisation

In order to aggregate the initial segments, we need to identify what domain content is presented in each segment. This is done during the segment characterisation step which links each video segment VS with a set of domain focus topics FTand a set of domain focus concepts FC (classes or individuals) that are presented in the transcript of the segment. The outcome is the characterised segment (see Definition 6) which includes information about: the video Id, beginning and end of a segment referring to its position in the video footage, segment's transcript, domain FT and FC. The segment characterisation enables linking related domain areas from different videos following the domain taxonomy.

To produce segment characterisation, we propose to combine two algorithms: Semantic Tagging (using the domain taxonomy) and Topic Classifier (using a language model). Then, the outcomes from both algorithms are combined to obtain the segment characterisation.

Semantic Tagging. This is the first characterisation step where we apply the semantic tagging algorithm which we developed in our work Mohammed &

3.5. VIDEO SEGMENTATION AND CHARACTERISATION FOR LEARNING (VISC-L) FRAMEWORK

Dimitrova (2020), see Figure 3.5. This semantic algorithm links each video segment to its focus topics and focus concepts by mapping the terms from the domain taxonomy to the text in the video transcript. It first pre-processes the transcript¹ : tokenising the transcript, cleaning it from stop words and punctuation, and selecting nouns and noun phrases to match them with the taxonomy terms. If there is a match, the taxonomy concept c_i is identified (tagged to the text), noting also the path to reach a top level concept (i.e. linking to a focus topic $T, c_i \in$ T). Based on Definition 4 and 5 from (Section 3.4), each segment VS_i is linked to a set of focus topics FT^1 and their corresponding focus concepts FC^1 ; where ¹ indicates that this is an output from the first segment characterisation algorithm).

A key challenge for this algorithm is word sense ambiguity. This is not that prominent with carefully selected videos. However, if the videos are selected automatically from open social spaces, there will be a need of word sense disambiguation. To do this, we disambiguate the focus topics based on context, which is done with the second step of segment characterisation which is Topic Classifier.



Figure 3.5: Video Segment Characterisation Pipeline

Topic Classifier. This is the second algorithm which identifies a domain topic based on the context of that topic to overcome the issue of word sense ambiguity. This issue happens when same word has different meaning when used in different domains, e.g. the word bank has two different meanings: either river bank or a financial institution. Following the latest development in

¹We have used the Natural Language Tool Kit (NLTK) https://www.nltk.org/[Accessed: May 2021]

3.5. VIDEO SEGMENTATION AND CHARACTERISATION FOR LEARNING (VISC-L) FRAMEWORK

natural language processing, we use Bidirectional Encoder Representations from Transformers (BERT) (Devlin *et al.*, 2018) as a topic classifier. BERT embeds pre-trained deep bidirectional representations from unlabelled text by jointly conditioning on both left and right context in all layers. Accordingly, it can be fine-tuned with just one additional output layer to create state-of-the-art models for different language tasks, topic classification in this case. First, the BERT model is fine-tuned using training data with domain topic labels (which is part of the input for VISC-L). Then, the fine-tuned model is used as a classifier to link each segment VS_i to its domain topics FT_i^2 (where ² indicates that this is an output from the second segment characterisation algorithm).

Characterisation Combining. This is the last step in the segment characterisation phase where the output from each algorithm (Semantic tagging and Contextual Classification) is combined. For each segment Ch_i , the outcomes from the two algorithms $Ch_i = \langle VS_{Id}, VS_s, VS_e, VS_t, FT^1, FC^1 \rangle$ and

 $Ch_{i} = \langle VS_{Id}, VS_{s}, VS_{e}, VS_{t}, FT^{2} \rangle$ are combined by intersecting the focus topics $FT_{i}^{1} \cap FT_{i}^{2}$ to find the common focus topic FT_{i} and selecting the focus concepts FC_{i} from FC_{i}^{1} that belong to FT_{i} . Hence, the final characterisation outcome is that each segment is characterised by $Ch_{i} = \langle VS_{Id}, VS_{s}, VS_{e}, VS_{t}, FT, FC \rangle_{i}$ (a set of focus topics FT_{i} and their concepts FC_{i}).

Algorithm 2 includes the steps we run to do the segment characterisation step of VISC-L and its tracing is in (Appendix B).

Algorithm 2: VISC-L- Algorithm (Segment Characterisation)
Data: List of video segments <i>L</i> which is the output of Algorithm 1 and
domain taxonomy topics = $\{T_1,, T_n\}$ where each T_i is a set of
concepts (classes or individuals).
Result: List of Characterised Video segments <i>LCh</i> .
1 /* LCh is a list of lists where each list includes: a segment information
[segmentId, start time, end time, transcript], a list of Focus Topics FT and a list of Focus Concepts FC */
i = 0: $FT = []$: $LCh = []$: Semantic Tagging Output = []:
Topic Classifier Output = []:
3 /* Loop through all video segments. Each segment will have two
characterisations, one from the Semantic Tagging and the other one from
the Topic Classifier characterisation which are parts of the the Segment
Characterisation steps of VISC-L */
4 while $i < length(L)$ do
5 $Semantic Tagging Output = [L[i], FT^1[i], FC^1[i]] /* Implement the$
Semantic Tagging step of VISC-L to get the first characterisation of
a video segment-list of Focus Topics (FT^1) and Focus Concepts
$(FC^1) */$
6 Topic Classifier Output= $[L[i], FT^2[i]] /*$ Implement the Topic classifier
step of VISC-L to get the second characterisation-list of Focus
$\frac{Topics(FT^2) *}{ETT^{[1]}} = TTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTT$
7 $FT[1] = \text{intersection}(FT[i], FT[i]) / *FT is the list of the same Frame Tania(a) assigned to the summary of the implementing the$
Focus Topic(s) assigned to the segment after implementing the Somentia Tagging and the Topic Classifier stops of VISC L +/
if $ET[i] \neq []$ then
[LCh append([L[i] FT[i] FC[i]]) /* The Focus Concents FC are
ertracted from the Semantic Tagging output (first segment
characterisation) which are belong to the Focus Topic $FT[i] */$
10 else
11 LCh.append([L[i], [], []]) /* No Focus Topics and Focus Concepts
are assigned to the segment when there is no matching between the
Focus Topics assigned to the segments after implementing the
Semantic Tagging and the Topic Classifier steps of VISC-L */
12 $i = i + 1;$

13 Return LCh

3.5.4 Segment Aggregation

After characterising the video segments and to decide how to aggregate these segments to complete the segmentation work, we have examined the result. We have noticed that most of the adjacent segments within a video have similar focus topic and concept. Accordingly, the next step in VISC-L, after segment characterisation, is to aggregate the adjacent segments based on the similarity in their focus topic and concept which corresponds to boundary finding (as described by Text Tiling). For that purpose, we have developed an aggregation algorithm based on Thematic Progression (TP) theory (Danes, 1974), see (Section 3.3.2).

Relating to video segments and their characterisation, we associate the focus topic with the segment's Theme (i.e. the main clause) and the focus concepts with the segment's Rheme (the part of the sentence where the Theme is developed). The Simple linear Theme pattern was selected as the most appropriate (as described in Section 3.3.2), as it allows for keeping a continuous focus topic and focus concept along the aggregated segments. Some segments can be without characterisation which can be because the speaker is silent or is digressing from the domain. If we look strictly for adjacent segments, these gap segments which break the topic flow will lead to starting a new aggregate. Following (Hearst, 1993), as described in Section 3.3.1, we smoothen the boundaries between adjacent blocks by removing small gaps using interpolation. If the segments before and after a gap segment have common focus concepts, it is assumed that the common concepts are spread across the three segments. Hence, the gap segment is interpolated in the aggregated segment. Examples of Linear Aggregation with Interpolation are presented in both Section 4.4.4 and Section 6.4.4.

Below we propose a linear aggregation with an interpolation algorithm (see Algorithm 3) where the tracing of this algorithm is in (Appendix C).

3.5. VIDEO SEGMENTATION AND CHARACTERISATION FOR LEARNING (VISC-L) FRAMEWORK

Algorithm 3: VISC-L–Linear Aggregation with Interpolation		
]	Data: List of characterised video segments LCh which is the output of	
	Algorithm 2. Each characterised segment LCh_i has the following	
	elements: at position 0 is the list of [segmentId, start time, end	
	time, transcript], at positions 1 and 2 are the lists of the Focus 1	
	Topics (FT) and the Focus Concepts (FC), respectively.	
]	Result: List of aggregated video segments $(LAgg)$.	
1 7	i = 0; $gap = []; newSeg = [] /* Temporary holder of the aggregates */$	
	$LAgg=[]; \theta = 30 /* Setting to 30 seconds duration */$	
2	AggFT = []; AggFC = []; /* AggFT and AggFC are the Focus Topic and	
	Focus Concept of the aggregates, respectively $*/$	
3	while $i + 1 < length(LCh)$ do	
4	$s_1 = LCh[i] /* Check the first segment s_1 has a FC */$	
5	if $s_1/2 = []$ then	
6	/* No Focus Concepts (FC), save s_1 as a single segment */	
7	$LAgg.append(s_1); i = i + 1; go to step 3;$	
8	if $i+1 < length(LCh)$ then	
9	$i = i + 1; s_2 = LCh[i] /* Take the adjacent segment */$	
10	else	
11	go to step 44 /* If it is the end of the list, do the required saving $*/$	
12	if $intersection(s_1[2], s_2[2]) \neq []$ then	
13	/* Aggregate the adjacent segment and save the FC to AggFC and	
	$F I \ lo \ Agg F I * /$ $A ag EC - intersection (a [2] a [2]); \ A ag ET - intersection (a [1] a [1]);$	
14	$Aggr C = \text{Intersection}(s_1[2], s_2[2]), Aggr I = \text{Intersection}(s_1[1], s_2[1]),$	
15	<i>newsey</i> .append(s_1), <i>newsey</i> .append(s_2), $s_1 = []$, $s_2 = []$, got to 23 /*	
	Exit this toop and start to read one segment at a time and compareits EC with the AgaEC $*/$	
16	else	
17	/* Fither save s ₂ as a gap segment or save s ₁ as a single segment	
1.	and make $s_1 = s_2$. In both cases read a new $s_2 */$	
18	$if (s_2[0]/2] - s_2[0]/1]) < \theta $ then	
19	$\begin{vmatrix} & (2i) & (2i$	
20	else	
21	$LAgg.append(s1); s_1 = s_2; i = i + 1; go to step 8;$	

22 /* This while loop is to compare the FC of a new segment with the AggFC. Either include the segment in the aggregate or save it as a gap segment */23 while i+1 < length(LCh) do $i = i + 1; s_2 = LCh[i];$ $\mathbf{24}$ if intersection(AggFC, $s_2[2]$) \neq // then $\mathbf{25}$ if $gap \neq //$ then 26 $newSeg.append(gap); newSeg.append(s_2); s_2 = []; gap = [];$ $\mathbf{27}$ else $\mathbf{28}$ $newSeg.append(s_2); s_2 = [];$ 29 else 30 if $duration(s_2) \leq \theta$ then 31 $gap.append(s_2); s_2 = [];$ 32 else 33 /* Before starting a new aggregate, call Function Saving 1 */ 34 Call Function Saving 1; $s_1 = s_2; s_2 = [];$ Go to step 5; /* Start a 35 new aggregate */ 36 Function Saving 1(): if $gap \neq //$ then 37 newSeg.append(gap); gap = [];38 if $newSeg \neq //$ then 39 LAgg.append([newSeg[0][0][0], newSeg[0][0][1], newSeg[-1][0][2]],40 AggFT, AggFC]) /* Save the id, start time of the first segment and the end time of the last segment in the aggregates list with the AggFT and AggFC */if $s_1 \neq []$ and s_1 not in newSeg and s_1 not in LAgg then $\mathbf{41}$ $LAgg.append(s_1); s_1 = [];$ $\mathbf{42}$ return gap, newSeg, s_1 , LAgg; $\mathbf{43}$ 44 Call Function Saving 1; **45** if $s_2 \neq []$ and s_2 not in newSeg and s_2 not in LAgg then $LAgg.append(s_2);$ 46 47 Return LAgg

3.5.5 Output

The output of VISC-L is a set of characterised video segments with: id, a start time, end time in the corresponding video and transcript. Each characterised video segment Ch_i has a set of domain focus topics FT_i (top classes in the taxonomy) and a set of concepts FC_i (classes or individuals) from the focus topics which are mentioned in the transcript of the video segment. These characterised video segments represent the input to the VIN-L framework which is used to generate domain related video narratives.

3.6 Video Narratives for Learning (VIN-L) Framework

Similarly to the VISC-L framework which is generic and can be applicable in different learning domains and contexts, we propose a generic framework for generating video narratives to support learning (VIN-L). The overall architecture of VIN-L framework is presented in Figure 3.6.



Figure 3.6: VIN-L Framework. The taxonomy is a list of topics which are top level in the taxonomy structure (black nodes), and a list of concepts (classes or individuals) C (white nodes).

Although Ausubel theory has been adopted by different researches as described

in (Section 3.3.3), the novelty of the video narratives is represented by the VIN-L framework that includes four algorithms each designed following Ausubel subsumption theory to generate four types of video narratives. These narratives, that link video segments from different videos, have been evaluated in two different domains in terms of their quality, perceived usefulness to support learning (see Chapter 5 and Section 6.5). VIN-L includes three main phases: Input, Subsumption Linking Algorithms and Output. The input is the aggregated video segments and the domain taxonomy. The Subsumption Linking Algorithms are designed based on Ausubel's subsumption Theory that defined in (Section 3.4), Definitions 7, 8, 9, 10 and 11. The output includes the four types of video narratives generated from the four types of Subsumption Linking Algorithms. The duration of the generated video narrative should be ≥ 3 minutes and ≤ 6 minutes as described in (Section 3.4), Definition 7.

3.6.1 Input

To be able to implement VIN-L, we first assume the availability of the domain related video segments characterised based on the domain taxonomy which, in our work, is the output of the VISC-L framework. Secondly, we assume the availability of the domain taxonomy that is utilised to implement the Subsumption Linking Algorithms.

Domain taxonomy. We are using the domain taxonomy defined in (Section 3.4), Definition 2. The Linked Open Data Cloud, as discussed in (Section 3.5.1), can be used or the taxonomy can be developed with experts (the latter is used in this work).

The Characterised Video Segments. The output of the VISC-L framework is video segments that have been characterised with domain FT and FC as defined in (Section 3.4), Definition 6. The segment characterisation is used to implement the four Subsumption Linking Algorithms.

3.6.2 Subsumption Linking Algorithms

Here we are explaining the implementation of Ausubel's Subsumption Theory Ausubel et al. (1968) which defined four types of linking (Derivative, Super-ordinate, Correlative and Combinational). The subsumption link definitions revolve on human cognitive structure that help in binding prior and new knowledge. Based on their definitions in (Section 3.4) we have designed four Subsumption Linking Algorithms (Derivative, Super-ordinate, Correlative and Combinational). Ausubel theory as a cognitive theory that is described in (Section 3.3.3) is different from learning style where the latter can be defined by, e.g. an interview with learners and can be assessed by using e.g. questionnaires, after using the learning materials. In this thesis, we are focusing on linking the video segments by generating video narratives based on how the information reside in the cognitive structure. Then, we evaluated the video narratives in terms of their quality, perceived usefulness and cognitive work load. From the evaluation studies we have obtained the information about the average number of experts and learners preferring one type of narratives over the other type due to their cognitive preferences. In future work, another type of analysis between video narratives, which is similar to comparing learning styles, can be conducted by embedding these video narratives in a learning platform.¹

Derivative Linking. It helps to learn a specific concept by exploring its generic concept first to be able to link them together. The design of the Derivative algorithm is based on its definition in (Section 3.4), Definition 8. Hence, referring to the taxonomy structure presented in (Figure 3.7), to learn the specific concept c_{22_1} , the Derivative algorithm first introduces the generic concept C_{22} via a group of video segments that focus on it and places these segments at the beginning of the narrative. Then, the Derivative algorithm select the video segments that

¹For example, in Zapalska & Dabb (2002) four types of learning styles have been mentioned: concrete (where learners use a concrete object to learn in a real way), abstract (learners pay attention to details), sequential (learners are organised and tend to follow instructions), and random (learners are not sequential but holistic). As a future work (Section 7.4.2), the narratives can be embedded in a learning platform to make an analogy between the Derivative linking and random learning style and between the Super Ordinate linking and the sequential learning style.

focus on the specific concept c_{22_1} and add these segments to the narrative. The algorithm below shows the steps we followed to implement Derivative subsumption linking to generate the Derivative narratives.

The input to the algorithm is the list of the characterised video segments and a list of the taxonomy concepts' paths. A path of a concept c_i , as defined in (Section 3.4), Definition 3, is: $PC(c_i) = \langle c_i, C_j, T_i \rangle$. To learn a specific concept c_i , the algorithm first iterate through the list of the characterised video segments and match their focus concepts with the concepts c_i from the concept path. Hence, if the FC of a segment includes C_j (the immediate parent concept of c_i) then this segment is selected for the first group in the Derivative Video Narrative (D). Whilst, if the FC of the segment includes the concept c_i (the concept to be learnt) then this segment is selected for the second group of the segments in the D. The duration of the selected segments for the D does not exceed six minutes (as described in Section 3.4), Definition 7. If the duration is $\leq = 6$ minutes then include all the collected segments in D; otherwise, we select some of the collected segments from both groups to prevent exceeding the 6 minutes duration of the D. Examples of the generated Derivative narratives in both domains are described in (Chapter 5 and 6). Algorithm 4 includes the steps we followed to generate the Derivative narratives and the tracing of this algorithm is in (Appendix D).



Figure 3.7: The domain taxonomy consists of: top concepts T that represent the main domain topics. The hierarchy of a concept (class or an individuals) represent its path to the main topic, e.g. the concept $c_{i1_1} \in C_{i1}$ and the latter is \subseteq of the topic T_i . Hence, the path of the concept c_{i1_1} is: $c_{i1_1} \rightarrow C_{i1} \rightarrow T_i$. The Subsumption links are illustrated using the domain taxonomy hierarchy. The Combinational linking used to learn concepts related to different topics (T_{i-1}, T_i) . The Correlative linking used to learn concepts related to the same topic (e.g c_{i1_1} and c_{i1_2}). The Derivative linking used to learn the specific concept c_{22_1} by first learning its generic concept C_{22} . The Super-Ordinate linking is used to learn the generic concept C_{12} by start learning its specific concept c_{12_1} or/and c_{12_2} .

Algorithm 4: VIN-L–Algorithm (Derivative Linking–D)

- **Data:** List of the aggregated video segments *LAgg*, which is the output of Algorithm 3 where each segment has: its information at position 0, the list of the Focus Topics at position 1, the the list of the Focus Concepts at position 2. List of taxonomy concepts' paths *PC* /* *Each concept c from the taxonomy is represented with a list of concepts where c is the first concept and the remaining concepts are its parents* */. The specific concept to be learnt *c*.
 - **Result:** D: List of video segments to learn the specific concept c by introducing first video segment focusing on its parent concept p.
- 1 D = []; i = 0; j = 0; p = " /* Variable to get the name of the parent of the concept c */
- **2** $\theta = 6$ /* Set 6 minutes as a max segment duration */
- **3** Lp =[] /* List of the segments that have the parent concept p of a concept c as the focus concept in their characterisation */
- 4 Lc =[] /* List of the segments that have the specific concept c as the focus concept in their characterisation */

5 while
$$j < length(PC)$$
 do

- **6** /* Finding the path in PC for the concept c */
- 7 | if PC[j][0] = c then
- **s** p = PC[j][1] / * Find the immediate parent concept p which is at position 1 in the path of the concept c * /

9
$$j = j + 1;$$

10 while i < length(LAgg) do

- 11 /* Find the characterised video segment where its second element includes p; otherwise check if the second element includes c */
- 12 | if $p \in LAgg[i][2]$ then
- **13** Lp.append(LAgg[i]);
- 14 else if $c \in LAgg/i/2$ then
- 15 Lc.append(LAgg[i])
- 16 i = i + 1;

$i = 0; duration p = 0 / * Save the duration of the segments that have P in their Focus Concept FC */$ $is while i < length(Lp) do$ $ig \ duration p = duration p + (Lp[i][0][2] - Lp[i][0][1]); i = i + 1;$ $i = 0; duration c = 0 / * Save the duration of the segments that have c in their Focus Concept FC */$ $i while i < length(Lc) do$ $ig \ duration c = duration c + (Lc[i][0][2] - Lc[i][0][1]); i = i + 1;$ $if duration p + duration c <= \theta then$ $if duration p + duration of the collected segments is \leq \theta, save these segments in the list of the video segments for the Derivative narrative D * / D.append(Lp); D.append(Lc);if duration of the selected segments is > \theta, split the time (\theta = \theta/2; i = 0; duration = 0; someLc=[]; someLp=[] /* The lists someLp and someLc include some of the segments from the Lp and Lc lists, respectively */ if duration = duration + (Lp[i][0][2] - Lp[i][0][1])); i = i + 1; i = 0; duration = 0; while i < length(Lc) do if duration = 0; while i < length(Lc) do if duration = 0; while i < length(Lc) do if duration = 0; someLc.append(Lc[i]); duration = duration + (Lc[i][0][2] - Lc[i][0][1])); i = i + 1; D.append(someLp); D.append(someLc); s Return D$	17	/* Ensure the duration of the collected segments does not exceed θ */		
their Focus Concept FC */ 18 while $i < length(Lp)$ do 19 \lfloor duration $p = duration p + (Lp[i][0][2] - Lp[i][0][1]); i = i + 1;$ 20 $i = 0;$ duration $c = 0 / *$ Save the duration of the segments that have c in their Focus Concept FC */ 21 while $i < length(Lc)$ do 22 \lfloor duration $c = duration c + (Lc[i][0][2] - Lc[i][0][1]); i = i + 1;$ 23 if duration $p + duration c <= \theta$ then 24 $/*$ If the duration of the collected segments is $\leq \theta$, save these segments in the list of the video segments for the Derivative narrative $D */$ D.append(Lp); D.append(Lc); 25 else 26 $/*$ If the duration of the selected segments is $> \theta$, split the time $(\theta = \theta/2)$ between the two groups of the collected segments, Lp and Lc */ 27 $\theta = \theta/2; i = 0;$ duration $= 0;$ some $Lc = [];$ some $Lp = [] / *$ The lists some Lp and some Lc include some of the segments from the Lp and Lc lists, respectively */ 28 while $i < length(Lp)$ do 29 $ $ if duration $\leq \theta$ then 30 $ $ $\int duration \leq \theta then30 \int duration = 0;31 uration = 0;33 while i < length(Lc) do34 if duration \leq \theta then35 wilk i < length(Lc) do34 if duration \leq \theta then35 wilk i < length(Lc) do34 if duration \leq \theta then35 wilk i < length(Lc) do34 if duration \leq \theta then35 wilk i < length(Lc) do34 if duration \leq \theta then35 wilk i < length(Lc) do34 if duration \leq \theta then35 wilk i < length(Lc) do36 i = i + 1;37 D.append(someLp); D.append(someLc);38 Return D$		i = 0; duration $p = 0 /*$ Save the duration of the segments that have P in		
<pre>18 while i < length(Lp) do 19</pre>		their Focus Concept FC */		
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	18	while $i < length(Lp) do$		
<pre>20 i = 0; durationc = 0 /* Save the duration of the segments that have c in their Focus Concept FC */ 21 while i < length(Lc) do 22 durationc = durationc + (Lc[i][0][2] - Lc[i][0][1]); i = i + 1; 23 if duration p + duration c <= 0 then 24 /* If the duration of the collected segments is ≤ 0, save these segments in the list of the video segments for the Derivative narrative D */ D.append(Lp); D.append(Lc); 25 else 26 /* If the duration of the selected segments is > 0, split the time ($\theta = \theta/2$) between the two groups of the collected segments, Lp and Lc */ 27 $\theta = \theta/2$; i = 0; duration = 0; someLc=[]; someLp=[] /* The lists someLp and someLc include some of the segments from the Lp and Lc lists, respectively */ 28 while i < length(Lp) do 29 if duration ≤ 0 then 30 duration = duration + (Lp[i][0][2] - Lp[i][0][1])); 31 i = i + 1; 32 i = 0; duration = 0; 33 while i < length(Lc) do 34 if duration ≤ 0 then 35 duration = duration + (Lc[i][0][2] - Lc[i][0][1]); 36 i duration = duration + (Lc[i][0][2] - Lc[i][0][1]); 37 D.append(someLp); D.append(someLc); 38 Return D</pre>	19	duration p = duration p + (Lp[i][0][2] - Lp[i][0][1]); i = i + 1;		
their Focus Concept FC */ 21 while $i < length(Lc)$ do 22 $\lfloor durationc = durationc + (Lc[i][0][2] - Lc[i][0][1]); i = i + 1;$ 23 if duration p + duration $c <= \theta$ then 24 $/*$ If the duration of the collected segments is $\leq \theta$, save these segments in the list of the video segments for the Derivative narrative $D */$ D.append(Lp); D.append(Lc); 25 else 26 $/*$ If the duration of the selected segments is $> \theta$, split the time $(\theta = \theta/2)$ between the two groups of the collected segments, Lp and Lc */ 27 $\theta = \theta/2; i = 0;$ duration = 0; someLc=[]; someLp=[] /* The lists someLp and someLc include some of the segments from the Lp and Lc lists, respectively */ 28 while $i < length(Lp)$ do 29 if duration $\leq \theta$ then 30 $\begin{bmatrix} someLp.append(Lp[i]); \\ duration = duration + (Lp[i][0][2] - Lp[i][0][1])); \\ i = i + 1; \\ $ 32 $i = 0;$ duration = 0; 33 while $i < length(Lc)$ do 34 $\begin{bmatrix} i f duration \leq \theta \ flen \\ someLc.append(Lc[i]); \\ duration = duration + (Lc[i][0][2] - Lc[i][0][1]); \\ someLc.append(Lc[i]); \\ duration = duration + (Lc[i][0][2] - Lc[i][0][1]); \\ someLc.append(someLp); D.append(someLc); \\ 38 Return D$	20	i = 0; duration $c = 0 /*$ Save the duration of the segments that have c in		
<pre>21 while i < length(Lc) do 22 durationc = durationc + (Lc[i][0][2] - Lc[i][0][1]); i = i + 1; 23 if duration p + duration c <= θ then 24 /* If the duration of the collected segments is $\leq \theta$, save these segments in the list of the video segments for the Derivative narrative D */ D.append(Lp); D.append(Lc); 25 else 26 /* If the duration of the selected segments is > θ, split the time ($\theta = \theta/2$) between the two groups of the collected segments, Lp and Lc */ 27 $\theta = \theta/2$; i = 0; duration = 0; someLc=[]; someLp=[] /* The lists someLp and someLc include some of the segments from the Lp and Lc lists, respectively */ 28 while i < length(Lp) do 29 if duration $\leq \theta$ then 30 if duration = 0; 31 while i < length(Lc) do 34 if duration = 0; 35 while i < length(Lc) do 34 if duration $\leq \theta$ then 35 if duration $\leq \theta$ then 35 if duration = duration + (Lc[i][0][2] - Lc[i][0][1])); 36 i = i + 1; 37 D.append(someLp); D.append(someLc); 38 Return D</pre>		their Focus Concept FC */		
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	21	while $i < length(Lc)$ do		
23 if duration $p + duration < = \theta$ then 24 $/*$ If the duration of the collected segments is $\leq \theta$, save these segments in the list of the video segments for the Derivative narrative $D * /$ D.append(Lp); D.append(Lc); 25 else 26 $/*$ If the duration of the selected segments is $> \theta$, split the time $(\theta = \theta/2)$ between the two groups of the collected segments, Lp and Lc * / 27 $\theta = \theta/2; i = 0;$ duration = 0; some Lc =[]; some Lp =[] /* The lists some Lp and some Lc include some of the segments from the Lp and Lc lists, respectively $* /$ 28 while $i < length(Lp)$ do 29 if duration $\leq \theta$ then 30 $\begin{bmatrix} if duration \leq \theta$ then 30 $\begin{bmatrix} some Lp.append(Lp[i]); \\ duration = duration + (Lp[i][0][2] - Lp[i][0][1])); \\ i = i + 1; \\ 32 i = 0; duration = 0; \\ 33 while i < length(Lc)$ do 34 $\begin{bmatrix} if duration \leq \theta$ then 35 $\begin{bmatrix} some L.append(Lc[i]); \\ duration = duration + (Lc[i][0][2] - Lc[i][0][1]); \\ i = i + 1; \\ 37 D.append(someLp); D.append(someLc); \\ 38 Return D$	22	$\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ $		
24 /* If the duration of the collected segments is $\leq \theta$, save these segments in the list of the video segments for the Derivative narrative $D *$ / D.append(Lp); D.append(Lc); 25 else 26 /* If the duration of the selected segments is > θ , split the time $(\theta = \theta/2)$ between the two groups of the collected segments, Lp and Lc */ 27 $\theta = \theta/2$; $i = 0$; duration = 0; some $Lc=[]$; some $Lp=[]$ /* The lists some Lp and some Lc include some of the segments from the Lp and Lc lists, respectively */ 28 while $i < \text{length}(Lp)$ do 29 if duration $\leq \theta$ then 30 $\begin{bmatrix} \text{if duration} \leq \theta \text{ then} \\ \text{some} Lp.\text{append}(Lp[i]); \\ \text{duration} = \text{duration} + (Lp[i][0][2] - Lp[i][0][1]));$ 31 $\begin{bmatrix} i = i + 1; \\ \text{some} Lc.\text{append}(Lc[i]); \\ \\ \text{duration} = \text{duration} + (Lc[i][0][2] - Lc[i][0][1]);$ 36 $\begin{bmatrix} \text{if duration} \leq \theta \text{ then} \\ \text{some} Lc.\text{append}(Lc[i]); \\ \\ \text{duration} = \text{duration} + (Lc[i][0][2] - Lc[i][0][1]);$ 36 $\begin{bmatrix} i = i + 1; \\ \text{J} \\ D.\text{append}(\text{some} Lp); D.\text{append}(\text{some} Lc); \\ \text{J} $	23	if $durationp + durationc <= \theta$ then		
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	24	/* If the duration of the collected segments is $\leq \theta$, save these segments		
$\begin{array}{ c c c c c } D.append(Lp); D.append(Lc); \\ \textbf{25 else} \\ \textbf{26} & /* If the duration of the selected segments is > \theta, split the time \\ (\theta = \theta/2) between the two groups of the collected segments, Lp and Lc \\ & */ \\ \textbf{27} & \theta = \theta/2; i = 0; duration = 0; someLc=[]; someLp=[] /* The lists \\ someLp and someLc include some of the segments from the Lp and Lc \\ lists, respectively */ \\ \textbf{28} & while i < length(Lp) do \\ \textbf{29} & & \text{if } duration \leq \theta \text{ then} \\ \textbf{30} & & & & \\ & & someLp.append(Lp[i]); \\ & & & duration = duration + (Lp[i][0][2] - Lp[i][0][1])); \\ \textbf{31} & & & \\ & & & i = 0; duration = 0; \\ \textbf{33} & \text{while } i < length(Lc) \text{ do} \\ \textbf{34} & & & \\ & & & \text{if } duration \leq \theta \text{ then} \\ \textbf{35} & & & \\ & & & & \\ & & & someLc.append(Lc[i]); \\ & & & & duration = duration + (Lc[i][0][2] - Lc[i][0][1]); \\ \textbf{36} & & \\ & & & & \\ & & & & li \ duration = duration + (Lc[i][0][2] - Lc[i][0][1]); \\ \textbf{36} & & \\ & & & & \\ & & & & li \ e i + 1; \\ \textbf{37} & & D.append(someLp); \ D.append(someLc); \\ \textbf{38} & \text{Return } D \end{array}$		in the list of the video segments for the Derivative narrative $D * /$		
<pre>25 else 26 /* If the duration of the selected segments is > θ, split the time ($\theta = \theta/2$) between the two groups of the collected segments, Lp and Lc */ 27 $\theta = \theta/2$; i = 0; duration = 0; someLc=[]; someLp=[] /* The lists someLp and someLc include some of the segments from the Lp and Lc lists, respectively */ 28 while i < length(Lp) do 29 lif duration $\leq \theta$ then 30 l someLp.append(Lp[i]); duration = duration + (Lp[i][0][2] - Lp[i][0][1])); 31 l i = i + 1; 32 i = 0; duration = 0; 33 while i < length(Lc) do 34 lif duration $\leq \theta$ then 35 l someLc.append(Lc[i]); duration = duration + (Lc[i][0][2] - Lc[i][0][1]); 36 l i = i + 1; 37 D.append(someLp); D.append(someLc); 38 Return D</pre>		D.append(Lp); D.append(Lc);		
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	25	else		
$ \begin{array}{c c} (\theta = \theta/2) \text{ between the two groups of the collected segments, } Lp \text{ and } Lc \\ */\\ \theta = \theta/2; i = 0; \text{ duration} = 0; \text{ some}Lc = []; \text{ some}Lp = [] /* \text{ The lists} \\ \text{ some}Lp \text{ and some}Lc \text{ include some of the segments from the } Lp \text{ and } Lc \\ \text{ lists, respectively } */\\ \textbf{28} \textbf{while } i < \text{ length}(Lp) \textbf{ do} \\ \textbf{29} \begin{bmatrix} \text{ if } \text{ duration} \leq \theta \text{ then} \\ \text{ some}Lp.\text{append}(Lp[i]); \\ \text{ duration} = \text{ duration} + (Lp[i][0][2] - Lp[i][0][1])); \\ \textbf{31} \begin{bmatrix} i = i + 1; \\ \text{32} \\ i = 0; \text{ duration} = 0; \\ \text{33} \textbf{while } i < \text{ length}(Lc) \textbf{ do} \\ \textbf{34} \begin{bmatrix} \text{ if } \text{ duration} \leq \theta \text{ then} \\ \text{ some}Lc.\text{append}(Lc[i]); \\ \text{ duration} = \text{ duration} + (Lc[i][0][2] - Lc[i][0][1]); \\ \text{ duration} = \text{ duration} + (Lc[i][0][2] - Lc[i][0][1]); \\ \end{bmatrix} \\ \textbf{36} \begin{bmatrix} \text{ some}Lc.\text{append}(Lc[i]); \\ \text{ duration} = \text{ duration} + (Lc[i][0][2] - Lc[i][0][1]); \\ \text{ some}Lc.\text{append}(\text{ some}Lp); D.\text{append}(\text{ some}Lc); \\ \textbf{38} \textbf{Return } D \end{array} \right$	26	/* If the duration of the selected segments is $> \theta$, split the time		
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		$(\theta = \theta/2)$ between the two groups of the collected segments, Lp and Lc		
$\begin{array}{c c c c c c c c c c c c c c c c c c c $		*/		
$someLp and someLc include some of the segments from the Lp and Lc lists, respectively */$ $mhile i < length(Lp) do$ $if duration \le \theta then$ $someLp.append(Lp[i]); \\ duration = duration + (Lp[i][0][2] - Lp[i][0][1]));$ $i = i + 1;$ $i = 0; duration = 0;$ $while i < length(Lc) do$ $if duration \le \theta then$ $someLc.append(Lc[i]); \\ duration = duration + (Lc[i][0][2] - Lc[i][0][1]);$ $i = i + 1;$ $D.append(someLp); D.append(someLc);$ $Return D$	27	$\theta = \theta/2; i = 0; duration = 0; someLc=[]; someLp=[] /* The lists$		
$lists, respectively */$ 28 while $i < length(Lp)$ do 29 if duration $\leq \theta$ then 30 if duration = duration + (Lp[i][0][2] - Lp[i][0][1])); 31 i = i + 1; 32 i = 0; duration = 0; 33 while $i < length(Lc)$ do 34 if duration $\leq \theta$ then 35 if duration $\leq \theta$ then 35 if duration = duration + (Lc[i][0][2] - Lc[i][0][1]); 36 if duration = duration + (Lc[i][0][2] - Lc[i][0][1]); 37 D.append(someLp); D.append(someLc); 38 Return D		someLp and someLc include some of the segments from the Lp and Lc		
28 while $i < length(Lp)$ do 29 30 30 31 31 31 31 31 31 31 31 31 32 <i>i = 0</i> ; duration = duration + (Lp[i][0][2] - Lp[i][0][1])); 33 34 35 35 36 36 37 37 38 Beturn D		lists, respectively */		
29 if duration $\leq \theta$ then 30 $arrow constant bound (Lp[i]);$ 31 $arrow constant bound (Lp[i][0][2] - Lp[i][0][1]));$ 31 $arrow constant bound (Lp[i][0][2] - Lp[i][0][1]));$ 32 $i = 0; duration = 0;$ 33 while $i < length(Lc)$ do 34 if duration $\leq \theta$ then 35 $arrow constant bound (Lc[i]);$ arrow constant bound (Lc[i]); arrow constant bound (Lc[i][0][2] - Lc[i][0][1]); 36 $arrow constant bound (Lc[i]);$ arrow constant bound (Lc[i]); arrow co	28	while $i < length(Lp)$ do		
30 $someLp.append(Lp[i]);$ $duration = duration + (Lp[i][0][2] - Lp[i][0][1]));$ 31 $i = i + 1;$ 32 $i = 0;$ duration = 0;33while $i < length(Lc)$ do34if duration $\leq \theta$ then35 $someLc.append(Lc[i]);$ $duration = duration + (Lc[i][0][2] - Lc[i][0][1]);$ 36 $i = i + 1;$ 37D.append(someLp); D.append(someLc);38Return D	29	if duration $\leq \theta$ then		
$ \begin{array}{ c c c c c } & & & & & & & & & & & & & & & & & & &$	30	someLp.append(Lp[i]);		
31 $i = i + 1;$ 32 $i = 0; duration = 0;$ 33while $i < length(Lc)$ do34if $duration \le \theta$ then35 $if duration \le \theta$ then36 $isomeLc.append(Lc[i]);$ $duration = duration + (Lc[i][0][2] - Lc[i][0][1]);$ 36 $i = i + 1;$ 37 $D.append(someLp); D.append(someLc);$ 38Return D		$ \qquad \qquad$		
32 $i = 0; duration = 0;$ 33while $i < length(Lc)$ do34if $duration \leq \theta$ then35 $\begin{bmatrix} someLc.append(Lc[i]); \\ duration = duration + (Lc[i][0][2] - Lc[i][0][1]); \\ i = i + 1; \\ 3736\begin{bmatrix} i = i + 1; \\ D.append(someLp); D.append(someLc); \\ 3838Return D$	31	i = i + 1;		
33while $i < length(Lc)$ do34if $duration \leq \theta$ then35 $ameLc.append(Lc[i]);$ $duration = duration + (Lc[i][0][2] - Lc[i][0][1]);$ 36 $i = i + 1;$ 37 $D.append(someLp); D.append(someLc);$ 38Return D	32	i = 0; duration = 0;		
34if $duration \le \theta$ then35 $amelc.append(Lc[i]);$ $duration = duration + (Lc[i][0][2] - Lc[i][0][1]);$ 36 $i = i + 1;$ 37 $D.append(someLp); D.append(someLc);$ 38Return D	33	while $i < length(Lc)$ do		
35 $someLc.append(Lc[i]);$ $duration = duration + (Lc[i][0][2] - Lc[i][0][1]);$ 36 $i = i + 1;$ 37 $D.append(someLp); D.append(someLc);$ 38Return D	34	if duration $\leq \theta$ then		
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	35	someLc.append(Lc[i]);		
36 $i = i + 1;$ 37D.append(someLp); D.append(someLc);38Return D		$ \qquad \qquad$		
 37 D.append(someLp); D.append(someLc); 38 Return D 	36			
38 Return D	37	D.append(someLp); D.append(someLc);		
	38	38 Return D		

3.6. VIDEO NARRATIVES FOR LEARNING (VIN-L) FRAMEWORK

The output from the algorithm is the video narratives which include a sequence of video segments ordered based on the definition of the D. These videos narratives with the segments only are not sufficient for learning. For this purpose, we augment them with text following the template in Figure 3.8 to be suitable for learning.

Text purpose	Description
1. Narrative Introduction	[Domain name] This is a collection of video segments from YouTube videos with [video type: tutorial/ patient stories, etc.]. They have been automatically segmented, characterised and linked to domain concepts. The videos in this page have been linked via [linking type] which means: To become aware of a specific concept from a topic, you will explore videos about its parent concept which is more generic. Become aware of the concept [specific concept name] by exploring its parent concept [generic concept name] from the topic [topic name]
2. Segments Introduction	Each segment will start from a time point in the video and will stop when the corresponding concepts are covered. You can re-watch the segment or watch other parts of the video, if you wish so.
3. Segment Characterisation	The next video segment mentions [generic concept name] (related to [topic name]). The segment is taken from a YouTube video with ID= [video Id] and starts at [segment's start time in the video] with duration of [the difference between segment's start time and end time].
4. Video Segment footage	[A video segment that will start from its starting time and stop at the end of its duration as decided by the characterisation algorithms. There are controls to: maximise segment's size, volume up and down, and skip forward and backword to watch any part of the video if needed.]
Parts 3 and 4 are repeated for group 2 of segments that introduces the specific concept	
5. Conclusion	You watched video segments about [generic concept name] which is a generic concept that includes [specific concept name]. These concepts are related to [topic name]. The segments were automatically extracted from popular YouTube videos related to the [domain name], using algorithms for segmentation, characterisation and linking based on a theory for concept learning. We hope you found the collection of video segments helpful.

Figure 3.8: The template used for generating the augmented Derivative Video Narrative.

Super-Ordinate Linking. It helps to learn a generic concept by exploring its specific concept(s) first to be able to link them together. The design of the Super-Ordinate algorithm is based on its definition in (Section 3.4–Definition 9). Hence, referring to the taxonomy structure presented in Fig. 3.7, to learn the generic concept C_{12} , the Super-Ordinate Linking algorithm first introduces the specific concept(s) c_{12_1} and/or c_{12_2} via a group of video segments that has it/them as a focus and places these segments at the beginning of the Super-Ordinate video Narrative (S). By doing this, these concepts are in the cognitive structure and ready to bind to its/their generic concept C_{12} . Hence, the Super-Ordinate algorithm select the video segments that focus on the generic concept C_{12} and add these segments to the S.

The input to the algorithm is the list of the characterised video segments and a list of the taxonomy concepts' paths. A path of a concept c_i , as defined in (Section 3.4–Definition 3), is: $PC(c_i) = \langle c_i, C_j, T_i \rangle$. To learn a generic concept C_j , the algorithm first iterate through the list of the characterised video segments and match their focus concept(s) with the concepts which are belong to the C_j . Hence, if the FC of a segment includes the c_i (the specific concept of C_j) then this segment is selected for the first group in the S. Whilst, if the FC of the segment includes the concept C_j (the generic concept to be learnt) then this segment is selected for the S does not exceed six minutes. If the duration of the selected segments for the S does not exceed six minutes. If the duration is $\leq = 6$ minutes then we include all the collected segments in S; otherwise, we select some of the collected segments from both groups to prevent exceeding the 6 minutes duration of the S. Algorithm 5 includes the steps we followed to generate the Super-Ordinate narratives and the tracing of this algorithm is in (Appendix E). Algorithm 5: VIN-L–Algorithm (Super-Ordinate Linking–S)

- **Data:** List of the aggregated video segments *LAgg*, which is the output of Algorithm 3 where each segment has: its information at position 0, the list of the Focus Topics at position 1, the the list of the Focus Concepts at position. List of taxonomy concepts' paths *PC* /* *Each concept c from the taxonomy is represented with a list of concepts where c is the first concept and the remaining concepts are its parents* */. The generic concept to be learnt *g*.
- **Result:** S: List of video segments to learn the generic concept g by introducing first video segments focusing on its specific concepts.
- 1 S=[]; i = 0; j = 0; sc = [] /* List to save the concepts that belong to the generic concept <math>g */
- 2 $\theta = 6$ /* Set 6 minutes as a max segment duration */
- **3** Lg =[] /* List of the segments that have the generic concept g as the focus concept in their characterisation */
- 4 Lc = [] /* List of the segments that have the concepts sc of the generic concept g as the focus concepts in their characterisation */
- 5 while j < length(PC) do
- 6 /* Finding the path in PC where the generic concept g is the parent of other concepts */
- 7 **if** PC[j][1] = g **then**
- **s** if PC[j][0] not in sc then

10
$$j = j + 1;$$

9

11 while i < length(LAgg) do

12 /* Find the characterised video segment where its second element include any of sc; otherwise check if the second element includes g */

- 13 for concept in sc do
- 14 if $concept \in LAgg[i][2]$ then
- 15 $\ \ Lc.append(LAgg[i])$
- 16 if $(g \in LAgg[i]/2])$ and (LAgg[i] not in Lc) then
- 17 Lg.append(LAgg[i])
- 18 i = i + 1;

_

16	/* Ensure the duration of the collected segments does not exceed θ */	
17	i = 0; duration $c = 0 / *$ Save the duration of the segments that have c in	
	its Focus concept FC */	
18	while $i < length(Lc)$ do	
19	durationc = durationc + (Lc[i][0][2] - Lc[i][0][1]); i = i + 1;	
20	i = 0; duration $g = 0 /*$ Save the duration of the segments that have g in	
	its Focus concept FC */	
21	while $i < length(Lg)$ do	
22	urationg = durationg + (Lg[i][0][2] - Lg[i][0][1]); i = i + 1;	
23	if $durationc + durationg <= \theta$ then	
24	/* If the duration of the collected segments is $\leq \theta$, save these segments	
	in the list of the video segments for the Super Ordinate narrative $S * /$	
25	S.append (Lc) ; S.append (Lg) ;	
26	else	
27	/* If the duration of the selected segments is $> \theta$, split the time	
	$(\theta = \theta/2)$ between Lg and Lc */	
28	$i = 0; duration = 0; \theta = \theta/2; someLc=[]; someLg=[] /* The lists$	
	someLc and someLg include some of the segments from the Lc and Lg	
	lists, respectively */	
29	while $i < length(Lc)$ do	
30	if duration $\leq \theta$ then	
31	someLc.append(Lc[i]);	
32	i = i + 1;	
33	i = 0; duration = 0;	
34	while $i < length(Lg)$ do	
35	if duration $\leq \theta$ then	
36	someLg.append(Lg[i]);	
	duration = duration + (Lg[i][0][2] - Lg[i][0][1]);	
37		
38	\mathbf{s} S.append(someLc); S.append(someLg);	
39	Return S	

3.6. VIDEO NARRATIVES FOR LEARNING (VIN-L) FRAMEWORK

The output from the algorithm is the video narratives which includes a sequence of video segments ordered based on the definition of the S. These videos narratives with the segments only are not sufficient for learning. For this purpose, we augment them with text following the template in Figure 3.9 to be suitable for learning.

Text purpose	Description
1. Narrative Introduction	[Domain name] This is a collection of video segments from YouTube videos with [video type: tutorial/ patient stories, etc.]. They have been automatically segmented, characterised and linked to domain concepts. The videos in this page have been linked via [linking type] which means: To become aware of a generic concept from a topic, you will explore videos about its children concept which is more specific. Become aware of the concept [generic concept name] from [topic name] by exploring more specific concepts [specific concept(s) name(s)].
2. Segments Introduction	Each segment will start from a time point in the video and will stop when the corresponding concepts are covered. You can re-watch the segment or watch other parts of the video, if you wish so.
3. Segment Characterisation	The next video segment mentions [specific concept name] (related to [topic name]). The segment has been taken from a YouTube video with ID= [video Id] and starts at [segment's start time in the video] with duration of [the difference between segment's start time and end time].
4. Video Segment footage	[A video segment that will start from its starting time and stop at the end of its duration as decided by the characterisation algorithms. There are controls to: maximise segment's size, volume up and down, and skip forward and backword to watch any part of the video if needed.]
Parts 3 and 4 are repeated for group 2 of segments that introduces the generic concept	
5. Conclusion	You watched video segments about [specific concept name] which is part of [generic concept name]. These concepts are related to [topic name]. The segments were automatically extracted from popular YouTube videos related to the [domain name], using algorithms for segmentation, characterisation and linking based on a theory for concept learning. We hope you found the collection of video segments helpful.

Figure 3.9: The template used for generating the augmented Super-Ordinate Video Narrative.

Correlative Linking. It helps to learn a specific concept by exploring its similar concept(s) first to be able to link them together. The design of the Correlative algorithm is based on its definition in (Section 3.4), Definition 10. Hence, referring to the taxonomy structure presented in Figure 3.7, to learn the specific concept c_{i1_2} , the Correlative algorithm select the video segments that focus on the concepts c_{i1_3} and c_{i1_1} , respectively, and place them in the Correlative Video Narrative (CR). Notice that these concepts (c_{i1_3}, c_{i1_1}) are similar to the specific concept c_{i1_2} and all of these concepts $\in C_{i1}$. By doing this, (c_{i1_3}, c_{i1_1}) are

in the cognitive structure and ready to be linked to their similar specific concept c_{i1_2} . Then, the Correlative algorithm select another group of video segments that focus on the specific c_{i1_2} and add these segments to the CR.

The input to the algorithm is the list of the characterised video segments and a list of the taxonomy concepts' paths. A path of a concept c_i , as defined in (Section 3.4), Definition 3 is: $PC(c_i) = \langle c_i, C_j, T_i \rangle$. To learn a specific concept c_i that has a generic concept C_j , the algorithm first iterate through the list of the concepts paths PC(C) to identify all the specific concepts that have C_j as their generic concept, e.g. c_{i-1} and c_{i+1} . Then, the algorithm iterate through all the characterised video segments to select the segments that have the specific concepts c_{i-1} and c_{i+1} as a focus. These segments are selected as first group of segments in the CR. The same process is followed to select the characterised video segments that have the specific concept c_i as a focus and place them as the second group of segments in CR. The duration of the selected segments for the CR does not exceed six minutes. If the duration is $\leq = 6$ minutes then we include all the collected segments in CR; otherwise, we select some of the collected segments from both groups to prevent exceeding the 6 minutes duration of the CR. Algorithm 6 includes the steps we followed to generate the Correlative narratives and the tracing of this algorithm is in (Appendix F).

Algorithm 6: VIN-L–Algorithm (Correlative Linking–CR) **Data:** List of the aggregated video segments *LAgg*, which is the output of Algorithm 3 where each segment has: its information at position 0, the list of the Focus Topics at position 1, the the list of the Focus Concepts at position 2. List of taxonomy-concepts' paths PC /*Each concept c from the taxonomy is represented with a list of concepts where c is the first concept and the remaining concepts are its parents */. The name of the concept to be learnt c and its parent concept name p. **Result:** CR: List of video segment to learn the specific concept c by introducing video segments that focus on similar concepts to cwhere c and its similar concepts have the same parent concept p. 1 $CR = []; i = 0; j = 0; \theta = 6 / * Set 6 minutes as a max segment duration */$ **2** s=[] /* List of the names of the similar concepts to c */**3** Ls=[] /* List of the segments that have any of concepts in the list s as thefocus concept in their characterisation */ 4 Lc = [] /* List of the segments that have c as the focus concept in their characterisation */ 5 while j < length(PC) do /* Finding the path in PC for the parent concept p */6 if PC[j]/[1] = p then 7 if $PC[j][0] \neq c$ then 8 /* Do not collect the concept c as similar to itself */ 9 s.append(PC[j][0]) /* Finding the specific concepts which are similar to c as p is their immediate parent */ j = j + 1;10 11 while i < length(LAqq) do /* Find the characterised video segment where its second element 12includes s; otherwise check if the second element includes c * / ck = 0 /* Variable used to loop through the s list /* 13 while k < length(s) do 14 if $(s/k) \in LAgg[i]/2]$ and (LAgg[i] not in Ls) then 15 Ls.append(LAgg[i]);16 k = k + 1;17 if $c \in LAgg[i]/2$ and LAgg[i] not in Ls then $\mathbf{18}$ Lc.append(LAgg[i]);19

16	/* Ensure the duration of the collected segments does not exceed θ */		
	i = 0; durations = 0 /* Save the duration of the segments that have s in		
	their Focus concept $FC * /$		
17	while $i < length(Ls)$ do		
18	$\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ $		
19	i = 0; duration $c = 0 /*$ Save the duration of the segments that have c in		
	their Focus concept $FC * /$		
20	while $i < length(Lc)$ do		
21	durationc = durationc + (Lc[i][0][2] - Lc[i][0][1]); i = i + 1;		
22	if $durations + durationc \le \theta$ then		
23	/* If the duration of the collected segments is $\leq \theta$, save these segments		
	in the list of the video segments for the Correlative narrative CR $*/$		
24	CR.append(Ls); CR.append(Lc);		
25	else		
26	/* If the duration of the selected segments is $> \theta$, split the time		
	$(\theta = \theta/2)$ between Ls and Lc */		
27	$i = 0; duration = 0; \theta = \theta/2; someLs = []; someLc = [] /* The lists$		
	someLs and someLc include some of the segments from the Ls and Lc		
	lists, respectively */		
28	while $i < length(Ls)$ do		
29	if duration $\leq \theta$ then		
30	someLs.append(Ls[i]);		
31			
32	i = 0; duration = 0;		
33	while $i < length(Lc)$ do		
34	if duration $\leq \theta$ then		
35	someLc.append(Lc[i]);		
36			
37	CR.append(someLs); CR.append(someLc);		
38	Return CR		

3.6. VIDEO NARRATIVES FOR LEARNING (VIN-L) FRAMEWORK

The output from the algorithm is the video narratives which includes a sequence of video segments ordered based on the definition of the CR. These videos narratives that include the segments only are not sufficient for learning. For this purpose, we augment them with text following the template in Figure 3.10 to be suitable for learning.

Text purpose	Description
1. Narrative Introduction	[Domain name] This is a collection of video segments from YouTube videos with [video type: tutorial/ patient stories, etc.]. They have been automatically segmented, characterised and linked to domain concepts. The videos in this page have been linked via [linking type] which means: To become aware of a specific concept from a topic, you will explore videos about a similar concepts. Become aware of the concepts similar to [specific concept name] from topic [topic name].
2. Segments Introduction	Each segment will start from a time point in the video and will stop when the corresponding concepts are covered. You can re-watch the segment or watch other parts of the video, if you wish so.
3. Segment Characterisation	The next video segment mentions [similar concept names] (related to [topic name]). The segment has been taken from a YouTube video with ID= [video Id] and starts at [segment's start time in the video] with a duration of [the difference between segment's start time and end time].
4. Video Segment footage	[A video segment that will start from its starting time and stop at the end of its duration as decided by the characterisation algorithms. There are controls to: maximise segment's size, volume up and down, and skip forward and backword to watch any part of the video if needed.]
Parts 3 and 4 are repeated for group 2 of segments that introduces the specific concept	
5. Conclusion	You watched video segments about [similar concepts names] which are similar to [specific concept name] related to [topic name].
	The segments were automatically extracted from popular YouTube videos related to the [domain name], using algorithms for segmentation, characterisation and linking based on a theory for concept learning. We hope you found the collection of video segments helpful.

Figure 3.10: The template used for generating the augmented Correlative Video Narrative.

Combinational Linking. It helps to learn two topics together by exploring the relationship between some of their concepts as they are mentioned together in the same video segment. The design of the Combinational Algorithm is based on its definition in (Section 3.4), Definition 11. Hence, to learn the topics FT_i and FT_j , the Combinational algorithm select a group of characterised video segments, from the input set of segments, that have the focus concepts Fc_i and Fc_j where $Fc_i \in FT_i$ and $Fc_j \in FT_j$.

For example, referring to the taxonomy structure presented in Figure 3.7, to learn the relationship between two topics T_1 and T_2 , the Combinational algorithm

include video segments that focus on the concepts, e.g. C_{12} and C_{22} mentioned together in the same video segment and they belong to the topics T_1 and T_2 , respectively. The full details of the Combinational algorithm are presented below and the tracing of this algorithm is in (Appendix G).

Algorithm 7: VIN-L–Algorithm (Combinational Linking–CO)		
Data: List of the aggregated video segments <i>LAgg</i> , which is the output of		
Algorithm 3 where each segment has: its information at position 0,		
the list of the Focus Topics at position 1, the list of the Focus		
Concepts at position 2. The name of the topics to be learnt t_1 and		
t_2 .		
Result: CO: List of video segments to learn the topics t_1 and t_2 by		
introducing video segments focusing on these topics.		
1 $CO=[]; i=0;$		
2 $j = 0;$		
3 $\theta = 6 /*$ Set 6 minutes as a max segment duration $*/$		
4 $Lt = [] /* List of the segments that have the topics t_1 and t_2 as the focus$		
topics in their characterisation */		
5 while $i < length(LAgg)$ do		
6 /* Find the characterised video segments where its first element		
includes both t_1 and $t_2 */$		
7 if $t_1 \in LAgg[i][1]$ and $t_2 \in LAgg[i][1]$ then		
8 $Lt.append(LAgg[i]) /* Each collected segment has, in its second$		
element, focus concepts that belong to t_1 and $t_2 */$		
9 $\lfloor i = i + 1;$		
10 /*Ensure the duration of the collected segments does not exceed θ */		
11 $i = 0$; duration $t = 0$ /* Save the duration of the collected segments that		
have t_1 and t_2 in their Focus Topics FT */		
12 while $i < length(Lt)$ do		
13 $durationt = durationt + (Lt[i][0][2] - Lt[i][0][1]);$		
14 $\lfloor i = i + 1;$		
15 if $durationt \leq \theta$ then		
16 $CO = Lt;$		
17 else		
18 /* If the duration of the collected segments is > θ , select some of the collected segment where their duration do not exceed $\theta */$		
i = 0		
y = 0; y = 0; duration = 0;		
while $i < length(Lt)$ and duration $\leq \theta$ do		
$22 \qquad \bigcirc CO.append(Lt[i]):$		
23 $duration = duration + (Lt[j][0][2] - Lt[j][0][1]):$		
24 $j = j + 1;$		
$= \frac{1}{25} \operatorname{Beturn} CO$		

3.6. VIDEO NARRATIVES FOR LEARNING (VIN-L) FRAMEWORK

The output from the algorithm is the video narratives which includes a sequence of video segments ordered based on the definition of the CO. These videos narratives with the segments only are not sufficient for learning. For this purpose, we augment them with text following the template in Figure 3.11 to be suitable for learning.

Text purpose	Description
1. Narrative Introduction	[Domain name] This is a collection of video segments from YouTube videos with [video type: tutorial/ patient stories, etc.]. They have been automatically segmented, characterised and linked to domain concepts. The videos in this page have been linked via [linking type] which means: You will explore videos which link concepts from two topics. Link the topics [topic 1 name] and [topic 2 name].
2. Segments Introduction	Each segment will start from a time point in the video and will stop when the corresponding concepts are covered. You can re-watch the segment or watch other parts of the video, if you wish so.
3. Segment Characterisation	You will watch video segment mentions [specific concept(s) name(s)] (related to [topic 1 name]) and [specific concept(s) name(s)] related to [topic 2 name]. The segment has been taken from a YouTube video with ID= [video Id] and starts at [segment's start time in the video] with a duration of [the difference between segment's start time and end time] seconds.
4. Video Segment footage	[A video segment that will start from its starting time and stop at the end of its duration as decided by the characterisation algorithms. There are controls to: maximise segment's size, volume up and down, and skip forward and backword to watch any part of the video if needed.]
Parts 3 and 4 are repeated for all the video segments included in the CO	
5. Conclusion	You watched video segments that link [topic 1 name] and [topic 2 name].
	The segments were automatically extracted from popular YouTube videos using algorithms for segmentation, characterisation and linking based on a theory for concept learning. We hope you found the collection of video segments helpful.

Figure 3.11: The template used for generating the augmented Combinational Video Narrative.

3.6.3 Output

These algorithms are implemented all at the same time. The output from each subsumption linking algorithm is a set of video narratives and each narrative is a set of video segments that have been linked together following Ausubel's Subsumption Theory. The generated narratives need to be augmented with text to make them more sufficient for learning. Hence, generated video narratives are augmented with text as presented in the text templates in Figures 3.8, 3.9, 3.10 and 3.11. Accordingly, each video narrative begins with an introduction identifying the

domain, focus topic, the type of linking and the focus concept. Then, it presents the video segments where each segment is preceded by its characterisation. The characterisation of each segment is the focus topic and concept, segmentId, the start time within the video and the segment duration in seconds. At the end of each narrative, there is a conclusion statement reminding the users of the focus topic and concepts illustrated in the video segments they have watched. Each video narrative has a duration that ≤ 6 minutes based on the recommendation in (Brame, 2016). The augmented narratives have been embedded in four groups of web pages generated automatically, each group correspond to the linking type used to generate them.

Examples of video narratives are presented in the evaluation studies (see Chapter 5 and Chapter 6). The quality and the perceived usefulness of these narratives has been evaluated when used to support domain learning.

3.7 Domains Used in this Thesis

In order to assess the applicability and to validate the frameworks, we need to apply them in practical domains. In this thesis, we have chosen two representative domains which are both in the area of soft skill learning. Soft skill learning is the main area to apply our work where videos are used for informal soft skill learning. Soft skills refers to a collection of inter personal skills and personal attributes (Dixon et al., 2010). Interpersonal skills represents the ability to communicate with others and being able to present and convey the ideas clearly. This will enhance the personal attributes and enhancing the self esteem to grab an opportunity and start actions (Vasanthakumari, 2019). Learning soft skills for students to prepare them for professional environments (Mainga et al., 2022) and to help them sustaining long term employability which requires more than a formal training. It requires informal support, self-study and self analysis which happens outside the professional environment (Crosbie, 2005; Tseng et al., 2019). To aid the informal support to learn soft skills, online learning and using videos for learning plays a key role in that (Cinque, 2017). Videos for learning help users to pay attention to soft skills details and the relation ship between them to enhance their self

regulation and motivation (Conkey et al., 2013). Learning such skills requires time and resources management to ensure good quality of learning materials (Crosbie, 2005). On the other hand, teaching soft skills requires wide knowledge of pedagogical knowledge to decide the best teaching and a knowledge in the latest ICT resources to select the best methods (Caeiro-Rodríguez *et al.*, 2021). In our work we focus on two domains. One domain is typically used in academic context, e.g. within a software engineering course, and the other domain is informal learning in every day living context, e.g. a health domain. In both domains we have tapped into international collaboration. Within the first domain, which is Giving Pitch Presentations, we are part of the AVW project which involves an international collaboration with the University of Canterbury. Within the second domain which is Health Related Quality of Life Awareness (HRQLA), our work is linked to the European project (InAdvance) which is focusing on palliative care support. In this section, we present an overview of the two domains which are further applied in Chapters 4 and 5 (Giving Pitch Presentations) and in chapter 6 (HRQLA).

3.7.1 Giving Pitch Presentations

Giving Pitch Presentations represents a key transferable skill that students should have for their academic or career purposes (Van Emden & Becker, 2017). They will be able to present and explain their work to large audiences, share their ideas, and be prepared for interviews and professional meetings. There are different collaboration applications like (Google Drive, Mendeley, Office 365, Dropbox, text apps as Whats App, Snapchat, etc.) to learn presentation skills. Adding to that, using social networks and messaging apps becomes a popular tools for sharing information and learning resources (Maican *et al.*, 2021). This resulted in having more self-efficacy learners who engage in online and collaborative learning (Alshare & Hindi, 2004) to boost up their presentation skills like: presenter engaged audience, hand gestures, eye gestures, verbal communication, voice, eye contact and presenter presence (Di Gangi *et al.*, 2017).

In this research, we have decided to aid students in learning how to pitch their presentation using videos by implementing our frameworks (VISC-L and VIN-L) in this domain. This decision has also been based on the successful design and usage of the video based learning platform AVW (Active Video Watching) (Lau et al., 2016; Mitrovic et al., 2016). AVW is a platform that allows people to watch videos and annotate them interactively. It can be set up using different domains, e.g. doctor training, introduction to university, empathy training and presentation skills. AVW has been designed by the research team in the University of Leeds, University of Canterbury and University of Adelaide focusing on helping students build up skills on delivering pitch presentations to a wide spectrum of audiences (Lau et al., 2016; Mitrovic et al., 2016). AVW is used by both teachers and learners. Teachers select a set of videos and define aspects to draw a learner's attention to, targeting presentation skills. For Giving Pitch Presentations, the AVW instantiation includes 4 YouTube tutorial videos on giving presentations. The aspects defined by teachers for these videos to help students be reflective are: "I am rather good at this", "I did/saw this in the past", "I didn't realise I wasn't doing it" and "I like this point". While the aspects defined by teachers to drag the students' attention to main concepts in the tutorials are: "Delivery", "Speech", "Structure" and "Visual aids". Students can be active with AVW while watching videos by commenting individually in the Personal Space using the aspects defined by the teacher, see Figure 3.12. Students' comments are recorded by the system and once they are approved by the teachers, they can be shared and browsed in the Social Space and can be rated by other students, see Figure 3.13. Two evaluation studies (Mitrovic et al., 2016, 2017) have been conducted to test the impact of using AVW spaces (Personal and Social) on students' learning. The results revealed that using the AVW spaces improved students' engagement in constructive learning. Inspired by this result and to further support learning presentation skills while watching videos, Dimitrova et al. (2017) showed that adding nudges is an opportunity for intervention to enhance engagement while watching videos. In Mohammadhassan et al. (2022) high quality comments written by students have been added as quality nudges to encourage students to write better comments to enhance their engagement while watching videos.

However, within the presentation skills domain, AVW studies have used four tutorials and four example videos. It is not sufficient for learning as we can not explore the domain deeply. In reality, learners can access much broader



Figure 3.12: AVW-Personal Space (Mohammadhassan et al., 2022).

Review video comments for "TUTORIAL 1: How to Give an Awesome (PowerPoint) Presentation"



Figure 3.13: AVW- Social Space (Mohammadhassan et al., 2022).

space of videos (using e.g. YouTube) which cause some challenges. For example, the amount of time needed to watch the videos to decide the most relevant ones, and the inabilities to recognise the key domain points regarding the lack of experience or domain knowledge. For that purpose, we have developed our frameworks to be able to work automatically with more videos that are broader and can cover more domain aspects. Our work includes automating the process of: collecting domain related videos from social learning platforms (e.g. YouTube), segmenting, characterising and linking the videos to aid learning. Accordingly, we first collect videos related to giving pitch presentations using the taxonomy of this domain to design a search schema. Secondly, we segment the collected videos and characterise their transcript using the domain taxonomy by implementing the VISC-L framework described in (Section 3.5). Finally, we use both the characterised segments (VISC-L output) and the domain taxonomy to link the related segments following a pedagogical theory by implementing the VIN-L framework described in (Section 3.6). This way, learners will be able to focus and learn specific related key points extracted and linked from different videos without needing to watch full videos.

The use of this domain in our work will be fully described in detail in Chapters 4 and 5.

3.7.2 Health Related Quality of Life Awareness (HRQLA)

To show the broad application of our approach and to validate the generality of VISC-L and VIN-L, we will consider a second domain for soft skills learning. For this purpose, we have chosen the HRQLA. Health related quality of life as mentioned by ¹ is the "individuals' perceptions of their position in life in the context of the culture and value systems in which they live and in relation to their goals, expectations, standards and concerns." This domain shows the link between the physical, psychological and social health needs used to guide specialists and health agencies to meet them if they are reported by patients, carers or specialists. The aspects of the health related quality of life domain have been proven to have a direct impact on patients with different needs. For example, patients with cancer who have left the hospital (Sitlinger & Zafar, 2018) need regular checks to maintain their health stability. In addition, elderly people need support from health authorities to maintain health circumstances and environments (Marques *et al.*, 2019). On the other hand, being aware of health related quality of life can help in preventing the spread of pandemics and epidemic diseases. For

¹https://iris.who.int/bitstream/handle/10665/77932/WHO_HIS_HSI_Rev.2012.03_ eng.pdf?sequence=1[Accessed: August 2022]

example, when patients diagnosed with COVID 19 leave hospitals, they need to be aware of the possible weakness symptoms that might affect them (Talman *et al.*, 2021); especially with elderly people. This is to prevent any impairments in their functionality (Walle-Hansen *et al.*, 2021).

Providing health related services and educating patients, carers, relatives and friends are essential in maintaining a positive health related quality of life. For example, providing educational components (Koekenbier *et al.*, 2016), exercise sessions and nurse interventions, called 'non-pharmacological interventions' showed a positive impact on a patients' health with different diseases, e.g. breast-cancer (Oei *et al.*, 2021).

Notwithstanding, the concepts of the health related quality of life should be clearly identified and differentiated from general quality of life. This is because general quality of life is a broad concept while health related quality of life focuses primarily on peoples' health and all the factors that are associated with it. Only 6% of researchers distinguish between the two domains (Haraldstad *et al.*, 2019). The identification of these concepts will make them easy to be observed and implemented by experts (Post, 2014).

Identifying health related quality of life concepts will support educating people about them. The recommendations from most researchers in the health sector state that educating patients (Ciasullo *et al.*, 2022) to be aware of their health needs will have a positive impact on their different requirements. For example: being aware of the key elements behind COPD exacerbation to avoid them (Hurst *et al.*, 2020), stopping the risk of menopausal syndrome on menopausal women by providing them with Multidisciplinary Health Education based on Lifestyle Medicine (MHELM) Li *et al.* (2023), and to prevent the risk of developing lower back pain in certain populations with special socio-demographic and levels of education parameters (Járomi *et al.*, 2021).

Motivated by the findings from the aforementioned research, our work aims to raise awareness of health related quality of life of patients with respiratory illness. There are different learners (users) in this domain. First, patients, carers, relatives, friends will use this domain to learn the main aspects of health which are: PHYSICAL HEALTH, PSYCHOLOGICAL HEALTH, SOCIAL RELATION SHIP, PERSONAL VALUES AND BELIEFS, ENVIRONMENT and LEVEL OF INDEPENDENCE. This will help them to measure the factors related to their health and enables them to communicate with professionals to maintain better health outcomes. From a professional's side, knowing these patients' health related quality of life aspects will aid them in making clinical decisions and to guide their researches.

Accordingly, we have firstly unified the main health related quality of life concepts identified by different researches by designing the domain taxonomy (the full design process is in Section 6.2). Secondly, we used this taxonomy to implement the search schema we defined in (Section 3.5) to collect domain related videos from YouTube. Thirdly, the transcript of the collected videos has been segmented, characterised and linked to generate video narratives as illustrated in (Sections 3.5 and 3.6). The generated narratives are presented to users as web pages that include video segment URLs to focus on the key domain concepts and to raise awareness of health related quality of life aspects.

3.8 Summary

In this chapter, we have introduced our research methodology, which allow us to address the main research question 'How to generate good quality narratives that can be used to support perceiving domain learning?'. These narratives aim to help learners focus on key domain concepts and understand the link between them. Our research is outlined in (Section 3.2) where we illustrated the main generic frameworks we designed to achieve our goal. The design of the frameworks has been underpinned by several theoretical models introduced in (Section 3.3); including the Text Tiling approach (used to guide segmenting and aggregating of videos), the Thematic Progression (used to guide the aggregating of video segments for learning), and Subsumption linking Theory (to guide the generating of video narratives).

The main definitions of the terms used throughout this thesis and explanations of the main frameworks are presented in (Section 3.4 Preliminaries).

To produce a generic work, we have designed two frameworks for video collecting, segmenting, characterising and linking. These frameworks with their associated algorithms have been illustrated in detail in (Sections 3.5 and 3.6).
Finally, to prove the generality of our methodology, we have selected two domains to implement the designed frameworks with their algorithms. These domains focus on soft skills (Giving Pitch Presentations and Health Related Quality of Life Awareness). These domains' full descriptions have been presented in (Section 3.7).

In the following chapters we present the implementation of VISC-L and VIN-L in the two chosen domains, and evaluate the validity of our approach with experts from the two domains and with advanced learners with the first domain.

Chapter 4

Implementation and Validation of VISC-L in the Presentation Skills Domain

4.1 Introduction

This chapter describes how the VISC-L framework, which is presented in Section 3.5, is implemented in our prime domain 'Giving Pitch Presentations' in order to validate VISC-L and assess its applicability in a practical context. The VISC-L implementation includes collecting, segmenting and characterising domain related videos to identify the key domain concepts presented in the videos. Having done so, we are be able to generate the video narratives from these video segments (our ultimate research goal) as described in (Chapter 5).

By implementing and evaluating VISC-L, this chapter will answer RQ1 and RQ2 (see Chapter 1).:

RQ1: How to automatically generate good quality characterisations of video segments that can be used for learning?

RQ2: Does the characterisation of video segments help in identifying and focusing on the key domain concepts?

In order to answer these research questions, we implement VISC-L in the prime domain (Section 4.4) and evaluate with experts (Section 4.5). Since our approach is knowledge driven, the implementation of VISC-L requires a domain taxonomy which is used to collect domain related videos (as described in Section 3.5.1) and to implement the VISC-L framework (as defined in Section 3.5). For the domain "Giving Pitch Presentations", we utilise the ontology designed by Abolkasim (2019), which is described in Section 4.2, and use it to collect domain related videos (described in Section 4.3) and to implement VISC-L (described in Section 4.4). To evaluate the video segment characterisation output of VISC-L we compare it with the Google video segment characterisation output (described in Section 2.3.1). The Google video segment characterisation has been selected because it was launched during this research and represented the state-of-the-art in video characterisation at the time the evaluation was conducted. The evaluation study (presented in Section 4.5) was done with domain experts, including people with experience in giving presentations. This allows validation of the characterisation and its potential to support learning.

4.2 Presentation skills ontology

'Giving Pitch Presentations' is the prime domain we adopted in this thesis to generate the video narratives. It represents a transferable skill that can assist learners in transmitting their message or to convince others of their ideas (Di Gangi *et al.*, 2017; Maican *et al.*, 2021). The computational representation of this domain is presented via an ontology designed by Abolkasim (2019) where the domain terms are extracted from the studies conducted in Abolkasim (2019) to extend the already built Body Language ontology terms. To convert the domain terms into an ontology, the semi-automatic ontology engineering approach had been utilised and the resulting ontology is called Presentation Skills Ontology (PreSOn)¹. To ensure the validity of this ontology, (Abolkasim, 2019) invited three experts from the institutional study skills training team at the University of Leeds. They have been asked to either agree or disagree with the selected concept, or suggest

¹https://archive.researchdata.leeds.ac.uk/982/[Accessed: June 2022]

new concepts. According to their feedback, the ontology has been improved and the final number of concepts has been agreed. The main Topics T of PreSOn (which are the top classes in the ontology) and the number of concepts (classes and individuals) per topic (each topic has several concepts linked to it, see Figure 3.3) is as follows: DELIVERY- 99, STRUCTURE- 67, VISUAL AID- 93 and PRESENTATION ATTRIBUTE- 27.

The taxonomy of PreSOn has been used successfully to measure the diversity profiling in a social cloud using the meta data of online objects and the user profiles who used these objects and wrote associated comments. According to the successful implementation of this ontology, we have decided to use it in our work and improve it by increasing the number of concepts to be 398 concepts to be used to implement VISC-L and VIN-L, as we illustrate in this chapter. Figure 4.1 shows an example of the the ontology main topics and Figure 4.2 shows the structure of the topic DELIVERY and its main concepts. The improved version of the PreSOn ontology, after implementing VISC-L framework, can be accessed using our GitHub account ¹.



Figure 4.1: Giving Pitch Presentations main topics based on the PreSOn ontology.

4.3 Data Collection

The main input data required to implement the VISC-L framework is the domain related videos. Any pre-selected domain related videos can be used (Mohammed & Dimitrova, 2020) or they can be collected automatically from available social platforms (like YouTube). In this section, we semi-automatically collect the

¹Presentation Skills Domain /Domain Ontology



Figure 4.2: Part of the PreSOn ontology representing the hierarchy of the topic DELIVERY and its concepts.

domain related videos by implementing the search schema described in (Section 3.5.1) by following the concept hierarchy in the domain ontology (PreSOn). For example, using a combination of $\langle Domain, T_i, c_i, "tutorial" \rangle$ as search terms, videos with tutorials related to the concept c_i from the topic T_i in the domain are collected. We have implemented this search schema using the library youtube-search-python ¹. To filter out the collected videos, we have manually followed the scoping review approach (Arksey & O'Malley, 2005) that is designed to review the literature (Oravec *et al.*, 2022; Prahladh & van Wyk, 2020).

Videos retrieved. We have implemented the search schema with a time condition so that each video duration should be >= 3 minutes and < 60

¹https://pypi.org/project/youtube-search-python/[Accessed:June2022]



Figure 4.3: The collected domain related videos have been filtered using the scoping review approach.

minutes–as learner's concentration span is reduced over time (Meseguer-Martinez *et al.*, 2017; Risko *et al.*, 2012). Consequently, 110 videos have been collected which we passed to the next stage of the scoping review.

Screened for domain relevance and transcript availability. Some of the collected videos have domain related words in their title but they are domain-irrelevant. Example of domain-irrelevant videos are, the videos: 'Architecture Presentation Tips -4 FUNDAMENTAL Principles' which is about how to draw an architecture project using PowerPoint, and 'Schooled by Kids: Presentation Skills, Part 1' which is a one minute video for primary school students practicing their presentations. Additionally, since we process the video transcripts only, we only include the videos that have their transcript automatically generated by YouTube. Based on these two conditions, we have excluded 47 videos. This part requires manual screening to ensure that the collected videos are relevant—this step is not automatic and it is part of the limitation is added to (Section 7.3).

Videos included. After applying all the filters (excluding irrelevant videos and checking for the transcript availability), the final outcome was 63 videos. These videos are with a corpus of 110594 tokens and an average duration of 660 (STDV=780) seconds, see Figure 4.3.

The taxonomy with the collected videos are passed as inputs to implement the VISC-L framework.

4.4 Implementation of VISC-L

In this section, we start implementing the framework for Video Segmentation and Characterisation for learning (VISC-L) proposed in (Section 3.5) in the prime domain 'Giving Pitch Presentations'. The Inputs are the collected domain related videos and the domain taxonomy. In the Initial Segments step, the videos are segmented by implementing the Text Tiling approach as proposed in (Section 3.5.2). In the Segment Characterisation step, the initial video segments are characterised by implementing the Semantic Tagging and Topic Classifier steps, as proposed in (Section 3.5.3). The characterised video segments are aggregated by implementing the Thematic Progression Theory as proposed in (Section 3.5.4). The final output is evaluated with experts by comparing the VISC-L output with Google's characterisation of the video segments as described in (Section 4.1).

4.4.1 Input

The VISC-L framework presented in (Section 3.5), requires three inputs. The first one is the **domain taxonomy**, $\Omega = \{C, H\}$ which includes the relevant domain concepts $C \neq \phi$ linked in a concept hierarchy H as proposed in (Section 4.2). We use $c_i \subset C_j$ to denote that c_i is a subclass of C_j . The top level concepts in the concept hierarchy define the main domain topics $\{T_1, ..., T_m\}$. Available ontologies - The Linked Open Data Cloud:¹ can be used if such domain taxonomy is not available.

The second input is the 63 videos collected as described in (Section 4.3) where their video transcript relates to the domain to be learnt. Each text line in any video transcript is associated with a timestamp that represents its start within the video. Before we commence the Initial Segmentation algorithm, we pre-process the video transcripts by restoring their punctuation. This is because we are aiming to use the end of sentence punctuation, [.,?,!], as one of the segment boundary determiners as we describe in the next section. For this purpose, we have used the restore punctuation tool² that has been designed by (Tilk & Alumäe, 2016) which is a 'bidirectional recurrent neural network model with an attention mechanism for restoring missing inter-word punctuation in unsegmented text'-Note that this tool can be deactivated by the designer at any time. The third input is domain related **training data** labelled with the domain topics. It is required in order to fine-tune the model used as a topic classifier to be able to identify the main topics in the video transcripts as part of the characterisation step. We obtained our training data from the six studies (Mitrovic et al., 2016, 2017) conducted when AVW Space was used as a learning platform by users, as described in (Section 3.7.1). The students, who used AVW Space as a learning platform to learn presentation skills, were able to write comments or rate other students' comments. The total number of participants who watched the videos was 38 and they wrote 2038 comments. These comments had been labelled by other students with the domain topics: DELIVERY, STRUCTURE and VISUAL AIDS. Note that the topic PRESENTATION ATTRIBUTE is missed from the labels–we have

¹https://lod-cloud.net/[Accessed: March 2021]

²http://bark.phon.ioc.ee/punctuator[Accessed: May 2021]

solved this issue in the semantic tagging step which we described later in this chapter. The total number of the comments labelled with the topics DELIVERY, STRUCTURE and VISUAL AIDS is 1134, 422 and 482 respectively. Although the data is unbalanced among the topics which is a limitation in this work, it enables us to fine-tune the topic classifier model.

For the Initial Segment step in VISC-L, we passed the 63 collected videos.

4.4.2 Initial Segmentation

Our initial video segmentation algorithm defined in (Section 3.5.2) implements the Text Tiling approach (Hearst, 1997b) starting with smaller units (6 text line sentences) and aggregating them to get larger coherent units (paragraphs). Hence, the algorithm starts by cutting the video transcripts into small segments that are used as a starting point for aggregation. Initial segmentation can be done by using a certain number of text lines (e.g. 6 utterance text lines which is the approach followed in this section) or by using pre-defined segments (e.g. high attention intervals from past interactions (Mohammed & Dimitrova, 2020)). Each transcript line has a corresponding time in the video footage which is used to decide the segment duration. Accordingly, each video segment have three types of data: the transcript (= 6 text lines), the starting time (the timestamp of the first text line in the video) and end time (the timestamp of the last text line in the video). We have implemented this algorithm on the 63 collected videos of the prime domain and generated 2382 segments. These segments have an average duration of 14 (STDV=6) seconds and an average number of 38 (STDV=49) segments per video. The generated segments are passed to the next phase of the the VISC-L framework which is the Segment Characterisation.

4.4.3 Segment Characterisation

In order to aggregate the initial segments, we need to identify what domain content is present in each segment. This is done during the segment characterisation step where we propose using two algorithms (Semantic Tagging and Topic Classifier) to link each video segment VS_i with a set of focus topics FT_i and a set of concepts FC_i . The final characterisation of the video segments will be as proposed in (Section 3.4–Definition 6), $Ch = \langle VS_{Id}, VS_s, VS_e, VS_t, FT, FC \rangle$

Semantic Tagging. To apply the Semantic Tagging algorithm, as demonstrated in (Section 3.5.3), two inputs are needed; the transcript of the 2382 segments generated from the Initial segmentation step in VISC-L, and the taxonomy (Abolkasim, 2019) of the prime domain. This algorithm first pre-processes the transcript ¹ by: (a) tokenisation; (b) cleaning from stop words and punctuation; (c) PoStagging ('Part-of-Speech tagged') to get the resulting nouns and noun phrases. The resulting nouns and noun phrases are to be semantically tagged to the taxonomy terms to decide the focus concept FC. Noting also the concept path to reach a top level concept (i.e. linking to a focus topic FT; $FC \subset FT_i$). As a result, each characterised video segment Ch_i is linked to a set of focus topics and their corresponding concepts; we denote this as $Ch = \langle VS_{\text{Id}}, VS_{\text{s}}, VS_{\text{e}}, VS_{\text{t}}, FT^{1}, FC^{1} \rangle$ where ¹ indicates that this is an output from the first segment characterisation algorithm. This algorithm also succeeds in noticing the topic, PRESENTATION ATTRIBUTE, and collecting its concept.

Characterisation Evaluation. In order to evaluate the characterisation of the single segments, we have asked an external expert to assess the accuracy of the characterisation of 137 random selected segments taken from the 2382 segments that we have characterised from all the videos, (see Appendix H). The expert was a researcher working on the same domain used in this work 'Giving Pitch Presentations'. We provided the expert with a sheet that includes the selected segments with their characterisation, and the topics and concepts in the domain taxonomy with their hierarchy to be familiar with the nature of the video segments' characterisation. The expert was asked to do the following: add new topics or concepts if it is thought to be missed from the characterisation, mark the assigned topic or concept to segment as wrong if she thought it was irrelevant or leave the characterisation if it was correct. The overall number of topics/concepts she assessed were 345. The expert accessed all the taxonomy structure and were

¹We have used the Natural Language Tool Kit (NLTK) https://www.nltk.org/[Accessed: May 2021]

able to see all the concepts and their super concepts. In her evaluation, the expert did not suggest to change the structure of the taxonomy but she either recommend adding new concepts, disagree or agree with the assigned concepts by the algorithm. After receiving the evaluation **result**, we have designed three types of responses; 'To Fix the Domain taxonomy', 'To fix the Semantic Tagging Algorithm', 'Ignore-the concept was picked for the segment' and 'Ignore-suggested new concept is irrelevant'.

'To Fix the Domain taxonomy' means we have added the the new suggested concepts by the expert which we found were missed from the domain taxonomy. There were 34 new suggested concepts added to the taxonomy. We have improved the **PreSOn taxonomy** in response to the expert evaluation and through our observation of the video transcript and the Semantic Tagging result.

'To fix the Semantic Tagging Algorithm' response happens with the two concepts, 'VIDEO' and 'SUIT', which were selected by the algorithm and disagreed by the expert. They have been selected because their video transcripts are domain related but the presenter used the concept 'VIDEO' to mention the video they are presenting but not how to use videos during presentation design. Additionally, they use the concept 'SUIT' with the meaning 'convenient' while in the domain it relates to the presenter's outfit. This issue we are aiming to avoid by enhancing the characterisation of the video segments with the aid of the Topic Classifier.

Furthermore, the 'Ignore-the concept was picked for the segment' response is to ignore the expert's disagreement with the assigned concepts. This is because the concepts assigned to the segments, by the algorithm, are included in the taxonomy and are presented in the segment's transcript. Furthermore, the experts were given the initial segments that characterised only by the Semantic Tagging algorithm not the final segments characterisation. The final characterisation is the combining of the Semantic Tagging and Topic classifier-that contextually characterise the video segments. The adjacent segments in a video with the same focus topics and concepts are aggregated and these aggregated segments with their common focus topics and concepts represent the final segments to be used to generate the narratives. Accordingly, these concepts should be kept for the segments' characterisation at this step. Hence, we have ignored 21 suggestions from the experts. The 'Ignore-suggested new concept is irrelevant' is when the expert suggested a new concept which is not presented in the transcript but the expert found that it relevant from the context of the segment. Accordingly, we decided to ignore 40 suggested concepts as it is impossible to assign them to the video segments characterisation because they have not been mentioned explicitly in the transcript.

Finally, to calculate the agreement between the expert responses and our responses, we ran Cohen's Kappa Coefficient (Cohen, 1960). We are using Python where we use a library¹ to calculate Cohen Kappa Coefficient. We have passed two lists to the code showing the agreement and disagreement between the expert evaluation and our responses. To create these lists and inspired by the Confusion Matrix (Luque et al., 2019), we have designed a combination of the two responses and include them into brackets where the first element in the bracket is our response to the expert evaluation and the second element is the expert evaluation. Each segment in the evaluation data has been assigned one of these brackets. The values within these brackets are as follows: (1,1) means that the expert agree with the Semantic Tagging characterisation, (1,0) means we ignore the expert's disagreement on the Semantic Tagging characterisation of a segment, (0,1) means we ignore the new suggested concept by the expert as they are irrelevant because they did not appear in the video transcript. We have saved the first element of each bracket in a list that represent our responses and save the second element of each bracket in another list representing the evaluator responses. These two lists have been passed to the Python code we designed to calculate the Cohen Kappa Coefficient where the inner agreement result was very high (90%).

On the other hand, after we run the Semantic Tagging algorithm, we found that some concepts within the ontology are not domain specific concepts. This caused an issue with the characterisation results as these concepts mentioned in the videos are in different contexts. For example, the concept 'Speaker' in the topic 'Visual Aid' is referring to a speaker as a device in the ontology while in the video transcript it refers to the presenter of the slides. Also, the concept 'Table' is used in the domain ontology for organising data while in the video transcript it has been mentioned many times for a piece of furniture. Consequently, we have deleted 8 concepts, 1

¹https://scikit-learn.org/stable/modules/generated/sklearn.metrics.cohen_ kappa_score.html[Accessed: May 2021]

from the Topic DELIVERY and 7 from the topic VISUAL AIDS. Additionally, we have used the transcripts of the collected videos to extend the PreSOn ontology which resulted in adding new concepts. Examples of the concepts added per topics are: DELIVERY (within the concept 'BODY POSITION' we have added 'STANDING', 'WALKING', 'LEANING'), STRUCTURE (within the concept 'STRUCTURE APPROACH' we have added 'TACTIC', 'STORY TELLING'), PRESENTATION ATTRIBUTE (within the concept 'IMPACT' we have added the concepts 'PERSUASIVE', 'REMOEMORABLE', 'MOTIVATIONAL'), and VISUAL AIDS (within the concept 'VISUAL AIDS DEVICE' we have added the concept 'VIRTUAL REALITY (VR)'). The final number of the concepts is as mentioned in (Section 4.2) is 398.

After we made the required amendments to the domain ontology, we re-implemented the Semantic Tagging algorithm.

A key challenge for this algorithm is word sense disambiguation. This is not that prominent with carefully selected videos (Mohammed & Dimitrova, 2020), however, if the videos are selected automatically from open social spaces, there will be a high risk of word sense ambiguity (words in transcripts are used in different contexts to the domain). For example, the word 'BODY COMPONENT' can have two different meanings if used in two different contexts, e.g. in *BIOLOGY* or in *PRESENTATION'S STRUCTURE*. Hence, we need to disambiguate the topics based on the context, which is done with the second algorithm which is the Topic Classifier.

Topic Classifier. To overcome the aforementioned word sense ambiguity issue, we need a topic classifier which identifies a domain topic based on its context. Following the latest development in natural language processing, we use Bidirectional Encoder Representations from Transformers (BERT) (Devlin *et al.*, 2018) as a topic classifier.

(a) Brief description of BERT. It is a powerful natural language processing (NLP) model developed by Google (Devlin *et al.*, 2018). We will use pre trained BERT models which will be fine tuned. BERT is a deep neural network model that is based on transformers in its architecture. A transformer has two main components (Encoder and Decoder) and can process a large amount of data in

parallel and BERT is based on the Encoder part only. Each encoder (Vaswani *et al.*, 2017) consists of two main components a self-attention layer and feed forward layer, where the output from the self-attention layer is the input of the feed forward layer.

The self-attention layer in BERT structure helps in recognising the context of the input. First, it passes the input sentence to an embedding layer where each word in the input will be converted to a vector using, e.g. word-to-vec. These vectors will be used to form a matrix (A) where each value is the dot product of two specific vectors. Then, another matrix (X) is formed using the word vectors generated from the embedding layer and a dot product between the two matrix (A*X) will performed to produce an output matrix (Y) which is a context aware matrix as each word knows its neighbours. This output layer (Y) will be passed to feed forward network where it is further processed to give it a richer representation. The encoders can be stacked N times to further encode the information where each layer will be able to learn different attention representations. Hence, by pre-training BERT on a large corpus of text data, it can develop a deep understanding of the structure and meaning of the language, which produces a highly effective tool for different NLP tasks. This will allow BERT to be fine-tuned with just one additional output layer to create models for different language tasks, topic classification in this research.

Within the Segment Characterisation phase of VISC-L, the BERT model is fine-tuned using training data with domain topic labels (which is part of the input for VISC-L). Then, the fine-tuned model is used as a classifier to link each segment VS_i to its domain focus topics FT_i^2 (where ² indicates that this is an output from the second segment characterisation algorithm). For that purpose, we need first to select a BERT model to be the topic classifier.

(b) Use of BERT as Topic Classifier. To select the best BERT model as a topic classifier for our work, we are going to run three experiments to select one from the three models: Roberta Base, Distill Bert and BERT-BASE-Uncased. These models are widely used for topic modelling and semantic analysis tasks. We fine-tune these models with the training data. Each BERT model type has the following architecture: number of **Parameters** which are the learnable variables available for the model, **Transformer Layers** which are used to transform a sequence of word representations to a sequence of contextualised words, **Hidden Size** which is a middle layer between the input and output layers used to assign weights to words to produce a desirable outcome, and **Attention Head** represents the size of a transformer block. The architecture of the selected BERT models is presented in the Table 4.1.

Table 4.1: Selected BERT models for topic classification. The information has been taken from (Adoma *et al.*, 2020).

Model Type	Transformer Hidden		Attention	Parameters
	Layers	Layers	Heads	
BERT-base-uncased	12	768	12	110 M
RoBERTa-base	12	768	12	$125 \mathrm{M}$
DistilBERT-uncased	6	768	12	66 M

The models will be fine-tuned using the domain related training data described in (Section 4.4.1). The model that shows the best results on the metrics (Precision, Recall, and F1 score) will be selected as the VISC-L topic classifier. Within the experiments, we will assess the performance of the models by selecting different values of hyper parameters (Maximum length, Batch size, Learning rate and Epoch). The group of hyper parameters where all the models showed improvement in the evaluation metrics will be selected and the model that gives higher values of these metrics will be selected as the VISC-L topic classifier. Before we commence the experiments, we will define the hyper parameters used. Maximum length e.g., 512 is the maximum number of tokens of the input. Accordingly, the input will be either truncated if it is above the maximum length or padded if it is below the maximum length. According to the small size of the labeled data we have, we used BERT BASE-Uncased-small model as one of the selected model to compare with instead of BERT-BASE-Uncased-large model. Batch size is the number of samples processed before the model is updated and is always set in powers of two. **Learning rate** is used to regulate the weights of the model concerning the loss gradient. An epoch is one cycle through the full training data (Devarakonda

et al., 2017). Inspired by (Adoma et al., 2020), we have designed three experiments where we used three groups of hyper parameters with different values: Group 1 (maximum length=500, batch-size=6, learning-rate= 5_e -5, epochs=1), Group 2 (maximum length=500, batch-size=16, learning-rate= 5_e -5, epochs=5), and Group 3 (maximum length=500, batch-size=16, learning-rate= 5_e -5, epochs=6).

To run the experiments, we first downloaded the Python libraries required for this purpose which are: **tensorflow**, created by Google as a library in Python that can be used to create deep learning models by using a wrapper library. We have used **Ktrain** as the wrapper library. The training data has been pre-processed using the libraries from the **Ktrain** model in Python to divide them into training and validation sets. Within these experiments we have used the models as multiple classifiers. Within each model, the training data is split into both training and test data in a ratio of 9:1 respectively. This ratio is decided because we have a limited training dataset. The main idea of splitting the dataset into a validation set is to prevent our model from over-fitting. Random-state=42 is used to enable reproducible results and allows us to get the same training and test sets across different executions. In this way, the training data will be constant for every run. The model is fine-tuned on the training set and, simultaneously, the model evaluation is performed on the validation set after every epoch. To implement the evaluation of the models, we have used the Ktrain library in Python ¹.

Within the first and second experiments, we have selected the hyper parameters Group 1 (maximum length=500, batch-size=6, learning-rate= 5_e -5, epochs=1) and Group 2 (maximum length=500, batch-size=16, learning-rate= 5_e -5, epochs=5) respectively to fine-tune the aforementioned BERT models. The results are presented in the Tables 4.2, 4.3, and 4.4

We can notice that the BERT models give better results in regard to the metrics (Precision, Recall, and F1 score) after applying the second group of the hyper parameters. The Roberta Base model showed improvement in the metrics results using the Group 2 hyper parameters except for the topic STRUCTURE where it did not improve the Recall metric. Whilst the Distill BERT model did not improve any of the metrics with the topic STRUCTURE, it did improve them with the other topics using Group 2 of the hyper parameters. This can be referred

¹https://pypi.org/project/ktrain/0.1.6/[Accessed:May2021]

Table 4.2: Roberta-BASE model as multiple classifiers after applying the hyperparameters Group 1 and Group 2.

Topic	Hyper parameters- Group 1			1 Hyper parameters- Group 2		
	Precision Recall F1-score		Precision	Recall	F1-score	
Delivery	0.88	0.79	0.83	0.88	0.93	0.91
Structure	0.53	0.71	0.61	0.69	0.63	0.66
Visual Aids	0.81	0.78	0.79	0.95	0.85	0.90

Table 4.3: Distill BERT-BASE-Uncased model as multiple classifiers after applying the hyper parameters Group 1 and Group 2.

Topic	Hyper parameters- Group 1			Hyper parameters- Group 2		
	Precision	Precision Recall F1-score		Precision	Recall	F1-score
Delivery	0.83	0.90	0.86	0.85	0.91	0.88
Structure	0.64	0.60	0.62	0.61	0.57	0.59
Visual Aids	0.97	0.78	0.87	1.00	0.85	0.92

Table 4.4: BERT-BASE-Uncased model as multiple classifiers after applying the hyper parameters Group 1 and Group 2.

Topic	Hyper parameters- Group 1			Hyper par	ameters-	Group 2
	Precision	Recall	F1-score	Precision	Recall	F1-score
Delivery	0.83	0.90	0.86	0.89	0.94	0.92
Structure	0.64	0.60	0.62	0.76	0.74	0.75
Visual Aids	0.97	0.78	0.87	0.97	0.85	0.91

to the lower presentation of the topic STRUCTURE in the training data with 422 labels compared to the other topics, DELIVERY- 1134 and VISUAL AIDS- 482. On the other hand, BERT-BASE-Uncased gave the highest results in regards to the metrics with all of the topics using Group 2 hyper parameters.

To further explore the performance of the BERT models, we have fine-tuned them using the Group 3 hyper parameters which differ from Group 2 in which the number of epochs is to be 6 instead of 5: (maximum length=500, batch-size=16, learning-rate= 5_e -5, epochs=6). The result is presented in the Table 4.5

Table 4.5: Applying Group 3 hyper parameters to fine-tune the BERT models: Roberta Base (R), Distill BERT (D), and BERT-BASE-Uncased (B).

Topic	Precision				Recall			F1-score		
	R	D	В	R	D	В	R	D	В	
Delivery	0.88	0.90	0.87	0.92	0.93	0.94	0.90	0.91	0.91	
Structure	0.66	0.75	0.74	0.66	0.69	0.66	0.66	0.72	0.70	
Visual Aids	0.98	0.91	1.00	0.87	0.89	0.87	0.92	0.90	0.93	

We found that Distill BERT has out-performed the Roberta Base and BERT-BASE-Uncased in regard to the Precision, Recall and F1-score metrics. Compared to the result of fine-tuning the BERT-BASE-Uncased with Group 2 hyper parameters, we found that BERT-BASE-Uncased out-performed Distill BERT in all the metrics except for the Precision metric of the topic DELIVERY and the Recall metric of the topic VISUAL AIDS. For that purpose, and for better performing results and execution time, we decided to select the BERT-BASE-Uncased model using the Group 2 of hyper parameters as our topic classifier. Inspired by the success of using BERT as a binary classifier (Paraschiv & Cercel, 2019) used in German tweets classification, we have decided to run a fourth experiment to compare between BERT-BASE-Uncased model as multi and binary classifiers. Accordingly, we ran BERT-BASE-Uncased as a binary classifier for each topic. Accordingly, we have three copies of the training data, a copy for each topic (DELIVERY, STRUCTURE, VISUAL AIDS) where the labels are either the topic name or 'No'. The results after using the BERT-BASE-Uncased model as a multi and binary classifier are presented in Table 4.6.

The results showed that the model as a binary classifier improved the Precision metric for the topics DELIVERY and all the metrics for the topic VISUAL AIDS. However, the model did not improve any of the metrics for the topic STRUCTURE and this can be referred to the imbalance in the labels of the topics in the training

Topic	Multiple Classifier			Binary Classifier		
	Precision Recall		F1-score	Precision	Recall	F1-score
Delivery	0.89	0.94	0.92	0.91	0.93	0.92
Structure	0.76	0.76 0.74 0.75		0.69	0.71	0.70
Visual Aids	0.97	0.85	0.91	1.00	0.87	0.93

 Table 4.6:
 BERT-BASE-Uncased model as multi and binary classifier results.

data. Based on these results, we have decided to use BERT-BASE-Uncased as a binary topic classifier.

To characterise the video segments using the selected topic classifier, we have passed on to it the 2382 video segments generated from the initial segmentation step. This process has been repeated three times, one for each topic. After that, we concatenated the characterisation results from the three runs. Having done so, each segment is now characterised with a set of focus topics FT^2 where ² means this is the characterisation using the topic classifier.

To get the final characterisation of each video segment, we have combined the Semantic Tagging and the topic classifier characterisations as described in the next section.

Characterisation Combining. To get the final segment characterisation, we ran the step of Combining the characterisation results identified from the semantic tagging and the topic classifier as explained in (Section 3.5.3). For instance, a segment VS_i has two characterisations, one from the semantic tagging algorithm $\langle FT_i^1, FC_i^1 \rangle$ and one from the topic classifier model FT_i^2 . The final characterisation of the segment VS_i is the result of combining the two segment's characterisations: $\langle FT \rangle = \langle FT_i^1 \cap FT_i^2 \rangle$ which is the common Focus Topic(s) assigned to the segments after running the the Semantic Tagging algorithm and the Topic Classifier model. This means, the focus topic is the one identified in both characterisations and the focus concepts belong to the agreed focus topic which can be extracted from the Semantic Tagging characterisation output. Hence the final characterisation of video segments is $Ch = \langle VS_{Id}, VS_s, VS_e, VS_t, FT, FC \rangle$.

Notice that the topic *PRESENTATION ATTRIBUTE* can only be recognised by the semantic tagging algorithm as mentioned in (Section 4.4.3–Semantic Tagging).

Characterisation Outcome. The final characterisation result revealed that 1848 segments have been characterised with a focus topic(s) and concept(s). However, there are 534 segments with no characterisation which happens when the presenter is silent or digresses from the domain. The characterisation results are presented in Table 4.7. According to the characterisation results, 32% of all characterised segments focus on only one topic which could help in learning the effect of the focus topic on the design of presentations. Whereas, 68% of the characterised video segments focus on multiple topics which could help in exploring the relationship between different topics and the effect of using them on the design of presentations. These segments can be aggregated together to get longer segments with the same focus topic and concept. By aggregating the video segments guided by the Thematic Progression theory, we will finish implementing the Text Tiling approach and the VISC-L framework.

 Table 4.7:
 Video Segment Characterisation output.

Focus Topic	Number of Segments	Segment Duration in s			n sec.
		Avg.	STDV	Min	Max
Delivery	243	13	3	7	25
PresentationAttribute	53	14	4	8	32
Structure	167	13	4	7	35
Visual Aids	114	14	5	8	41
Multi Focus Topics	1271	15	6	6	68

4.4.4 Segment Aggregation

Following the Text-Tiling approach (Hearst, 1997b), small segments are aggregated into larger segments. To maintain the flow of information within adjacent segments, we apply the aggregation algorithm based on the Thematic Progression (TP) Theory (Bloor & Bloor, 2013) described in (Section 3.2.2). Accordingly, we ran the third step of VISC-L on the characterised segments by implementing the

Linear Aggregation With Interpolation algorithm based on the focus topic and concept, see (Section 3.5). The input to the aggregation (Algorithm 3) is the video segments with their final characterisation. The algorithm compares the adjacent segments Ch_i , Ch_{i+1} within a video and checks their focus concepts $\langle FC_i \rangle$, $\langle FC_{i+1} \rangle$. It checks whether they intersect with each other and have common concept(s) $\langle FC \rangle = \langle FC_i \cap FC_{i+1} \rangle$. If $\langle FC \rangle \neq \phi$ then these segments are aggregated and their focus concepts are $\langle FC \rangle$ and their focus topics are extracted from the segments characterisation which are related to the $\langle FC \rangle$. The duration of the aggregates starts from the beginning of the first segment and finishes at the end of the last segment. If $\langle FC \rangle = \phi$, then the second segment Ch_{i+1} is a gap segment as long as its duration is $\leq \theta$ -The value of θ represents the segment duration threshold and in this thesis we have selected θ to be 30 seconds which is the average duration of a video segment. The algorithm checks the intersection between the first and third segment $\langle FC \rangle = \langle FC_i \cap FC_{i+2} \rangle$. If $\{FC\} \neq \phi$ then the segment Ch_i, Ch_{i+1}, C Ch_{i+2} are aggregated-this is the reason we call it aggregation with interpolation. Otherwise, if $\langle FC \rangle = \phi$, the first segment Ch_i is saved as a single segment and a new aggregation starts from the segment Ch_{i+1} which is considered the first segment, see Figure 4.4.

4.4.5 Output

The output of VISC-L is a set of aggregated video segments with a start and end time in the corresponding video. Each aggregated video segment Ch_i is characterised with a set of domain focus topics FT_i (top classes in the taxonomy) and a set of focus concepts FC_i from the focus topics which are mentioned in the transcript of the video segment (for every $FC_i \subset FT_i$ and FC is a set of classes or individuals). The aggregation result revealed that the number of segments decreases to 969 (where the original number was 2382), see Figure 4.5. This showed that many adjacent segments have the same focus topic and concept. This is proved with the increase in the percentage of segments (73%) that focus on one topic. In line with the aggregation, the size of the new aggregates has been increased with an average duration of 36 (STDV=35) seconds. Furthermore, the predominant topics, after aggregating the segments, are Delivery (29.1%)

Start Time	End Time	Video segment transcript	Focus concept	Focus Topic	
2:45	3:00	So what I 'm going to do is first of all go through some basic about power point and then we 're going to think about how you structure a talk in order to lead a member of the audience through the data in a way that they will understand each piece that you 're presenting and be able to really understand what the take message is . First of all, what font should you use ?	PowerPoint Message Font	Visual aids Structure Visual Aids	
3.00	3.06	It turn out that some of the default font in power point and keynote are exactly wrong . The font you should use is a Sans Serif font .	PowerPoint Font	Visual aids Visual aids	
3.06	3.13	nan		-	— -1
3.14	3.20	What is a Sans Serif font ? San Serif font are font without little doober line and stuff at the bottom of letter . And that 's to be contrasted with Serif font which have all the little doober at the bottom of each letter .	Font	Visual Aids	
3.20	3.27	nan		-	-
3.28	3.39	So why do you use one and not the other ? Well visual psychophysicist have discovered that using a Sans Serif font when you 're projecting text onto a screen like this make it easier for the audience to read the text quickly. These font down here the Serif font this is the font if you have novel you were going to read it in bed	Font Text	Visual aids Visual aids	
3.39	3.50	it 's very easy to read . And that 's why book are printed with Serif font but you should choose and there are a range of choice that you can make	Font	Visual aids	Gap segme no focus to
3.50	3.59	one of the Sans Serif font for your scientific presentation . How big should your font be ?	Font	Visual aids	no focus co
3.59	4.04	Again be careful of some of the default font size . One of the default in Power Point is astronomically huge	Font PowerPoint	Visual aids Visual aids	≤ 30 second
4.04	4.11	bigger than this one right here it 's like 42 . It 's way too big ! You do n't need that . So the range of font size that you should use are shown here .	Font	Visual aids	
1.19	4.28	Anywhere from 18 down to 36 with one exception . We often in our scientific presentation will want to insert a reference to a published paper which is n't really the focus of the slide but we want to have it up there for scientific accuracy.	Slide	Visual aids	
4.34	4.41	So under those circumstance you might use a 14 point font to put your reference at the bottom of the slide . slide .	Font Slide	Visual aids Visual aids	
1.42	4.48	Other than that just work in this range and you 'll be good . Avoid using capital letter . This make it clear why. It 's actually really hard to read all capitol . And secondly in this age of email I think we all know that capitalizing everything		-	
4.49	4.57	is the visual equivalent of shouting at someone . So it 's just not polite . In addition when you 're thinking about title		-	_]
4.57	5.04	or how you would actually capitalize word use a sentence format . In other word do n't capitalize every word	Format	Presentation Attribute	

Figure 4.4: Example of aggregating adjacent segments by interpolating the gap segments. For example, we have aggregated some of the segments in one of the collected videos which has the YouTube video ID= Hp7Id3Yb9XQ-this is taken from the segment characterisation which saved the segment Id that is created using the video Id as described in Algorithm 1. The aggregation process is as follows: the segment with duration [3.06-3.13] is with duration < 30 seconds so it is aggregated with the previous segments. The same process is followed to aggregate the segments with duration [3.20-3.27], [4.19-4.28], [4.42-4.48] and [4.49-4.57] with the previous segments.

of the segments) and the topic Structure (20.6% of the segments). There is a noticeable decrease in the number of segments that focus on multiple topics (27.03%). Nevertheless, the topics Delivery and Structure stand out as the more correlated topics among other topics which highlighted that they are necessary to understand each other. On the other hand, the topic Presentation Attribute appears alongside other topics in the aggregates more than it being a single focus topic. This indicates that this topic is better to be demonstrated by presenting its relationship with other topics, see Table 4.8.

To evaluate the Quality, Perceived Usefulness, Learning effect and the Cognitive

V	/ideo Collection using YouTube		Video Seg	mentation ; fra	and Chara mework	cterisation VI	SC-L
Search Schema < Domain name, Topic, Concept>			Initial Segments #Segments	AVG Duratio	n STD	v	
Filtering Criteria			2382	14s	6	_	
Duration Video has	>= 3 minutes s transcript		Segment Aggreg #Segments A	sation AVG Duration	STDV		
Output			969 3	36s	35		
63	660s	780	Filtering Criteria Video views >=100 do not include do Output 247 segments con	OK and delete main related ning from 33	e the first ar information videos	nd last segment า	ts as they
			Topic	#Seg	gments /	AVG Duration	STDV
			Delivery	86	(52s	48
			Presentation Attr	ibute 12	(53s	63
			Structure	54	4	45s	29
			Visual Aid	43	:	88s	87
			Multi	52	4	45s	24

Figure 4.5: VISC-L output using the videos with views $\geq 100K$ after restoring the punctuation of their transcript

Work Load of the video segments' characterisation to support learning, we have conducted our user study in which we compared VISC-L video segment characterisation with Google's video characterisation, see (Section 4.1).

4.5 User Evaluation

.

The evaluation study is an initial exploratory study which aims to assess the Quality, Perceived Usefulness, Learning Effect and Cognitive Work Load of the characterised video segments when used for learning. These types of assessment have been used in other work to assess the effect of new learning strategy on users. First we assessed the Quality of the characterised video segments inspired by Kriksciuniene *et al.* (2019) by asking the participants whether the provided

Focus Topic	Number of Segments	Segment Duration in sec			n sec.
		Avg.	STDV	Min	Max
Delivery	282	42	47	6	390
PresentationAttribute	48	29	20	8	93
Structure	200	32	24	7	127
Visual Aids	177	42	32	7	199
Multi Focus Topics	262	30	30	6	321

 Table 4.8: Video Segment Aggregation Data.

segments are useful to learn from and the characterisation of the video segments helped them to recognise and focus on domain key areas. For that purpose we have used the LIKERT scales (Taherdoost, 2019). Inspired by (Dimitrova & Mitrovic, 2022), where they assess the acceptance of the new technology in learning, we use TAM questionnaire (Davis, 1989) to assess the participants' Perceived Usefulness of learning while watching the characterised video segments, and use NASA-TLX instrument (Hart, 2006) to check the participants' perception of Cognitive Work Load while using the characterised video segments for learning. The Learning Effect of the characterised video segments on participants' domain comprehension have been assessed using the pre and post tests. For this reason, users have been invited to participate in this evaluation study where the evaluation process includes comparing the video segment characterisation generated by VISC-L with the Google's characterisation (See Section 2.3.1).

4.5.1 Study Setup

To commence this evaluation study, we invited 18 potential users to assess the characterisation of a selective number of videos following the evaluation procedure. All the details are described below.

Participants. According to the domain of this thesis which is for learning, we decided to invite learners or educators to participate in the study. This decision is based on the results of the evaluation studies (Mitrovic *et al.*, 2016, 2017) conducted on AVW within the presentation skills domain where it is found that

postgraduate participants provided better responses. Consequently, we decided to invite postgraduate researchers from the University of Leeds or PhD holders from industry. Only 18 participants responded (10 male, 7 female and 1 other); 16 PhD students from the University of Leeds and 2 from Industry. 12 participants were 24-29 years old, 1 participant was 18-23 years old and 5 participants were above 30 years. The training level is varied: 13 have some training on giving presentations, the rest either have a lot of training or have received no training before. Their presentation experience is varied: 10 have a medium level, 5 have either an experienced level or little experience. 12 participants are native English speakers. 10 participants watch YouTube videos every week for learning and every day for other purposes, whereas the rest watch YouTube occasionally.

The selection of the participants is done using a convenience sampling which is a non-probability sampling method which is some of the limitations of this study (Section 7.3). The participants were researchers and lecturers who were available at the time of the study and showed willingness to participate in the study as not of all the invited participants responded to the invitation.

The study was approved by the ethics committee of the Faculty of Engineering and Physical Sciences, University of Leeds. We asked the participants to leave their emails so their time will be compensated by $\pounds 50$ each¹ (as token of appreciation) and the emails were deleted afterwards. This decision is inspired by (Largent & Lynch, 2017) where they defined the payment to the evaluators participated in a study by: "Payment may also be offered to compensate research participants for the fair value of their time, their willingness to accept research-related burdens, and their contributions to research". The amount of payment depends on the time and the number of questions that participants need to answer. In this study the payment was $\pounds 50$ each while in the study conducted in (Section 6.6) the participants were paid $\pounds 30$ as they did not do the pre and post tests.

¹The amount of $\pounds 50$ that each participant received as a voucher was decided to ensure a fair reward considering both the duration of the study (around 3 hours) and the budget limitation for this PhD research.

Materials. To be able to conduct the surveys, we have selected 8 videos from the data set. The selection of these videos is based on the following filtering criteria: their popularity (where we divided the number of the video's views by the days since its creation to the date of the data collection), and the video's duration which be between > 4 and < 6 minutes so the study will not last for more than one hour. The surveys have been designed using Google Forms to assess the segment's characterisation in regard to its: Quality, Learning Effect, Perceived Usefulness and Cognitive Work Load. To do so, we have compared the characterisation of the same video segments using VISC-L and Google characterisation. We have designed 2 Google forms (see Appendix I), Form 1 and Form 2, where the order of the videos in both forms is the same. However, in Form 1, the first 4 videos have been characterised by Google while in Form 2 these videos have been characterised by VISC-L. The second 4 videos in Form 1 have been characterised by VISC-L while in Form 2 they have been characterised by Google. We sent these forms separately to 20 different participants however not all of them responded. Accordingly, Form 1 was completed by 8 participants while Form 2 was completed by 10 participants. The surveys are fully compliant with the General Data Protection Regulation (GDPR) and have been reviewed and approved by the Research Ethics Committee of the Faculty of Engineering and Physical Sciences, University of Leeds, UK. The structure of the forms is the same and as follows: a consent form, a section to collect user profiles, a pre-test section where the participants list all the terms they know about the domain topics, and the last two parts. Part 1 (whether it is Form 1 or 2), has three sections. Section 1 includes the footage of 4 videos where each video is followed by the characterisation of the recognised segments by one of the algorithms, followed by the quality questions designed using the LIKERT response labels (Taherdoost, 2019) and post-test where they will list the domain related terms they recognised while watching the videos. Section 2 is the Perceived Usefulness questions to assess the characterised video segments and their positives and negatives and in Section 3, the Cognitive Work Load scales. The structure of **Part 2** of the survey is the same as **Part 1**, but the characterisation of the videos will be done by using the other algorithm. The Perceived Usefulness of the characterised video segments is designed following the TAM questionnaire

(Davis, 1989) while the Cognitive Work Load scales have been designed using a combination of NASA-TLX scales (Hart, 2006).

Procedure. To commence the evaluation study, we sent the evaluation form link to the participants. Each participant should go through the following steps in the evaluation form: (1.) Read and accept the consent form, (2.) Complete a short pre-study section to collect their profiles, (3.) Watch several suggested video segments with a characterisation generated using one of the algorithms (VISC-L or Google), (4.) Give feedback on the video segments and provided characterisations, (5.) Provide a short video summary, (6.) Give feedback on the usability and usefulness of the recommended video segments for learning about giving presentations, (7.) Repeat [3-6] with segments generated by the other algorithm (Google or VISC-L).

In the following section, we analyse the outcome from the evaluation studies where we aggregate the outcomes of the two surveys. We report and analyse the results of the Quality, Learning effect, Perceived Usefulness, Cognitive Work Load and the positives and negatives of the video segments' characterisation generated by VISC-L in comparison with Google's characterisation.

4.5.2 Results: Quality of the Segment Characterisations

We analysed the Quality of the characterisation of the video segments based on the aggregated experts' responses to be able to answer RQ1.

RQ1: How to automatically generate good quality characterisations of video segments that can be used for learning?

To assess **Quality**, we asked the participants to rank whether the segments were useful and the characterisation was helpful for learning about presentation skills. Referring to the Table 4.9, more than half of the participants agreed that the video segment were useful with a slight difference between Google and VISC-L characterisation. Google segments found to be more useful than VISC-L with an AVG difference of 0.08. Similarly, nearly half of the participant agreed that the characterisation of the video segments is helpful where Google was higher than VISC-L characterisation with an AVG difference of 0.19. Additionally, around quarter of the participants found that video segments were not useful and VISC-L characterisation found to be less helpful than Google with an AVG difference of 0.16. These results indicate that more than half of the participants agreed that the positives (usefulness and the help of the video segment characterisation) is more than its negatives with both VISC-L and Google characterisation.

Table 4.9: Quality of the characterisation of video segments' characterisation using VISC-L and Google. Hence, the values of each quality category (Useful, Not useful, Helpful, Unhelpful) of each video characterised by VISC-L is aggregated from both surveys and the final result is presented in the table as an average (STDV). The same steps were followed for the videos characterised by Google in both surveys. Notice that there is 1 segment characterised by VISC-L but had not been watched by one of the experts and 11 segments characterised by Google that had not been watched by an expert.

Method	Useful	Not useful	Helpful	Unhelpful
VISC-L	$0.51 \ (0.50)$	0.29(0.45)	0.48(0.50)	0.43(0.49)
Google	0.59(0.49)	0.23(0.42)	0.67(0.47)	0.27(0.45)

To assess the **Learning effect** of watching the video segments, we compared the participants' domain terms mentioned in the pre-test with the new domain terms mentioned after watching the video segments generated by either VISC-L or the Google algorithms. During the pre-test, an average of 6.78 terms (STDV=5.67) were mentioned by the participants. After watching the VISC-L segments, the participants named an average of 9.89 (STDV=8.48) new terms, while after watching the Google segments, the participants named an average of 8.72 (STDV=8.36) new terms. There is no significant difference between the segments with a characterisation that led to identifying new domain terms.

4.5.3 Results: Usefulness of the Segment Characterisations

In this part we analyse and compare between the Perceived Usefulness results of the characterised video segments using VISC-L and Google based on the aggregated experts' responses to be able to answer:

RQ2: Does the characterisation of video segments help in identifying and focusing on the key domain concepts?



Figure 4.6: Evaluation results of the Perceived Usefulness of the segment characterisation using the VISC-L and Google characterisations.

Perceived Usefulness. The comparison considered whether participants managed to identify and link main points in the videos to the topics in the domain and to identify key points and focus on them. The results presented in Figure 4.6 showed that in general, the characterised segments were LIKELY to meet their goals. The participants preferred the characterisation generated by Google more because the language used was extracted directly from the transcript and was easy to recognise in the video, while in VISC-L the characterisation was referring to key domain topics form the domain ontology that related to

the key concepts mentioned in the transcript. Figures 4.7, 4.8 and 4.9 show the difference in the video segment characterisations generated by VISC-L and Google.



Figure 4.7: Example 1 of characterising a video segment using VISC-L and Google. The segment has been taken from one of the collected domain related videos with ID= dEDcc0aCjaA, and starts at minute 2.09 and ends at minute 4.20 from the video footage.

Cognitive Work Load. In this part, we assess the Cognitive Work Load expected while watching the characterised video segments using VISC-L and Google. For this purpose, we have analysed the experts' responses on the NASA-TLX sub-scale items after watching the characterised video segments. We have assessed the Mental Load, Effort, Frustration, and Performance sub-scales and the results are presented in Figure 4.10 as an average using the range of scores starting from 1 (the lowest) to 20 (the highest). The participants were asked to provide comments to justify their scores. They reported that the characterised video segments generated with both VISC-L and Google had **low** Mental Load and required **low** Effort with average scores < 10. This is because the segments were short, easy to watch and the characterisation helped the participants to focus on a single topic. Meanwhile, **high** Mental Load and Effort was reported because



Figure 4.8: Example 2 of characterising a video segment using VISC-L and Google. The segment has been taken from one of the collected domain related videos with ID= ADJAcyTq1us, and starts at minute 16.55 and ends at minute 17.55 from the video footage.



Figure 4.9: Example 3 of characterising a video segment using VISC-L and Google. The segment has been taken from one of the collected domain related videos with ID= Hp7Id3Yb9XQ, and starts at minute 2.45 and ends at minute 5.04 from the video footage.

some of the video content added little or no new knowledge, or the description was not in-line with the video or did not specify the focus topics. For that reason, 12 participants scored ≥ 10 for Mental Load and 7 scored ≥ 10 for Effort while watching the segments characterised by VISC-L. On the other hand, there are 8 and 9 participants who scored ≥ 10 for Mental Load and Effort respectively while watching the segments characterised by Google.

Some examples of the participants' positive and negatives feedback regarding the Mental Load and Effort while watching the characterised video segments by VISC-L are: for Mental Load **positives**, "The descriptions helped to signpost what was being taught in a section", and for Mental Load **negatives**, "the short description limits my experience of watching the videos". For Effort **positives**, "Easy, very simple and passive videos often with widely known knowledge and little new information", and for Effort **negatives**, "There were a number of points to remember and the short descriptions were not specific for many of the segments."

Whereas, some participants' positive and negative feedback regarding the Mental Load and Effort while watching Google's characterised video segments is: for Mental Load **positives**, "It was very easy to watch and read the suggested segments", and for Mental Load **negatives**, "The videos were easy to listen to and understand what they were presenting, but segmenting made it slightly more to think about". For Effort **positives**, "It was easy to click the videos and begin and focus on the segments", and for Effort **negatives**, "however, there was also a small amount of physical demand required for stopping and starting the videos to remember things and take notes".

With regard to Frustration, for VISC-L, 6 participants reported high Frustration ≥ 15 because they found some of the segments' characterisation did not align with the actual content. While with Google, 4 participants gave high Frustration ≥ 15 because they found some segments' start time was inaccurate or the characterisation was incomplete. Examples of participants' feedback regarding the Frustration while watching characterised video segment by VISC-L are: **positives**, "It is good to watch the videos if the person have time", **negatives**, "Some segments were not relevant or incorrectly labelled, which was frustrating as it meant putting in more thought to decide what the point of the talk was". Whereas, participant feedback regarding the Frustration while watching characterised video segment by Google is: **positives**, "The videos where very good and useful to me", and **negatives**, "It was frustrating when the segments started in the middle of a sentence, as the description was not enough to work out what was going on".

Regarding Performance, there was similar feedback for both segments generated with VISC-L and Google. Participants gave **high** Performance, 9 scores were ≥ 15 for VISC-L and 6 scores were ≥ 15 for Google, as they found the segments were very good at explaining the key terms and helped them to think of the domain topics. Whereas, 9 scores were < 15 for VISC-L and 12 scores were < 15 for Google as participants did not enjoy some of the video content and did not feel they learned new things. Examples of participant feedback regarding the Performance while watching the characterised video segments by VISC-L are: **positives** "I learnt aspects of giving presentations especially on Delivery and Visual Aids", **negatives** "The segment description included the terms but was often wrong about the content within the segments". Whereas, participant feedback regarding the Performance while watching characterised video segment by Google is: **positives**, "I learned quite a lot from the videos", and **negatives**, "I think I remember many points but it was hard to think which were related to structure, delivery etc".

The results presented above showed that the participants preferred the Google characterisation which was found to be more user friendly and clear as it is taken directly from the video transcript. Notwithstanding, the conceptual characterisation of VISC-L is also noticed positively by users and this characterisation represents the key to generate the narratives (as described in Sections 5.2 and 6.5). These results are not statistically significant at p < 0.05 after running Mann Whitney U Test with regard to the Quality, Learning effect, Perceived Usefulness and Cognitive Work Load of the video segment characterisation. This is due to the limited number of participants in the evaluation study. The number of the participants could not be increased in this study according to the scope of this PhD research which is considered to be one of the limitations in this thesis. Nevertheless, the number of the participants could be increased in future study, see (Section 7.3).

Qualitative Feedback from Experts. The participants were asked what they found positive or negative when watching the characterised video segments



Figure 4.10: Mental Load estimation of the segment characterisation using VISC-L and Google. The score range is 1-Low to 20-High.

generated by VISC-L or Google. For both VISC-L and Google Positives the participants found that the segments offered them a strategy for learning and were good to help them focus as these segments are short and with description. For VISC-L Negatives the participants noticed that the characterisation was scripted and not in a natural way while for Google Negatives they found that some descriptions did not match with the video content, and commented that there were too many segments within some videos, and pointed at inaccurate starting times. Examples of the positive and negative feedback regarding the Quality of the characterised video segments using VISC-L: positives, "focused on specific areas rather than too much information", "It was helpful to target certain aspects of the video relating to areas I lack experience in", and the negatives, "some were irrelevant and not useful. The descriptions were very inaccurate", "They were very brief and sometimes unclear what they related to specifically". Whereas, examples of the positive and negative feedback regarding the Perceived Usefulness of the characterised video segments using Google: positives, "it helped to identify the main points and what the video was going to cover", "The description felt

natural and tailored for each video and the description shows the main points for the video", and the negatives, "Some phrases were not depicting the actual learning point", "Some of the descriptions seemed unfinished so it was difficult to work out what they meant".

4.5.4 Discussion

We have implemented our generic video segmentation and characterisation framework VISC-L to support learning as presented in (Section 3.5). It was applied in a presentation skills domain to answer research questions RQ1 and RQ2 identified in this thesis (see Chapter 1).

An evaluation study examined the Quality, Perceived Usefulness, Learning effect and the Cognitive work Load of video segmentation and characterisation by comparing VISC-L and Google. The results from the study give two indications. Firstly, they indicate that using characterised video segments could improve a learners' domain knowledge, as the learners were able to identify new domain terms. Secondly, the results showed that there is no statistically significant difference between VISC-L and Google video segmentation and characterisation. With regards to the learning effect, for both VISC-L and Google there was improvement in learning because there were new unique terms mentioned in the summaries after watching the videos. Hence, the study provides support for using segmentation with characterisation to support learning. The Quality with both VISC-L and Google shows that their generated characterised segments were helpful. The Perceived Usefulness of segmentation and characterisation with Google was slightly better than VISC-L. Participants' feedback indicated that the format used to present the characterisation has influenced the usefulness- the natural language descriptions offered by Google were easier to follow than the list of concepts offered in the VISC-L interface.

In response to the feedback from the experts, after evaluating the aggregated characterised video segments, we have improved the initial segmentation algorithm. We have decided the segment boundary to be the end of sentence punctuation [.,?,!]. We have also selected the videos with views $\geq 100K$ to ensure that the content of the videos is more engaging. This is in response to some of the

participant's feedback where they did not like the video content, e.g. "some of them were really outdated and boring", "some of the videos were not engaging as (they) presented a different style of presenting to how I would do it". Accordingly, 33 videos match the number of views filtering criteria and for them we re-ran the initial segmentation algorithm with the new condition (using the aforementioned punctuation to decide the segment boundary) and the result was 2788 segments with a duration in seconds of AVG=5 (STDV=6) with an average number of segments of 86 (STDV=66) per video. After we re-ran the Aggregation algorithm on these segments, the result was 247 segments as presented in the Figure 4.5. These new segments are used as an input for VIN-L as described in (Section 5.2) and the segments and the narratives quality and usefulness for learning is assessed in (Section 5.3) by experts and workers from Amazon Mechanical Turk.

In Chapter 6, we combine VISC-L with the Google approach: VISC-L to extract the concepts and Google to formulate titles when we apply VISC-L in the healthcare- focusing on patient quality of life awareness. Now having characterisations in the forms of terms linked to a domain ontology, in the next chapter we implement the VIN-L framework presented in (Section 3.6) for connecting video segments to create video narratives (combining several segments) to focus on specific domain concepts .

4.6 Summary

In this chapter we have presented the implementation of the VISC-L framework on the prime domain, 'Giving Pitch Presentations'. The aim is to prove the applicability of the VISC-L framework. We have automatically collected, segmented and characterised videos related to Presentation Skills. The characterisation of the video segments have been evaluated in comparison with Google's characterisation. For that purpose, we have invited 18 PhD participants from the University of Leeds and Industry to participate in this study. The aim of this evaluation study is to be able to answer the following research questions:
How to automatically generate good quality characterisations of video segments that can be used for learning?

Does the characterisation of video segments help in identifying and focusing on the key domain concepts?

The initial results are promising which point towards potentially answering these research questions. The users after watching the characterised video segments using VISC-L found them have a potential to help them: identify domain key terms, focusing on these terms, and improving their domain learning be recalling new domain terms different from the domain terms they knew before watching these segments.

Chapter 5

Implementation and Validation of VIN-L in the Presentation Skills Domain

5.1 Introduction

The aim of this chapter is to prove the applicability of the VIN-L framework, designed in (Section 3.6), that is underpinned by Ausubel's Subsumption Theory to generate video narratives to support learning presentation skills. Generating video narratives represents our main goal in this thesis; consequently, the work in this chapter aims to answer the following research questions from this thesis (chapter 1):

RQ3: How to generate good quality narratives that can be used to support perceiving domain learning?

RQ4: Are the generated narratives useful to support perceiving domain learning?

To be able to answer these questions, we generate the domain related video narratives and evaluate them with experts, who have experience with giving presentations and teaching others how to give presentations, and with advanced learners. The expert evaluators are domain experts from the University of Leeds and the University of Canterbury in New Zealand and their responses are analysed to answer RQ3. In addition, we use Amazon Mechanical Turk as a crowd source to evaluate the video narratives with users and analyse their responses. Thus, we can compare between the expert and learners answers of RQ3, and use the learners' responses to answer RQ4. Consequently, we implement the subsumption linking algorithms from the VIN-L framework that take, as an input, the characterised video segments (VISC-L outputs from Chapter 4) and link them to generate the narratives. According to the approach of this research which is a knowledge approach, the design of the VIN-L framework presumes that the domain taxonomy is available. We use the taxonomy to implement the subsumption linking algorithm part of VIN-L which is presented in (Section 3.6).

The structure of this chapter is as follows: the implementation of VIN-L to generate video narratives is presented in (Section 5.2), the generated video narratives are evaluated with experts in (Section 5.3) to answer RQ3, in (Section 5.4) the generated video narratives are evaluated with learners using Amazon Mechanical Turk to answer RQ4, and the summary of the chapter is presented in (Section 5.5).

5.2 Implementation of VIN-L

Generating video narratives is the aim of this research for which we designed the VIN-L framework, as shown in (Section 3.6), and in this chapter we prove its applicability by applying it to the presentation skills domain. This is to help learners identify and link together the domain's key points presented in the generated video narratives. The full implementation of VIN-L is presented next.

Input. To apply VIN-L in the presentation skills domain, the following inputs are required: the domain taxonomy which we presented in (Section 4.2) and the 247 characterised video segments which were the output of VISC-L as presented in (Chapter 4). These segments are linked together to generate the narratives by implementing the subsumption linking algorithms as is shown next.

Subsumption Linking Algorithms. We use the subsumption linking algorithms designed in (Section 3.2.3) (Algorithms 4, 5, 6, 7) to implement the linking types defined by Ausubel's subsumption theory; Derivative, Super Ordinate, Correlative and Combinational. Referring to Figure 3.6 column 2, we demonstrate the implementation of the linking algorithms.

Filtering Criteria To ensure that the generated video narratives fulfill their target of helping learners identify the key domain points and link them, we have applied a duration filtering criteria on them. This filtering criteria imposes that the video narrative duration is ≥ 3 minutes and ≤ 6 minutes, see (Section 3.4–Definition 7). Additionally, we generated narratives where the number of video segments included in them is ≥ 2 and ≤ 6 to ensure that learners are not watching too few or many segments per narrative.

Output. After applying the conditions to control the generating of the video segments, we generated 85 video narratives with a minimum average duration of 199 (STDV=86) seconds and a maximum average duration of 290 (STDV=11) seconds. They are distributed among the four narrative types as follows: 15 Derivative narratives, 13 Super Ordinate narratives, 51 Correlative narratives and 6 Combinational narratives, this is presented in Figure 5.1. The number of Correlative video narratives is higher than the video narratives generated for other types of narratives because of the repetition issue. This means the same video narrative is generated for all the sibling concepts within a parent concept. On the other hand, a few Combinational narratives have been generated because we only have 4 topics and we generate one narrative per pair of topics. Hence, no different narratives focus on the same pair of topics.

These generated video narratives are a sequence of video segments that are ordered based on their linking type. However, these generated narratives include only the video segments and are not sufficient for learning. Accordingly, we augment them with text following the templates identified for each narrative type as presented in (Section 3.6). We illustrate the output with several examples.

Subsumption Linking Algorithms								
Video Narrative generation VIN-L framework								
Four Algorithms								
Narrative Type	<u>#Narratives</u>	AVG Duration	<u>STDV</u>					
Derivative	22	199s	86					
Super Ordinate	14	224s	57					
Correlative	80	202s	59					
Combinational 6		198s	106					
Filtering Criteria: 2<=Number of segments per narrative <=6 180 s<=Duration of narrative <=300 s								
Output								
Narrative Type	<u>#Narratives</u>	AVG Duration	STD\					
Derivative	15	199s	86					
Super Ordinate	13	237s	31					
Correlative	51	239s	35					
Combinational	6	290s	11					

Figure 5.1: The output after applying the four Subsumption Linking Algorithms. Four types of video narratives have been generated: Combinational, Correlative, Derivative, Super Ordinate. The number of narratives per type with their duration Average and Standard Deviation (STDV) is reported.

An example of one of the augmented video narratives is presented in Figure 5.2.

The full list of the narratives is available in our research GitHub link 1 .

An example of the Derivative narrative is shown in Figure 5.3 and aims to support learning the concept 'GRAPH'. It first introduces the parent concept 'FIGURE' to the cognitive structure by providing a segment focusing on it. Then,

¹https://github.com/Generating-Video-Narratives/Generating-Video-Narratives. git

5.2. IMPLEMENTATION OF VIN-L

Learning about Presentation Skills:

Link DELIVERY and VISUALAID.

You are given a collection of video segments about DELIVERY and VISUALAID. These video segments have been extracted automatically from popular YouTube videos related to presentation skills. Each segment will start from a time point in the video and will stop when the corresponding concepts are covered. You can re-watch the segment or watch other parts of the video, if you with so.

You will now watch a video segment that mentions FEEDBACK (related to DELIVERY) and SLIDE (related to VISUALAID). The segment is taken from a YouTube video with ID=Hp7ld3Yb9XQ, and starts at 20.32 with duration of 40 seconds.



You will now watch a video segment that mentions SMILE, AUDIENCE (related to DELIVERY) and PICTURE, SLIDE (related to VISUALAID). The segment is taken from a YouTube video with ID+ h6sm47j-Am4, and starts at 4.16 with duration of 33 seconds



You watched several video segments that link DELIVERY and VISUALAID relevant to Presentation Skills.

The video segments were automatically extracted from popular YouTube videos on presentation skills, using algorithms for segmentation, characterisation and linking based on a theory for concept learning. We hope you found the collection of video segments helpful.

Figure 5.2: Example of a video narrative augmented (Section 3.4–Definition 12) using the template we designed in (Section 3.6.2).

the narrative provides a video segment that focuses on the specific concept 'GRAPH' to help in linking both concepts together.

An example of the Super Ordinate narrative is shown in Figure 5.4 that aims to support learning the concept 'POSTURE'. It first introduces the specific concept 'SHOULDER' to the cognitive structure by providing a segment focusing on it. Then, the narrative provides a video segment that focuses on the generic concept 'POSTURE' to help in linking both concepts together.

An example of the Correlative narrative is shown in Figure 5.5 and aims to support learning the concept 'CONNECTION' by introducing its similar concept(s). The narrative first introduces the similar concepts 'PAUSE' and 'FOCUS' to the cognitive structure by providing segments focusing on them. Then, the narrative provides a video segment that focuses on the specific concept

5.2. IMPLEMENTATION OF VIN-L

Learning about Presentation Skills:

Learning GRAPH from VISUALAID

GRAPH is part of FIGURE. You are given a collection of video segments about FIGURE and GRAPH related to VISUALAID. These video segments have been extracted automatically from popular YouTube videos related to presentation skills. Each segment will start from a time point in the video and will stop when the corresponding concepts are covered. You can re-watch the segment or watch other parts of the video, if you wish so.

The next video segment mentions FIGURE (related to VISUALAID). The segment is taken from a YouTube video with ID= Hp7Id3Yb9XQ , and starts at 16.56 with a duration of 43 seconds.



The next video segment mentions FIGURE (related to VISUALAID). The segment is taken from a YouTube video with ID=Hp7Id3Yb9XQ, and starts at 21.2 with a duration of 107 seconds.



The next video segment mentions GRAPH (related to VISUALAID). The segment is taken from a YouTube video with ID= MnIPpUTCRC, and starts at 1.49 with a duration of 107 seconds.



You watched several video segments about (FIGURE) and (GRAPH) related to VISUALAID.

The video segments were automatically extracted from popular YouTube videos on presentation skills, using algorithms for segmentation, characterisation and linking based on a theory for concept learning. We hope you found the collection of video segments helpful.

Figure 5.3: Example of the Derivative Narrative to support learning showing that to learn the specific concept 'GRAPH', its generic (parent) concept 'FIGURE' must be introduced first before introducing the specific concept. Both concepts belong to the topic 'VISUAL AIDS'.

'CONNECTION' to help in linking all the concepts together.

An example of the Combinational narrative is shown in Figure 5.6 and shows the relationship between the topics 'STRUCTURE' and 'PRESENTATION

5.2. IMPLEMENTATION OF VIN-L

Learning about Presentation Skills:

Learning POSTURE from DELIVERY

SHOULDER is part of POSTURE. You are given a collection of video segments about SHOULDER and POSTURE belong to DELIVERY. These video segments have been extracted automatically from popular YouTube videos related to presentation skills. Each video segment will start from a time point in the video and will stop when the corresponding concepts are covered. You can re-watch the video segment or watch other parts of the video, if you wish so.

The next video segment mentions SHOULDER (relates to DELIVERY). The segment has been taken from a YouTube video with ID=1VCaJ4dSHak, and starts at 3.44 with a duration of 42 seconds.



The next video segment mentions SHOULDER (relates to DELIVERY). The segment has been taken from a YouTube video with ID= 7MWaeOHDBOg, and starts at 0.35 with a duration of 149 seconds.



The next video segment mentions POSTURE (relates to DELIVERY). The segment has been taken from a YouTube video with ID= d812a7qG9Kw, and starts at 3.14 with a duration of 38 seconds.



You watched a video segment about (SHOULDER) which is part of (POSTURE) belong to DELIVERY.

The video segments were automatically extracted from popular YouTube videos on presentation skills, using algorithms for segmentation, characterisation and linking based on a theory for concept learning. We hope you found the collection of video segments helpful.

Figure 5.4: Example of the Super Ordinate Narrative to support learning showing that to learn the generic concept 'POSTURE', its specific concept 'SHOULDER' must be introduced first before introducing the generic concept. Both concepts belong to the topic 'DELIVERY'.

ATTRIBUTE' by identifying the relationship between their concepts mentioned in the same segment.

Learning about Presentation Skills:

Learning CONNECTION from DELIVERY

PAUSE, FOCUS are similar to CONNECTION. You are given a collection of video segments about PAUSE, FOCUS related to DELIVERY. These video segments have been extracted automatically from popular YouTube videos related to presentation skills. Each segment will start from a time point in the video and will stop when the corresponding concepts are covered. You can re-watch the segment or watch other parts of the video, if you wish so.

The next video segment mentions PAUSE (related to DELIVERY). The segment is taken from a YouTube video with ID= yoD8RMq2OkU, and starts at 0.13 with a duration of 92 seconds.



The next video segment mentions FOCUS (related to DELIVERY). The segment is taken from a YouTube video with ID= yoD8RMq2OkU, and starts at 13.3 with a duration of 51 seconds.



The next video segment mentions CONNECTION (related to DELIVERY). The segment is taken from a YouTube video with ID=Hp7ld3Yb9XQ, and starts at 27.32 with a duration of 23 seconds.



You watched several video segments about (PAUSE, FOCUS) which are similar to (CONNECTION) related to DELIVERY. The segments were automatically extracted from popular YouTube videos on presentation skills, using algorithms for segmentation, characterisation and linking based on a theory for concept learning. We hope you found the collection of video segments helpful.

Figure 5.5: Example of the Correlative Narrative to support learning showing that to learn the specific concept 'CONNECTION', its similar concepts 'PAUSE' and 'FOCUS' must be introduced first before introducing the specific concept. All concepts belong to the topic 'DELIVERY'.

Learning about Presentation Skills:

Link STRUCTURE and PRESENTATION ATTRIBUTE.

You are given a collection of video segments about STRUCTURE and PRESENTATION ATTRIBUTE.

These video segments have been extracted automatically from popular YouTube videos related to presentation skills. Each segment will start from a time point in the video and will stop when the corresponding concepts are covered. You can re-watch the segment or watch other parts of the video, if you wish so.

You will now watch a video segment that mentions STORY (related to STRUCTURE) and MEMORABLE (related to PRESENTATION ATTRIBUTE). The segment is taken from a YouTube video with ID= yoD8RMq2OkU, and starts at 3.05 with duration of 41 seconds.



You will now watch a video segment that mentions START, BEGINNING (related to STRUCTURE) and ATTENTION (related to PRESENTATION ATTRIBUTE). The segment is taken from a YouTube video with ID= frIxD1/Xn44, and starts at 4.41 with duration of 42 seconds.





The video segments were automatically extracted from popular YouTube videos on presentation skills, using algorithms for segmentation, characterisation and linking based on a theory for concept learning. We hope you found the collection of video segments helpful.

Figure 5.6: Example of the Combinational Narrative to support learning the topics 'STRUCTURE' and 'PRESENTATION ATTRIBUTE' by introducing video segments that focus on different concepts belong to these topics.

5.3 Experimental Setup

We conducted two studies, Study-1 with experts and Study-2 with learners, to be able to answer RQ3 and RQ4. In both studies, the participants assessed the Quality and the Perceived Usefulness of a number of selected video narratives. Additionally, in Study-2, the participants also assessed the Cognitive Work Load and the possible Learning Effect of the generated video narratives. An overview of the materials used in this study and the evaluation result are presented in Figure 5.7.

Subsumption Linking Algorithms				Evaluation Study						
Video Narrative generation VIN-L framework				Evaluation Study with Experts				Evaluation Study with Learners		
Four Algorithms Narrative Type #Narratives AVG Duration STDV Derivative 22 199s 86 Super 14 224s 57 Ordinate 202s 59 Correlative 80 202s 59 Combinational 6 198s 106			Availity for learning Avroatic Stream S) is:	Quality for learning The result is statistically significant and the overall Quality for the narratives (out of 5) is: Narrative type AVG STDV Derivative (N=126) 3.64 1.06 Super Ordinate(N=98) 4.01 0.99 Correlative(N=114) 4.23 0.70			
Filtering Criteria: 2<=Number of segments per narrative <=6 180 s<=Duration of narrative <=300 s				Filtering Criteria: From each narrative type select the video narrative which has average quality > =4				Ombinational(N=96) 4.0 0.80 Outcome: All research questions have been answered		
Output				Output				positively:		
Narrative Type	<u>#Narratives</u>	AVG Duration	<u>STDV</u>	Narrative Type	#Narrative:	AVG Duration	STDV	RQ3:How to generate	good au	ality narratives that
Derivative	15	199s	86	Derivative	12	247s	34	can be used to support domain learning?		
Super Ordinate	13	237s	31	Super Ordinate	7	239s	33	RO4: Are the generate	ed narrat	ives useful to suppo
Correlative	51	239s	35	Correlative	21	246s	30	domain learning?		
Combinational	6	290s	11	Combinational 3 217s 117						

Figure 5.7: Implementation of VIN-L in the Giving Pitch Presentations domain.

The study was approved by the ethics committee of the Faculty of Engineering and Physical Sciences, University of Leeds.

Participants. For Study-1, we invited 11 domain experts who had substantial experience in giving presentations or had taught/assessed presentation skills. The participants included: 9 PhD researchers (7 from the University of Leeds, UK and 2 from the University of Canterbury, New Zealand) who were experienced with giving presentations- they all had given numerous presentations for conferences and had been involved in teaching; 2 academics (1 retired associated professor from the University of Leeds, and 1 professor from the University of Canterbury,

New Zealand) who had over 30 years of experience in teaching, presenting to a broad audience, and advising others on presentation skills, see Figure 5.8 for a summary of the domain expertise spread.

Convenience sampling was used for selecting participants. Following the ethics approval from the University of Leeds, each participant was given a $\pounds 50$ Amazon voucher as gratitude for their participation, as described in (Section 4.5.1).



Figure 5.8: Experts' profiles.

For Study-2, we used the Amazon Mechanical Turk marketplace 1 to invite workers to participate in the evaluation study to help in evaluating the generated video narratives. Accordingly, we have created an account and assigned as a 'requester' with Amazon Mechanical Turk marketplace which allowed us to create a project in which we designed 4 batches. A batch, as presented in (Appendix K), is where the Amazon workers (hereafter known as advanced learners) around the world can access the study link and participate in it. We have used some conditions to decide the workers who can participate in our study. First, they can speak English and more than 100 participants are English native speakers. Second, they are Master workers where 95% of their hits is approved in the surveys they participated in. Referring to Figure 5.10, those participants are advanced learners not actual learners as they have ample presentation skills knowledge (80%) which enables them to give useful feedback to the questions asked about learning this skill. However, their domain expertise might seen as a limitation because their domain knowledge might affect the pre-test result where they are able to notice the domain aspects clearly, as described in (Section 5.5). Their hits (answers) are saved in the batch answers which can be downloaded as an Excel sheet for analysis after the completion of the study. The number of

¹https://www.mturk.com/[Accessed:January 2023]

advanced learners who participated in the study was 237. This number is 61.5%of the usually required number 385^{-1} . We could not ask for the 385 number of participants from the beginning because of the budget constraints of this PhD research. This number has been filtered out by excluding the duplicated answers and the responses where learners did not answer the pre-test or post-test as this affects the analysis of the video narrative's learning effect. The number of learners, after applying the filters, was 217. Further discussion about the limitation of using this number of participants is discussed in (Section 5.6.5). We collected the learners' profiles, see Figures 5.9 and 5.10, showing that more than half of the learners were males from the age group ≥ 30 years, are employed and can speak English fluently. Additionally, more than half of the participants watch YouTube videos for learning and for other purposes. More than a third of the participants have a lot of training on presentation skills through training in the universities to pitch project presentations. Accordingly, they are classified as medium-to-advanced learners and their experience in presentation skills means they are able to give useful feedback to the questions asked about learning this skills.

¹https://www.calculator.net/sample-size-calculator.html?type=1&cl=95&ci=5& pp=50&ps=1000000&x=Calculate[Accessed: March 2024]



Figure 5.9: Learners' profiles which include: Age group, Gender, English native speakers and Job sector.



Figure 5.10: Learners' profile including data about: how often they watch YouTube videos, how often they watch YouTube for learning, the type of presentations they have given, the amount and type of training they had about presentation skills.

Materials. To provide public access for the video narratives, we converted the video narratives into web pages which we included in Google forms to be shared with the evaluators. The Google forms we designed have the same structure in both studies with a slight difference which is described below.

For Study-1, we designed five Google forms as evaluation surveys (survey-1, survey-2, survey-3, survey-4 and survey-5) and we distributed the 85 generated video narratives among them. We sent the survey links to 20 participants in a ratio of 4:1 to ensure that each video narrative is evaluated by 4 experts. However, only 11 have responded and for that reason the number of participants per survey is varied: 1 expert was assigned to survey-1, 4 experts were assigned to survey-2 and 3 experts have been assigned for each of the surveys 3, 4 and 5. Each video narrative has been evaluated by at least 2 experts except for the 19 narratives included in survey-1 which have been evaluated by 1 expert. The URLs of these narratives can be found in our GitHub account ¹. The surveys are fully compliant with the General Data Protection Regulation (GDPR).

The structure of the surveys is as follows: a consent form, a section to collect user profiles and then the narrative section which is divided into four groups where each group represents a narrative type (see Appendix J). In each narrative group there are two parts; the Quality categories after each video narrative link, and the Perceived Usefulness scale items at the end. The Quality categories aim to assess the quality of each video narrative and are designed using the LIKERT response labels (Taherdoost, 2019). The Perceived Usefulness of the generated narratives is assessed using a TAM questionnaire (Davis, 1989). At the end of each survey, we have asked the experts to type feedback to identify the possible positives and negatives of the video narratives they have watched. These two parts (the Quality categories and the Perceived Usefulness scale items) have been repeated for each narrative type.

For Study-2, we filtered out the video narratives based on the expert evaluation by selecting the video narratives that have a quality rank ≥ 4 , meaning it is a good quality narrative. This filtering of the video narratives might be seen as

¹https://github.com/Generating-Video-Narratives/Generating-Video-Narratives. git

biased and thus a further limitation of this study, see (Section 5.6.5). However, this filtering process is required to exclude the lower quality video narratives and reduces the number of video narratives to be evaluated with learners. This helps us using the resources of our study (number of participants and budget) efficiently. We have only selected 8 videos (2 videos from each narrative type) from the filtered video narratives, see Figure 5.11. We also considered selecting a variety of topics and concepts to avoid redundancy of the information presented to the users. The selected video narratives for the study, as presented in the Table 5.1, are included in 4 surveys, one for each video narrative type (Derivative, Super Ordinate, Correlative and Combinational). These surveys are designed using Google Forms with a structure similar to the expert surveys (see Appendix J). The difference from the expert surveys is that we included pre-test questions to collect the domain terms that learners know prior to watching each video narrative. Additionally, we have included a post-test to collect new domain terms that the learners recognised after watching the video narratives. Furthermore, to determine the learner acceptance of the generated video narratives as learning materials, we have included an extra part in the evaluation study to estimate the Cognitive Work Load resulting from using video narratives to support learning.

Table 5.1: The selected video narratives based on experts' evaluation study where the overall Quality ranking of a video narrative was ≥ 4 which is \simeq Agree. The narrative types are: (Combinational, Correlative, Derivative, Super Ordinate) and their duration, in seconds, is presented in the table as an average and standard deviation.

Narrative Type	No.Narratives	AVG Duration	STDV
Combinational	2	180.5	138
Correlative	2	244.5	11
Derivative	2	241	23
Super Ordinate	2	243.5	76

Evaluation Study with Experts							
Quality for learn	ning						
The overall Qua	ity for	the nar	ratives (out of 5)	is:			
Narrative type		AVG	<u>STDV</u>				
Derivative (N=33)		4.32	0.60				
Super Ordinate(N	=31)	4.08	0.82				
Correlative(N=112)		3.79	0.82				
Combinational(N	=33)	4.11	0.86				
Filtering Criteria From each narrat narrative which h	i: ive typ as ave	e select t rage qual	he video ity > =4				
Output							
Narrative Type	#Na	rratives	AVG Duration	STDV			
Derivative	12		247s	34			
Super Ordinate	7		239s	33			
Correlative	21		246s	30			
Combinational	3		217s	117			

Figure 5.11: Quality of the generated narratives of the Giving Pitch Presentations Domain as evaluated by experts. We have used the quality average value of ≥ 4 as filtering criteria to select from the video narratives which their quality match this criteria to be used for the evaluation study with learners.

Procedure. The evaluators in both studies (who participated via different Google forms links) followed the same procedure which is: (1.) Complete a short pre-study section to collect their profiles; (2.) Click on the video narrative link which will open in a new tab. They can also copy the link and open it in a browser themselves; (3.) Go through the video narrative page by following the descriptions and watch the suggested video segments which have a duration of 3 to 5 minutes; (4.) Go back to the screen with the survey and complete the scoring form for each video narrative and give suggestions, if applicable, for improving the narrative; (5.) Repeat steps 2-4 for each each video narrative link included in the survey. After each narrative type (D, S, CR and CO), complete the: (6.) Perceived Usefulness section; (7.) Further Feedback section. The difference in Study-2 is that we have a pre-test section which is part of the learner profile step, and a post-test to be completed after watching the video narratives. Additionally, in Study-2, there is a section for Cognitive Work Load. Example of the Google form used for the evaluation study is in (Appendix J).

5.4 Findings from Study-1: Experts

We analysed the outcome from the evaluation studies where we aggregated the outcomes of the 5 surveys. We reported and analysed the results of the Quality and Perceived Usefulness, and presented a qualitative analysis of the positive and negative feedback written by the experts after watching the generated video narratives.

5.4.1 Quality of the Video Narratives

We have aggregated the responses from 11 experts per video narratives who participated in 5 studies (different experts participate in different studies where they evaluated different number of video narratives from different types), see Table 5.2 and Figure 5.12. The aggregated responses are used to analyse the quality of the generated video narratives in order to answer RQ3.

Table 5.2: Expert Evaluation Results of the Quality of the video narratives. Each narrative type (CO, CR, D, S) has a number of assessments (N) and the Median, AVG and STDV scores of each Quality question based on the responses from the experts (1= Strongly Disagree, 2=Disagree, 3=Neutral, 4=Agree, 5=Strongly Agree). The difference between the narrative types is based on the Quality category and has been calculated using the Kruskal-Wallis test. The * indicates when p<0.05, the difference between the video narratives is statistically significant.

Video Narrative Quality	СО	CR	D	S	All
Categories	N=33	N=112	N=33	N = 31	N=209
Q1–The introduction text in the	5.0	4.0	4.0	4.0	4.0
video narrative gives a clear idea	4.39.0	3.88	4.12	3.94	4.0
of the aim of the narrative.	(0.79)	(1.07)	(0.82)	(1.03)	(1.0)
*Q2–All video segments are	4.0	3.0	4.0	4.0	4.0
clearly linked to the presentation	3.64	3.36	4.21	3.58	3.75
skills areas mentioned in the	(1.14)	(1.11)	(0.70)	(1.26)	(1.12)
introduction.					
Q3–The descriptions which	4.0	3.0	4.0	4.0	4.0
introduce each video segment	3.97	3.42	3.64	3.74	3.59
provide a useful summary.	(1.02)	(1.08)	(1.03)	(1.12)	(1.08)
*Q4–All video segments provide	4.0	4.0	5.0	5.0	4.0
relevant content for presentation	4.0	3.85	4.48	4.10	4.01
skills learning.	(1.12)	(1.16)	(0.83)	(1.19)	(1.13)
Q5–The concluding text at the end	4.0	4.0	5.0	4.0	4.0
of the video narrative provides an	3.91	3.66	4.21	3.90	3.82
appropriate summary.	(1.13)	(1.11)	(1.05)	(1.25)	(1.14)
OVERALL Quality–AVG score for	4.15	3.70	4.27	3.84	4.00
all Quality questions per narrative	4.01	3.63	4.13	3.85	3.80
type.	(0.59)	(0.64)	(0.66)	(0.57)	(1.09)
*USEFULNESS-This video	5.0	4.0	5.0	5.0	5.0
narrative is useful for learning	4.30	4.10	4.67	4.35	4.26
presentation skills.	(0.95)	(1.01)	(0.69)	(1.02)	(0.98)

5.4. FINDINGS FROM STUDY-1: EXPERTS



Figure 5.12: Quality of the generated narratives of the Giving Pitch Presentation Domain as evaluated by the experts.

Quality categories of the video narratives. In our analysis of the quality categories of the video narratives, we first refer to the column 'All' in Table 5.2 which shows an overview of the quality of each category based on the experts' responses. The experts found that the majority of the video segments have good quality with a median of 4.0 and an average of 3.80 (Agree). They also agreed that the video narratives provide content related to the domain (Q4) with a median of 4.0 and an average score of 5. Furthermore, they agreed that the domain's key areas are clearly mentioned in the introduction (Q1, with a median and average of 4.0) which helps the experts notice them in the content. As a result, the conclusion statement (Q5) was also found to clearly summarise the key domain

areas which are presented in the introduction and noticed in the segments (with a median of 4.0 and an average score of 3.82). Additionally, the experts found that the segments are linked clearly to the domain (Q2 with a median of 4.0 and an average of 3.75) and the description clearly describes the segment's content (Q3 with a median of 4.0 and an average score of 3.59). To assess whether these values are statistically significant, we ran the Kruskal-Wallis test. We found that there is a statistically significant difference in categories Q4 and Q2 among the narrative types with (p<.05). To identify which narrative type(s) caused this difference, we ran the Mann Whitney U Test to run a pair-wise comparison, as presented next.

Comparison between the narrative types. To analyse the difference in the quality of the narratives, we ran the Mann Whitney U test for a pair-wise comparison. The result was statistically significant with p < 0.05 between the narrative types D and CR. This indicates a statistically significant agreement among the experts that the Derivative narratives are better than the Correlative narratives in providing conceptual learning. Hence, Derivative type of linking (from generic to specific) seems to be more appealing than Correlative linking (learning similar sibling concepts). Experts' comments about the two narratives types CR and D, indicate that it is easier to notice the generic and specific concepts in the video segments in the D narratives, e.g.: " so clearly the segments are semantically very related to the topic."; and further clarification is needed in the CR narratives, e.g.: "I do understand why you use the word 'similar' in the context of the ontological hierarchy, but you need to think about does it make sense to the learner... may need some rephrasing.". There were no statistically significant results among the other pairs of video narratives. It should be noted that when comparing video narratives, we did not consider participants' learning styles or cognitive preferences, this is further discussed in Section 7.4.2.

5.4.2 Perceived Usefulness of the generated Video Narratives

In this part, we analyse the Perceived Usefulness results of the generated video narratives based on the aggregated experts' responses to be able to answer RQ4.

We aggregated the expert responses from all surveys on the Perceived Usefulness of the generated video narratives and they are reported in Table 5.3 and presented in Figure 5.13. We ran the Kruskal Wallis, Mann Whitney and Spearman's RHO tests but the results showed that there is no statistically significant difference among the narrative types in regard to the Perceived Usefulness scales. Consequently, we analyse the experts' responses presented in Table 5.3.

By referring to column 'All' in Table 5.3, we can conclude that the experts found the video narratives 'Quite Likely' to be useful when used to learn presentation skills. According to the scores of the individual narrative types, the D narratives scored the highest as they enabled the experts to identify key domain areas (PU1); be able to focus on them (PU2), identify the name of the focus topic and concept mentioned in the video narrative (PU3) and link the focus concepts together (PU4). However, the CR narratives scored slightly below the other narrative types where the experts found that linking concepts in breadth (sibling concepts) might misdirect a learner's attention as the type of linking is not clear. Similarly, the link between different concepts from different topics mentioned in the same video segments (CO narratives) causes confusion. According to the overall results of the Perceived Usefulness scores among the narrative types (median 6.06 and average 5.85 (STDV=0.86) out of 7), we can infer that the video narratives are 'Quite Likely' to be useful when used to learn presentation skills. Based on size of the study and the participant's expertise, we can deduce that the video narratives have a potential to answer RQ4.

5.4.3 Qualitative feedback from experts

In this part we summarise the experts' written feedback where they expressed what they found positive and negative about the video narratives.

Positives. These positives are classified into two groups; (a) Support focusing on specific domain concept(s) and (b) Support linking between domain concept(s).

(a) Support focusing on specific domain concept(s). The experts found that the CO narratives provide an overview of the concepts from different topics

Table 5.3: Evaluation Results of the Perceived Usefulness of the narratives. Each narrative type (CO=Combinational, CR=Correlative, D=Derivative, S=Super Ordinate) has 11 expert assessments and a Median, AVG and STDV score of their assessments as per the Perceived Usefulness category (1= Extremely Unlikely, 2= Quite Unlikely, 3= Slightly Unlikely, 4= Neutral, 5= Slightly Likely, 6= Quite Likely and 7= Extremely Likely).

Perceived Usefulness scale items	CO	CR	D	S	All
PU1–Identify key points related to	6.0	6.0	6.0	6.0	6.0
giving presentations.	6.09	6.09	6.09	5.82	6.02
	(0.83)	(0.94)	(1.04)	(1.25)	(1.02)
PU2–Focus on a certain key	6.0	6.0	6.0	6.0	6.0
point(s) at a time.	5.73	5.73	6.09	5.64	5.80
	(1.10)	(1.10)	(1.04)	(1.29)	(1.10)
PU3–Identify key topics/concepts	6.0	5.0	7.0	6.0	6.0
to learn.	5.82	5.64	6.36	5.82	5.91
	(1.17)	(1.03)	(0.81)	(1.08)	(1.02)
PU4–Link main points mentioned	6.0	6.0	6.0	7.0	6.25
in the video.	5.45	5.45	6.0	5.73	5.66
	(1.57)	(0.93)	(1.0)	(1.62)	(1.28)
Overall Perceived Usefulness	5.77	5.86	6.18	5.82	6.06
(AVG score for all Perceived	5.77	5.73	6.14	5.75	5.85
Usefulness scales per narrative	(0.40)	(0.59)	(0.44)	(0.60)	(0.86)
type.)					



Figure 5.13: Perceived Usefulness of the generated narratives of the Giving Pitch Presentations domain evaluated by experts.

and the relationship between them; "serves as an anchor to build on or start from if one topic is new", "connecting two concepts from different presentation skills areas helps one understand the idea in more breadth."

The CR narratives were found to aid focusing on relevant key domain areas by showing examples; "situates the concept within related concepts and areas", "they provide examples."

The D narratives were found to facilitate conceptual learning; "short, focused, saves time, and provides the context (the whole video) if we wish to go in more depth", "I think this generation process is very useful in enhancing derivative concept learning. So, it could make the teaching process and understanding of new concepts easier by providing such a way."

The S narratives were found to be helpful in providing in-depth learning; "they go deeply into more specific related topics", "some give actual key points that support the relevant skills." (b) Support linking between domain concept(s). The CO narratives support in-breadth learning by linking different concepts from different topics together: "connecting two concepts from different presentation skills areas helps one understand the idea in more breadth", "they link up pretty well."

The **CR** narratives provide examples which clarify the link between the target concept and its siblings, "I understand the concept more when it is linked to similar concepts from the same class. It is also felt that the videos are much more related and cohesive as compared to the earlier videos", "relevant videos are easy to understand and emphasize the most important point."

The **D** narratives were found to provide in-depth learning starting from the generic to specific concepts using short segments linked in one narrative: "knowing that a specific concept is part of a more generic concept. When watching the video covering the generic concept, a deeper learning may be triggered in the learner as he/she might try to work out the relevance of the selected segment", "it is more tangible".

The **S** narratives were found to support learning a generic concept by providing in-depth learning about it (its specific concepts): "defining the relationships between concepts, and somehow simplifying the concept by starting to explain parts of the whole idea", "they go deeply into more specific related topics".

Negatives. There are some issues associated with using video narratives for learning as specified by the experts which we have separated into two groups: "Confusion and distraction" and "Lack of elaboration".

(a) Confusion and distraction. In the CO narratives, sometimes the link between the concepts might not be obvious which causes attention-distraction: "real links might not be there", "it is possible though since there are multiple concepts being discussed that the learner might be confused (about) or might not be able to focus on the main point or topic."

With the **CR** narratives, focusing on more concepts within a narrative, even when they are similar, might cause distraction as the link between them might be vague: "might distract or confuse if concepts are on same level but not very similar", "it is sometimes confusing." With the **D** narratives, the linking of concepts from generic to specific might limit learning by not showing other related concepts: "a rather limited viewpoint is presented by just linking a specific concept to a generic concept. May be more useful to optionally reveal the other siblings under that same generic concept so that the learner understands the size of the learning space." Furthermore, the experts found that some segments start by carrying on what was previously mentioned in the video footage before starting to illustrate the main focus concept which caused confusion: "sometimes start of the video contains previous points" statement which could be confusing."

With the **S** narratives, a learning issue became apparent when the linking between the specific and generic concept was not clear enough which also caused confusion. This happened when the segments were short and did not elaborate enough on the focus concepts: "the sole use of linear style of linkage only reveals a fraction of the knowledge tree. There are usually other sibling concepts covered in the chosen segments and might confuse a beginner learner", "there is a need to explain or describe properly how specific concepts are linked to the main concept being learned."

(b) Lack of elaboration. With the CO narratives, it is found that the duration of the segments which focus on multiple concepts might not offer enough elaboration on these concepts: "it is possible though since there are multiple concepts being discussed that the learner might be confused or might not be able to focus on the main point or topic", "occasionally contains videos with less explanations." A similar issue appears with the CR narratives as the experts found that: "some segments were not stand-alone in explaining a concept well... can see that it was picked purely because the word was mentioned"

With the **D** narratives, learning only one specific concept from a generic concept might cause confusion by not mentioning the other specific concepts within the same generic concept: "a rather limited viewpoint is presented by just linking a specific concept to a generic concept. May be more useful to optionally reveal the other siblings under that same generic concept so that the learner understands the size of the learning space", "all specific features might not be covered." A similar issue was also noticed with the **S** narratives: "the sole use

of linear style of linkage only reveals a fraction of the knowledge tree. There are usually other sibling concepts covered in the chosen segments and might confuse a beginner learner", "I was quite confused as to what the specific concepts were in the 2nd and 3rd set of videos. It seems that for super-ordinate narratives, there is a need to explain or describe properly how specific concepts are linked to the main concept being learned"

According to the experts' evaluation results and considering their feedback, we decided to filter out the generated video narratives before evaluating them with learners. We have selected the video narratives based on their Quality result which should be ≥ 4 , meaning there is an agreement among the experts that these narratives have the potential to support domain learning, see Figure 5.11.

5.5 Findings from Study-2: Potential learners

We analysed the quality of the generated video narratives based on the aggregated learners' responses to be able to answer RQ3 and RQ4 based on the learners' point of view. This includes assessing the video narratives considering their: Quality, Perceived Usefulness, Learning effect, Cognitive Work Load and qualitative feedback.

5.5.1 Quality of the Video Narratives

The number of participants from the four surveys was 217 (CO-48, CR-57, D-63 and S-49) and each participant provided two responses as he/she watched two video narratives per survey. Accordingly, the number of the aggregated responses was 434 and they are reported in Table 5.4 and presented in Figure 5.14. The aggregated learners' responses of each narrative type per quality category are reported as the Median, Average (AVG) and Standard Deviation (STDV) scores, and the N value indicates the number of learner responses per narrative type. In our analysation of the learners' responses, we present two comparisons. One comparison is between the quality categories and the other comparison is between

the video narrative types.

Table 5.4: Evaluation Results of the quality of the narratives. Each narrative type (CO=Combinational, CR=Correlative, D=Derivative, S=Super Ordinate) has a number of assessments (N) and the Median, AVG and STDV score of each Quality question based on the responses from the learners (1=Strongly Disagree, 2=Disagree, 3=Neutral, 4=Agree, 5=Strongly Agree). The difference between the narrative types is based on the Quality category and has been calculated using the Kruskal-Wallis test. The * indicates when p<0.05, the difference between video narratives is statistically significant.

Video Narrative Quality	CO	CR	D	S	All
Categories	N=96	N=114	N=126	N=98	N=434
*Q1 –The introduction text in the	4.0	4.0	4.0	4.0	4.0
video narrative gives a clear idea	3.98	4.20	3.72	3.89	3.94
of the aim of the narrative.	(0.62)	(0.65)	(0.89)	(0.96)	(0.81)
*Q2-All video segments are	4.0	4.0	4.0	4.0	4.0
clearly linked to the presentation	4.07	4.32	3.67	4.18	4.05
skills areas mentioned in the	(0.80)	(0.66)	(1.09)	(0.93)	(0.93)
introduction.					
*Q3–The descriptions which	4.0	4.0	4.0	4.0	4.0
introduce each video segment	3.94	4.15	3.65	3.94	3.91
provide a useful summary.	(0.88)	(0.69)	(1.09)	(1.07)	(0.97)
*Q4–All video segments provide	4.0	4.0	4.0	4.0	4.0
relevant content for presentation	4.08	4.36	3.50	4.23	4.02
skills learning.	(0.78)	(0.67)	(1.17)	(0.92)	(0.98)
*Q5–The concluding text at the	4.0	4.0	4.0	4.0	4.0
end of the video narrative provides	3.93	4.13	3.67	3.83	3.88
an appropriate summary.	(0.95)	(0.84)	(1.04)	(1.07)	(0.99)
OVERALL Quality–AVG score for	4.0	4.0	4.0	4.0	4.0
all Quality questions per narrative	4.0	4.23	3.64	4.01	3.96
type.	(0.80)	(0.70)	(1.06)	(0.99)	(0.93)
*USEFULNESS-This video	4.0	4.0	4.0	4.0	4.0
narrative is useful for learning	4.19	4.31	3.60	4.16	4.0
presentation skills.	(0.79)	(0.63)	(1.13)	(1.05)	(1.01)



Figure 5.14: Quality of the generated narratives of the Giving Pitch Presentations domain as evaluated by learners.

5.5. FINDINGS FROM STUDY-2: POTENTIAL LEARNERS

Comparison between the Quality categories of the video narratives.

We refer to the column 'All' in Table 5.4 where we can get an overview of the quality of each category based on the 434 responses. The learners found that segments within the video narratives and their content are with good quality and related to the domain (AVG of Q2 is 4.05 and AVG of Q4 is 4.02). Additionally, the learners found that the characterisation (Q3) of the video segments enables the identifying of the domain focus topic(s) and concept(s) that was mentioned in the introduction (Q1) and conclusion (Q5) of the video narratives; (AVG of Q1= 3.94, AVG of Q3= 3.91 and AVG of Q5= 3.88 out of 5). We can conclude that the overall quality of the video narratives is good ($3.96 \simeq 4$). Additionally, the learners agreed the the video narratives are useful (AVG ≥ 3.60) for learning. We ran the Kruskal Wallis test to compare the responses on each Quality category collected from each narrative type. The result was found to be statistically significant with a p value < 0.05 on all Quality categories. To further investigate the difference in the Quality results, we do a pair-wise comparison between the narrative types by running the Mann Whitney U Test.

Comparison between the narrative types. To analyse to which extent the narratives were different from each other based on their quality, we ran the Mann Whitney U test to compare between narratives pairs. The result was statistically significant with p < 0.05 except for the narrative pairs CO-S and CR-S. Referring to the definition of these narratives types, with CR and S, the focus is only on one topic and they mention different concepts belonging to it (in-breadth linking with CR and in-depth linking with S). Similarly with CO, defined as different concepts from specific topics being mentioned in the same narrative to show a relationship between said concepts. This can be expressed as learning via giving examples, as shown in the S and CR narratives. According to the average quality values presented in Table 5.4, the CR narratives seem to be the most preferable narratives to be used for learning over the CO, D or S narratives. On the other hand, the least preferable narratives are of type D where learning a specific concept starts by learning its generic concept first (starts learning from generic to specific). It seems that the learners did not benefit from the conceptual learning presented with D narratives.

In this comparison we did not consider comparing the video narratives with other learning styles which is counted as a limitation, similar to the limitations mentioned in (Section 5.4.1).

In addition to the Quality evaluation of the video narratives, we have also evaluated the Perceived Usefulness and the Cognitive Work Load of using the video narratives for domain learning. The results are analysed in the next section.

5.5.2 Perceived Usefulness of the generated video narratives

In this part, we analyse the Perceived Usefulness results of the generated video narratives based on the aggregated learners' responses to be able to answer RQ4.

We have aggregated the 217 learners' responses from the four surveys on the Perceived Usefulness of the generated video narratives and they are reported in Table 5.5 and presented in Figure 5.15. We ran the Kruskal Wallis test and the results showed that there is a statistically significant difference among the narratives types in regard to the Perceived Usefulness scales. To identify the narratives that caused this difference, we ran the Mann Whitney U Test.

Perceived Usefulness. Referring to the column 'All' in Table 5.5, we can get an overview of the Perceived Usefulness of each narrative type based on the 217 aggregated learners' responses. We can conclude that the learners found that the video narratives are (=5.62 \simeq 6) 'Quite Likely' to be useful when used to learn presentation skills. According to the scores of the individual narrative types, the CR narratives score the highest as they enable the learners to identify key domain areas (PU1), focus on them (PU2), identify the name of the focus topic and concept (PU3) and link them together (PU4). Similarly, the links between different concepts from different topics, mentioned in the same video segments, were obvious in the CO narratives. However, the D narratives score slightly below the other narrative types especially when the learners were not sure how to link (PU4) the generic concept to the specific concept. We took into consideration that the difference in the Quality categories is statistically significant with p ≤ 0.05 meaning that there is a difference in the Quality of the narratives.

Table 5.5: Evaluation Results of the Perceived Usefulness of the narratives. The narrative type (CO=Combinational, CR=Correlative, D=Derivative, S=Super Ordinate) have assessed by the learners and the Median, AVG and STDV of their assessments per each Perceived Usefulness category (1= Extremely Unlikely, 2= Quite Unlikely, 3= Slightly Unlikely, 4= Neutral, 5= Slightly Likely, 6= Quite Likely and 7= Extremely Likely). The difference between the narrative types is based on the Perceived Usefulness scales and has been calculated using the Kruskal-Wallis test. The * indicates when p<0.05, the difference between video narratives is statistically significant.

Perceived usefulness scale items	СО	CR	D	S	All
	N = 48	N=57	N=63	N=49	N=217
*PU1–Identify key points related	6.0	6.0	5.0	6.0	6.0
to giving presentations.	5.71	6.09	4.75	5.78	5.54
	(0.92)	(0.79)	(1.79)	(1.30)	(1.38)
*PU2–Focus on key points at a	6.0	6.0	5.0	6.0	6.0
time.	5.83	6.19	4.75	5.86	5.62
	(0.91)	(0.90)	(1.94)	(1.26)	(1.47)
*PU3 –Identify key topics and	6.0	6.0	5.0	6.0	6.0
concepts to learn.	6.15	6.26	4.78	5.92	5.73
	(0.87)	(0.81)	(1.71)	(1.35)	(1.40)
*PU4–Link main points	6.0	6.0	5.0	6.0	6.0
mentioned in the video.	5.92	6.12	4.75	5.82	5.61
	(0.85)	(0.73)	(1.70)	(1.30)	(1.35)
Overall perceived usefulness	6.0	6.0	5.0	6.0	6.0
(AVG score for all Perceived	5.90	6.17	4.75	5.84	5.62
Usefulness scales per narrative	(0.89)	(0.81)	(1.79)	(1.30)	(1.40)
type)					



Figure 5.15: Perceived Usefulness of the generated narratives of Giving Pitch Presentation Domain as evaluated by the learners.

To ensure which video narrative(s) are different among the other narrative types, we ran the Mann Whitney test for a pair-wise comparison between the narratives. The results showed that there is a statistically significant difference among all the narrative pairs except for CO-S and CR-S. Based on the learners' feedback on the positives about CO, CR and S, we found that there is an agreement among the learners that these video narratives facilitate learning as presented in some of their responses; CR "easy to understand and absorb many key points.", CO "I like the brevity of the points. This makes things stand out and makes them easy to remember.", and S "Concept Learning Theory: the video narratives are created based on a theory for concept learning, suggesting a structured and methodical approach to presenting information effectively.". On the other hand, we found that the scores of the type D narratives were less than the scores for Usefulness of the other narrative types. Referring to the learners' feedback about what they did not like about the D narratives: "lack of human creativity: the narratives generated by AI models lack human creativity and intuition. This can result in predictable or formulaic story lines, which may limit engagement and fail to evoke an emotional connection with learners", "The generated derivative narratives can also be too generic and not tailored enough to the individual needs of the learners. Additionally, they can be difficult for learners to interpret as the generated narratives lack clarity and may fail to explain the concepts in a way that is meaningful to the learner." Next, we investigate whether this difference among the Quality and Usefulness of the video narratives affects the domain learning.

Learning effect of the video narratives. We compare between the learners' responses on the pre-test and post-test to assess the effect of the video narratives on domain learning. We have processed the terms of the pre-test and the post-test using our Semantic Tagging algorithm (Part of the VISC-L characterisation phase) to find a match between the learners' terms and taxonomy terms. Then, we counted only the new terms found in the post-test to show the possibility of using the video narratives to enhance domain learning. The results presented in Table 5.6 show that the learners managed to recall new domain terms after watching the video narratives. This result is in-line with the Quality scores presented in Table 5.4 where the learners found the video narratives are of good quality and hence useful for domain learning.

To help in clarifying in more depth the difference between the narrative types, we analyse the learners' feedback on the Cognitive Work Load of using the video narratives to support domain learning.

Cognitive Work Load. The aim of this part in the evaluation study is to assess the Cognitive Work Load expected while using the generated video narratives for domain learning. For this purpose, we have analysed the learners' responses on the NASA-TLX sub-scale items after watching the video narratives—we assessed the Mental Work Load, Effort, Frustration and Performance sub-scales and the results are presented in Figure 5.16. The results show that the D video narrative scores are the highest in all types of Work Load, whereas, the scores of the CO, CR and S video narratives were close to each other, to some extent, and generally lower than the scores of the D narratives.
Table 5.6:	The doma	ain related t	erms recalle	d by the le	earners a	fter watchi	ng the
video narrat	tives (Con	binational,	Correlative	Derivativ	e, Super	Ordinate)	

Narrative Type	Average (STDV) of new	Number of learners with
	domain terms per learners	New terms
Combinational	4 (4)	40 out of 48
Correlative	4 (5)	52 out of 57
Derivative	4 (5)	49 out of 63
Super Ordinate	4 (3)	45 out of 49

Referring to the learners' feedback about the Mental Work Load, Effort, Frustration and Performance scales resulting from watching the D narratives, some learners said: "had to remember details from the videos, and pay attention to the overall narrative structure and thematic elements so I could gain the most knowledge from them", "I had to focus on the material and take notes in order to comprehend and remember what I was watching", "I felt I needed more explanation and context around the concepts being explained in order to really understand them", "I was able to identify a number of new terms that I was not familiar with before. This is a good indication that I was able to learn new information from the videos. The clarity of the definitions: the definitions of the new terms were clear and easy to understand. This made it easy for me to learn and remember the new terms."

The feedback demonstrates the results in Figure 5.16 showing that regardless the high Mental Work Load, Effort and Frustration, the learners managed to increase their performance. On the other hand, the learners' feedback for the Mental Work Load, Effort and Frustration of the CO narratives is > 10 and the reason is presented in some of their responses: "I had to pay attention and think about what was said, but not at some high level. It wasn't super complicated stuff", and for the performance Work Load some said "there were a lot of useful phrases and ideas with which I may have already been familiar under different analogous terms.". With the CR narratives, the learners found it challenging to remember the concepts mentioned and recognise the link between them as mentioned in some of the responses: "Content Complexity: If the concepts discussed in the videos are complex or new to the viewer, it might require more mental effort to understand and process the information".

In this evaluation study we did not consider the learning styles of the participants or their cognitive preferences which we consider as a future work, see (Section 7.4.2). In the next part we summarise the positives and negatives of the video narratives based on the learners' feedback. This helps us draw some insights on how to improve the narratives in future work and to which extent the design of the narratives was successful.



NASA-TLX work Load scales

Figure 5.16: Cognitive Work Load estimation of the generated video narratives to support learning the Giving Pitch Presentations domain based on the evaluation study with the learners. The score range is 1-Low to 20-High.

5.5.3 Qualitative feedback from learners

We analysed the feedback written by the learners after they watched the video narratives to further assess their quality and effect to support learning. The learners found that the video narratives support domain learning by providing clarity through helping them to focus on specific domain areas, enhancement of their engagement, and the clarity of the transition between the segments to facilitate concept learning.

Positives. We have grouped the learners' positive feedback into three groups: (a) Clarity, (b) Engagement and (c) Transition between segments.

(a) Clarity. The clarity of the video narratives is shown by the video segments' content and textual format to present information which is delivered by various delivery methods. These segments that focus on a specific topic(s) and concept(s) provide concise and understandable learning materials. This enables the learners to identify specific presentation skill areas to learn and link to their prior knowledge.

There are 23 learners out of 48 who evaluated the **CO** video narratives and appreciated that these narratives gave them presentation delivery ideas and recommended steps on how to implement these ideas.

For the **CR** video narratives, 17 learners out of 57 appreciate that these narratives are understandable and provide some suggestions on how to improve pitching presentations.

There are 10 out of 63 learners who found that \mathbf{D} video narratives provide understandable learning materials that are easy to absorb and use for self-reflection while practicing for delivering presentations.

24 out of 49 learners noticed that the \mathbf{S} video narratives help them link new learning materials to their previous experience and gave them useful presentation ideas on what to focus on and what to prevent.

(b) Engagement. Maintaining a learner's engagement while watching the video narratives is one of the most vital aspects. This includes helping the learners be more organised while learning specific concepts and noticing the relationships between them to be able to apply them in their presentations. Furthermore, the different audiovisual video content supports a wide range of different learning styles.

With the **CO** video narratives, 7 out of 48 learners expressed their engagement with the video narratives as this narrative type offers ideas on how to organise

presentations and understand which presentation's attribute to focus on while preparing for pitching.

For the **CR** video narratives, 12 out of 57 learners found that these narratives gather various related ideas from different videos where they have been delivered differently to facilitate their learning.

The engagement with the \mathbf{D} video narratives is noticed by 21 out of 63 learners who suggest that these narratives can be beneficial for different learners based on their needs. If used in an educational environment, they will help in providing inclusiveness.

For the **S** video narratives, 5 out of 49 learners noticed that this type of narrative provides small parts of learning materials that are structured in a story-line which helps in enhancing learning specific concepts.

(c) Transition between segments. The relationship between the focus topic(s) and concept(s) mentioned in the segments represents a dynamic digital story for learning. The segments within the video narratives help in highlighting key areas and provide direct access to them which helps in easily digesting the information and reducing distraction.

With the **CO** video narratives, 8 out of 48 learners noticed that this type of narrative links several concepts from specific topics which helps learners understand the relationship between these concepts in a more memorable way.

For the **CR** video narratives, 13 out of 57 learners noticed that this narrative presents the similar concepts in a story-line way which helps in recognising the relationship between them.

The **D** video narratives have been found by 4 out of 63 learners to help in expanding their domain experience by providing a clear connection between the concepts mentioned within the video narratives.

Similarly, with the \mathbf{S} video narratives, 6 out of 49 learners liked the structure of the concepts by linking them to their generic concept which helps to quickly learn some areas in the domain without losing concentration.

Negatives. There are some aspects in the video narratives where the learners found that improvement would be beneficial to boost the quality of the video

narratives to better support learning. We have grouped the negative feedback of the learners into four groups: (a) Segment duration and ending point, (b) Segment Focus Topic and Focus Concept, (c) Video content and (d) Learner interactivity with the video narratives.

(a) Segment duration and ending point. The duration of the segments and their ending point within the videos caused some issues as shown the learners' feedback. With the CO narratives, these issues were: "not enough detail", "the clips sometimes stopped earlier than I think they maybe should have, but then I could always investigate the rest on my own." Meanwhile, with CR, the issues were: "sometimes the information was not obvious as the video played and I had to watch a couple times", "I could see some of the clips being a bit too shortened and not having proper information." With D, the issue was: "there is not enough time to successfully implement differentiated instruction." With S, the learners found that: "the videos cut off too quickly for me. I wanted to listen to more of the presenter's talk because they seemed to be very helpful."

(b) Segment Focus topic and Focus concept. With CO, this issue was: "some concepts are not relatable", "some videos couldn't connect the link properly and one of them seemed like it was chosen randomly." With CR, the learners did not notice this issue but they did notice it with the D narratives as they said: "while they provide a great overview of the main points and the relationships between them, they may miss some of the finer details that could be important for a successful presentation." With S, the learners found that: "some of the times they really did not deliver in the video segments according to the topics."

(c) Video content. The quality of the video itself might not suit many learners. With CO, the learners found that: "the worst part is that people are less likely to check the wrong facts when they are presented them in the form of story", "they can be a little dull, and they are teaching me to be the opposite." With CR, the poor delivery of the information affects the learning: "for videos about presentations, some of them were boring and with no charisma." With S, the learners also mention this issue: "the quality of the video might differ which

affects the quality of the learning process."

(d) Learner interactivity with the video narratives. More interactivity tools would be beneficial for learners. With CO, they said: "can encourage passive learning, where learners simply watch the content without actively engaging with the material. This can limit critical thinking and problem-solving skills that active participation and hands-on activities promote." With CR, the learners found that: "interactive elements like quizzes, activities, and immediate feedback might be missing in the video-based learning approach, limiting the range of engagement". With S, the learners found: "Lack of Interaction: video narratives might not allow for direct interaction with instructors or peers, limiting opportunities for asking questions or discussing concepts."

However, with **D**, the learners did not notice the issues associated with the video contents or the learner interactivity. Alternatively, they asked for more clarification about the relationship between the generic and focus concept in the text used within the video narratives; "they lack the nuance and creativity present in human-created stories. Generated narratives are often formulaic and lack the emotional depth and resonance of real stories. Additionally, because of the data-driven approach used to generate these narratives, they may lack context relevant to a given situation or environment, which could lead to a poorly informed decision-making process. Lastly, the literature surrounding the use of generated narratives in education is still in its infancy, meaning that there is limited research-based evidence to support its use in educational practice", "Lack of human creativity: the narratives generated by AI models lack human creativity and intuition. This can result in predictable or formulaic story-lines, which may limit engagement and fail to evoke an emotional connection with learners."

5.6 Discussion

We have implemented the VIN-L framework in our prime domain 'Giving Pitch Presentations'. The input for this framework was the characterised video segmentsthe VISC-L output from Chapter 4. The domain taxonomy used to characterise the video segments in Chapter 4 is also used to link the video segments to generate the video narratives. The generated video narratives have been evaluated using two separate evaluation studies; Study-1 and Study-2. Study-1 involved 11 domain experts who are PhD researchers and a retired lecturer from the University of Leeds, along with PhD researchers and a professor from the University of Canterbury, New Zealand. Study-2 was conducted by inviting 217 workers (learners) via the Amazon Mechanical Turk marketplace to assess the generated video narratives to support perceiving domain learning. The aim of the evaluation studies was to be able to answer RQ3 and RQ4 when video narratives are used with learners with medium to advanced expertise but not beginners.

We discuss the effectiveness of the generated video narratives to support perceiving domain learning and the difference between them by comparing the video narrative evaluation results and feedback from both studies with experts and learners. We did not compare like with like and we did not compare with learning styles or cognitive preferences which can be considered as future work, see (Section 5.6.5).

5.6.1 RQ3: Quality of the video narratives.

In this part we discuss to what extent we have answered RQ3 by combining the evaluation results from Study-1 and Study-2. Both experts and learners expressed that the quality categories Q2 and Q4 of the generated video narratives differ significantly. For Q2, which is 'video segments are clearly linked to the presentation skills', mentioned in the introduction, there was not an unanimous agreement on it. This means that the introduction was not clear. Although there was different opinions about the quality of the introduction, they did not unanimously agree on which video narrative had the worse quality; either CR from the experts point of view or D from the learners point of view. The experts' qualitative data about the video narratives, and not enough **elaboration** provided on the focus concept in the video segments. A sample of the expert qualitative data is: "might distract or confuse if concepts are on same level but not very similar", "some segments were not stand-alone in explaining a concept well. Can see that it was picked

purely because the word was mentioned", "few of the videos are not about what the descriptions said." Similarly, the learners found that the D narratives had this issue as well which is expressed in the learners qualitative data: "Lack of human creativity: the narratives generated by AI models lack human creativity and intuition. This can result in predictable or formulaic story-lines, which may limit engagement and fail to evoke an emotional connection with learners", "there is not enough time to successfully implement differentiated instruction."

According to the qualitative data of the experts and learners, the narratives' introduction **text** needed more clarification. The clarification process includes clarifying in more depth the learning objectives of the video narratives especially for the D and CR video narratives. This includes clarifying the relationship between the **Focus topic and Focus concept** mentioned in the introduction. Although both experts and learners agree that they can notice the Focus topic and Focus concept mentioned in the video segments and link it to the introduction description, there are some exceptions found in the CR and D narratives. For example, the experts managed to notice the hierarchical connection between the Focus topic and Focus concepts based on their domain experience but the learners were not able to notice that. Hence, for the learners with a lack of domain knowledge, additional text in the introduction is necessary to better articulate the conceptual connection between the generic and specific concepts and notice the nature of the relationship in more depth.

The experts noticed a difference in two Quality criteria (Q2 and Q4) which is not the case with the learners who did not concur on any of the Quality criteria. We noticed that there is a statistically significant difference in the Quality of the generated video narratives when assessed with learners. For that reason, we ran the Mann Whitney U Test for a pair-wise comparison between the narrative types, (a) CR and CO and (b) D and S, based on the expert and learner results to identify which video narrative(s) is/ are different from the other types based on the two evaluations.

(a) CR and CO video narratives. The strength, as judged by the learners, is that these narratives are useful as they diversify the learning experience by highlighting different aspects of specific domain areas. Hence, with the CR

narrative, they see same level concepts and with the CO narrative, they see the relationship between concepts from different topics. With both narrative types, there is a required improvement in the introduction and conclusion. This will help in better conceptualising the Focus topic and Focus concepts. On the other hand, the experts were more criticising about the CR and CO narratives as they found that linking many concepts together provides broad cover and a lack of in-depth focus.

(b) **D** and **S** video narratives. The experts prefer these narrative types for learning whereas the learners had some concerns about them. Referring to the definition of these narratives where the hierarchical Up-Down or Down-Up relationship was found to be useful for conceptual learning based on the experts' feedback, the learners found them challenging to grasp. This can be solved by clarifying said hierarchical relationships in the introduction.

The recommended changes to the introduction of the video narratives have been addressed when implementing VIN-L in the second domain. This was done by adding a shape to help visualise the type of linking in the introduction, thus clarifying the conceptual relationship between the Focus topics and concepts based on the domain taxonomy. Additionally, a further in-depth study can be conducted as future work to select the best text style for the introduction of the video narratives.

Furthermore, the duration of the selected segments should be ≥ 30 seconds and adding more segments to the video narratives should help in providing greater elaboration on the Focus topic and Focus concepts.

5.6.2 RQ4: Usefulness of the generated video narratives.

From the experts, we attained the Perceived Usefulness and from the learners, we analysed the actual usefulness and the Cognitive Work Load. Referring to Figures 5.13 and 5.15 (the actual numbers used to draw the pie-charts are in the tables included in Appendix L), the experts and learners had ranked the video narratives highly. They found that the video narratives are useful in helping to identify key domain points and in being able to focus in on the key points.

They also agreed that the narratives help them identify concepts to learn. The experts' perceptions were homogeneous, however, the learners agreed that the video narratives are useful but they found differences in the Perceived Usefulness among the video narratives. The learners had different perceptions about the narratives' Quality and this has been reflected in the Perceived Usefulness results. To identify the narrative type(s) that caused these differences, we ran the Mann Whitney U Test for a pair-wise comparison between the narratives. This showed that the D narrative result is significantly different from the other narrative type results. The Quality of the D narratives affected the grasping of the Focus Topic and Focus concept which affected this narrative's usefulness for learning.

Below, we will summarise the strengths and areas for improvement as recommended by the experts and learners.

5.6.3 Strengths

The video narratives, when used with learners with medium to advanced expertise to support domain learning, have a potential to enhance domain knowledge. Some insights have been inferred when analysing the evaluation responses from both experts and learners. The experts found that the video narratives offer a good focus on key domain areas and have a potential to support the linking between these areas. The learners also noticed some strengths in the video narratives; the videos provide a clear explanation of the domain's key points, and the transition between the segments helped them notice the link between the concepts.

5.6.4 Areas for video narratives' improvement

The experts and learners who evaluated the Derivative narratives found some issues in their quality that affects their usefulness for learning. Experts also found some issues in the Correlative video narratives that provide in-breadth learning and the learners found that the conceptual learning in D narratives needed more clarification. This can be seen clearly in the learners results about the narrative's Cognitive Work Load. They showed that the Frustration scale with the D narratives was the highest which can be linked to the lowest Quality and Usefulness rank of this type. Hence, more Mental Work Load and Effort is needed while using this narrative type.

Further improvements of the narratives, as recommended by the experts and learners, include: improving the text used to augment the video narratives to clarify the relationship between the Focus topic(s) and Focus concept(s) which is described in (Section 7.4.1), and selecting more video segments from the same videos that have the same Focus topic and Focus concept to elaborate more on them. Also, increasing the text window of the video segments to be more than 6 lines so the segment itself will add more elaboration on the Focus topic and concept, see (Section 5.6.5). According to the expert evaluation, there is a need for future work to conduct a study where we will automatically generate different summarisation styles—which we have not done in this study. Then, evaluate these summarisation styles with experts and learners to assess their effectiveness in conveying the narratives' learning objectives, see (Section 7.4.1).

5.6.5 Limitations of the studies

Expert-Participants convenient sampling. The selection of the experts number is done using a convenience sampling which is a non-probability sampling method. The participants were researchers and lecturers who were available at the time of the study and showed willingness to participate in the study.

Learner-participants selected in the study. The ideal number of participants from Amazon Turk, which will enable us using parametric statistics, is 385 participants according to sample size calculation. The actual number of the participants is a representative number which we see as a sufficient number in order to demonstrate some statistical significance using non parametric statistics.

Hence, the smaller size is selected in this study due to budget constraints and our decision for using the most expensive Amazon workers as they are more experienced and reliable participants. Those participants have domain knowledge which can be useful to give better quality answers for the questions about the learning effect of the video narratives. However, this knowledge can affect the result of measuring the learning effect of the video narratives as those participants managed to give high number of domain related concepts which are to some extent similar in both pre-test and post-test. Accordingly, those participants are classified as medium-advanced learners. Further research would be needed to validate the usefulness of the narratives for beginner learners who have no experience with presentation skills.

Comparing between both studies (expert and learners). We have used the experts results to filter out the video narratives that are used with learners study. This is to ensure that the good quality narratives are going to be presented to the learners and it also enables us to narrow down the number of the video narratives used with the Amazon Mechanical Turk workers. Hence, we are not comparing like to like in both evaluation studies (with experts and learners).

Sample Size of Study-2. The participants in the learner study are medium to advanced learners but not beginners. Those participants with high experience gave us good feedback about the narratives and appreciate them. However, we can not conclude that the narratives can be useful for beginner learners as we did not embed the study in a module which is considered as a future work, see (Section 7.4.2).

Experimental design. We do not have similar system to compare with. The only possible way is to compare between learners who watch only the video narratives with learners who watch the whole YouTube videos. This comparison is not applicable according to the time constrain where the learners who use YouTube will watch hours of video footage which is not fair for them. accordingly, we decided to compare between the narratives themselves.

Segment window size. The segment window size of 6 lines is not the final segment size as these segments are aggregated with adjacent segments with the same Focus Topic and Concept. We notice that we can increasing the size of starting with 6 text lines and deciding the segment cut-point to be where the nearest end of sentence punctuation is found. By doing this we aim to tackle the issue of segment duration and to avoid the abrupt segment ending. This decision on initial segments has been implemented in (Chapter 6).

5.7 Summary

In this chapter we have proved the applicability of the VIN-L framework by implementing it in our prime domain, 'Giving Pitch Presentations'. Our contribution to this domain is through generating video narratives relating to presentation skills by re-using the available YouTube videos and linking different parts of them for domain learning purposes. Thus, we have generated 85 domain related video narratives. A filtered group of these narratives has been used in two evaluation studies. The first study was conducted with experts from the University of Leeds and the University of Canterbury in New Zealand. The second evaluation study was conducted with learners invited using the Amazon Mechanical Turk marketplace and both studies aimed to answer the two research questions:

How to generate good quality narratives that can be used to support perceiving domain learning?

Are the generated narratives useful to support perceiving domain learning?

These research questions have been addressed in both studies and the findings showed some positives and areas for improvement. The results revealed that the generated narratives are of good quality and the video segment content is related to the domain key concepts. Furthermore, the video narratives are potentially useful if they are used as learning materials to identify key domain concepts and to realise the links between them. The experts results showed that the D narratives were preferred over the other types, and the CR narratives were the least preferred type. However, the result was not significantly different due to the limited number of experts who participated in the study. On the other hand, the learners preferred the CR narratives over the other types and the D narratives were the least preferred type. The result was significantly different except for the video narrative pairs CO-S and CR-S. According to the feedback from both experts and learners, areas for improvement have been indicated and are mostly about the quality of the narratives. From the experts' point of view, more elaboration on the Focus topic(s) and Focus concept(s) can be achieved if more domain videos are used. Whereas, from the learners' point of view, the reshaping of the text used inside the video narratives can be enhanced to be more user friendly. Hence, this can improve the interactivity of the learners with the video narratives to improve their learning outcomes. With regards to the usefulness, the overall ranking was positive from both the experts and learners. We found that improving the quality of the narratives will enhance their usefulness as whenever there were positives and negatives in the Quality, they were noticed in the Usefulness as well. The recommended improvements to the quality are addressed in our second domain in Chapter 6.

The learners managed to identify new domain terms after watching the video narratives but this cannot be used to ensure deep learning as can be otherwise assessed in different studies where the video narratives are parts of modules dedicated to teach presentation skills. This requires a larger study which is beyond the scope of work in this thesis.

For the next domain, the following improvements have been incorporated: including figures in the augmented narratives introduction, improving the semantic tagging algorithm to include n-grams to recognise long domain terms in the video transcript, increasing the size of the segments to be ≥ 6 lines as the segment cut-point is where the nearest end of sentence punctuation is found. By doing this we aim to tackle the issue of segment duration and abrupt ending.

To prove the generality of our methodology, in the next chapter we implement VISC-L and VIN-L in a different domain: Health Related Quality of Life Awareness.

Chapter 6

Implementation and Validation of VISC-L and VIN-L in the Health Domain

6.1 Introduction

In this chapter, we aim to prove the generality of our work by implementing our frameworks VISC-L and VIN-L in another domain as described in (Chapter 3). The second domain is called Health Related Quality of Life Awareness. The generated video narratives are used to learn domain topics and concepts and raise awareness of health related quality of life of patients need who live with chronic illness-respiratory illness. Our approach is knowledge driven and requires a taxonomy of domain knowledge. Since such a taxonomy is not available, we build it. Building the taxonomy manually caused our method to be semi-automatic; however, domain taxonomy can be created automatically using automatic ways which enables our method to be fully automatic. Similarly to the Presentation Skills domain (Chapter 4 and 5), we use this taxonomy to collect domain related videos and to aid the implementation our frameworks.

The application of VISC-L and VIN-L is part of the **InAdvance project**¹. The goal of this project is to "propose a new model of palliative care (PC) based

¹https://www.inadvanceproject.eu/[Accessed:April 2023]

on early integration and personalised pathways addressed specifically to older people with complex chronic conditions." Our contribution to **InAdvance project** is to help with the communication when addressing patients, which is part of ecosystem support. There is a need to analyse the palliative care conversations as they relate to a patient's quality of life and are complex as they can happen in different contexts. Consequently, the awareness of health related quality of life challenges faced by patients in need of palliative care will increase. Raising the awareness of this domain would benefiting the general public, patients' relatives and medical students. Consequently, in this chapter we generate video narratives to support learning main Health Related Quality of Life aspects and raise awareness of theses concepts and their possible impact on patients' health. For that purpose, we aim to answer the following research questions:

RQ3: How to generate good quality narratives that can be used to support perceiving domain learning?

RQ4: Are the generated narratives useful to support perceiving domain learning?

To be able to answer these questions, we generate the domain related video narratives and evaluate them with InAdvance project team and researchers from the University of Leeds. Accordingly, we first have collected videos where elderly patients with respiratory illnesses share their stories and tutorial videos designed by domain professionals. After that, we have implemented the VISC-L framework defined in (Section 3.5) to produce characterised video segments from the patient story videos. After that, we have implemented the VIN-L framework defined in (Section 3.6) where the characterised video segments (VISC-L output) were linked to generate video narratives. Additionally, the design of the frameworks presumes that the domain taxonomy is available. We have developed the domain taxonomy which is a key input for both VISC-L and VIN-L. It is used to automatically collect domain related videos by implementing the search schema presented in (Section 3.5.1). It guide the characterised video segments to generate the video narratives, the domain taxonomy help in implementing the subsumption linking algorithms presented in (Section 3.6).

The structure of this chapter is as follows: the building of the domain taxonomy is presented in (Section 6.2), in (Section 6.3) we use the domain taxonomy to collect the required domain related videos, (Section 6.4) presents the implementation of the VISC-L framework and in (Section 6.5) we present the implementation of the VIN-L framework. To evaluate the Quality (RQ3) and the Perceived Usefulness and the Cognitive Workload (RQ4) of the generated narratives, we have conducted an evaluation study with experts. The study and the results are presented in (Section 6.6). The work in this chapter is summarised in (Section 6.7).

6.2 Health Domain Taxonomy

Our approach is using computational knowledge to automate the generating of the video narratives which is our ultimate goal in this thesis. Therefore, any domain we tackle should be represented computationally via a domain taxonomy. Health taxonomies represent health related knowledge where they provide the description of the main medicinal terminologies (concepts in the domain taxonomy) and the relationships between them. In this way, health knowledge will be shareable and accessible to maintain effective communication between specialists and patients. Here we revise the design of health ontologies; although in our work we focus on the domain taxonomy only.

6.2.1 Background

To facilitate the communication between patients and service providers, health terminologies should be specified and their structure should be identified. This evokes the need to build domain related ontologies and taxonomies with a focus on their quality. To build a "Smart Health-Care Ontology" (Tiwari & Abraham, 2020) and to maintain its quality, different tools have been utilised: Themis, Test-Driven Development Onto (TDD)onto, Protégé and OOPs. Themis and TDDonto have been used to verify the ontology and Protégé and OOPs have been used to validate the modelled knowledge of the ontology.

Researchers found it is easier to build ontologies for specific aspects in the domain where the required concepts are collected using different applications and technologies. For example, Shahzad *et al.* (2021) designed an ontology to find the differences and similarities between different health services and their activities based on the "Internet Of Things" health applications, e.g., in Disease, Monitoring, Telemedicine, and Remote Health. The building of such an ontology is to provide the construction that links the similarities and differences among health services in one place for the benefit of both patients and specialists. Similarly, Mohamad *et al.* (2021) presents an overview of the main concepts related to the Virtual Reality Training (VRT) for medical education in the healthcare domain. The aim is to help in establishing a strategy to design an ontology to aid the development of VRT for medical education.

Nevertheless, Spoladore & Pessot (2021) stated that among the most promising technologies used to manage health data in health systems are ontologies and semantic webs. However, the methodologies used to build such ontologies are not free of deficiencies (feasibility study, project management, reuse, and stakeholders involvement). To explore the possible solutions to the aforementioned deficiencies, a comparison between three methodologies (life cycle, waterfall, agile) have been conducted. The findings showed that each methodology managed to tackle the aforementioned deficiencies based on the approaches and tools used in them. Accordingly, a collaborative work between the stakeholders and tools to be used to build health ontologies.

For health related quality of life, there is no identified taxonomy that links all the domain concepts. Hence, we consolidated knowledge from different researches to design the domain taxonomy. This taxonomy identified the domain classes, topics (top classes in the domain taxonomy) and individuals. As we mentioned in (Section 3.4–Definition 2, we will refer to the domain taxonomy classes and individuals as concepts) and the top concepts in the taxonomy as topics. The taxonomy we build in this section focuses on health related quality of life of patients with respiratory illness. Though, it can be used with any chronic illness by adding illness-specific physical symptoms.

The design process of the health related quality of life taxonomy is presented below.

6.2.2 Taxonomy Design

The aim of this taxonomy is to provide a computational representation of the health related quality of life for patients with respiratory illnesses. Hence, it will identify the main concepts and their hierarchy. This taxonomy supports researches that consider health text as a source of information or input data in their work. It aids the process of identifying the presence of the domain concepts in different parts of the text and links these parts using the concepts hierarchy. This will allow for extracting different insights that can be adopted by health providers to determine which services to offer, e.g, how to maintain a patient's health, understand their disabilities, and to prevent possible diseases.

To build the health related quality of life taxonomy, we have been inspired by the iterative ontology development methodology (Fernández-López, 1999; Noy *et al.*, 2001). Since our aim is to build the domain taxonomy only, we are going to follow part of the steps of the iterative ontology development methodology as presented in Figure 6.1.



Figure 6.1: Iterative Taxonomy Development Methodology inspired by the ontology building steps (Fernández-López, 1999; Noy *et al.*, 2001).

The ontology has been saved using GitHub¹

Initial Phase. In this phase, we will identify the: Purpose, Scope, Level of Formality, Intended Users and the Intended Use. We will also add the step Requirements.

(a) Purpose. The aim of this taxonomy is to support automating the analysis of conversations with patients/carers or patient/carer stories. With this analysis we can identify the main aspects related to the patient's quality of life. Additionally, this taxonomy can provide the knowledge source for analysing patient stories text corpora and conversations, and can also be helpful when supporting Personal Care (PC) needs. For this propose, it is essential to identify and provide the key terms (concepts) that relate to the health related quality of life domain taxonomy.

(b) Scope. The scope of the taxonomy is the Quality of Life of patients with Chronic Illnesses. The main chronic illness that we focus on in this thesis is the respiratory illness, but other groups of chronic illnesses can also be applied (e.g. cardiovascular). The main domain terms identified are linked to the WHO Health Related Quality of Life Knowledge sources (WHOQOL-100²) and to the relevant ontologies (MESH ³, COPD ontology ⁴).

(c) Level of formality. We decided to represent the domain purely as a taxonomy which means we will not consider some of the design aspects used when building ontologies (functions, individuals or instances and axioms).

(d) Intended users. They are those who would like to develop an analysis of patients' corpora (patient texts and stories) and want to identify the health related quality of life concepts present (in the text). Furthermore, people who

git[Accessed: January 2023]

¹https://github.com/Health-Related-Quality-of-Life-Ontology/Ontology.

²https://www.who.int/[Accessed:October 2022]

³https://www.ncbi.nlm.nih.gov/mesh/68008171[Accessed:October 2022]

⁴https://bioportal.bioontology.org/ontologies/COPDO/?p=

summary[Accessed:October 2022]

seek in depth knowledge about health related quality of life for patients can also use this domain taxonomy as a guide.

(e) Intended uses. It can be used for text analytics, as guidance for application materials, for training (what aspects to include in the training), can be used for advice for other people who need it, and can be used by clinics to train patient carers about aspects of the patient's quality of life.

(f) Requirements. We need to make sure the domain taxonomy focuses on respiratory illness but it should be possible to expand it further to include other chronic illnesses.

Design Phase. In this phase, we will identify the main concepts, sub concepts and their hierarchy in the domain. The concepts in the domain taxonomy are linked as Top-Down where the top level concepts are the taxonomy topics, the second level includes the generic concepts and the remaining levels include the specific concepts within the generic concepts.

To identify the main domain concepts and their granularity, we used the following health related websites as knowledge sources: Assessing Health-related Quality of Life (HRQoL) in achondroplasia¹, Needs Assessment Tool: Progressive Disease (NAT: PD) User Guide², Physical Health Components³. Accordingly, six domain topics (top level concepts in the taxonomy) have been identified: PHYSICAL HEALTH, PSYCHOLOGICAL HEALTH, LEVEL OF INDEPENDENCE, SOCIAL RELATIONSHIP, ENVIRONMENT and PERSONAL VALUES AND BELIEFS. The main topics have been linked to their sub concepts following the granularity structure identified in the aforementioned health websites.

To evaluate the accuracy of the defined topics, concepts and their hierarchy, we have passed the taxonomy structure through iteration feedback–3 iterations.

¹https://www.beyondachondroplasia.org/en/about/welcome[Accessed:November 2022] ²https://www.caresearch.com.au/Portals/20/Documents/Health-Professionals/

NeedsAssessmentTool-ProgressiveDiseaseCHeRP.pdf[Accessed:November 2022] ³https://study.com/learn/lesson/what-is-physical-health.

html[Accessed:November 2022]

The evaluation, following the iteration feedback, was with our InAdvance partners, where one evaluator participated in the first iteration, one evaluator participated in the second iteration and four evaluators participated in the third iteration. First, we assessed the taxonomy structure with the first evaluator who is a project officer at (Age¹, Brussels, Belgium). The feedback we had got was to make the following changes: expanded PERSONAL VALUES AND BELIEFS by adding 'LIVING ALONE', expand SOCIAL RELATIONSHIP by adding 'LONELINESS', 'ISOLATION', and add the sub concept 'SENSE OF ISOLATION' to the generic concept 'NEGATIVE FEELING' within the topic PSYCHOLOGICAL HEALTH. Additionally, we replaced the concept 'CLIMATE' within the generic concept 'PHYSICAL ENVIRONMENT' with the specific concepts; 'HEAT', 'DRAUGHT', 'HUMIDITY', 'COLD'and 'FLOOD'.

After the changes we have made to the domain taxonomy, we commenced the second iteration to get the second feedback to check the validity of the resulting taxonomy. For this purpose, we have asked the second evaluator, who is working in Medical Informatics Education in the School of Medicine in Aristotle University of Thessaloniki, Greece, for feedback. In response, new concepts have been recommended; the 'ACTIVITIES OF DAILY LIVING' and the 'INSTRUMENTAL ACTIVITIES OF DAILY LIVING' to the topic LEVEL OF INDEPENDENCE. We researched these generic concepts and found the most related concept to add to the domain taxonomy is 'ACTIVITIES OF DAILY LIVING' which includes 5 sub concepts: 'TOILETING', 'SHOWERING', 'BATHING', 'FOOD PREPARATION' and 'CAREGIVER ASSISTANCE'. After we made the changes, we sent the domain taxonomy to four doctors, who are also members of the InAdvance project, for a third feedback to evaluate the latest changes. The results stated that the domain taxonomy is now complete, well structured and ready to be implemented.

Implementation Phase. To implement the domain taxonomy, we used Protégé (Musen, 2015) the most widely used tool for ontology building. The main domain topics and their sub concepts, which have been proved by experts, have

¹https://www.age-platform.eu/[Accessed:November 2022]

been linked as a taxonomy and are ready to be used in any software program to process health text, see Figures 6.2, 6.3 and 6.4.



Figure 6.2: Main Topics of the in the Health Related Quality of Life Awareness taxonomy.



Figure 6.3: The concepts that belong to the topics ENVIRONMENT, LEVEL Of INDEPENDENCE, PERSONAL VALUES AND BELIEFS and SOCIAL RELATIONSHIP in the Health Related Quality of Life Awareness taxonomy.

The final number of the domain concepts are 212 and the main domain topics are: PHYSICAL HEALTH, PSYCHOLOGICAL HEALTH, LEVEL



Figure 6.4: The concepts that belong to the topics PHYSICAL HEALTH and PSYCHOLOGICAL HEALTH in the Health Related Quality of Life Awareness taxonomy.

OF INDEPENDENCE, SOCIAL RELATIONSHIP, ENVIRONMENT and PERSONAL VALUES AND BELIEFS.

6.3 Data Collection

To generate health domain related video narratives, we first need the required data to be able to implement the VISC-L and VIN-L frameworks. The required data is: the domain taxonomy, patient story videos and training data.

For video collection, any pre-selected domain related videos can be used if they are associated with their transcript. Otherwise, we can use the domain taxonomy to collect videos from available social platforms (like YouTube) as illustrated in (Section 3.5.1) which we are using for this domain. We have used the search schema to collect two types of domain related videos: Patient Stories and Tutorial Videos. Although we are collecting the videos automatically, we manually filtered the collected videos which makes this step as a semi-automatic and not fully automatic way, see (Section 7.3). The selection of the most suitable domain related videos is described in the next section.

6.3.1 Patient Stories.

To collect health related quality of life videos with patient stories, we utilised the domain taxonomy concepts to implement the search schema we designed (see Section 3.5.1). Using a combination of \langle Domain, t_i , c_i , 'patient story' \rangle as search terms, videos with patient stories related to the topic t_i and focus on the concept c_i have been collected. We have implemented this search schema using the library YouTube-search-python¹. According to this library and to make sure we are collecting context related videos, we have to associate the search schema with the number of videos to be collected for each search term. For this purpose, we ran a test code and asked for number of videos ≥ 5 to be collected for each search term. We found that the program starts to collect videos which have the concept in their titles, but are in a different context. Consequently, we ran another experiment with number of videos = 4 but we still got the same issue. For that reason, we decided that the number of videos to be collected ≤ 3 per each search term. Inspired by the scoping review approach in (Damarell *et al.*, 2019), we have filtered the collected videos and the results are presented in Figure 6.5.

Videos retrieved. The number of first collected patient story videos, after implementing the search schema, was 543. We noticed that there were 295 duplicated videos which we excluded. The number of unique videos was now 248 which were passed on to the next stage of the scoping review.

Screened for relevancy and transcript availability. We found some irrelevant videos that includes some of domain related words in the title but they are domain irrelevant. For example, the videos: '10 Life Lessons From Buddha

¹https://pypi.org/project/YouTube-search-python/[Accessed: December 2022]



Figure 6.5: Filtering the collected domain related patient story videos using the scoping review approach.

(Buddhism)', which is talking about what it is Buddha; 'Stem Cells: Medical Miracle Or Science Gone Too Far? (Medical Documentary) — Real Stories' which is mainly about what stem cells are. Since we process the video transcripts only in our work, we decided to filter the 248 videos manually and select the videos that are associated with their transcripts that have been generated automatically by YouTube. Accordingly, 188 videos have been discarded, as they were not associated with their transcript-irrelevant videos.

Videos included. After applying all the filters (excluding the duplicated videos, and check for transcript availability condition), the final outcome was 60

videos which are unique and were associated with their transcript. These videos are the patient story videos which are passed to the VISC-L framework to be segmented, characterised and aggregated.

A sample of Patient Story videos, that are going to passed as input to VISC-L, are presented in the Table 6.1.

Table 6.1: Sample of the collected Patient Story videos. Note that due to the YouTube policy the creator of the videos can delete them at any time.

Video title	Summary of the video
Becca's lung cancer story ¹	Becca, was given just weeks to live but
	everything changed after the biopsy results.
	She has now come back to her normal life.
Lung Cancer: Patient Stories ²	Patients living with lung cancer share their
	stories as well as the reasons why they chose
	to receive care at specific hospitals.
Young women can get lung	Joanna was diagnosed with stage 4 lung
cancer: Joanna's story 3	cancer and she had symptoms for around
	14 months.

6.3.2 Tutorial Videos

The last main input data required to implement VISC-L is the domain related training data. This type of data is required to fine-tune the Topic Classifier model within the characterisation phase in the VISC-L framework. Such training data is not available for this domain and for that reason we decided to create it automatically. Accordingly, we have to collect domain related tutorials to be able to use them to generate the training data. To collect the domain tutorials, we used the search schema combination of \langle Domain, t_i , c_i , 'tutorial' \rangle . Before passing these video tutorials to VISC-L, we had to filter them out as we did with

¹https://www.youtube.com/watch?v=MxmrjLQvg7E[Accessed: December 2022]

²https://www.youtube.com/watch?v=rQutyErqBG0[Accessed: December 2022]

³https://www.youtube.com/watch?v=wwiwjZw7d6o[Accessed: December 2022]



Figure 6.6: Filtering the collected domain related tutorial videos using the scoping review approach.

the patient story videos, using the scoping review approach as presented in Figure 6.6

Video retrieved. After implementing the search schema, we collected 592 videos. We noticed that some of the videos were not related to the domain context in spite of the presence of some of the domain concepts in their title. For example, the videos with the titles 'Natural Resources and Environment' and 'Improve Quality of Life and Environment with Cutting-Edge Building Technologies' .These irrelevant videos have been collected mistakenly because of a homonyms issue (words that are spelled the same but have different meanings).

Hence, 310 videos have been excluded from the collected videos.

Screened for relevancy and transcript availability. The outcome from the previous step was 282 domain related videos which have been screened manually to check the availability of their transcript, generated automatically by YouTube. Hence, 21 videos, have been excluded as they were not associated with their transcript.

Videos included. After applying all the filters, the final outcome was 261 videos which are domain related tutorial videos associated with their transcripts that have been automatically generated by Youtube. These videos are passed to VISC-L to generate the required training data.

A sample of the collected tutorial videos with a summary about each video is presented in table 6.2.

Table 6.2: Sample of the collected domain related tutorials. Note that due to the YouTube policy, the creator of the videos can delete them at any time.

Video title	Summary of the video		
General impacts of green spaces	The effect of green space, urban and		
on physical and mental health $^{\rm 1}$	sustainability on mental health, physical		
	health and psychology.		
Lecture 2.3: Social and	The role of social and environment factors		
Environmental Determinants of	on everyday health.		
Health 2			
Advanced Practice in Health and	The advance practice in health and social		
Social Care Quality of Life 3	care related to the quality of life.		

The summary of the collected videos (patient stories and tutorials), after passing the scope review, is presented in Table 6.3. For each video type, we

¹https://www.youtube.com/watch?v=ibyEHKnFTQ[Accessed: December 2022]

²https://www.youtube.com/watch?v=w2CmfIHNmV4[Accessed: December 2022]

³https://www.youtube.com/watch?v=cgc5uGqIgH8[Accessed: December 2022]

recorded the number of videos collected, the average (standard deviation) duration of the videos in seconds and the corpus size of each video type—the total number of tokens included in the transcripts of all videos in a group.

Table 6.3: Summary of the collected videos related to the domain of HealthRelated Quality of Life Awareness–Respiratory Illness

Video Type	No.Videos	AVG.(STDV) Duration in sec.	Corpus Size
Patient stories	60	572 (612)	10122 tokens
Tutorial Videos	261	1762 (2501)	1349205 tokens

These videos with the domain taxonomy are passed as input to the VISC-L framework as described in the next section.

6.4 Implementation of VISC-L

The proposed framework for Video Segmentation and Characterisation for Learning (VISC-L) in (Section 3.5) is implemented in this chapter for the health domain. It includes five main steps: Input, Initial Segments, Segment Characterisation, Segment Aggregation and Output. The implementation of these steps are presented below.

6.4.1 Input

The VISC-L framework is based on three assumptions as in (Chapter 4). The main group of videos we are using is the patient story videos which we used to generate the video narratives and the taxonomy of this domain is built as presented in (Section 6.2). For the health domain, we do not have training data but we generate it from the tutorial video transcripts we collected as presented in (Section 6.2). The process of labelling these transcripts is presented in (Section 6.4.3). Additionally, the transcript of the collected videos is pre-processed (adding punctuation) similar to the work in (Chapter 4). The punctuated video

transcripts are ready to be passed to the Initial Segmentation phase of VISC-L.

6.4.2 Initial Segmentation.

Our video segmentation algorithm is the same one as described in (Section 4.4.2) but we improve it in this section. This is in responding to the expert evaluation of VISC-L implementation in the first domain where they complained that the segments are short and they would like to watch more. Hence, the initial segmentation algorithm starts by selecting a certain number of text lines (6). The video segment boundary is decided if the last line (the 6th text line) in the text-window includes one of the following, [.,?,!]. Otherwise, the algorithm add an extra line(s) to the text-window and stop when the required punctuation is found. Consequently, the size of the video segment varies with ≥ 6 text lines. The resulting video segment has three types of data: the transcript (≥ 6 text lines), starting time (the timestamp of the first text line in the video transcript) and end time (the timestamp of the last text line in the video transcript). The segmentation algorithm has been applied on the patient story and tutorial videos separately. After applying the segmentation algorithm on the 60 patient story videos, we generated 1483 segments with an average duration of 15 (STDV=7) seconds with an average number of segments, 68 (STDV=54), per video. On the other hand, after applying the segmentation algorithm on the tutorial videos, 25959 segments have been generated with average duration of 15 (STDV=28) seconds and an average number of segments, 297 (STDV=268), per video.

The generated segments are ready to be passed to the next phase of the the VISC-L framework which is Segment Characterisation.

6.4.3 Segment Characterisation

The aim of the Segment Characterisation phase is to aid the aggregation of the initial segments of the patient story videos, while also allowing us to generate training data from the tutorial video segments. For that purpose, we need to identify what domain content is presented in each video segment. This is done during this phase where each characterised video segment in a patient story video characterised video segment $Ch = \langle VS_{Id}, VS_s, VS_e, VS_t, FT, FC \rangle$ is linked with a set of focus topics FT and a set of concepts FC, where $FC \subset FT$.

The Segmentation Characterisation step in VISC-L includes two algorithms: Semantic Tagging and Topic Classification.

Semantic Tagging. The semantic tagging algorithm we designed in (Mohammed & Dimitrova, 2020) and illustrated in (Section 3.5) is used at this phase to characterise both the patient story and tutorial video segments. This algorithm has two inputs: the transcript of the video segments and the domain taxonomy illustrated in (Section 6.2).

(a) Patient Story Video Segments. At this stage in semantic tagging, we have used, as an input, the 1483 segments generated from the Initial Segmentation step on the patient story videos. The transcript of each segment has been tokenized, cleaned and PoStagged ('Part-of-Speech tagged') to get the resulted nouns and noun phrases. These nouns and noun phrases have been semantically tagged with the domain taxonomy terms to identify the focus topics and concepts in the video segments. The output is the characterised video segments with the domain focus topics and concepts $Ch = \langle VS_{Id}, VS_s, VS_e, VS_t, FT^1, FC^1 \rangle$, where ¹ means this the first characterisation result.

After screening the output, we found that some of the domain taxonomy terms have been mentioned in the video segment's transcript but in a different context. For example, the concept 'COMMUNITY' which is part of the topic SOCIAL RELATIONSHIP, is mentioned in the video as part of community centre name. This causes word sense ambiguity issue. This issue emerged when we collected a large number of domain related videos automatically where in some of these videos the speaker either digress from the topic or talks to the audience. To solve this issue, as presented in (Section 3.5.3), we use a Topic Classifier model and to do so we need training data. Such data is not available for this domain and we have to generate it. The process of generating the training data to fine tune the Topic Classifier model is presented in the next section. (b) Tutorial Video Segments. To be able to fine-tune the Topic Classifier model to solve the word sense ambiguity issue, domain related training data is needed. For this purpose we apply the semantic tagging algorithm on the tutorial video segments to identify the focus topic(s) in them. Accordingly, we have passed the 25959 tutorial video segments resulted from the initial segmentation phase on the tutorial videos. These video segments have been characterised by: tokenizing their transcript, cleaning them from stop words and PoStagging them to get the nouns and noun phrases. These nouns and noun phrases have been semantically tagged with the domain taxonomy terms to identify the focus topics in the tutorial video segment transcripts. The output was 10441 tutorial video segments characterised with domain topics and 15518 discarded segments as they were irrelevant to the domain. These 10441 labelled video segments focus on different domain topics as presented in the Table 6.4 which is used as training data in the next step of the characterisation phase–Topic Classifier.

Table 6.4: Tutorial Video segments after applying the Semantic Tagging algorithm. The output is the training data to be used to fine-tune the BERT model in the Topic Classifier step in the VISC-L framework.

Focus Topic(s)	No.Segments	Duration in sec. AVG (STDV)
Physical Health	3399	15.0 (5.0)
Psychological Health	2131	16.0 (21.0)
Environment	1358	15.0 (5.0)
Social Relationship	2219	15.0 (8.0)
Level Of Independence	592	15.0 (6.0)
Personal Values and Beliefs	742	16.0 (5.0)

(c) Expert Evaluation of the Semantic Tagging Algorithm. We have conducted the evaluation study of the Semantic Tagging Algorithm with one domain expert, as we did in (Section 4.4.3). The evaluation of the Semantic Tagging algorithm output is crucial as it will affect the quality of the overall characterisation result. This is because the characterised patient story video segments will affect the selection of the focus concept. Meanwhile, the

characterised tutorial video segments are used to fine-tune the Topic Classifier. The fine-tuned Topic Classifier is implemented on the patient story video segments to provide their second characterisation. Both patient story video segment characterisations are combined to get the final characterisation result. For this purpose, we have invited one expert to evaluate the characterised video segments of both video groups (patient stories and tutorials). The expert, who has participated in the evaluation process, is a member of the InAdvance project, and is different from the evaluators in (Section 6.6.1), who works as a health care psychologist and is an expert in the health related quality of life of patients with respiratory illnesses.

The **materials** are 160 segments from which (80 segments have been selected from the patient stories, and the other 80 segments have been selected from the tutorial videos). The expert has been provided with an Excel sheet that includes the video segments from both groups of videos with their characterisations, (see Appendix M). Additionally, we have provided her with domain taxonomy topics and concepts as in Figures 6.2, 6.3 and 6.4.

The **procedure** of the evaluation process that required to be followed by the expert is: if agree with the labelled data then do nothing, if disagree with the labelled data or to add a new/missed concept then highlight the concept in red, and if not sure about the assigned concept or the suggested concept then highlight in orange.

After receiving the evaluation **result**, we have designed three types of responses, similar to the evaluation study in (Section 4.4.3–Characterisation Evaluation step); 'To Fix the Domain taxonomy', 'To fix the Semantic Tagging Algorithm' and 'Ignore–the concept was picked for the segment'. The response 'To Fix the Domain taxonomy' means we have added the the new suggested concepts by the expert which we found were missed from the domain taxonomy. There were 4 new suggested concepts added to the taxonomy.

While the 'To fix the Semantic Tagging Algorithm' response is where we decided to modify our algorithm to be able to pick the multi-word concepts suggested by the expert and were missed from the characterisation, e.g, 'ACCESS TO GREEN SPACES' from the topic ENVIRONMENT. For this purpose, we have used the n-gram library (where n=2,3 and 4 to collect the concepts with compound words) from TextBlob¹, which is a free open source in the Python software function.

Furthermore, the 'Ignore-the concept was picked for the segment' response is to ignore the expert's new suggested concept/ disagreement with the assigned concepts. This is because the concepts assigned to the segments, by the algorithm, are included in the taxonomy and presented in the segment-transcript, similar to our work in (Section 4.4.3-Characterisation Evaluation step). Hence, we have ignored 9 suggestions from the experts.

Finally, to calculate the agreement between the expert responses and the output from the semantic tagging algorithm, we ran Cohen's Kappa Coefficient (Cohen, 1960), following the exact steps in (Section 4.4.3–Characterisation Evaluation step). Accordingly, the inner agreement result was very high (96%), (see Appendix M).

Thereafter, we made the required amendments to the Domain Taxonomy and Semantic Tagging Algorithm and re-implemented them to the patient stories and tutorial videos. To further improve the taxonomy concepts, we checked the transcript of each segment with its characterisation and found new domain related concepts present in the transcript but were missed from the taxonomy. We decided to add these concepts to the domain taxonomy. For example, we have added the super concept 'RESPIRATORY ILLNESS' to the topic PHYSICAL HEALTH. 'RESPIRATORY ILLNESS' now has the following sub concepts: 'ASTHMA', 'CHEST-PAIN', 'COPD', 'COVID', 'LUNG-CANCER', 'LUNG DISEASE', 'PNEUMONIA', 'PULMONARY-FIBROSIS', 'PULMONARY-HYPERTENSION' and 'TUMOR'. Additionally, we have added the concept 'PULMONARY-REHAB' to its super concept 'TREATMENT' and both belong to the topic LEVEL OF INDEPENDENCE. We also added the concept 'SORENESS' within the super concept 'PAIN' and both concepts belong to the topic PHYSICAL HEALTH. Finally, we have added the concept 'GRIEF', 'HYPOCHONDRIAC' and

¹https://buildmedia.readthedocs.org/media/pdf/textblob/latest/textblob. pdf[Accessed: January 2023]

'SADNESS' to their super concept 'NEGATIVE FEELING' and all concepts belong to the topic PSYCHOLOGICAL HEALTH. Consequently, the final number of concepts within the taxonomy 212.

The characterised tutorial video segments are passed to the second step of the characterisation phase of VISC-L to be used as training data for the Topic Classifier model.

Topic Classifier. This is the second step of the characterisation phase in VISC-L which considers the context of the segments to aid the resolving of the aforementioned issue of word sense ambiguity.

To be able to implement the BERT-BASE-Uncased model (Topic Classifier) used in (Section 4.4.3–Topic Classifier step), we need to fine-tune it with domain related training data. For this purpose, we use the characterised tutorial video segments labelled with their domain topic as training data, see Table 6.4. We fine tuned the BERT-BASE-Uncased model as a binary classifier for each topic in the domain. As we described in (Section 4.4.3–Topic Classifier step). We used the fine tuned model as a Topic classifier to characterise the patient stories segments. The characterisation output of each segment after implementing the Topic Classifier model is FT_i^2 , where ² means this characterisation is the second segment characterisation which identifies the focus topic only.

The characterisation output of the patient story video segments resulting from the Semantic Tagging algorithm and the Topic Classifier is ready to be combined in the final characterisation step of the VISC-L framework.

Combining the Characterisation. This is the last step of the characterisation phase of VISC-L where the characterisation of the segments is decided by the intersection between the semantic tagging algorithm results and the topic classifier result as follows: $\langle FT_i \rangle = \langle FT_i^1 \rangle \cap \langle FT_i^2 \rangle$. Thus, the output of the characterisation step for each video segment is the focus topic identified by the Semantic Tagging algorithm which is similar to the topic identified by the Topic Classifier. The focus concept(s) of the agreed focus topic can be extracted from the Semantic Tagging output $\langle FT_i, FC_i \rangle$. Following the work in (Section
Table 6.5: Evaluation result after fine tuning the BERT-BASE-Uncased model when applied as a binary classifier in the health domain–Health Related Quality of Life Awareness.

Topic	Precision	Recall	F1-score
Physical Health	0.98	0.99	0.99
Psychological Health	0.99	1.00	1.00
Environment	0.99	0.99	0.99
Social Relationship	1.00	0.98	0.99
Level of Independence	0.99	0.95	0.97
Personal Values and Beliefs	0.99	1.00	0.99

4.4.4) we aggregate the adjacent segments in the next section.

6.4.4 Segment Aggregation

We have implemented the Segment Aggregation step of VISC-L on the patient stories video segments in this domain following the methodology step described in (Section 3.5.4) which is implemented in (Section 4.4.4). An example of the aggregated segments is presented in Figure 6.7, where the fifth segment has been aggregated with the segments preceded. This segment matched the aggregation conditions as it is without characterisation and its duration is less than 30 seconds-the segment duration threshold used in this thesis as an aggregation condition.

6.4.5 Output

The last step in the VISC-L framework is the final output which is the 306 aggregated segments with an average duration of 56 (STDV=36) seconds, and their full details are presented in Table 6.6.



Figure 6.7: Example of aggregating adjacent segments with interpolating the gap segment.

Table 6.6: Linear aggregation with the interpolation results of the patient stories video segments.

Focus Topic(s)	No.Segments	Duration	Duration
		(AVG) in sec.	(STDV) in sec.
Physical Health	127	62	41
Psychological Health	30	44	23
Environment	14	57	47
Social Relationship	108	53	31
Level Of Independence	24	49	22
Personal Values and Beliefs	3	62	33

It is clear that the number of initial segments decreases noticeably after applying the aggregation algorithm. In contrast, the average duration of the aggregated segments is greatly increased as presented in Table 6.7. Table 6.7: Summary of the generated patient stories video segments after implementing the Initial Segment and the Linear Aggregation with Interpolations steps from VISC-L.

Video Segments	No.Segments	AVG.Duration	STDV Duration
Initial Segments	1483	15	7
Aggregation Result	306	56	36

The aggregation algorithm helps in better recognising the areas in the videos that elaborate the same focus topic and concepts. This can be beneficial to users to help them notice the main domain points.

More examples of VISC-L output which is the characterisation of video segments are in Figures 6.8, 6.9, 6.10.



Figure 6.8: Example 1 of a segment characterisation by VISC-L. The segment has been taken from one of the patient story videos with YouTube video ID= hQQ5xQGZxdA-this is taken from the segment characterisation which saved the segment Id that is created using the video Id as described in Algorithm 1. It starts at 04 : 41 with duration of 128 seconds.

To be able to generate video narratives from the characterised video segments, we need to ensure that these videos can be uploaded to Google Cloud as we are



Figure 6.9: Example 2 of a segment characterisation by VISC-L. The segment has been taken from one of the patient story videos with YouTube video ID= wwiwjZw7d6o-this is taken from the segment characterisation which saved the segment Id that is created using the video Id as described in Algorithm 1. It starts at 01 : 14 with duration of 118 seconds.

using Google Drive to generate URLs for our video narratives to be used by users. Consequently, only 52 videos have been uploaded successfully to Google Cloud and 8 videos have been discarded with their 139 segments. Thus, the remaining video segments used to generate the narratives numbered 167 and focus on the main health topics as presented in the VISC-L output in Figure 6.15.

These characterised segments are used as the input to implement the subsumption linking algorithms as we describe in the next section.

6.5 Implementation of VIN-L

The aim of our research is to generate video narratives to help learners identify key domain points in the videos and be able to link them together. For this purpose we have designed the VIN-L framework to generate the video narratives which is underpinned by the Ausubel Subsumption theory for meaningful learning, as described in Section 3.6. The full VIN-L framework implementation steps are



Figure 6.10: Example 3 of a segment characterisation by VISC-L. The segment has been taken from one of the patient story videos with YouTube video ID=GV3Rew061FA-this is taken from the segment characterisation which saved the segment Id that is created using the video Id as described in Algorithm 1. It starts at 01 : 31 with a duration of 216 seconds.

presented below.

6.5.1 Input

To be able to implement the VIN-L framework in this health domain, we need the domain taxonomy which has been designed in (Section 6.2) and the 167 characterised patient stories video segments which are the VISC-L framework output. The video segments are linked by implementing the subsumption links defined in Section 3.6. The implementation of the subsumption linking algorithms on the video segments is presented next.

6.5.2 Subsumption Linking Algorithms

To guide the linking of the video segments, the subsumption theory of meaningful learning designed by Ausubel has been utilised as described in (Section 3.2.3). Four types of subsumption linking have been identified by Ausubel which are going to be used to generate the video narratives: Derivative, Super Ordinate, Correlative and Combinational. To implement these subsumption links, the four algorithms (Algorithms 4, 5, 6, 7) designed in (Section 3.6) are going to be implemented.

Referring to Figure 3.6, we demonstrate the implementation of each algorithm in column 2 titled 'Subsumption Linking Algorithms' following Ausubel's definition, as we did in (Section 5.2.2). Additionally, to augment the generated narratives with text, we have addressed the recommendation of improving the introduction part of the video narrative by including a figure clarifying the conceptual relationship between the Focus topic and the Focus concepts.

Derivative Linking is when the learning is determined by having the generic concept in the cognitive structure first before introducing one of its specific concepts (target concepts) to bind them together, (see Algorithm 4).

An example of the derivative narratives is in Figure 6.11 that shows one of the derivative narratives generated. In this example, the Derivative narrative aims to support learning the target concept 'EXERCISE'. It starts by firstly inculcating its generic concept 'REHABILITATION' into the cognitive structure then providing the segments that focus on the target concept 'EXERCISE'.

Super Ordinate Linking helps learning a concept by showing examples-its sub-concepts. with Super Ordinate Linking, the learning starts with the specific concepts before learning their parent concept, (see Algorithm 5). An example of the Super Ordinate narrative is presented in Figure 6.12. The video narrative aims to raise awareness of the generic concept 'ENERGY'. It starts by firstly inculcating one of its specific concepts, 'STRENGTH', into the cognitive structure then providing the segments that focus on the target (parent) concept 'ENERGY'.

<text><text><text><text><image><image><text><text><text><text><text><text><text>

Health-related Quality of Life Needs of Patients with Long-term Respiratory Illnesses

The video segments were automatically extracted from popular YouTube videos on Healthy related Quality of life-Respiratory Illness, using algorithms for segmentation, characterisation and linking based on a theory for concept learning. We hope you found the collection of video segments helpful.

Figure 6.11: Example of the Derivative Narrative to support learning showing that to learn the concept 'EXERCISE', its generic (parent) concept 'REHABILITATION' must be introduced first before introducing the specific concept. All of the concepts belong to the topic 'ENVIRONMENT'.

Correlative Linking helps in learning a concept by showing different aspects related to it (its sibling concepts), (see Algorithm 6). An example of the generated

Health-related Quality of Life Needs of Patients with Long-term Respiratory Illnesses

This is a collection of video segments from YouTube videos with patient stories.

They have been automatically segmented, characterised and linked to domain concepts.

The videos have been linked via "Super Ordinate Linking" which means:

To become aware of a generic concept (Green oval) from a topic T, you will explore videos about its children concept which are more specific (Orange ovals).



Became aware of the generic concept ENERGY from PHYSICAL HEALTH by exploring more specific concepts STRENGTH.

Each video segment will start from a time point in the video and will stop when the corresponding concepts are covered.

You can re-watch the video segment or watch other parts of the video, if you wish so.

The next video segment mentions STRENGTH (relates to PHYSICAL HEALTH). The segment has been taken from a YouTube video with ID=VyGp12XbPzs, and starts at 0.59 with a duration of 81 seconds.



The next video segment mentions ENERGY (relates to PHYSICAL HEALTH). The segment has been taken from a YouTube video with ID= wArV8Ho_C1A , and starts at 1.34 with a duration of 51 seconds.





The segments were automatically extracted from popular YouTube videos related to the Health-related Quality of Life Needs of Patients with Long-term Respiratory Illnesses, using algorithms for segmentation, characterisation and linking based on a theory for concept learning.

We hope you found the collection of video segments helpful.

Figure 6.12: Example of the Super Ordinate Narrative to support learning showing that to learn the generic (parent) concept 'ENERGY', its specific concept 'STRENGTH' must be introduced first before introducing the parent concept. All of the concepts belong to the topic 'PHYSICAL HEALTH'.

Correlative narratives is presented in Figure 6.13. In this example the Correlative narrative aims to support learning the targeting concept 'BELIEF'. It starts by introducing its sibling concept 'MEDITATION' and both concepts belong to the

topic 'PERSONAL VALUES AND BELIEFS'.

Health-related Quality of Life Needs of Patients with Long-term Respiratory Illnesses

This is a collection of video segments from YouTube videos with patient stories.

They have been automatically segmented, characterised and linked to domain concepts.

The videos in this page have been linked via "Correlative Linking" which means:

To become aware of a specific concept (Green oval) from a topic T, you will explore videos about a similar concept (Orange oval).



Become aware of the concepts similar to BELIEF from topic PERSONAL VALUES AND BELIEFS.

Each segment will start from a time point in the video and will stop when the corresponding concepts are covered. You can re-watch the segment or watch other parts of the video, if you wish so.

The next video segment mentions MEDITATION (related to PERSONAL VALUES AND BELIEFS). The segment is taken from a YouTube video with ID= HBFn82ffSVE, and starts at 4.25 with a duration of 108 seconds.



The next video segment mentions BELIEF (related to PERSONAL VALUES AND BELIEFS). The segment is taken from a YouTube video with ID=mSYXddFV+Ho , and starts at 0.52 with a duration of 36 seconds.



You watched a video segment about (MEDITATION) which is similar to (BELIEF) related to PERSONAL VALUES AND BELIEFS.

The segments were automatically extracted from popular YouTube videos related to the Health-related Quality of Life Needs of Patients with Long-term Respiratory Illnesses, using algorithms for segmentation, characterisation and linking based on a theory for concept learning. We hope you found the collection of video segments helpful.

Figure 6.13: An example of one of the generated Correlative Narratives to support learning shows that to learn the concept 'BELIEF', a similar concept to it must be introduced first which is 'MEDITATION' and both concepts belong to the topic 'PERSONAL VALUES AND BELIEFS'.

Combinational Linking aims to aid learning different concepts that belong to two different topics together to conceive the relationship between them, (see Algorithm 7). An example of the Combinational narrative is presented in Figure 6.14 and shows the relationship between the topics 'SOCIAL RELATIONSHIP' and 'PSYCHOLOGICAL HEALTH' by identifying the relationship between their concepts mentioned in the same segment.

Filtering Criteria. After following the improvement recommendation from (Chapter 5) which are to increase the video segment window size and the number of video segments within the video narrative. First, the segment window size boundary is decided when the end of sentence punctuation presents at the end of the 6th text line of the segment window size or at the end of the first text-line(s) after the 6th text-line. Second, select the video segments from the same video if they have the same Focus Topic and Focus Concept of the video narrative. This is to ensure that the generated video narratives fulfill their target by helping learners identify the key domain points and link them but without exceeding the video narrative duration condition (≥ 3 minutes and ≤ 6 minutes) described in (Section 3.4–Definition 7) which is implemented in (Chapter 5).

6.5.3 Output

The final number of the generated video narratives, after applying the duration filtering criteria, is 59 video narratives with a minimum average duration of 143 (STDV=92) seconds and a maximum average duration of 349 (STDV=29) seconds. These narratives are distributed among the narrative types as follows (3 Derivative narratives, 3 Super Ordinate narratives, 43 Correlative narratives and 10 Combinational Narratives). The reason we have more Correlative narratives than any other type is because the same Correlative narrative is generated for a concept and its siblings. On the other hand, there are few Derivative narratives because not all of the generic (parent) concepts of the specific concepts were present as a focus in the video segments. Similarly, there are few Super Ordinate narratives because not all of the specific concepts of the generic (parent) concepts of the generic

Health-related Quality of Life Needs of Patients with Long-term Respiratory Illnesses

This is a collection of video segments from YouTube videos with patient stories.

They have been automatically segmented, characterised and linked to domain concepts.

The videos in this page have been linked via "Combinational Linking" which means:

You will explore videos which link concepts from two topics (Green ovals).



Link the topics SOCIAL RELATIONSHIP and PSYCHOLOGICAL HEALTH.

Each segment will start from a time point in the video and will stop when the corresponding concepts are covered

You can re-watch the segment or watch other parts of the video, if you wish so.

You will now watch a video segment that mentions CHILD (related to SOCIAL RELATIONSHIP) and THINKING (related to PSYCHOLOGICAL HEALTH). The segment is taken from a YouTube video with ID=ZoEE@U6L-m8. and starts at 3.31 with duration of 83 seconds.



You will now watch a video segment that mentions WIFE (related to SOCIAL RELATIONSHIP) and HOPE, SCARED (related to PSYCHOLOGICAL HEALTH). The segment is taken from a YouTube video with ID= cfyb39S_CZw, and starts at 1.2 with duration of 44 seconds.



You watched video segments that link SOCIAL RELATIONSHIP and PSYCHOLOGICAL HEALTH.

The video segments were automatically extracted from popular YouTube videos, using algorithms for segmentation, characterisation and linking based on a theory for concept learning.

We hope you found the collection of video segments helpful.

Figure 6.14: An example of a Combinational Narrative that supports the learning the topics 'SOCIAL RELATIONSHIP' and 'PSYCHOLOGICAL HEALTH' by introducing video segments that focus on different concepts belonging to these topics.

were presented as a focus in the video segments. The full details of the final output of the generated video narratives are presented in Figure 6.15.

Input			VISC-L				Subsumption Linking Algorithms				
Video Collection using YouTube			Video Segmentation and Characterisation VISC-L framework				Video Narrative generation VIN-L framework				
Search Schema < Domain name, Topic, Concept, Patient Story> Videos with patients' stories Filtering Criteria Videos relevant to the domain and have transcript with corpus size=10122 tokens Output		Initial Segments Formation #Segments AVG Duration STDV National Stream St		Four Algorithms	s <u>#Narratives</u>	AVG Duration	<u>STDV</u> 42 70 1465				
				Super Ordinate Correlative	5 3 51	578s					
		Filtering Criteria Exclude 8 videos that could not be used when generating HTML links			Combinational 10 192s 176 Filtering Criteria:			176			
<u>#Videos</u> 60	AVG Duration 442s	<u>STDV</u> 508	Output 167 segments o <u>Topic</u>	oming from 52 <u>#Segments</u>	videos <u>AVG Duration</u>	<u>stdv</u>	Output Narrative Type	#Narratives	AVG Duration	STD	
Tutorial Videos		Physical Health	80	61s	39	Derivative	3	349s	29		
Filtering Criteria Videos relevant to the domain and have transcript with corpus size= 1249205 tokens		Psychological Health	16	375	13	Super Ordinate Correlative	3 43	335s 268s	22 45		
		Environment	10	62s	56	Combinational	10	143s	92		
Output	utput / <u>ideos AVG Duration STDV</u> 51 2192s 1273		Level Of Independence	15	59s	34					
<u>#Videos</u> 261			Personal Values and Beliefs	3	56s	37					
			Social Relationships	43	50	27					

Figure 6.15: The output after implementing VIN-L following the research methodology.

These narratives are evaluated with experts to assess their Quality, Perceived Usefulness and Cognitive Work Load to raise awareness of the health related quality of life of patients living with a chronic illness-respiratory illness. The evaluation process and the results are presented in the next section.

6.6 Expert Evaluation

The main objective of this evaluation study is to assess the possibility of using the generated video narratives in the domain Health Related Quality of Life Awareness. For this reason, domain experts have been invited to participate in this evaluation study to assess the Quality, Perceived Usefulness and Cognitive Work Load of the video narratives.

6.6.1 Study Setup

To commence this evaluation study, we invited six experts to assess a selected number of video narratives following the evaluation procedure. All the details are given below.

Participants. We have followed the same process of using convenience sampling when selecting domain experts as described in (Section 5.6.5) and we have invited six domain experts. Three of whom are members of the InAdvance project (introduced in Section 6.1), two of them are researchers in AI related to patients quality of life and one of them is a nurse specialist also in patient' quality of life. The other 3 experts are from the University of Leeds who are PhD researchers (in their third year) focusing on AI related to patient quality of life. Their domain knowledge enables them to provide good quality answers for the questions about the quality of the video narratives and their potential usefulness if used to support domain learning. Following the ethics approval from the University of Leeds, see (Section 4.5.1), each participant was given a £30 Amazon voucher as gratitude for their participation as the study duration is around one hour which is less than the expert study in (Chapter 5) where its duration was around three hours.

Video Narratives Selection Criteria. To select the best video narratives for the evaluation study, we have filtered out the poor quality and duplicated video narratives. Poor quality video narratives mean that more than half of the included segments are too short (duration < 30 seconds) with more than one focus topic and concept or the focus concept is used in different context. The latter case happened with one Combinational video narrative that includes one segment where the focus concept is used in a different context. Hence, this video narrative will be excluded and nine out of ten Combinational video narratives are included in the evaluation study. According to the duplication condition in the filtering criteria, some video narratives are duplicated; mainly in the Correlative video narratives group where concepts are linked to their siblings. This means the same video narrative is regenerated for all the siblings of the specific concept. Thus, 30 Correlative narratives have been excluded and 13 have been included in the evaluation study. On the other hand, we have included all the Derivative and super ordinate video narratives in the study. No duplicate was found in Derivative and Super Ordinate video narratives because we generate a single video narratives to learn a specific or a generic concept.

Materials. We have conducted two evaluation surveys where we used the 28 video narratives that passed the filtering criteria successfully. The first survey had been conducted with the experts from the InAdvance project where they evaluated 16 video narratives (3 Derivative, 3 Super-Ordinate, 6 Correlative and 4 Combinational). The URLs of these narratives can be accessed using our GitHub account¹ and the evaluation form is in (Appendix N) and both surveys have the same structure. In the second survey, the experts from the University of Leeds participated and evaluated 12 video narratives (5 Combinational and 7 Correlative). The surveys are fully compliant with General Data Protection Regulation (GDPR) and have been reviewed and approved by the Research Ethics Committee of the Faculty of Engineering and Physical Sciences, University of Leeds, UK.

Both surveys have been designed using Google Forms with the following structure: a consent form, a section to collect user profiles, then the narrative section which is divided into four groups with each group representing a narrative type. In each narrative group there are three parts; the Quality categories after each video narrative link, the Perceived Usefulness and Cognitive Work Load scale items at the end. The Quality categories aim to assess each video narrative link and are designed using the LIKERT response labels (Taherdoost, 2019). To determine the user acceptance of the generated video narratives as learning materials, we have included two extra parts in the evaluation study.

¹https://github.com/Generating-Video-Narratives/Generating-Video-Narratives. git[Accessed: April 2023]

One is to assess the Perceived Usefulness of the generated narratives using a TAM questionnaire (Davis, 1989). The second is to estimate the Cognitive Work Load required while using video narratives as a source of learning through a combination of NASA-TLX scales (Hart, 2006). Additionally, we have asked the experts to type feedback to identify the possible positives and negatives of the video narratives they have watched. These three parts (the Quality categories, the Perceived Usefulness and the Cognitive Work Load scale items) have been repeated for each narrative type.

Procedure. To commence the evaluation study, we sent the evaluation form link of form-1¹, and form-2² to the participants. Both surveys have the same structure but different URLs, and in the second survey we have only evaluated the 5 Combinational and 7 Correlative video narratives. Each participant should go through the following steps in the evaluation form: (1.) Read and accept the consent form; (2.) Complete a short pre-study section to collect their profiles; (3.) Click on the video narrative link which will open in a new tab. They can also copy the link and open it in a browser themselves; (4.) Go through the video narrative page by following the descriptions and watch the suggested video segments which have a duration of 3 to 6 minutes; (5.) Go back to the screen with the survey form and complete the Quality and the Perceived Usefulness sections; (6.) Repeat steps 3-5 for each of the four video narrative types.

The study was approved by the ethics committee of the Faculty of Engineering and Physical Sciences, University of Leeds. As the work is funded by the InAdvance project, we did not pay the InAdvance experts but we did pay the University of Leeds participants £30 each as a token of gratitude for the in participation in the evaluation study.

In the following sections we analyse the outcome from the evaluation studies where we aggregate the outcomes of the two surveys. We report and analyse the results of the Quality, Perceived Usefulness, Cognitive Work Load and the

¹https://forms.gle/TwCs2BsbgPkT4vhv9[Accessed: April 2023]

²https://forms.gle/FApPjumbUihRuX1W6[Accessed: April 2023]

positives and negatives of the generated video narratives.

6.6.2 Results: Quality of Video Narratives

We analysed the quality of the generated video narratives based on the aggregated experts' responses to be able to answer RQ3.

How to generate good quality narratives that can be used to support perceiving domain learning?

The 84 aggregated responses from both evaluation surveys are reported in Table 6.8 and presented in Figure 6.16, the responses values of the evaluators is in (Appendix O). The aggregated experts' response per narrative type and per quality category is reported as the median, average (standard deviation) scores. The N value indicates the number of expert-responses per narrative type; each video narrative was assessed by 3 experts.

Table 6.8: Evaluation Results of the quality of the narratives. Each narrative type (CO=Combinational, CR=Correlative, D=Derivative, S=Super Ordinate) has a number of assessments (N) and the Median, AVG and (STDV) of each quality question based on the responses from the experts (1= Strongly Disagree, 2=Disagree, 3=Neutral, 4=Agree, 5=Strongly Agree).

Video Narratives Quality	СО	CR	D	S	All
Categories	N=27	N=39	N=9	N=9	N=84
Q1–The introduction text in the	4.0	4.0	4.0	4.0	4.0
video narrative gives a clear idea	4.0	4.3	3.9	3.6	4.7
of the aim of the narrative.	(1.0)	(0.9)	(1.1)	(0.7)	(0.93)
Q2–All video segments clearly	4.0	4.0	4.0	4.0	4.0
link to the health related quality	4.0	4.1	4.0	4.3	4.8
of life areas mentioned in the	(1.0)	(0.9)	(1.0)	(0.5)	(0.95)
introduction.					
$\mathbf{^{*Q3}}$ -The descriptions which	4.0	5.0	4.0	4.0	4.0
introduce each video segment	3.7	4.4	3.4	3.7	4.0
provide a useful summary.	(1.0)	(0.8)	(0.7)	(0.7)	(0.91)
Q4–All video segments provide	4.0	5.0	5.0	4.0	5.0
relevant content for health related	4.0	4.4	4.4	4.4	4.25
quality of life areas.	(1.2)	(0.8)	(0.7)	(0.5)	(0.97)
Q5–The concluding text at the	4.0	4.0	4.0	4.0	4.0
end of the video narrative provides	3.8	4.2	3.7	3.9	3.96
appropriate summary.	(0.9)	(0.9)	(0.7)	(0.6)	(0.84)
OVERALL QUALITY–AVG score	4.0	4.4	4.2	4.0	4.0
for all Quality categories per	3.9	4.3	3.9	4.0	4.7
narrative type.	(1.0)	(0.9)	(0.8)	(0.6)	(0.93)
USEFULNESS–video narrative is	5.0	5.0	5.0	5.0	5.0
useful for raising awareness of	4.3	4.6	4.8	4.7	4.54
quality of life needs of patients	(1.1)	(0.6)	(0.4)	(0.5)	(0.77)
living with chronic respiratory					
illnesses.					

In our analysis of the experts' responses, we do two comparisons. One

comparison is between the quality categories and the other comparison is between the narrative types.

Comparison between the quality categories of the video narratives. We refer to the column (All) in Table 6.8 where we can get an overview of the quality of each category based on the aggregated experts' responses.

The experts found that a majority of the video segments within each video narrative provide content related to the domain (Q4) with a median 5 and an average score of 4.25 out of 5 (agree to strongly agree). As a result, the video segments are also found to be clearly linked to the health domain areas defined in the introduction of the video narrative (Q2), with a median of 4 and an average score of 4.8 (agree to strongly agree). Based on that, the experts found that the video narrative introduction part clarifies the aim of the narratives (Q1) with median of 4 and an average score of 4.7 (agree to strongly agree). The experts also agree with the quality of the description (characterisation) of the segments (Q3) with median of 4 and an average score of 4 (agree). However, the conclusion statement (Q5) was found to be of lower quality than the other quality categories with a median of 4 and an average score of 3.96 (neutral to agree). The overall video narrative quality indicates that the experts found the generated video narratives are of a good quality with a median score of 4 and an average score of 4.7 (agree to strongly agree). Based on that, there are pointers towards answering RQ3 positively. Additionally, the experts found the generated video narratives have a potential to be useful when used to raise awareness of Health Related Quality of Life needs with a median score of 5 and an average score of 4.54 (agree to strongly agree).

Comparison between the narrative types. We are referring to the columns in table 6.8. The experts found no huge difference in the quality of the Derivative, Super Ordinate and Correlative narrative types as the median score in all of them was 5 and the average score was ≥ 4.6 and ≤ 4.8 which is agree to strongly agree. This refers to the design of these narratives as the link between the concepts is obvious and all the concepts within a narrative belong to the same topic. However, in the Combinational narratives, this link between the

Quality categories со CR D s All Q1: The introduction text in the video narrative gives a clear idea of the aim of the narrative. Q2: All video segments clearly link to the healthrelated quality of life areas mentioned in the introduction. Strongly agree Q3: The descriptions which introduce each video Agree segment provide a useful summary. Q4: All video segments provide relevant content for health-related quality of life areas. Strongly disagre Q5: The concluding text at the end of the video narrative provides appropriate summary. OVFRALL Quality - AVG score for all Quality categories per narrative type. Usefulness

concepts was less obvious as the concepts belong to different topics, see Figure 6.16.

Figure 6.16: Evaluation results of the quality of the video narratives.

To add more explanations to the difference among the quality questions mentioned above, we run different tests (Kruskal-Wallis, Mann Whitney and Spearman's RHO). These tests help in confirming/ repudiating that the differences mentioned above are statistically significant.

The difference in the quality categories among the narratives types is not statistically significant, as the result shown after running Kruskal-Wallis (non parametric test). To compare between pairs of narrative types based on the quality categories (Q1-Q5) result, we have run Mann Whitney Test. The

results showed that difference between the CR and CO narratives is statistically significant. The experts found that two focus concepts per a video segment affect the quality of the defined learning in the narrative introduction. For instance, the relationship between two focus concepts from different focus topics is not clear as one focus concept could be mentioned more than the other focus concept. Example of experts feedback on CO narratives "I don't see any negatives, however it doesn't show which concepts/topics are more important than others.", "It does not always highlight all the points made in the video segment." Additionally, the difference between the quality categories and the usefulness of the video narratives is not statistically significant as shown by the result of running Spearman RHO test (Spearman $r_s = -0.10541$).

Further investigation between the narratives types has been done by comparing their perceived usefulness.

6.6.3 Results: Perceived Usefulness of the generated Video Narratives

In this part, we analyse the Perceived Usefulness results of the generated video narratives based on the aggregated experts' responses to be able to answer:

Are the generated narratives useful to support perceiving domain learning?

The 18 aggregated responses from both studies on the Perceived Usefulness of the generated video narratives are reported in Table 6.9 and presented in Figure 6.17.

We could not calculate the Kruskal Wallis, Mann Whitey and Spearman RHO tests because of the number of responses of the narrative types (D and S) which were < 5. Consequently, we compare between the Perceived Usefulness categories and between the narrative types based on the experts' responses.

Table 6.9: Evaluation Results of the Perceived Usefulness of the narratives. Each narrative type (CO=Combinational, CR=Correlative, D=Derivative, S=Super Ordinate) has number of experts' assessments (N) and the Median, AVG and (STDV) of their assessments per each Perceived Usefulness category (1= Extremely Unlikely, 2= Quite Unlikely, 3= Slightly Unlikely, 4= Neutral, 5= Slightly Likely, 6= Quite Likely and 7= Extremely Likely).

Perceived usefulness scale items	CO	CR	D	S	All
	N=6	N=6	N=3	N=3	N=18
PU1–Identify key points related to	6.0	6.50	7.0	7.0	6.5
patient quality of life.	5.83	6.33	6.67	6.67	6.28
	(1.17)	(0.82)	(0.58)	(0.58)	(0.87)
PU2–Focus on key points at a	6.0	6.0	6.0	6.0	6.0
time.	5.83	5.83	5.33	6.0	5.78
	(0.75)	(0.41)	(1.15)	(0)	(0.63)
PU3–Identify key topics/concepts	6.0	6.50	7.0	7.0	6.5
to be aware of.	6.0	6.50	6.67	6.67	6.39
	(0.89)	(0.55)	(0.58)	(0.58)	(0.68)
PU4–Link main points mentioned	6.5	6.0	6.0	6.0	6.0
in the videos.	6.0	5.83	5.67	5.33	5.78
	(1.26)	(1.47)	(0.58)	(1.15)	(1.13)
Overall Perceived Usefulness	5.75	6.25	6.50	6.50	6.13
usefulness–AVG score for all	5.92	6.13	6.08	6.17	6.06
Perceived Usefulness scales per	(0.65)	(0.44)	(0.72)	(0.58)	(0.54)
narrative type.					

Perceived Usefulness. We refer to the column (All) in Table 6.9 where we can get an overview of the Perceived Usefulness of each narrative type based on the 18 experts' responses.

The median and the average experts' responses showed that they found the generated narratives help them to be aware of the key topics and concepts (PU3) with a median of 6.5 and an average score of 6.39 out of 7 (Quite likely to Extremely likely). This means that the video narratives enable the experts to identify key points related to patients' health related quality of life (PU1) with a median of 6.5 and an average score of 6.28 (Quite likely to Extremely likely). Hence, the experts thought that the video narratives can help to focus on these key points (PU2) and link them together (PU4). The overall Perceived Usefulness values of the generated narratives state that the narratives are Quite likely to raise awareness of health related quality of life if used with users. As a result, there are pointers towards the usefulness of the video narratives to support perceiving domain learning if used with learners to help answering RQ4.

We are referring to the columns in Table 6.9. the average scores showed that the experts found that the generated narratives (CR, D and S) are quite likely (AVG ≥ 6.25) to help the users be aware of the domain concepts. This can be traced to the design of the CR, D and S narratives which focus on specific concepts within one topic. Meanwhile, the CO narratives are slightly likely (AVG= 5.92) to be used to raise awareness of the domain concepts as they link different concepts from different topics. As a result, the relationship between the concepts might not be as obvious as with the other narratives.

These results have been visualised in Figure 6.17.

Cognitive Work Load. The aim of this part in the evaluation study is to assess the Cognitive Work Load expected while using the generated video narratives for the second domain. For this purpose, we have analysed the experts' responses on the NASA-TLX sub-scale items after watching the video narratives. We have assessed the Mental Work Load, Effort, Frustration and Performance sub-scales and the results are presented in Figure 6.18. The range of scores is from 1 (the lowest) to 20 (the highest). We illustrate the scores presented in Figure 6.18 based on the experts feedback. It is noticeable that the video narratives



Figure 6.17: Evaluation results of the Perceived Usefulness of the video narratives.

have a positive impact on the experts' ability to identify domain concepts. The number of experts, who scores ≥ 15 out of 20 on the Performance scale, were 12 and all of the score is ≥ 10 . Experts express these Performance scores in their feedback. Example of positive feedback "Using videos is a good way to notice quality of life needs of patients living with respiratory illnesses so I felt I was successful in identifying new terms", "I think the method used generates quite successful awareness.". Example of negative feedback "Some of the videos linked the concepts really well, but some were not very relevant", "The videos were extremely useful at identifying key terms listed in the introduction. However, not all narratives linked the areas of quality of life well. For example, in the narrative with treatment being a part of independence, the treatment was mentioned, but the quality of life area 'independence' was not really addressed."

On the Mental Work Load scale, 16 experts scores were ≤ 5 and 2 experts scores were > 5. Example of positive feedback "I felt virtually no mental demand from this method of raising awareness, other than feeling sympathy for those telling their stories.", "Not very demanding, but needed to keep my focus at all times." Example of negative feedback "Linking concepts in this section was a bit more demanding and required more concentration, as the links were not as obvious and did not apply to each segment", "It is simple but i need to be more focus to understand the key points."

For the Effort scale, 16 experts score ≤ 5 while 2 experts score ≥ 6 . Experts' feedback illustrated these scores. Example of positive feedback: "Watching of the videos did not require much effort. Some of patients' experiences were interesting to listen to", "I feel like taking in information from videos makes the process very easy. Because there are lots of videos, it can be a little more tiresome but otherwise a great way to convey information.". Example of negative feedback: "More effort was needed watching these videos as due to looking for the links between the concepts."

Finally, for the Frustration scale, 15 experts score ≤ 5 and 3 experts score ≥ 6 . Example of experts' positive feedback: "I didn't feel frustrated", "I would not say the videos were frustrating to watch, but they did cause slight discomfort when listening about difficulties of living with the disease." Example of experts' negative feedback: "Sometimes one of the topics in the description was not the important topic or mentioned briefly. This was often the cases with 'Thinking' and the video with 'child' as a concept", "In the case of the videos where you expected a link between the issues and really there wasn't any, it was frustrating due to created expectations."

Qualitative feedback from experts. We summarise the positive and negative feedback from the experts to highlight the overall strength and weakness points in the generated video narratives.

For the Combinational narratives, the experts' found that the strength points are the linking of two concepts from different topics to help to have a clear view about them and how they interact with each other. However, the weakness point is that the video narratives sometimes focus on one concept more than others which affects the quality of the link between the concepts. Example of the experts' positive feedback: "Most of the video segments provided some insight into the quality of life areas that usually are very important to patients. It is great that the videos can be segmented to automatically select the topics of interests", "It shows how different areas of quality of life needs often interact with each other." Example of the expert negative feedback: "In some cases, one issue is mentioned much more than the other. as a result, the video is mostly focused on one of the two issues and not necessarily the link between them", "It does not always highlight all the points made in the video segment."

With the Correlative narratives, the experts found that the video segments provide an elaboration on a key point by exploring related concepts. Meanwhile, some video segments have been selected because the concepts are mentioned in them, but with no deep elucidation. Example of the experts' positive feedback: "Most of the identified videos were correctly identified and the links were obvious", "Some of the videos provide useful understanding of how different areas of quality of life can be linked. It is probably useful for patients to see that these dependencies relate to other people as well." Example of the expert negative feedback: "In some of the videos it seems that they only mention the point but do not delve into it", "Some of the videos were not that relevant to the introductory narrative."

The Derivative narrative was found useful to help in focusing on key concepts and show different levels of needs but in some segments, the relationship between the parent and its specific concept was not clear. Example of the experts' positive feedback: "you can easily focus on specific segments of a video rather than going through the entire videos, which can at times be quite lengthy", "It shows the different level's of needs that in the end turns the entire quality of life." Example of the experts' negative feedback: "Sometimes, the way a work is linked to its parent is not objective and needs to come out of the general concept of the video."

Finally, the Super Ordinate narratives were found to help in focusing on key domain concepts and being aware of them, but more domain details were missed in some segments. Example of the experts' positive feedback: "I felt like it was more targeted to the specific areas", "It helps to be aware of the key points in a concept." Example of the experts' negative feedback: "Does not take into advantage the general context."



Figure 6.18: Cognitive workload estimation of the generated video narratives. The scores range is from 1-Low to 20-High.

6.6.4 Discussion

We have implemented the VISC-L and VIN-L frameworks in the second domain (Health Related Quality of Life awareness). The work in this domain is part of the InAdvance project. For this domain, we have designed a taxonomy which has been evaluated by experts from InAdvance project and used to implement the frameworks. The characterised video segments generated after implemented the VISC-L framework have been evaluated with an expert from In Advance project. Additionally, the video narratives generated after implementing the VIN-L framework have been evaluated with experts from InAdvance project and PhD researchers in the health domain from the University of Leeds.

The aim of the work conducted in this chapter is to answer RQ3 and RQ4. For RQ3:

How to generate good quality narratives that can be used to support

perceiving domain learning?

The results as presented in (Section 6.6.2) showed that the overall quality of the video narratives was good with an average quality 4.0 out of 5.0. This include assessing the quality of the video narrative's: Introduction, the link of the video segments to the domain, the description of the video segments, the content of the video segments and the conclusion statement. No statistically significant difference found between the quality categories among the video narratives and between the narrative types according to the difference in their quality categories.

To answer RQ4:

Are the generated narratives useful to support perceiving domain learning?

The results as presented in (Section 6.6.3) showed that the video narratives have a potential to raise awareness of Health Related Quality of Life needs with average score of 6.0 out of 7. This includes assessing the: clarity of the introduction statement, the video segments are linked to the domain aspects mentioned in the introduction, the characterisation of the video segments, the segments contents are linked to the domain and the narratives is clearly summarised via a conclusion statement.

However, there are some limitations related to the implementation of the frameworks (VISC-L and VIN-L) in this domain, and due to the evaluation studies conducted. For the VISC-L implementation, first the health domain was not presented via taxonomy; for that purpose we have designed the domain taxonomy as presented in (Section 6.2). The number of videos that passed the filters of (transcript availability and domain relevancy) were 60 patient stories only to be used as an input for VISC-L. This affect the coverage of the domain topic's concepts within the collected videos: PHYSICAL HEALTH (23 concepts covered out of 57), PSYCHOLOGICAL HEALTH (9 concepts covered out of 42), ENVIRONMENT (8 concepts covered out of 29), LEVEL OF INDEPENDENCE (4 concepts covered out of 40), SOCIAL RELATIONSHIP (11 concepts covered out of 19), and PERSONAL VALUES AND BELIEFS (5 concepts covered out of 19).

Additionally, three concepts only have their generic concepts presented as focus in the video segments which affect the number of Derivative narratives (3 only). Three generic concepts have some of their focus concepts present as a focus in the video segments which affect the number of Super Ordinate narratives (3 only). Additionally, many video segments focus on the same domain concepts because of their popularity, e.g, the concepts related to Physical Health are presented as a focus in the majority of the video segments. Adding to that, the duration of the narratives hinders us from selecting all possible video segments that focus on the target concept. Consequently, only 71 segments out of 306 have been used to generate the video narratives. Furthermore, some topics have not been widely represented in the video segment regarding to their unpopularity, e.g. Personal Values and Beliefs and Level Of Independence topics. As a consequence, we could not generate more video narratives to cover many of their concepts.

To compare the implementation of VISC-L and VIN-L in the two domains: in the first domain has been presented with a taxonomy which we used in our work while we build the taxonomy for the second domain. The number of the collected videos of the first domain was 63 tutorials with corpus size of 110594 tokens while with the second domain, the number of the collected videos was 60 with corpus size of 10122 tokens. The VISC-L output of the first domain was 247 characterised segments used as an input for VIN-L to generate: 51 Correlative narratives, 6 Combinational narratives, 15 Derivative narratives and 13 Super Ordinate narratives. While in the second domain, the VISC-L output was 306 segments used as an input to VIN-L to generate: 43 Correlative narratives, 10 Combinational narratives, 3 Derivative narratives and 3 Super Ordinate narratives. The VIN-L output of the first domain has been evaluated with experts and learners while with the second domain it was evaluated with experts only due to the time scale of this PhD research. In both domains, VIN-L expert evaluation results showed that it generates good quality of video narratives which has some pointers towards answering RQ3. In the second domain, the experts agreed that the video narratives have a potential to be support learning which is the same in the first domain when the learners proved that the generated video narratives also have a potential to support perceiving domain learning–RQ4. A future studies can be

conducted to compare between the video narratives and the learning style theory, see (Section 7.4.2).

Thirdly, there was no high Cognitive Work Load (Mental, Effort, Frustration) with using video narratives, and the score of the Performance was high as the video narratives have a potential to be useful in supporting perceiving domain learning.

To summarise, in both domain the video narratives showed potential pointers towards helping experts focus on the key domain topics and concepts and being aware of some domains concepts. There is an agreement among the experts from both domains that there is a need for more elaboration on the focus topics and concepts within the video segments included in the video narrative. This can be tackled by selecting more domain specific videos by re running the search schema to collect more videos per search schema as now it is limited to 3 videos per a search schema and this number can be increased. Accordingly, more new concepts can be assigned for segment characterisation which can give more elaboration to the: generic concept when Super Ordinate narratives are used, domain topics when more concepts within the focus topics can be presented within the Combinational narratives, focus concept when more similar concepts can be presented within the Correlative narrative. In the second domain, the experts did not complain about the segments' abruption which we can explain by the segment boundary condition we use in this domain, see (Section 6.4.2).

The main limitation of the work in this domain is the evaluation study with learners which we have not conducted according to the thesis time constraints. This is affected us from answering the RQ4 with learners but instead the expert evaluation study gave pointers towards the potential usefulness of using the video narratives to support perceiving the domain learning. The other possible limitations are related to the: type of the experimental study which is just to compare between the narrative type not with other learning theories, and the participants size selection in the evaluation study. These issues are similar to limitations described in (Section 5.6.5) and added as limitations to (Section 7.3) and future work in (Section 7.4).

6.7 Summary

In this chapter we have presented the implementation of the research methodology in the Health Related Quality of Life of Awareness (HRQLA) domain. The aim is to prove the generality and the applicability of the VISC-L and VIN-L by implementing them in completely different domains related to soft skills. This domain is funded by the InAdvance project that proposes a new model of palliative care for elderly people. Our contribution to this project is by generating video narratives to be used to raise awareness of Health Related Quality of Life needs of patients having chronic illness–respiratory illness. We have generated 59 domain related video narratives. These narratives have been evaluated by experts from the InAdvance project and from the University of Leeds to answer the following research questions:

How to generate good quality narratives that can be used to support perceiving domain learning?

Are the generated narratives useful to support perceiving domain learning?

The results revealed that the generated narratives are of good quality and the video segment content is related to the domain key concepts. Furthermore, the video narratives have a potential to be useful if used as learning materials to identify key domain concepts and to realise the linking between them. The Cognitive work Load while watching the video narratives is very low in the scales (Mental, Effort and Frustration) while the Performance scale was high. The generated video narratives as evaluated by experts showed pointers to positively answer RQ3 and have a potential to answer RQ4.

Chapter 7

Conclusion and Future Work

7.1 Synopsis

This research aims to support video based learning to encourage learners to be interactive indirectly with the video content by identifying the domain key points and the relationships between them. For this purpose, our ultimate goal is to develop an automatic generic approach for generating video narratives to support learning.

To achieve this goal, in Chapter 1, we formulated four research questions: **RQ1** How to automatically generate good quality characterisations of video segments that can be used for learning?; **RQ2.** Does the characterisation of video segments help in identifying and focusing on the key domain concepts?; **RQ3.** How to generate good quality narratives that can be used to support domain learning?; and **RQ4.** Are the generated narratives useful to support domain learning?

In Chapter 2, we conducted background research to introduce the context of this thesis and presented some of the challenges associated with it. We reviewed the possible approaches to address the defined challenges, which are to segment and characterise videos for learning, and use narratives in learning. The gaps in the reviewed approaches have been identified and our approach has been defined and we explained our ultimate goal to solve the identified challenges. In Chapter 3, we presented our methodology and the theories we adopted to design our generic and novel frameworks. Additionally, we have presented two domains and demonstrated their importance as domains for learning. These domains were used to implement our frameworks to prove their generality and applicability.

In Chapter 4, we implemented the first framework in the first domain. Accordingly, we demonstrated how we implemented the theories we have adopted to design the framework. For that purpose, we have implemented the three core algorithms we designed for this framework. The output from the first framework has been evaluated with experts and was used as an input for the second framework.

In Chapter 5, we have implemented and evaluated the second framework which represents the ultimate goal of this thesis. The output of the framework has been evaluated with experts and learners to prove its Quality and Usefulness for learning.

In Chapter 6, we implemented the first and the second frameworks in the second domain to prove their generality. The output has been evaluated with domain experts. The work in this domain is part of the InAdvance European Project, for which we have also designed a taxonomy which is publicly accessible and can be used as a domain knowledge representation.

In this chapter we have summarised the work conducted in this thesis, presented our contributions and detailed our future work.

7.2 Contributions

By addressing the research questions, which we presented in Chapter 1, we have contributed to:

• *Video segmentation and characterisation.* We proposed the generic video segmentation and characterisation framework VISC-L to support learning

as described in (Section 3.5) and implemented in (Sections 4.4 and 6.4). Within this framework we have designed an algorithm for video transcript segmentation to generate initial segments where we adopted the Text Tiling approach. Since we are using the video transcript only to generate the segments, this makes our algorithm generic and can be applied in any soft skills educational context. To finalise the generation of the segments, we characterised the initial segments using a domain taxonomy. We used the semantic tagging algorithm we designed to identify the domain related terms in the video transcript by mapping them to the concepts from the domain taxonomy. For contextual characterisation to solve word sense ambiguity, we used the BERT model. To finalise the characterisation work, we combined the semantic and contextual characterisations. The adjacent segments that show a match in their focus topics and concepts were aggregated to finalise the segmentation work.

An evaluation study within the first domain examined the Quality and the Perceived Usefulness of video segmentation and characterisation by comparing VISC-L with Google methodology. The results from the study gave two indications. Firstly, they indicated that using characterised video segments could help learners identify new domain terms in the summaries they had written after watching the videos. Secondly, the results showed that there was no statistically significant difference between VISC-L and Google video segmentation and characterisation. The Perceived Usefulness of segmentation and characterisation with Google was slightly better than VISC-L. Participants' feedback indicated that the format used to present the characterisation has influenced the usefulness-the natural language descriptions offered by Google were easier to follow than the list of concepts offered in the VISC-L interface. Nevertheless, it is our conceptual characterisation that enables us to generate the video narratives while Google characterisation can not be used to generate the narratives using VIN-L. The Quality with both VISC-L and Google shows that their generated characterised segments were helpful. Having video segments characterised with domain concepts allow us to use the hierarchy of these concepts to link related segments together. We combined VISC-L with Google's approach: VISC-L to extract the concepts and Google to formulate titles. We also applied VISC-L in our healthcare domain, focusing on patient quality of life awareness.

• Video narratives for learning. We have developed a novel generic framework (VIN-L) to link different segments from different videos following the Ausubel Subsumption Theory of meaningful learning as described in (Section 3.6) and implemented in (Chapter 5 and Section 6.5). Consequently, we have generated four types of video narratives.

An evaluation study within the first domain with domain experts examined the Quality and the Perceived Usefulness of the generated video narratives. The second evaluation study involved advanced learners from Amazon Mechanical Turk in which we examined the Quality, Perceived Usefulness and the Cognitive Work Load of the generated video narratives. The results showed that there is a statistically significant difference among the video narratives in terms of their Quality and Perceived Usefulness. With regards to Learning Effect, the results from the study with the advanced learners showed that there was new unique terms mentioned in the summaries after watching the video narratives. The Perceived Usefulness results showed that participants' feedback (expert and advanced learner) indicated that the text format used to present the concepts and topics to be learnt, the type of linking used and the characterisation of the segments has influenced the Usefulness. The Cognitive Work Load from the study with the advanced learners showed that the video narratives were easy to watch, less frustrating than watching full videos, and their performance was high. We responded to the recommendations from both studies and added more clarification to the type of linking used for each narrative type. This includes adding a figure to visualise the type of linking used which is inspired from the taxonomy structure. We have added this change to the video narratives we generated in the second domain (the healthcare domain). The expert study showed that the linking was clear and no issues were raised about its clarity.

• Video based learning environments. We have two main contributions to this area. Firstly, by presenting an automatic way to search and collect

domain related videos from online video platforms (e.g. YouTube). This included designing a search schema (Section 3.5.1) that is based on a domain taxonomy and implementing it using Python libraries in (Sections 4.3 and 6.3). This way of collecting domain related videos can help teachers to collect videos quickly and then run a quick manual check to decide their suitability as we did in (Sections 4.3 and 6.3). Secondly, we have evaluated VISC-L in the first domain to prove its generality and usefulness to aid using video segments for learning. Additionally, we have presented a pedagogical validation of the VIN-L framework in the first domain (Chapter 5) and the results proved that there pointers towards the good quality of the generated narratives and they have a potential to support perceiving domain learning. An expert evaluation study for the video narratives in the health domain (Section 6.6) is also presented, showing the narratives' good quality and potential usefulness to raise awareness of health related quality of life needs.

7.3 Limitations

In our work, there are limitations related to the following:

- The data we used is limited to video transcripts. However, the visual contents could enhance the segmentation and characterisation of the videos. This would require an improvement to the design of the algorithm to manipulate different video modalities.
- Our second domain is not computationally represented via a taxonomy, consequently we build this taxonomy manually which makes the VISC-L implementation semi-automatic. However, taxonomy automatic generation methods can be utilised as described in (Section 3.5.1).
- The collected videos need to be screened manually to exclude irrelevant videos. This part is not automated yet which could be automated in future by comparing the similarity, between the domain topics and concepts with the words in the meta data associated with the collected videos (e.g. title, key words, video description). To decide which videos to exclude, an

experiment study is needed to decide the similarity threshold and to evaluate the relatedness of the collected videos to their domain.

- Limited number of videos which affect the coverage of the domain concepts-not all domain concepts presented in the videos.
- Limited number of experts and users participated in the studies affect the result to be not statistically significant. Moreover, we could not ask for more users from Amazon Turk to participate according to the limited funding available for the study. The number of the participants in the evaluation studies can be increased in future studies to give more statistically significant results.
- We have not evaluated the video narratives in the health domain with learners. This is due the scope of this thesis where we were able to prove the quality and the usefulness of the video narratives of the first domain with experts and learners.
- The video narratives provide indirect interactivity for the learners where we enable them to identify and recognise the relationships between different domain concepts. However, we did not provide active watching tools for learners.
- The textual format added to augment the video narratives needs more clarification to meet different students' needs and learning abilities.

The possible ways to overcome these limitations will be outlined in our future work.

7.4 Future Work

In response to the limitations of this thesis identified above, we discuss the future work (Immediately implementable and Long term).
7.4.1 Immediately implementable

According to the aim of this thesis which is to prove that the generated video narratives support learning, we would like to prove this with the second domain as well. This is can be done by commencing an evaluation study with learners who could be university students or workers from Amazon Mechanical Turk. All the resources which are: the video narratives of the second domain, the Google forms and the Mechanical Turk study design are all ready to be used. The reason for not being able to commence this study in this thesis and its implication on answering the RQ4 in the second domain based on learners responses is explained in (Section 6.6.4). Another study can be commenced to investigate different methods to automatically generate user-friendly textual data to better display the domain related information included in the video narratives and segment characterisations. We have already characterised the video segments automatically and augmented the video narratives with text. The source code which can be used to add the new amendments to both the video segment characterisation and video narratives is available on our GitHub account. Additionally, the Google forms to be used to evaluate the characterisation of the video segments (Section 4.5) and video narratives (Sections 5.3 and 6.6) are designed and ready to be re-used.

7.4.2 Long term

In the long term, the generated narratives can be included as learning materials in a learning platform, e.g, Active Video Watching (AVW). This will help in providing extra resources to help in further elaborating the domain's key areas. Also, the video narratives can be provided as personalised feedback based on a learner's learning style to help in clarifying domain related misconceptions. Accordingly, an evaluation study can be conducted to evaluate the difference between the video narratives types as style for learning with beginner learners to overcome the limitations described in (Sections 5.6.5 and 7.3).

With our video narratives evaluation we focus on one factor which is the familiarity with the skills. However, there could be other factors to consider in the evaluation study: learning styles, motivational styles (self regulated and non self regulated learners), automatically generate different summarisation styles, and familiarity with the interface which may influence how people are interacting with those narratives. This requires further researches to investigate them.

Furthermore, the work could be applied to generate narratives focusing on different chronic illnesses to raise awareness of health related quality of life needs. This is because the taxonomy we designed is generic and any specific chronic illness physical symptoms can be added. As a result, related video narratives would be generated and used as informative materials for patients, carers and specialists to explore illnesses from a patients' point of view. In our work, we have used domain tutorials delivered by expert to fine tune the classifier used in VISC-L; these tutorial segments can also be added to the video narratives. This would help in providing learning items and patient experience on different aspects of a specific illness.

References

- ABOLKASIM, E.N.A. (2019). Semantic Approach to Model Diversity in a Social Cloud. Ph.D. thesis, University of Leeds. 111, 118
- ABU-RASHEED, H., EFTHYMIOU, Y., FATHI, M., GHADAMIGHALANDARI,
 P., MEDINA, J.L., GARCÍA, C.O., TSARDANIDIS, G., ZENKERT, J. &
 ZGERAS, G. (2023). Supporting remote students through utilizing web-based exercise-templates and a mobile learning chatbot for creating and interacting with learning materials. In *European Conference on Technology Enhanced Learning*, 668–673, Springer. 34
- ADOMA, A.F., HENRY, N.M. & CHEN, W. (2020). Comparative analyses of bert, roberta, distilbert, and xlnet for text-based emotion recognition. In 2020 17th International Computer Conference on Wavelet Active Media Technology and Information Processing (ICCWAMTIP), 117–121, IEEE. 123, 124
- AL-TAWIL, M., DIMITROVA, V. & THAKKER, D. (2020). Using knowledge anchors to facilitate user exploration of data graphs. *Semantic Web*, **11**, 205–234. 65
- ALLEN, J.F. (1983). Maintaining knowledge about temporal intervals. Communications of the ACM, 26, 832–843. 46
- ALSHARE, K. & HINDI, N.M. (2004). The importance of presentation skills in the classroom: Students and instructors perspectives. *Journal of Computing Sciences in Colleges*, **19**, 6–15. 103
- AQQAL, A., RENSING, C. & STEINMETZ, R. (2007). The macro design as an own task in webtools production: Ideas, concepts and a tool. In *Creating*

New Learning Experiences on a Global Scale: Second European Conference on Technology Enhanced Learning, EC-TEL 2007, Crete, Greece, September 17-20, 2007. Proceedings 2, 420–425, Springer. 23, 48

- ARCHAMBAULT, L., LEARY, H. & RICE, K. (2022). Pillars of online pedagogy: A framework for teaching in online learning environments. *Educational Psychologist*, 57, 178–191. 34, 52
- ARKSEY, H. & O'MALLEY, L. (2005). Scoping studies: towards a methodological framework. International journal of social research methodology, 8, 19–32. 113
- AUSIN, M.S., MANIKTALA, M., BARNES, T. & CHI, M. (2021). Tackling the credit assignment problem in reinforcement learning-induced pedagogical policies with neural networks. In *International Conference on Artificial Intelligence in Education*, 356–368, Springer. 34
- AUSUBEL, D.P. (1963). The psychology of meaningful verbal learning. Grune & Stratton. 51
- AUSUBEL, D.P. (2012). The acquisition and retention of knowledge: A cognitive view. Springer Science & Business Media. 53, 64
- AUSUBEL, D.P., NOVAK, J.D., HANESIAN, H. et al. (1968). Educational psychology: A cognitive view, vol. 6. Holt, Rinehart and Winston New York. 11, 25, 53, 55, 64, 85
- AVRON, U., GERSHTEIN, S., GUY, I., MILO, T. & NOVGORODOV, S. (2022). Automated category tree construction in e-commerce. In Proceedings of the 2022 International Conference on Management of Data, 1770–1783. 72
- BARZOLA, V., ALVARADO, E., LOJA, C., VELEZ, A., SILVA, I. & ECHEVERRIA,
 V. (2022). Analyzing speech data to detect work environment in group activities.
 In International Conference on Artificial Intelligence in Education, 357–361,
 Springer. 35
- BENKADA, C. & MOCCOZET, L. (2017). Enriched interactive videos for teaching and learning. In 2017 21st International Conference Information Visualisation (IV), 344–349, IEEE. 40

- BIANCHI, C. & RAGONESE, R. (2023). Accounting semiotically for new forms of textuality and narrativity in digital brand storytelling. Advances in Brand Semiotics & Discourse Analysis, 33. 23, 48, 52
- BLOOR, M. & BLOOR, T. (2013). The practice of critical discourse analysis: An introduction. Routledge. 128
- BOURGUET, M.L. (2022). Measuring learners' self-regulated learning skills from their digital traces and learning pathways. In European Conference on Technology Enhanced Learning, 500–506, Springer. 36
- BOURINA, H.V. & DUNAEVA, L.A. (2019). Role of hypertext in teaching foreign languages. *E-Learning and Digital Media*, **16**, 110–121. 50
- BRAME, C.J. (2016). Effective educational videos: Principles and guidelines for maximizing student learning from video content. CBE—Life Sciences Education, 15, es6. 69, 102
- BYWATER, J.P., FLORYAN, M. & CHIU, J.L. (2021). Discs: A new sequence segmentation method for open-ended learning environments. In *International Conference on Artificial Intelligence in Education*, 88–100, Springer. 23, 30, 33, 36, 42
- CAEIRO-RODRÍGUEZ, M., MANSO-VÁZQUEZ, M., MIKIC-FONTE, F.A., LLAMAS-NISTAL, M., FERNÁNDEZ-IGLESIAS, M.J., TSALAPATAS, H., HEIDMANN, O., DE CARVALHO, C.V., JESMIN, T., TERASMAA, J. et al. (2021). Teaching soft skills in engineering education: A European perspective. *IEEE Access*, 9, 29222–29242. 103
- CAGLIERO, L., CANALE, L. & FARINETTI, L. (2019). Visa: A supervised approach to indexing video lectures with semantic annotations. In 2019 IEEE 43rd Annual Computer Software and Applications Conference, vol. 1, 226–235, IEEE. 44
- CASTRO, M.D.B. & TUMIBAY, G.M. (2021). A literature review: efficacy of online learning courses for higher education institution using meta-analysis. *Educ. and Info. Techn.*, 26, 1367–1385. 23, 40

- CHIU, P.S., CHEN, H.C., HUANG, Y.M., LIU, C.J., LIU, M.C. & SHEN, M.H. (2018). A video annotation learning approach to improve the effects of video learning. *Innovations in Education and Teaching International*, 55, 459–469. 30, 40
- CIASULLO, M.V., LIM, W.M., MANESH, M.F. & PALUMBO, R. (2022). The patient as a prosumer of healthcare: insights from a bibliometric-interpretive review. *Journal of Health Organization and Management*. 107
- CINQUE, M. (2017). MOOCs and soft skills: a comparison of different courses on creativity. *Journal of E-learning and Knowledge Society*, **13**. 102
- COHEN, G. & COHEN, A. (2023). Exploring students' problem-solving challenges in self-regulated learning through training video prompts. In *European Conference on Technology Enhanced Learning*, 536–541, Springer. 22, 34
- COHEN, J. (1960). A coefficient of agreement for nominal scales. *Educational and* psychological measurement, **20**, 37–46. 120, 215
- COLAÇO, A., OLIVEIRA, A.P. & ZAGALO, N. (2022). Know me: Promoting gender equality in education through an interactive digital narrative. In International Conference on Technology and Innovation in Learning, Teaching and Education, 213–226, Springer. 49
- COLASANTE, M. & DOUGLAS, K. (2016). Prepare-participate-connect: Active learning with video annotation. Australasian Journal of Educational Technology, 32. 30, 40
- CONKEY, C.A., BOWERS, C., CANNON-BOWERS, J. & SANCHEZ, A. (2013). Machinima and video-based soft-skills training for frontline healthcare workers. GAMES FOR HEALTH: Research, Development, and Clinical Applications, 2, 39–43. 103
- CROSBIE, R. (2005). Learning the soft skills of leadership. Industrial and commercial training, **37**, 45–51. 102, 103

- DAMARELL, R.A., MAY, N., HAMMOND, S., SLADEK, R.M. & TIEMAN, J.J. (2019). Topic search filters: a systematic scoping review. *Health Information & Libraries Journal*, **36**, 4–40. 205
- DANES, F. (1974). Functional sentence perspective and the organization of the text. Papers on functional sentence perspective, 23, 106–128. 10, 15, 55, 62, 63, 80
- DAS, A. & DAS, P.P. (2019). Automatic semantic segmentation and annotation of MOOC lecture videos. In Int. C. on Asian Digital Libraries, 181–188, Springer. 23, 24, 30, 42, 47
- DAS, A. & DAS, P.P. (2020). Semantic segmentation of MOOC lecture videos by analyzing concept change in domain knowledge graph. In Int. C. on Asian Digital Libraries, 55–70, Springer. 45
- DAVIS, F.D. (1989). Perceived usefulness, perceived ease of use, and user acceptance of information technology. *MIS quarterly*, 319–340. 132, 135, 160, 231
- DEVARAKONDA, A., NAUMOV, M. & GARLAND, M. (2017). Adabatch: Adaptive batch sizes for training deep neural networks. arXiv preprint arXiv:1712.02029. 123
- DEVER, D.A., SONNENFELD, N.A., WIEDBUSCH, M.D. & AZEVEDO, R. (2022). Pedagogical agent support and its relationship to learners' self-regulated learning strategy use with an intelligent tutoring system. In *International Conference* on Artificial Intelligence in Education, 332–343, Springer. 34
- DEVLIN, J., CHANG, M.W., LEE, K. & TOUTANOVA, K. (2018). BERT: Pre-training of deep bidirectional transformers for language understanding. arXiv:1810.04805. 78, 121
- DEY-PLISSONNEAU, A., LEE, H., PRADIER, V., SCRINEY, M. & SMEATON, A.F. (2021). The l2l system for second language learning using visualised zoom calls among students. In *European Conference on Technology Enhanced Learning*, 409–413, Springer. 50

- DI GANGI, P.M., GOH, S.H. & LEWIS, C.C. (2017). Using social media to support presentation skill development in traditional classroom environments. *Journal of Organiz. and End User Comput. (JOEUC)*, **29**, 68–91. 103, 111
- DIMITROVA, V. & MITROVIC, A. (2022). Choice architecture for nudges to support constructive learning in active video watching. *International Journal* of Artificial Intelligence in Education, **32**, 892–930. 132
- DIMITROVA, V., MITROVIC, A., PIOTRKOWICZ, A., LAU, L. & WEERASINGHE, A. (2017). Using learning analytics to devise interactive personalised nudges for active video watching. In *Proceedings of the 25th conference on user modeling*, *adaptation and personalization*, 22–31. 104
- DIXON, J., BELNAP, C., ALBRECHT, C. & LEE, K. (2010). The importance of soft skills. *Corporate finance review*, 14, 35. 102
- DODSON, S., ROLL, I., HARANDI, N.M., FELS, S. & YOON, D. (2019). Weaving together media, technologies and people: Students' information practices in flipped classrooms. *Information and Learning Sciences*. 30, 40, 46
- DUTTA, A. & ZISSERMAN, A. (2019). The via annotation software for images, audio and video. In Proceedings of the 27th ACM international conference on multimedia, 2276–2279. 30, 40, 46
- ESTERL, N., BERGER, S. & NISTOR, N. (2022). Digital media in schools during the covid-19 lockdown: Teachers' experiences with choosing teaching strategies. In European Conference on Technology Enhanced Learning, 507–513, Springer. 23, 35
- FAJTL, J., SOKEH, H.S., ARGYRIOU, V., MONEKOSSO, D. & REMAGNINO,
 P. (2019). Summarizing videos with attention. In Computer Vision-ACCV
 2018 Workshops: 14th Asian Conference on Computer Vision, Perth, Australia,
 December 2-6, 2018, Revised Selected Papers 14, 39–54, Springer. 38
- FERNÁNDEZ-LÓPEZ, M. (1999). Overview of methodologies for building ontologies. In Proc. 16th International Joint Conference on Artificial Intelligence (City

Conference Center, Stockholm, Sweden, August 1999), CEUR Publications (1999). 19, 199

- FESSL, A., MAITZ, K., DENNERLEIN, S. & PAMMER-SCHINDLER, V. (2021). The impact of explicating learning goals on teaching and learning in higher education: Evaluating a learning goal visualization. In *European Conference on Technology Enhanced Learning*, 1–15, Springer. 50
- FÖRSTER, M., MAUR, A., WEISER, C. & WINKEL, K. (2022). Pre-class video watching fosters achievement and knowledge retention in a flipped classroom. *Computers & Education*, **179**, 104399. 32
- GABBAY, H. & COHEN, A. (2022). Exploring the connections between the use of an automated feedback system and learning behavior in a MOOC for programming. In *European Conference on Technology Enhanced Learning*, 116–130, Springer. 34
- GHINASSI, I. (2021). Unsupervised text segmentation via deep sentence encoders: a first step towards a common framework for text-based segmentation, summarization and indexing of media content. In ACM International Conference on Interactive Media Experiences (IMX 2021). 60, 61
- GREENHALGH, T. & HURWITZ, B. (1999). Why study narrative? *BMJ*, **318**, 48–50. 48
- GUO, S., SUN, T., GONG, J., LU, Z., ZHANG, L. & WANG, Q. (2022). Remote co-teaching in rural classroom: Current practices, impacts, and challenges. In CHI Conference on Human Factors in Computing Systems, 1–16. 50
- HALLIDAY, M.A.K.M.A.K. & MATTHIESSEN, C.M.I.M. (2014). Halliday's Introduction to functional grammar. Routledge, London. 62
- HAO, J., ZHAO, L., MILISAVLJEVIC-SYED, J. & MING, Z. (2021). Integrating and navigating engineering design decision-related knowledge using decision knowledge graph. Advanced Engineering Informatics, 50, 101366. 24, 52

- HARALDSTAD, K., WAHL, A., ANDENÆS, R., ANDERSEN, J.R., ANDERSEN, M.H., BEISLAND, E., BORGE, C.R., ENGEBRETSEN, E., EISEMANN, M., HALVORSRUD, L. *et al.* (2019). A systematic review of quality of life research in medicine and health sciences. *Quality of life Research*, 28, 2641–2650. 107
- HARIDAKIS, P. & HANSON, G. (2009). Social interaction and co-viewing with youtube: Blending mass communication reception and social connection. *Journal of broadcasting & electronic media*, **53**, 317–335. 22, 30, 32
- HART, S.G. (2006). Nasa-task load index (nasa-tlx); 20 years later. In Proceedings of the human factors and ergonomics society annual meeting, vol. 50, 904–908, Sage publications Sage CA: Los Angeles, CA. 132, 135, 231
- HE, X., WANG, J., ZHANG, Q. & JU, X. (2020). Improvement of text segmentation texttiling algorithm. In *Journal of Physics: Conference Series*, vol. 1453, 012008, IOP Publishing. 59
- HEARST, M.A. (1993). Texttiling: A quantitative approach to discourse segmentation. Tech. rep., Citeseer. 10, 55, 58, 59, 61, 80
- HEARST, M.A. (1997a). Text tiling: Segmenting text into multi-paragraph subtopic passages. *Computational linguistics*, 23, 33–64. 58, 61, 73
- HEARST, M.A. (1997b). Texttiling: A quantitative approach to discourse segmentation. Computational Linguistics, 23, 33–64. 117, 128
- HSIN, W.J. & CIGAS, J. (2013). Short videos improve student learning in online education. Journal of Computing Sciences in Colleges, 28, 253–259. 22
- HURST, J.R., SKOLNIK, N., HANSEN, G.J., ANZUETO, A., DONALDSON, G.C., DRANSFIELD, M.T. & VARGHESE, P. (2020). Understanding the impact of chronic obstructive pulmonary disease exacerbations on patient health and quality of life. *European journal of internal medicine*, **73**, 1–6. 107
- JÁROMI, M., SZILÁGYI, B., VELÉNYI, A., LEIDECKER, E., RAPOSA, B.L., HOCK, M., BAUMANN, P., ÁCS, P. & MAKAI, A. (2021). Assessment of health-related quality of life and patient's knowledge in chronic non-specific low back pain. *BMC Public Health*, **21**, 1–8. 107

- JING, W. (2015). Theme and thematic progression in english writing teaching. Journal of Education and Practice, 6, 178–187. 63
- JUNE, S., YAACOB, A. & KHENG, Y.K. (2014). Assessing the use of youtube videos and interactive activities as a critical thinking stimulator for tertiary students: An action research. *International Education Studies*, 7, 56–67. 22
- KATAGALL, R., DADDE, R., GOUDAR, R. & RAO, S. (2015). Concept mapping in education and semantic knowledge representation: an illustrative survey. *Proceedia Computer Science*, 48, 638–643. 65
- KHAN, M.L. (2017). Social media engagement: What motivates user participation and consumption on youtube? *Computers in human behavior*, **66**, 236–247. 22, 30, 32
- KHATTER, H., SHARMA, A., KUSHWAHA, A.K. et al. (2022). Web scraping based product comparison model for e-commerce websites. In 2022 IEEE International Conference on Data Science and Information System (ICDSIS), 1–6, IEEE. 38
- KIM, H.J., SO, H.J. & SUH, Y.J. (2023). Examining the impact of flipped learning for developing young job seekers' AI literacy. In International Conference on Artificial Intelligence in Education, 817–823, Springer. 32
- KITHMI, A. et al. (2016). Semantic video search by automatic video annotation using tensorflow. In Manufacturing & Industrial Engineering Symposium (MIES). 44
- KOĆ-JANUCHTA, M.M., SCHÖNBORN, K.J., TIBELL, L.A., CHAUDHRI, V.K. & HELLER, H.C. (2020). Engaging with biology by asking questions: Investigating students' interaction and learning with an artificial intelligence-enriched textbook. *Journal of Educational Computing Research*, 58, 1190–1224. 43
- KOEKENBIER, K., LEINO-KILPI, H., CABRERA, E., ISTOMINA, N., STARK,
 Å.J., KATAJISTO, J., LEMONIDOU, C., PAPASTAVROU, E., SALANTERÄ,
 S., SIGURDARDOTTIR, A. *et al.* (2016). Empowering knowledge and its connection to health-related quality of life: A cross-cultural study: A concise and

informative title: Empowering knowledge and its connection to health-related quality of life. *Applied Nursing Research*, **29**, 211–216. 107

- KÖHLER, D., PÜNTER, W. & MEINEL, C. (2023). The "how" matters: Evaluating different video types for cybersecurity MOOCs. In *European* Conference on Technology Enhanced Learning, 149–163, Springer. 32
- KRIKSCIUNIENE, D., SAKALAUSKAS, V. & LEWANDOWSKI, R. (2019). Evaluating the interdependent effect for likert scale items. In Business Information Systems Workshops: BIS 2019 International Workshops, Seville, Spain, June 26–28, 2019, Revised Papers 22, 26–38, Springer. 131
- KUSWOYO, H., SUJATNA, E.T.S., RIDO, A. & INDRAYANI, L.M. (2020). Theme choice and thematic progression of discussion section in engineering English lectures. In *Proceedings of the 4th International Conference on Learning Innovation and Quality Education*, 1–10. 63
- LANGE, C. & COSTLEY, J. (2020). Improving online video lectures: learning challenges created by media. International Journal of Educational Technology in Higher Education, 17, 1–18. 22, 33
- LARGENT, E.A. & LYNCH, H.F. (2017). Paying research participants: regulatory uncertainty, conceptual confusion, and a path forward. Yale journal of health policy, law, and ethics, 17, 61. 133
- LAU, L., MITROVIC, A., WEERASINGHE, A. & DIMITROVA, V. (2016). Usability of an active video watching system for soft skills training. In *Proc. 1st Int. Workshop on Intelligent Mentoring Systems, ITS.* 104
- LECUYER, G., RAGOT, M., MARTIN, N., LAUNAY, L. & JANNIN, P. (2020). Assisted phase and step annotation for surgical videos. *International journal of computer assisted radiology and surgery*, 1–8. 52
- LEE, H., LIU, M., SCRINEY, M. & SMEATON, A.F. (2021). Usage-based summaries of learning videos. In European Conference on Technology Enhanced Learning, 414–418, Springer. 23, 49

- LEE, H.Y. & LIST, A. (2019). Processing of texts and videos: A strategy-focused analysis. Journal of Computer Assisted Learning, 35, 268–282. 36
- LEE, Y., LIMBU, B., RUSAK, Z. & SPECHT, M. (2023). Role of multimodal learning systems in technology-enhanced learning (tel): A scoping review. In European Conference on Technology Enhanced Learning, 164–182, Springer. 35
- LI, Y., HE, H., WANG, J., CHEN, Y., WANG, C., LI, X., DAI, A., LIU, Y., XI, X., HUANG, J. et al. (2023). Effect of multidisciplinary health education based on lifestyle medicine on menopausal syndrome and lifestyle behaviors of menopausal women: A clinical controlled study. Frontiers in Public Health, 11. 107
- LINSON, A., XU, Y., ENGLISH, A.R. & FISHER, R.B. (2022). Identifying student struggle by analyzing facial movement during asynchronous video lecture viewing: Towards an automated tool to support instructors. In *International Conference on Artificial Intelligence in Education*, 53–65, Springer. 35
- LIU, C., KIM, J. & WANG, H.C. (2018). Conceptscape: Collaborative concept mapping for video learning. In Proceedings of the 2018 CHI conference on human factors in computing systems, 1–12. 65
- LIU, X., SONG, Y., LIU, S. & WANG, H. (2012). Automatic taxonomy construction from keywords. In Proceedings of the 18th ACM SIGKDD international conference on Knowledge discovery and data mining, 1433–1441. 72
- LIU, Z. & LAUW, H.W. (2022). Towards aligning slides and video snippets: Mitigating sequence and content mismatches. In *International Conference on Artificial Intelligence in Education*, 533–539, Springer. 23, 24, 30, 41, 42, 47, 52
- LUQUE, A., CARRASCO, A., MARTÍN, A. & DE LAS HERAS, A. (2019). The impact of class imbalance in classification performance metrics based on the binary confusion matrix. *Pattern Recognition*, **91**, 216–231. 120

- MACHADO, C.T. & CARVALHO, A.A. (2020). Concept mapping: Benefits and challenges in higher education. The Journal of Continuing Higher Education, 68, 38–53. 51
- MAICAN, C., CAZAN, A.M., LIXANDROIU, R., DOVLEAC, L. & MAICAN, M.A. (2021). Exploring the factors influencing the use of communication and collaboration applications. *Journal of Organizational and End User Computing* (*JOEUC*), **33**, 94–124. 103, 111
- MAINGA, W., MURPHY-BRAYNEN, M.B., MOXEY, R. & QUDDUS, S.A. (2022). Graduate employability of business students. Administrative Sciences, 12, 72. 102
- MAJUMDAR, R., PRASAD, P., KADAM, K., GATARE, K. & WARRIEM, J.M. (2023). La-reflect: A platform facilitating micro-learning and its multimodal learning analytics. In *European Conference on Technology Enhanced Learning*, 731–735, Springer. 22, 34
- MANN, W.C. & THOMPSON, S.A. (1987). Rhetorical structure theory: A theory of text organization. University of Southern California, Information Sciences Institute Los Angeles. 49
- MARQUES, A., PERALTA, M., SANTOS, T., MARTINS, J. & DE MATOS, M.G. (2019). Self-rated health and health-related quality of life are related with adolescents' healthy lifestyle. *Public health*, **170**, 89–94. 106
- MAZIRIRI, E.T., GAPA, P. & CHUCHU, T. (2020). Student perceptions towards the use of youtube as an educational tool for learning and tutorials. *International Journal of Instruction*, 13, 119–138. 50
- MCQUIGGAN, S.W., ROWE, J.P., LEE, S. & LESTER, J.C. (2008). Story-based learning: The impact of narrative on learning experiences and outcomes. In Intelligent Tutoring Systems: 9th International Conference, ITS 2008, Montreal, Canada, June 23-27, 2008 Proceedings 9, 530–539, Springer. 24, 31, 48

- MESEGUER-MARTINEZ, A., ROS-GALVEZ, A. & ROSA-GARCIA, A. (2017). Satisfaction with online teaching videos: A quantitative approach. *Innovations* in Education and Teaching International, 54, 62–67. 22, 114
- MHAMDI, C. (2017). What can video add to the learning experience? challenges and opportunities. International Journal of Information Technology and Language Studies, 1, 17–24. 32
- MICHEL, C. & PIERROT, L. (2022). Towards modelling the technology integration in elementary school. a diachronic study of teachers' digital practices during and after covid-19 lockdown. In *European Conference on Technology Enhanced Learning*, 201–214, Springer. 33
- MICULICICH, L. & HAN, B. (2023). Document summarization with text segmentation. arXiv preprint arXiv:2301.08817. 60, 61
- MIRRIAHI, N., JOVANOVIĆ, J., LIM, L.A. & LODGE, J.M. (2021). Two sides of the same coin: video annotations and in-video questions for active learning. *Educa. Techn. Research and Develop.*, 69, 2571–2588. 23, 40
- MIRZABABAEI, B., MAITZ, K., FESSL, A. & PAMMER-SCHINDLER, V. (2023). Interactive web-based learning materials vs. tutorial chatbot: Differences in user experience. In *European Conference on Technology Enhanced Learning*, 213–228, Springer. 34
- MITROVIC, A., DIMITROVA, V., WEERASINGHE, A. & LAU, L. (2016). Reflective experiential learning: Using active video watching for soft skills training. In *Proceedings of the 24th international conference on computers in education*, Asia-Pacific Society for Computers in Education. 23, 41, 46, 104, 116, 132
- MITROVIC, A., DIMITROVA, V., LAU, L., WEERASINGHE, A. & MATHEWS, M. (2017). Supporting constructive video-based learning: requirements elicitation from exploratory studies. In Artificial Intelligence in Education: 18th International Conference, AIED 2017, Wuhan, China, June 28–July 1, 2017, Proceedings 18, 224–237, Springer. 104, 116, 132

- MOHAMAD, U.H., AHMAD, M.N., BENFERDIA, Y., SHAPI'I, A. & BAJURI, M.Y. (2021). An overview of ontologies in virtual reality-based training for healthcare domain. *Frontiers in Medicine*, 8, 698855. 198
- MOHAMMADHASSAN, N., MITROVIC, A. & NESHATIAN, K. (2022). Investigating the effect of nudges for improving comment quality in active video watching. *Computers & Education*, **176**, 104340. 16, 104, 105
- MOHAMMED, A. & DIMITROVA, V. (2020). Characterising video segments to support learning. In Proceedings of the 28th Inter. Conf. on Computers in Education. Asia-Pacific Society for Computers in Education (in print). 76, 112, 117, 121, 212
- MONGKHONVANIT, K., KANOPKA, K. & LANG, D. (2019). Deep knowledge tracing and engagement with MOOCs. In *Proceedings of the 9th international* conference on learning analytics & knowledge, 340–342. 22
- MUSEN, M.A. (2015). The Protégé project: a look back and a look forward. AI matters, 1, 4–12. 202
- NAUL, E. & LIU, M. (2020). Why story matters: A review of narrative in serious games. *Journal of Educational Computing Research*, **58**, 687–707. 49
- NOVAK, J.D. (1964). Importance of conceptual schemes for science teaching. *The* Science Teacher, 10–10. 51
- NOY, N.F., MCGUINNESS, D.L. *et al.* (2001). Ontology development 101: A guide to creating your first ontology. 19, 199
- OEI, S.L., THRONICKE, A., MATTHES, H. & SCHAD, F. (2021). Assessment of integrative non-pharmacological interventions and quality of life in breast cancer patients using real-world data. *Breast Cancer*, 28, 608–617. 107
- OGHAZ, T.A., MUTLU, E.Ç., JASSER, J., YOUSEFI, N. & GARIBAY, I. (2020). Probabilistic model of narratives over topical trends in social media: A discrete time model. In *Proceedings of the 31st ACM Conference on Hypertext and Social Media*, 281–290. 49

- ORAVEC, N., MONNIN, C., GREGORA, A., BJORKLUND, B., DAVE, M.G., SCHULTZ, A.S. & CHUDYK, A.M. (2022). Protocol for a scoping review to map patient engagement in scoping reviews. *Research Involvement and Engagement*, 8, 1–8. 113
- OTTO, C., YU, R., PARDI, G., VON HOYER, J., ROKICKI, M., HOPPE, A., HOLTZ, P., KAMMERER, Y., DIETZE, S. & EWERTH, R. (2021). Predicting knowledge gain during web search based on multimedia resource consumption. In *International Conference on Artificial Intelligence in Education*, 318–330, Springer. 36
- PALAIGEORGIOU, G., PAPADOPOULOU, A. & KAZANIDIS, I. (2019). Interactive video for learning: A review of interaction types, commercial platforms, and design guidelines. In *Technology and Innovation in Learning, Teaching and Education: First International Conference, TECH-EDU 2018, Thessaloniki, Greece, June 20–22, 2018, Revised Selected Papers 1*, 503–518, Springer. 36
- PARASCHIV, A. & CERCEL, D.C. (2019). UPB at GermEval-2019 Task
 2: BERT-Based Offensive Language Classification of German Tweets. In KONVENS. 126
- PINA, J., MAZUR, R., ELLIS, G., RUDNITSKY, A., MCGINNIS-CAVANAUGH, B., ELLIS, S., HUFF, I. & FORD, C. (2022). Adapting a narrative curriculum to a remote format in the context of socially distanced middle school education resulting from covid-19. Journal of Pre-College Engineering Education Research (J-PEER), 12, 4. 49, 52
- PINANDITO, A., PRASETYA, D.D., HAYASHI, Y. & HIRASHIMA, T. (2021). Semi-automatic concept map generation approach of web-based kit-build concept map authoring tool. *International Journal of Interactive Mobile Technologies*, 15. 24, 31, 51
- POST, M. (2014). Definitions of quality of life: what has happened and how to move on. *Topics in spinal cord injury rehabilitation*, **20**, 167–180. 107

- PRAHLADH, S. & VAN WYK, J. (2020). Protocol for a scoping review of the current data practices in forensic medicine. *Systematic Reviews*, 9, 1–8. 113
- PRITONI, M., PAINE, D., FIERRO, G., MOSIMAN, C., POPLAWSKI, M., SAHA, A., BENDER, J. & GRANDERSON, J. (2021). Metadata schemas and ontologies for building energy applications: a critical review and use case analysis. *Energies*, 14, 2024. 72
- RABIN, E., SILBER-VAROD, V., KALMAN, Y.M. & KALZ, M. (2019). Identifying learning activity sequences that are associated with high intention-fulfillment in MOOCs. In *European Conference on Technology Enhanced Learning*, 224–235, Springer. 35
- RECKE, M.P. & PERNA, S. (2021). Emergent narratives in remote learning experiences for project based education. *Electronic Journal of e-Learning*, **19**, 59–70. 31, 49
- RICHTBERG, S. & GIRWIDZ, R. (2019). Learning physics with interactive videos-possibilities, perception, and challenges. In *Journal of Physics: Conference Series*, vol. 1287, 012057, IOP Publishing. 23, 36
- RISKO, E.F., ANDERSON, N., SARWAL, A., ENGELHARDT, M. & KINGSTONE, A. (2012). Everyday attention: Variation in mind wandering and memory in a lecture. Applied Cognitive Psychology, 26, 234–242. 22, 114
- ROBIN, C., O'NEILL, J. & BUITELAAR, P. (2017). Automatic taxonomy generation: A use-case in the legal domain. In *Language and Technology Conference*, 318–328, Springer. 72
- ROCHAN, M. & WANG, Y. (2019). Video summarization by learning from unpaired data. In Proceedings of the IEEE/CVF Conference on Computer Vision and Pattern Recognition, 7902–7911. 38
- ROHLOFF, T. & MEINEL, C. (2018). Towards personalized learning objectives in MOOCs. In Lifelong Technology-Enhanced Learning: 13th European Conference on Technology Enhanced Learning, EC-TEL 2018, Leeds, UK, September 3-5, 2018, Proceedings 13, 202–215, Springer. 23, 36

- SCHLOTTERBECK, D., URIBE, P., JIMÉNEZ, A., ARAYA, R., MOLEN MORIS, J.V.D. & CABALLERO, D. (2021). Tarta: Teacher activity recognizer from transcriptions and audio. In *Int. Conf. on Artificial Intelligence in Education*, 369–380, Springer. 23, 24, 30, 33, 36, 41, 46
- SCHULTEN, C., MANSKE, S., LANGNER-THIELE, A. & HOPPE, H.U. (2020). Bridging over from learning videos to learning resources through automatic keyword extraction. In *International Conference on Artificial Intelligence in Education*, 382–386, Springer. 45
- SEO, K., DODSON, S., HARANDI, N.M., ROBERSON, N., FELS, S. & ROLL, I. (2021). Active learning with online video: The impact of learning context on engagement. *Computers & Education*, 165, 104132. 36
- SERRANO-IGLESIAS, S., GÓMEZ-SÁNCHEZ, E., BOTE-LORENZO, M.L., ASENSIO-PÉREZ, J.I., RUIZ-CALLEJA, A. & VEGA-GORGOJO, G. (2021). Demonstration of scarlett: A smart learning environment to support learners across formal and informal contexts. In *European Conference on Technology Enhanced Learning*, 404–408, Springer. 31, 50
- SHAHZAD, S.K., AHMED, D., NAQVI, M.R., MUSHTAQ, M.T., IQBAL,
 M.W. & MUNIR, F. (2021). Ontology driven smart health service integration.
 Computer Methods and Programs in Biomedicine, 207, 106146. 198
- SHAO, Z., LI, Y., WANG, X., ZHAO, X. & GUO, Y. (2020). Research on a new automatic generation algorithm of concept map based on text analysis and association rules mining. *Journal of ambient intelligence and humanized* computing, **11**, 539–551. 31, 52
- SIKOS, L.F. (2017). RDF-powered semantic video annotation tools with concept mapping to Linked Data for next-generation video indexing: a comprehensive review. *Multimedia Tools and Applications*, **76**, 14437–14460. 23, 24, 30, 44, 47
- SITLINGER, A. & ZAFAR, S.Y. (2018). Health-related quality of life: the impact on morbidity and mortality. Surgical Oncology Clinics, 27, 675–684. 106

- SOSNOVSKY, S. (2022). Enriching intelligent textbooks with interactivity: When smart content allocation goes wrong. 43
- SPICER, D.B., THOMPSON, K.H. & KILGALLEN, S.M. (2020). Making the connection: using concept mapping to bring the basic sciences to the diagnosis. *Journal of Osteopathic Medicine*, **120**, 524–528. 31, 51
- SPOLADORE, D. & PESSOT, E. (2021). Collaborative ontology engineering methodologies for the development of decision support systems: Case studies in the healthcare domain. *Electronics*, **10**, 1060. 198
- SPRENGER, D.A. & SCHWANINGER, A. (2021). Technology acceptance of four digital learning technologies (classroom response system, classroom chat, e-lectures, and mobile virtual reality) after three months' usage. *International Journal of Educational Technology in Higher Education*, 18, 8. 32
- TAHERDOOST, H. (2019). What is the best response scale for survey and questionnaire design; review of different lengths of rating scale/attitude scale/likert scale. *Hamed Taherdoost*, 1–10. 132, 134, 160, 230
- TALMAN, S., BOONMAN-DE WINTER, L., DE MOL, M., HOEFMAN, E., VAN ETTEN, R., DE BACKER, I. et al. (2021). Pulmonary function and health-related quality of life after covid-19 pneumonia. *Respiratory medicine*, 176, 106272. 107
- TAMMETS, K., SARMIENTO-MÁRQUEZ, E.M., KHULBE, M., LAANPERE, M. & LEY, T. (2022). Integrating digital learning resources in classroom teaching: Effects on teaching practices and student perceptions. In *European conference* on technology enhanced learning, 379–392, Springer. 22, 35
- TANG, C.L., LIAO, J., WANG, H.C., SUNG, C.Y. & LIN, W.C. (2021). Conceptguide: Supporting online video learning with concept map-based recommendation of learning path. In *Proceedings of the Web Conference 2021*, 2757–2768. 24, 31, 51, 52

- TILK, O. & ALUMÄE, T. (2016). Bidirectional recurrent neural network with attention mechanism for punctuation restoration. In *Interspeech*, 3047–3051. 116
- TIWARI, S. & ABRAHAM, A. (2020). Semantic assessment of smart healthcare ontology. International Journal of Web Information Systems, 16, 475–491. 197
- TORRE, I., MIRENDA, L., VERCELLI, G. & MASTROGIOVANNI, F. (2022). Prerequisite graph extraction from lectures. In International Conference on Artificial Intelligence in Education, 616–619, Springer. 46
- TSENG, H., YI, X. & YEH, H.T. (2019). Learning-related soft skills among online business students in higher education: Grade level and managerial role differences in self-regulation, motivation, and social skill. *Computers in Human Behavior*, **95**, 179–186. 102
- TSENG, S.S. (2021). The influence of teacher annotations on student learning engagement and video watching behaviors. Int. J. of Ed. Techn. in HE, 18, 1–17. 23, 36, 40
- TUSHEV, M., EBRAHIMI, F. & MAHMOUD, A. (2022). Domain-specific analysis of mobile app reviews using keyword-assisted topic models. In *Proceedings of* the 44th International Conference on Software Engineering, 762–773. 38
- VAN ALTEN, D.C., PHIELIX, C., JANSSEN, J. & KESTER, L. (2020). Self-regulated learning support in flipped learning videos enhances learning outcomes. Computers & Education, 158, 104000. 32
- VAN EMDEN, J. & BECKER, L. (2017). Presentation skills for students. Bloomsbury Publishing. 103
- VASANTHAKUMARI, S. (2019). Soft skills and its application in work place. World Journal of Advanced Research and Reviews, **3**, 066–072. 102
- VASWANI, A., SHAZEER, N., PARMAR, N., USZKOREIT, J., JONES, L., GOMEZ, A.N., KAISER, L. & POLOSUKHIN, I. (2017). Attention is all you need. Advances in neural information processing systems, **30**. 122

- VILLAGRÁN, I., HERNÁNDEZ, R., FUENTES, J., TORRES, G., SILVA, D., ARAYA, N., DELGADO, M., MIRANDA, C., NEYEM, A., VARAS, J. et al. (2022). What kind and how many?: Exploring feedback in remote training of procedural skills in physiotherapy. In European Conference on Technology Enhanced Learning, 631–637, Springer. 22, 34
- VILLALOBOS, E., PÉREZ-SANAGUSTIN, M., SANZA, C., TRICOT, A. & BROISIN, J. (2022). Supporting self-regulated learning in bl: Exploring learners' tactics and strategies. In *European Conference on Technology Enhanced Learning*, 407–420, Springer. 34
- WALLE-HANSEN, M., RANHOFF, A.H., MELLINGSÆTER, M., WANG-HANSEN, M.S. & MYRSTAD, M. (2021). Health-related quality of life, functional decline, and long-term mortality in older patients following hospitalisation due to covid-19. BMC geriatrics, 21, 1–10. 107
- XUE, Y., SONG, Y., LI, C., CHIANG, A.T. & NING, X. (2017). Automatic video annotation system for archival sports video. In 2017 IEEE Winter Applications of Computer Vision Workshops (WACVW), 23–28, IEEE. 38
- YACOBSON, E. & ALEXANDRON, G. (2023). How do teachers search for learning resources? a mixed method field study. In *European Conference on Technology Enhanced Learning*, 489–503, Springer. 22, 33
- YAROWSKY, D. (1992). Word-sense disambiguation using statistical models of roget's categories trained on large corpora. In COLING 1992 Volume 2: The 14th International Conference on Computational Linguistics. 59
- YOON, M., LEE, J. & JO, I.H. (2021). Video learning analytics: Investigating behavioral patterns and learner clusters in video-based online learning. *The Internet and Higher Education*, **50**, 100806. 36
- YUAN, L., TAY, F.E., LI, P., ZHOU, L. & FENG, J. (2019). Cycle-sum: Cycle-consistent adversarial lstm networks for unsupervised video summarization. In *Proceedings of the AAAI Conference on Artificial Intelligence*, vol. 33, 9143–9150. 38

ZAPALSKA, A.M. & DABB, H. (2002). Learning styles. Journal of Teaching in International Business, 13, 77–97. 85

Appendix A

Algorithm 1 tracing (Initial Segments)



	Wo-G3UQ-3U2 00.23 00.42								
$\label{eq:constraints} \begin{split} & textCounter + 1 \ /* \ Keep \ adding \ morv \ text \ line \ to \ find \ env \ of \ the \ punctual \ textCounter \ + 0 \ env \ but \ k \ but \ but$			hile textCounter + θ < length(Vi) and textCounter < length(Vi) do /* length (Vi): returns the number of text lines in the video transcript. The loop should stop if the number of the remaining text lines in the The loop should stop if the number of the remaining text lines in the	transcript + $\theta \ge tength V_{V_i}$. The loop also should stop if there are no remaining lines where textCounter $\ge length \langle V_i \rangle_{\theta} /$ \rightarrow if textLines = f_i then	$segmentSlow Time=V_{k} [textCounter];$ $n = n + 1; segmentId=string(V_{ki} + n);$ $- \text{for } i = 0 \text{ to } \vartheta \text{ do}$	<pre>/ textbackappend(vi[textCounter]); textCounter = textCounter+1 /* Check the segment boundary */</pre>	If intersection proncutations. V. [instruct-ul], \$\ell\$ of them segmentBudTime=V, [instCounter-1], \$\ell\$ of them segmentBudTime=V, [instCounter-1], Lapend[segmentId, segmentStartTime, segmentBudTime, instimas]); instimations = [] /* To start new video segment */	textLine append(V ₁ [textCounter]); textLine append(V ₁ [textCounter]); textCounter = textCounter+1 /* Keep adding more text line to find may diffe punctuationList */ \vec{f} textCounter + $\theta \geq length(V_1)$ then \vec{f} textCounter to a new segment to include the tremaining text lines or	add the remaining text lines to the current segment */ fit textlines =// then segmentStartTime=// [textCounter]; n = n + 1; segmentLesting(V _n +n); for i = textCounter - 1 to length (V;) do textLines.append(V[i]); textCounter = textCounter +1;
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	0.42 [[0 00, 0.13] [' My did Waa Jaways the rock of the family.''So seeing tim ruggle to brenthe, it was really heart-reading', it wad an waa aways so are really heart-reading', it wad an waa hava so are really heart-reading', it wad a waa baways so are really heart and are and an are to a route a a draw.'' Net waars allow are to broate a rulew.'' Net ways and the to be thoreant a titte more asaly.'' It ewas not the couch, lei and a the more asaly.'' It ewas not the couch. Heat and the more asaly.'' It was not the broat and and orgen mark.'' the was not the couch. Heat and the things'.' It really was a doubt to bo those things'.' It really was a doubt of the was something that to be done.'' He was in and out of bouil heaptabla a couple of times.'1]		т в Ф	2-2U3	· • • •	11 12		9 LL 1	
				0.45 3 mWo-G3UC					
[* He started to have less energy. He would 11 lose his breach hilder more ensity.", He mass." The score, the hard the oxygen mass." The warn table to do those mass." The warn table to do those do do so gring downill serve quickly, and we knew something had to be done."] [* He started to have less energy. He would 12 [* He started to have less energy. He would we knew something had to be done."] [* He started to have less energy. He would we have something that do be done."] [* He started to have less here do those mass.", He want se host do those things.", It even have a short to us.", My was in and out of local hosp table a couple was in and out of local hosp table a couple of time."]	re is a full stop at Vt[textCounter-1] > , He was in and out of lo optials a couple of times'		textCounter=12, and 12+6< 39 (length[Vt])		("Nobody was really able to accurately 13 diagnose him."]	[Nobody was really able to accurately 14 diagnose him.', I think that was probably the scarlest thing is not knowing what it	web, 1 Wobody was really able to accurately 15 diagnose him. "I think that was probably the scarlest thing is not knowing what it was." We dian't know what was going to homoon."	Thopsen's are really able to accurately 16 (hotobody, war really able to accurately 16 and accurately in the tark was possibly the scartistring is not knowing what it as a scartistring is not knowing what is a scartistring is not knowing what was poing to make it?]	[Yobody was really able to accurately 17 groups elimi, 17(th that was probably the scantes/thing is non knowing what it was. ; We didn't know what was going to a specify was a species of the species of the going to make it ??, We knew we reeded a place that had more experience ']
4 10	÷ 2		m		0	1	7	m	4
1.00	1.1.	1.19 S.	place. 1.23 he	1.28 ike	ire. 1.33	l 1.37 f what	1.40	or 1.45	1.49 im.
So they recommended the Temple Lung Center. And that was the best decision we ever made.	Immediately voi could refer a affective at range. From the doctors to the testing that was done there.	To even just sitting in the waiting room with the other lung patients	We just knew he was in the right He was going to get the best care could.	So I think at Temple Lung Center they just tend to see more cases I	his. Things that are a little bit more ra So they seem to have more knowledge.	Just right off the bat they seemed to have a better understanding of	was going on with him.	They really came up with a plan f him, how they could make him better,	how they could treat him, and if, like they found the right mix for h

He started to have less energy. He would loss its breach a little note easily. He would not the couch He had the covgen mask. He wast table to do have a ping, it really was a shock to its. Wo did was gring down little y quickly and we know something had to be done le wast and something had to be done le wast and out of local hospitals a cupied frimes.





And that was the best decision we ever made, immediately to could feel a difference at female. From the doctors to the testing that was done there. We just knew he was in the fight jatch He was going to get the best care he could so timits at female lung center they just feed to see more cases like his.



1.53 01:33

Things that are a little bit more rare. So they searn to have more involvedge usit efficient the start into serve involvedge usit efficient in understanding of what was going on with him. They really came up with a part for how they could rate thim and left like they found the right min, and found the right min. You know, it was a combination of rougen were provided the right mean.



Appendix B

Algorithm 2 tracing (Segment Characterisation)



Scannic Togging stay of VISC-L to get the Pineburnet have Scannic Togging stay of VISC-L to get the Pineburnet have a circle segment-tak of Four Torios (FT) and Four Consepter of $(EC) \rightarrow i$ and Four Consepter of the transformer for Topic chastlere app of VISC-L to get the accord domatrication tak of four Pine (TT) = 1, $Pine (TT) = 1$, $Pine (TT) = 1$, $Pine (TT) = 1$, Pine (TT) = 1, $Pine (TT) = 1$, $Pine (TT) = 1$, $Pine (TT) = 1$, Pine (TT) = 1, $Pine (TT) = 1$, $Pine (TT) = 1$, $Pine (TT) = 1$, Pine (TT) = 1, $Pine (TT) = 1$, $Pine (TT) = 1$, $Pine (TT) = 1$, Pine (TT) = 1, $Pine (TT) = 1$, $Pine (TT) = 1$, $Pine (TT) = 1$, $Pine (TT) = 1$, Pine (TT) = 1, $Pine (TT) = 1$, $Pine (TT) = 1$, $Pine (TT) = 1$, $Pine (TT) = 1$, Pine (TT) = 1, $Pine (TT) = 1$, $Pine (TT) = 1$, $Pine (TT) = 1$, $Pine (TT) = 1$, Pine (TT) = 1, $Pine (TT) = 1$, $Pine (TT)$	Assumint: Signation that the second	Assumatic Togener of the start of the function of the start of comparison of the start of the transmitter togener start of the start o	while $i < length(L)$ do
exholding a ways the color 0.3.1, by Me at ways anyor the color of the family. So selectly family benefitied with the select of the family. So selectly family benefitied with the select of the family. So selectly family benefitied with the select of the family so select of the term with the ways always very active. A family select of the family so select of the method of the family sole of the select of the method family select of the family sole of the ways and select or the family sole of the ways and select or the select of the family sole of the family sole of the select of the family sole family select of the family sole of the ways and select or the family sole of the select model and to be done. He ways in and out something that to be done, the way in and out sole of the family sole of the family sole of the select model with the select of the family sole of the select of the family sole of the select of the select of the select of the select of what it ways we define the or ways are been been what it ways we define the or ways are been been what it ways we define the or ways are been been been what it ways we define the select of the what it ways we define the way are been been been what it ways we define the select of the precommended the family we produce the family in the select of the select of the select of the precommended the family we produce the family in the precommended the family we produce the family in the precommended the family we produce the family in the select of the select of the select of the select of the select of the precommended the family we produce the family we produce the family in the select of the s	[mWo GLUQ; 2010, 0.10, 0.19, [Wy dat was " anotyce the cost clubbart." Another another service with an unward the location. The way clubbart. Independent, the folia dot with us, joint data independent, and independent with the second independent in the folia dot with us, clubbart. "The way rule is coor force mbar, it really would not be keen in a litter more cashy." In would is a block in a litter more cashy." We don'th a coor force mbar, it really we don't with data it would will do a segrer would not block in the folia dot with a counted with it coor force mbar." Use second will do a the gory of the notification of the second watt it way will dot it the word theorem watt it way will dot it the word theorem watt it way will dot it the word theorem watt it way we dot it the word theorem watt it way we dot it the word theorem watt it way we dot it theorem with the word word to be the latting to be forcer the precommended the Tampie under care for the precommended the Tampie under care word with the prove while the and theorem with the and theorem watt way we dot it theorem with the and the precommended the Tampie under care for the precommended the Tampie under care word with the prevention of the care word the second word the matt way we dot with the second word with the prevention of the care word with the word word the the second word with the word word word word with the second word with the word word word wo	expedition [into-ed.U.q.1.0.00.0.13.0 [hyt dat was control = strong for to exit of himly. So neglity him strongle to home, h. we make himly. So neglity for the strong for the himly. So neglity and independent he himly and way vary parter and the strong strong strong strong strong into any strong strong strong strong into any strong strong strong strong strong strong strong strong strong into any strong strong strong strong downling strong strong strong downling strong strong strong strong strong strong strong downling strong strong strong str	X
mpel Tresonationschafter (Personationschafter (Personationschafter) [5]	(105) [Physical H can th) 1]	(UK) mair: (aler) f: Ysocalfeatenning) f: Socalfeatenning f: Socalfeat	(IVS)
The state was the brack owner previous data was the brack owner previous and a mudic immediately you could field fill the state of the statement of the stateme	Thing: that are a fittle bit more are: I Ending: they are a fittle bit more are: I Subsequent private privation I Subsequent private I	Another pails the walk would get ["Percenvirul and contingent of the set of weat would get ["Percenvirul and the set of ender and the set of ender and the set of end the set of ender the set of ender the set of end the set of ender the set of ender the set of end the set of ender the set of end the set of ender the s	I < length
a mweetuq:u1 103 1126	4 mwoe3uq4U3 133	2 2 3 5 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	υ
1.28 And the how of the back advances in the detail of the event much is minimal advances of the detail of the event we part for and the detail of the event we part to the detail of the event of the part of the back and the part of the back and the order of the part	133 Things that are a little bit more are solver, sum to have more boowledge, suit shift affer that they served to have a setter they served to have a setter understanding of what was gear with a plant for this how they with a plant for this how they could meak him fert to have four boow it van a combaction of modelse, pulmonery therapy, organ.	2.22 And they really here what would be a waited to where a wait to the and the down price and series and series price and series and se	
mwo.GIUQ.14 1.03	mikeGBUQ.2US 1.33	MIN-GBUQAUG 1.58	



Appendix C

Algorithm 3 tracing (Segment Aggregation)






Appendix D

Derivative Algorithm tracing

| Derivative Algorithm Trading. None: we have added the duration to the segment characterisation to be added as part of
the video caracterism any unrest advances any unrest advances to added as part of | 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 | Stars 1 & 4 Colors (Colors Stars 1 + Colors 2 + Colors | (40) 1 | $\begin{array}{c c c c c c c c c c c c c c c c c c c $ | torsa kotaran bernarda an antara bernarda an antara | · ten ten · ten ten <thten< <="" th=""><th>Image: Instantion of the second sec</th><th>Image: Second second</th><th>\u03e4 \u03e4 \</th><th>Discrete(1) I <thi< th=""> I <thi< th=""><th>1 1</th><th>$\frac{1}{n} = \frac{1}{n} \sum_{i=1}^{n} \frac{1}{n} \sum_{i=1$</th><th>matrix matrix matrix<</th><th>Mathematical Structure A structure s</th><th>Exercise 1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.</th><th>$\begin{array}{c c c c c c c c c c c c c c c c c c c$</th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th></thi<></thi<></th></thten<> | Image: Instantion of the second sec | Image: Second | \u03e4 \ | Discrete(1) I <thi< th=""> I <thi< th=""><th>1 1</th><th>$\frac{1}{n} = \frac{1}{n} \sum_{i=1}^{n} \frac{1}{n} \sum_{i=1$</th><th>matrix matrix matrix<</th><th>Mathematical Structure A structure s</th><th>Exercise 1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.</th><th>$\begin{array}{c c c c c c c c c c c c c c c c c c c$</th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th></thi<></thi<> | 1 | $\frac{1}{n} = \frac{1}{n} \sum_{i=1}^{n} \frac{1}{n} \sum_{i=1$ | matrix matrix< | Mathematical Structure A structure s | Exercise 1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1. | $\begin{array}{c c c c c c c c c c c c c c c c c c c $ | | | | | | | | | |
|--|--|---|--|--|--
--
--|---|--

--
--|--|---|--
--|--|---|---|---|---
---|---|---|--|--|
| | Topic
Environment | Prysk izal Hwathh
Psychol of a del-wathh
Psychol of cart wathh
Divention of cart wathh | Environment
Provinsionent
Environment
Lavel270 referenderen
Lavel270 referenderen | acrobineden fildeeu J | Level201 répresendence
Provide des la contrevent
répresentations la pr | Personal/Ail usa Arcibieled's
Eval connenses

 | Privice landstreament
Privice and calification
Psychol and calification | Psychologi calrectift
Crist recrearce
Presonal una fordibilitif
Presonal result
Presonal result | | Soci allelation in fro
Environment
Economisticon fro
Personal Autor in Calibert
Level27 informations
 | Lowid2) released
Evaluation
Evaluation
Lowid2/released
Evaluation
Evaluation
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provident
Provi | LavelOf networkeroaeroae
Frysk sai health
Phydrod oal califeadh
Phydrod oal califeadh
 | A construction of the cons | Presonal kut lan Andbalach | End rowneed
LawShin Algeoreaderoo
Social Algebratron Hip
LawShin Ndependeroo
Physical Headth | According to the second | calityre(dor new | wour Physical Health
Interaction
Leaded in Observatives
Leader prior Physical Health
Resider prior on
Physical Health
Physical Health | krativesth
Prvi krativesth
Eval roverst
Leadoff ridspendense
Prva kativesth
Leadoff ridspendense
Prva kativesth | Environment
Environment
Pressonatival uss herobleder's
Social Hellenth
Press'stal Hellenth
Press'stal Hellenth | catalytican mu www.mu wather.
Physical headen
Physical headen
visur Physical headen
catalytication
catalytication ha
 | Lower Inspection
Psychological isolah
Psychological isolah
Psychological isolah
Psychological isolah
Psychological isolah
Psychological isolah | Pavota di cale calh
Presonalizi una molleder
Pavota del cale calh
Pavota del cale calh
Pavota del cale calh
Socialitado ni fic | Psychol og cal-teath
Environment
Laveloff referendense
Laveloff referendense
Provisel freath | Level2r reproduction
Psychol (at Arteal)
Psychol (at Arteal)
Psychol (at Arteal)
Psychol (at Arteal)
Environment |
| Concept Path | Parent | a de la constante d | | | Physical Symptoms |

 | | PPos tai Symptoms | |
 | Precision of the constant of t | Medical SefCare
Behavior
Regaby Veteri
Regaby Veterior
Dock traase | reservences
reservences
Reserved reservences
Barboring
Santoshing
Cohedormetorens
Reserved housed
Reserved housed
 | Pred allon n | Ac quirty of connection
Ac quirty of connection
Personabilistic contep
Dati M, M-VIE
Personabilistic contep | Ped Bion
Ped Bion
Pation
Ped Bion
Ped Bion
Ped Bion
Ped Activity
Treatment
Treatment
Cognitivity
Ped Activity
Treatment | Propriedon Articles
Propriedon Processing
Programmer Processing
Scalade UN
Programmer Conference
Programmer Programmer
Programmer Programmer
Development | Dist Beth
Phyliol Environment
Phyliol Environment
Pain
Colombingtones
Bethal our | Perginatory I hered Anglocom Physics
Economic Status
Langel scale
Frances
Privated is un
 | Periodia bation
Sofradity
Periodia dati conhip
Energy
Negati vefetal ng | Francesmetacora
Francesal sour
Energi
WorkCapaday
WorkCapaday
Beth
DaliMuking
Personalibili di comhip | Logial soue
Negati soue
Post Inder certer
Post Inder certer
Presentatival contrito
Negativarites fre
Perhedal taul co | Neigebavefreel ng
Red gan -
Post Usef reefre
Neigebavefreel ng
Post Usef reefre | Negativerfeed ng
Economic Status
Lagati saure
Medical Sefficine
Peac | Lagual suae
Pregativerificed rig
Pred ap.n.
Social Arcti ur
Cognishi verkunction
Perhabiti tuzi on |
| | Connept
Accumento | Behaviour
Biodyrnege
Coendi wifurction
Dahitu vig | Economidiatus
Erente
Frenci alti aue
Liegad sue | Medicalisel ficare | Model In
Present of the Int
Other Symptoms
Pean
Presential distorting | Physical Code of Trace C

 | Physic allowators
Postové eeleg | II draudal baadoon
Peril atta di atta | | Sociality
SocialZine
Spitruality
Treatment
 | WorldSpeedury
AccessfordSpeedpace
AccessfordShift
AccessfordShift
AccessfordShift
Active | A different ing Amoor A firtuarity
A too for
A moor
A moor
A moor | Agreenteenoon
Agreenteenoon
Motree
Babrierg
Bo welt suues
Bo welt suues
Bo welt suues
Bo welt suues
Bo on or hol
 | Buddisim | C. aret of formation
C. aret formation
C. aregiven sols tance
C. aregiven
O Ma | Christen
Chrostean
Chrostean
Chrostean
Chroste
Community
Common And Lation | C could
C could
C could
C could
C could fine the
D could fine the
D countries
D countries | Disestion
Dist
Distelencess
Disconfort
Drowsfress
Drow | Dyspress
Education
Enduration
Enduration
Enduration | Esercie
Estreristry
Farity
Farity
Foor | read
France
Prancial
Prantike
Prantike
Prantike
Prantike
Prantike
 | P surveyal Han
Curicoun
Galaci
Galaci
Galaci
Gara | HeadBroken
Hendulen
Hopelensnes
Humbrous
Humbrous | Hypochondvise
Income
Inchetizere
Indexis
Incomeria | I reazeros
I retadality
Language Officulty
Learning
Learning |
| | Focu Topic | and the second se | | | |

 | | | |
 | |
 | | | | | | | | |
 | | | | | | | | |
| | feoar Topic Fooar Concept
La Nathebiopendance | | Laved Divideoendary e | | |

 | | | |
 | |
 | | | | Lavel2/holiperderce | | | | |
 | | | | End to resure | Psychol og ask eadh. | So dial Relations hip | | |
| | fo au Conspt.
Treatment | ł | db diefs
1th
Treatment | | |

 | | 5 5 | | 5
 | 0.00 | ace
 | 5 | | | Treatment | | | | |
 | | | 5 | Training | Therefore | ton
Interley | | |
| Input: LAgg | Ro cur Topic
So cùr Reekcordry | Endro meet | Pres cerafitabasian
Pres cerafitabasian
Pres calerceatin | | Physical-Health
Physical-Health |

 | So dal Peterbordhy | Pay holo ge al ted | | Payr holo git all teal
 | Level Of religions | LevelOff depend
 | Paycholo gcalited | | Physical teach | Phys all teach | | | | |
 | So dal Peterborning | Enviro meerit.
So dal Relationship | Psycholo (Kahted | Physical Health
Physical Health | Physic calification
So call Relationershy | So dal Relationship
Psychologic altred
LevelOffvologication | | Physicality |
| | focur Compt | Training | ce
Terepie
Nope
LuveGeneer | | h
Tunor
LungCaroar |

 | xe
Caregoer | Thinking.
Hope | | xe fear
Prazroria
 | Transment | Tratriet
 | Thinking
Tumor
Bioletis | 9 | Proven Skeep | Breedth | že
 | Bell effs
Child | xe Rehabilitation
h | The Mrg. | turo
turganor
xe | LangCarroor
Greedsh M | xe Rierd
Nartaheabh
Traitreit | | xe
h Breegh
xe
xe |
| | focurtopic
Pychologicaheaf | Physical-test th
SocialR of adorading
Physical-test th
Physical-test th | Physical House (1)
Level C4 relignerate
Social R d aboreho
Social R d aboreho
Social R d aboreho | Physic all eal th | Physic advect th
Soc table of Arrando
Physic Model of Arrando
Soc table of a Archeveryo
Soc table of a Archeveryo | Soc tailit d autocrafip
Soc tailit d autocrafip

 | Investor entreation
Investor Cathologenetic
Physic alife edits | SocialR distorthy
PhysicalHeal th
PhysicalHeal th
PhysicalHeal th
PhysicalHeal th
PhysicalHeal th | | Sociality of actionary
Lovel Chrodopander
Sociality of actornary
Physicial Head Un
Physicial Head Un
 | Environment
PhysicalHealth
PhysicalHealth
RysicalHealth
RysicalHealth
PhysicalHealth | SocialN distorting
Physicial Health
Physicial Health
SocialN distorting
 | Sociality de aboretion
Socialité de aboretion
Prysicial Healt
Prysicial Health
Prysicial Health
Prysicial Health | Level Ofhidsparder | Physic alfred th
Prychologic alfreed th
SocialR 4 actionnhy
Physic alfred th
SocialR 4 actionnhy
SocialR 4 actionnhy
SocialR 4 actionnhy
SocialR 4 actionnhy | Physical Health
Physical Health
Physical Health
Physical Health
Physical Health
Physical Health
Physical Health
Physical Health | Sociality advection
Sociality advection
Physicial end th
Physicial end th
Physicial end th
Physicial end th
Physicial end th
 | Socialit d aborehip
Rev cont Valassento
Revica atteat (1)
Revica atteat (1)
Revica atteat (1)
Revica atteat (1) | Physical-test th
Level CPhysicpartier
Physical-test th
Physical-test th
Physical-test th | Physical Health
Physical Health
Rhysical Health
Physical Health | Physicales of the
Physical of a
Social flat a blocrahip
Social flat a blocrahip
Physical of the
Level Official perder | Preyoc advectory
Social of abooration
Social of abooration
Preyocial reador
Environment
Social of abooration
Social of abooration
Social of abooration | SocialR d aborration
Level CPMAbpender
Diversersers
SocialR d aborration
RhysicalHeal (1)
PhysicalHeal (1) | Brwinownert
Socialiti of attorning
Socialiti of attorning
Physicalitical th
Physicalitical th | Soc LaR 4 actorning
R ych chogyc adhrand
R ych chogyc adhrand
Lovel Christigeneter
Lovel Christigeneter
Lovel Christigeneter |
| | Dundion Focusion
0.4 40 Thinkie | 124 41 Lung aree, Pan
223 06 Man
225 51 Lung aree, treng
243 159 Lung aree, treng | 4 20 56 Linguese
5 43 56 Linguese
5 41 156 156 104
2 116 136 04d
1 23 82 04d
1 23 82 04d | 314 80 Preumonia
4.99 82 LungGarcer | 649 128 Jaroc
228 36 We
847 37 Yufang
301 82 We Gomeanty
047 35 fixed | 1123 25 Caragher
2180 24 Caragher
2180 24 Caragher

 | 10.00 Strategy | 511 26 Community 113 75 Skeep, Rkn 3.6 116 Pain Skeep 3.16 98 / Pain Pain 6.37 44 Skeep 2.19 138 LungGancer | | 112 55 Mars/Fanity
312 118 Treamert
312 35 56 fold
042 42 ShortnessOftweath
115 49 Copd
 | 2 15 215 217 Exercise
3 19 16 Copid
0 52 47 Protroeo20Reath
3 19 50 Longbrease
3 21 50 Longbrease
3 21 50 Longbrease
4 22 66 Longbrease | 443 13 Renhy
1.29 97 Lunglancer
2.49 30 Hope
0.49 48 Friend
 | 2.17 Se rotaney around
4.23 2.2 Caregove
1.25 0.0 Provencia, Ret
2.28 50 LongGrave
6.14 100 Mediation
6.14 20 Mediation
6.14 20 LongCare | 1.55 D5 Pratment | 0% 41 Couph
1.3 & Thrivity
1.3 % Thrivity
301 % LungLinee
321 #7 Hubbard
122 a0 Entriby | 407 26 Cord
407 26 Cord
223 81 Enrormona
323 81 Enrormona
323 81 Romotin
422 55 Community Farthy
422 55 Community Farthy
422 55 Community Farthy
422 105 Cord | 219 38 Holland
427 87 Proceedia
519 88 Proceedia
518 89 Proceedia
266 89 Proceedia
181 80 Couph
181 82 Proceedia
 | 256 40 Fanly
0.5 41 Christian
2.29 8 Minosuphystersion
2.41 5 Rect
3.41 5 Rect
3.41 5 Rect
3.41 5 Rect
3.41 5 Rect | 2.115 56 Drug
8.39 41 Treatment
4.00 25 Houtinent
2.01 123 Vaping
2.47 41 Drug
2.47 41 Drug | 5.07 216 Device
5.07 216 Device
2.234 152 Copd
1.16 46 Family
2.31 231 Copd
2.31 231 Copd | 110 31 Long
200 75 Famby Prince
4.1 50 Mont, Friend
4.1 50 Mont, Friend
8.8 Mr/mont/Hypettersion
1.83 61 Rest, Friendersion
3.5 53 Treatment | 0.07 N. Uughoesee
2.54 A. Ruchy, Old
2.121 81 Fanhy,
2.08 45 Stroking, Cough
3.54 45 Dad
1.126 35 Fanhy, Princi
1.126 35 Fanhy, Princi | 5 01 24 Friend
214 126 EndOuffecture
334 45 Endouffecture
2.27 147 Ond
045 45 Parto
129 45 Abroho Det | 2.14 del Training
047 ez Hosband
4.17 sel Wife
0.05 50 Part Paug
201 36 LungGarcor | 2.4 36 W/fe
3.94 50 Treatment
5.37 31 Connertation
0.88 89 Pain
2.29 102 Treatment
2.20 102 Treatment |
| | niaue VoicelD Sant End
0 Nonre(Long)51 0 | Marrel LongTea Marrel L | | 11 h02640826400 1.54
12 h02640826405 8.17 | 11 hogpsegmeiler 4.41 13 hogpsegmeiler 6.22 14 hogpsegmeiler 6.22 15 hogpsegmeiler 6.22 16 hogpsegmeiler 6.22 17 hogpsegmeiler 10.12 | 18 h000x000x4449 10.48 1
19 h000x000x4449 10.48 1
19 h000x000x4446 | 21 h005s02sa4464 14.01 | 22 10236-03354667 14-45 - 28 372864 ntubil 0 0 28 372864 ntubil 1 54 28 372864 ntubil 1 54 37 28 372864 ntubil 1 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 | | 29 wwwiej Bur/160.0 0.17
30 wwwej Bur/160.6 1.14
31 wwwej Bur/160.13 3.14
32 ub/96.M.2011 0.0
33 ub/96.M.2013 0.46 | 34 uo Yea M. 2015 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. | 40 PERFONDUCII 4.25
41 0682769561 0.02
42 0560269561 2.19
43 056027654610 2.19
44 01_019776961 2.19 | 40 00 (01999/0002) 2.19
47 00 (01999/00021) 4.01
48 04 (01999/00021) 4.01
49 049/000210 4.25
50 049/000210 4.25
51 049/000210 4.55
51 049/00000000000000000000000000000000000 | 0 IVM06446A946 88 | 54 weV984e_CLAI 0.13 56 weV984e_CLAS 0.55 56 weV984e_CLAS 0.55 57 weV984e_CLAS 1.28 56 weV984e_CLAS 2.28 57 weV984e_CLAS 2.28 56 weV984e_CLAS 2.28 57 weV98e_CLAS 2.28 58 weV98e_CLAS 2.28 50 weV98e_CLAS 2.28 50 weV98e_CLAS 2.28 | 0 (1992)
0 (199 | | 75 GUTOrgifici and
77 e00Yea, 771 0.00
78 e00Yea, 771 0.00
78 e00Yea, 7710 2.11
79 aver/20043 0.00
80 aver/20043 0.43 | 01 alWeFV 200155 119
02 alWeFV 200155 119
03 alWeFV 2001512 2.58
04 relationcommun. 2 45
05 relationcommun. 2 0
05 relationcommun. 2 00
05 relationcom | 61 04184-0011-65 131
68 Nerds_c_geno1 0.02
69 Nerds_c_geno1 2.20
60 Nerds_c_geno10 2.30
91 041000Mpc1 0 | 86 (phrosonyles 1.13
96 (phrosonyles 1.13
96 (phrosonyles 1.13
96 (phrosonyles 1.13
97 (phrosonyles 1.13
97 (phrosonyles 1.13
97 (phrosonyles 1.13
98 (phrosonyles 1.13
99 (phrosonyles 1.13
90 (phrosonyles 1.13
91 (phrosonyles 1.13
92 (phrosonyles 1.13
93 (phrosonyles 1.13
94 (phrosonyles 1.13
95 (phrosonyles 1.13
96 (phrosonyles 1.13
97 (phrosonyles 1.13
97 (phrosonyles 1.13
98 (phrosonyles 1.13
99 (phrosonyles 1.13
99 (phrosonyles 1.13
90 (phrosonyles 1.13
91 (phrosonyles 1.13)
91 (phrosonyles 1.13) | 00 30-set Servicus 0.0 010 30-set Servicus 0.0 010 30-set Servicus 2.2 010 30-set Servicus 2.2 010 0-set Servicus 2.3 010 0-set Servicus 2.5 010 0-set Servicus 2.5 010 0-set Servicus 4.15 010 0-set Servicus 0.41 010 0-set Servicus 0.41 | 108 CAN(5),2508.3 6,39
107 AMM29507W81 0.08
108 AMM29507W85 0.01
108 AMM29507W85 0.0
119 E89453.4564 0.0
111 E89453.45646 0.0 | 112 (594 SA.4°64/7 1.3
113 (194 Set SA.4°64/7 1.3
115 (194 Set SA.4°64/7 1.2
115 (195 Set | 117 Weil Concessor 2.04
119 Weil Concossor 2.04
119 Weil Concossor 2.04
120 Weil Concossor 2.05
121 Zusteppacka 1.04
122 Zusteppacka 0.47 |

Anticipation</

Appendix E

Super Ordinate Algorithm tracing

| Super Colored Ryport Your of the address caloried regret deviced as perforeation aperturbation of perforeation aperturbation of the advector o | I k u uk u uzenta uzenta uzenta OdgutStyper biblio data kata kitaki k | 0 for KUI 4(10 and 11 and 12 a | s again to the state of the sta | Dear 1 = 4 ± 0.4 \pm 0 | a constraint and the constraint of the providence of the providenc | Deepy Father and American Am
American American A | attraction of the state of the | Profit Decrementation of the second s | | Control
Data () Particle Advectored advector | Deep. J Reenth. 2014 01 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 | And Alianti and | ters a second to the second seco | 1000 km | e transmission of the state of | | ρ ² (We που που που που ματική τη ποι ματική τη ματική | 1 a white (2,11,2) and (2,11,1) | 6 Remain and a second s | Streption Num. Const. Name Num. Nam Num. Name Num. Name | D That the second secon | (Pergy Payor)
(Pergy Payor)
Analysi 1 | E. E | Devised Network Network (Network Of Statement Of Statemen | In a state of the | N and Microsoft (Light de
Instance and | Brendi Dina Antonio anto anto anto anto anto anto anto ant | 2.4.179mm4 | 26.44. (More and American Americ
American American Amer
American American A | Development of the second seco | (orrentiac) Simplify (10) (10) (10) (10) (10) (10) (10) (10) | Protoments,
protoments,
proverset,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protoments,
protom | 1 (Microsoft Microsoft Mic | Units Endowing and a second se | [Pergs] [Pergs] <t< th=""><th>00000000000000000000000000000000000000</th><th>to defined and the effect of the effect of and and and the set of the effect of and another of the effect of and another of the effect of another</th><th>0 1000000000 100000 100000000</th><th>model material m</th><th>Natural Manual Man
Manual Manual Manu</th><th>Development Development Developme</th><th>0.0404b [10] [10] [10] [10] [10] [10] [10] [10]</th><th>Instruction Instruction Instruction Instruction</th><th>Investory (Investory) (Investo</th><th></th><th></th></t<> | 00000000000000000000000000000000000000 | to defined and the effect of the effect of and and and the set of the effect of and another of the effect of and another of the effect of another | 0 1000000000 100000 100000000 | model material m | Natural Manual Man
Manual Manual Manu | Development Developme | 0.0404b [10] [10] [10] [10] [10] [10] [10] [10] | Instruction Instruction Instruction Instruction | Investory (Investory) (Investo | | |
--	---	--	--
--|--|---|---|---
--|--|---|--|---
--|--|--|--|--|--|--
---|---|--|------------|--|--|--
--|--|--
--
--|--|---|---|---|---
---|--|---|--
--|--|
| | | Notic
Environment 0 | Physical teath | Prychological trait in the second sec | Physic alterath Loop
Environment
 | Prysical eath 92 | Invicoment Loop
Lavid Oholoperclarco | Laved Officialization 26 | 1000 |
 | Lavel OProlependence 193
Lavel OProlependence | | Pryskiał esth | Previously values Andreaders
Brain comment | dom | Prysic alterath
Prysic alterath | (log
 | Ps whological Haith.
Environment | Resonal Valuesh ndbelefs | Provide diet exacts | Tipob | Provide det cuelto.
Provide det cuelto. | SocialPlicationship 6 | Britom et
 | Soci alteriation drup
Resound Valvasch nelletier's
Land Phonoscoletier | Lavel Of volgerolarios | Binitions est | 1000 | Land O'holoperderoe
Breitore et | 11 | scens Physical each |
 | Level Of volgeerdance
Resid area to | | R whological each
 | | RyrbologisHeath
RyrbologisHeath
Stores Ryscateath
SyrbologisHeath
 | Rysic all early
Lave Ohodpendence | 8 | Renord VA usA rollielefs
score: Physic alteath
score: Physic alteath | Physic alt-cath
Personal Val useA individers
Environment
Level Otvolgenederco | SocialPetatorenia Loop
Thusid Ohrobenolence
Physicatrealin
SocialPetatorenia | Provod VM und v Milleters | Renoral VA wash millineler's
 | Revision of
Devision of
Devision of
Devision of
Devision of Devision
Devision of Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision
Devision | COTT IN THE ACTION AND AND AND AND AND AND AND AND AND AN | | | |
| Concept Path | | Parent | |
 | Dehaviour | | | |
 | | | | Physical Symptoms
Physical Symptoms | |
 | | | | | |
 | Physic loads yn pricen s | | | | | Physical Network and | | WerkDeparts
SocialZare | have can be do a la
 | Pain Prain Phasic dSarra | | Medical SelfCare
Behaviour | | Negali wrfed reg

 | | Negati wifed ing
Booly mage
OtherSymptoms Frysk dSym
Negati wifed ing | Respiration Biress
Dates Liverg | | Spirtual ky
OtherSyngtons Prostores Provides yn | Pergiratory Bress
Religion
As qui ringi rformation
Legal suue | Pressonabledationedato
Dada U váng
Prespiratory Biness
Prvia nevalitedationedato
 | nekon
Pan
Pan | Religion | Physicitocon eff
(Physicitocon eff
Toxobol
Physicitocon
Registration
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon
Physicitocon | Pano
Pano
Contropations Physicalism
Contropations Physicalism
Contropations
Contropations
Contropations |
| | | Concept
As guiring relormation | Dehaviour | Cognitive unction
Delivatives
 | Diet
Fconomic Status | Units | Financialiss up | Logisti soure |
 | | Med affetCare
Models | A may be a set of the | OtherSymptoms
Pain | Physical Creation and | | Physical Symptoms | The manufacture of the second
 | Rehabilitati an | Pedigion | Peopleatory III ness | | Peoplestary II nas Symptoms
Peop | 5 odaWc Ev Nr | Sociatizare
 | Spirituality | WorkGook by | An constitution reprofession | | Ac consider itsy
Ac consider ty/Official | Shaki Ye | Accedition |
 | Addressing/Minor Min erts
Microsoft | | heger
 | | ira kity
Appearance
Appetitat cas
Apprehensi an
 | Asthra.
Bathra | | Refer
Rowel kaues
Browth Mesones | Brondhik
Buddhian
Carehformation
Carehfun | Caregiver
Caregiver/s sistence
ChestPain
Chard | Creates
Creates | Overh
 | Comuce
Comuce
Comuce
Commission
Comparison
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count
Count | a monore
Dispertent
Dispertent
Dispertent
Dispertent
Crossfurce
Crossfurce
Crossfurce | | | | | | |
| | | Fours Topic Focul Consignt Focul Topic
Level3Redependence | |
 | Dreforment Onld Social/Mationality | | | Loss 12 through control for co |
 | | | | | |
 | | | | | |
 | | | | | | | | |
 | | | | |

 | | | | | | | | |
 | | e treitigendent e | | |
| | | Feature etc.
Treature etc. | |
 | Noti e | | Dedet | Testrat |
 | | | | | |
 | | | | 4 | 4 |
 | | | £ | | | | | |
 | | | | |

 | | | 4 | | | | | |
 | | Treatment | | |
| Input: LAgg | | t Foous Topic
SocialBieldti on hip | Erst rare are |
 | Physical table | | Personal Valuesko
Psychologica Haal | Physical each |
 | | | | Physical with | COMPANY STATE |
 | and more the production of | du san malautanon r | | Psychological and | Psychologica H eah |
 | | | PsytrologicaHeat | | | | | Level O'Volgerete |
 | | | L and O'ndepende | |

 | | | Psychological and | | | | Projected enter-
 | | Physical (with | | Socialitetati on hip
Ervi romenek
Socialitetati on hip |
| | | Fooul Topic
Psychological-Health Muni | Physical (cab) | Physical teach
 | LevelOf religendence Pain
Presidenth | Level 34 roli perdenze | Socia/PM attorn/rip Temple
Socia/PM attorn/rip Highe | Socialityl attorned to LunaCancer | a state of the sta | LOUGH LEVEL & LOUGH LEVEL
 | Physical watch
Device leaders | Contraction of the Contraction | Psychologicableath
Sociableathorder p Tumor | sociality attorney possible attorney possible attorney p | | PhysicaMeath
Level04 relignedence | rightener ment
 | Sociality attionable p
Physical Hostin | Physical each Think of | Physical Health Hope | | rthe kult of the start | Sociality4 Alticerary p | Level04 rokperdence Faur
 | SociaPM attornet/ p
PhysicaM earth Disconcession
Disconcession | Erekoneerk | PPost kalet ealth. | | Prociation for Treatment | Language Street Area | Physicial calls |
 | Sociality allocate p
Presidents Treatment | | Physical auth
 | | Psychological Health
Scotal Material For
Physical each
Social Pol Atlond For
 | Scotaling attorney p
Provision autor | | Physical teach
Pres control developed left
Physical teacher | Physical wath
Level 31 rationations
Physical wath
Psycrological Weath | SocialPatriceArp trings a way
Physicial auto-
Proceedia a second p
SocialPatriceArp | Free university | Erniktonnierk Streight
 | Рудования
Сокрания
Сокрания
Поликие
Поликие
Поликие
Сокрания
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие
Поликие | Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Processment
Proces |
| | | I Value Mancelo Barri Eral Duration Focul Corropt
D Manuri/Digg713 0 0.4 40 Forking | 1 Meerplay26 043 124 41 Longlaroot Pan | 3 Meeringagette 11, 12, 12, 12, 14, 12, 14, 14, 14, 14, 14, 14, 14, 14, 14, 14
 | 5 roustreption 335 4.01 36 Yearner
6 roustreption 4.01 4.59 56 languaren | 7 rQuyerq60017 5.01 5.43 42 lvavarce | 0 revo.60.02-01 0 2.16 136 0ad
9 revo.60.02-06 2.22 2.31 9 0ad | 10 POCEADZIMM 0.01 1.23 ED Careford | 11 hotocontraction 1 to 1 1 to 00 features
 | I LOOMANNAMENT OF ALL O | 12 PODGeogradits 3.17 4.19 88 Language
13 Poncentravits 4.44 6.40 10 Taxoo | 14 Indocementary was not the same | 10 holdsaccentration and 1 and | 10 housements total total and total and total 10 housements 10 housements 10 housements 11.0 h | | 20 h00546/204668 12.39 13.24 6 Turor
21 h0056/204686 13.35 14.02 27 Yearner | AL PROPRESSION | 23 hQG#QGI#M67 14.45 15.11 26 Community
24 hY3604 retuelt 0 1.13 78 8epp.Pain | 25 w189Aretues 1.54 3.5 1.6 Part Step
 | 20 v/864.rd.udd8 4.37 5.16 20 Pán | | 22 M780ALML023 5.53 6.17 44 3440
20 VM002M41E1 0.01 2.19 136 LMM24007 | 20 www.pw/bloot 0.17 1.12 % Mure, Family | 20 www.pic.000 1.14 5.12 1.26 Yearnerd | 31 www.mp2w/bloo13 3.14 3.5 8 Ch46
22 Concedenzor1 0 0.42 4 Second Streath
20 concentrons And 115 40 Second Streath | au ouoreneerse and
 | 36 united/02019 3.03 3.19 36 Codd | | 30 JISSPERULCI 0.05 0.52 47 Socirios S20wath
27 JISSPERULCI 0.56 0.55 1.46 50 Langliseare | adou no veo veo men antalasci o | 39 jasovina.co 3.36 4.32 46 lungolense | | 40 [105/4/11.021 4.25 4.43 15 firely
41 1/603140-041 0.02 1.30 97 Londoneer |
 | 42 U4613V40948 1.43 2.18 B Turror
 | | 43 1/4431/46/#610 2.19 2.49 20 Nope
44 01/3/1/4/M31 0.21 0.49 86 1/mod
55 01/3/1/4/M2 0.21 1.49 86 P.Atason/Brook
66 01/3/1/4/MT 2.19 1.17 28 Nukand | 47 00 DifferMB11 4.01 4.23 22 Carejuer
40 Hiffed2PDVI3 0.15 1.15 00 Proventia Post | | 40 H6F6479/17 1.18 2.33 S Lungaroor
50 H6F6479/13 4.25 6.14 10 Madation
51 H6F6479/125 6.15 6.14 25 Prantora 24
 | C2 HERVERYERD 7.31 7.51 20 Languagement
50 Januaryshi 2014 0.1145 5 Presarrent
56 www.Herc.CAN 0.13 0.54 41 Coult
56 www.Herc.CAN 0.55 1.3 15 Printing | 66 www.elev.c.Lob 1.14 2.28 N. Mure
57 www.elev.c.Lob 2.28 3.01 N. Uwetaneor
68 www.elev.c.Lob 3.24 3.51 at h.b.ben
68 www.elev.c.Lot 3.51 at h.b.ben
68 www.elev.c.Lot 3.51 at h.b.ben
69 www.elev.c.Lot 3.51 at h.b.ben
69 www.elev.c.Lot 3.51 at h.b.ben
60 www.elev.c.Lot 3.51 at h.b.ben
70 www.elev.c.Lot 3.51 at h.b.ben
70 www.elev.c.Lot 3.51 www.elev.c.Lot 3.51 at h.b.ben
70 www.elev.c.Lot | B removation on use he control vigitations II vigitations | 02 \\\000000000000000000000000000000000 | 0 0 0 0 0 0 0 0 0 0 0 0 | 0 монка (ут. 11 32.1
 32.1 32.1 32.1 33.1 |

NameNa

tyg hologic a health Nga calina th Pogc Poologi cal Head Pr whomself Tealing Fain Fear, Hope Tairing Tarak eg (койтейн кулу) Руудийн авт) Гуудийн авт) Physical each Socialized on Np Socialized on Np Psychological each Psychological each Physical each or the second second second with with Stergh) Stergh Mon Market Mark Lung Caroor Grandbrid Prinned Mertahlaabh Treatrained Therefore, Turner LungCaroner Montender
 Montender 84892

Appendix F

Correlative Algorithm tracing





| Locatherations:
Locatherations:
Participations:
Data and the second
Data and the second
Location data code on
Location data code on | Physical-Math
Physical-Macath
Physical-Macath
Physical-Macath
(Environment
Physical-Macath
Physical-Macath | Location in contract memory
(Decretories)
(Environment
(Environment)
(Environment)
(Environment)
(Environment)
(Environment)
(Environment)
(Environment)
(Environment)
(Environment) | Level 2/1 referencience
(Level 2/1) referencience
(Level 2/1) referencience
200 Lathout referencience
200 Lathout Level 2/1 referencience
200 Lathout Level 2/1
Dec 10 board Level 1/1
Constitution and Level 1/1
Constitution and Level 1/1
Constitution and Level 1/1
Dec 200 Level 2/1 (Level 1/1) | Part of All search Part of All search Part of All search Part to Nage 4 Headh <li< th=""><th>Thosis alkeath.
Tevelitin networksno
Practic section of the section
of the section of the section
Social section of the section
the section of the section of the section
to section of the section of the section
of the section of the section of the section of the section
of the section of th</th><th>Parch bagraf Headh
(Sacrto bagraf Headh)
(Sacrto bagraf Headh)
(Tacrist Headh)
(Trosis Aleadh)
(Trosis aleadh)
(Trosis aleadh)
(Trosis aleadh)
(Trosis aleadh)
(Trosis aleadh)</th><th>Levelor Independence
Processit Jakon Carlo
Parcito Ingel Literahh
Parcito Ingel Literahh
Processit Jakon Carlo
Level Citi ndependence
Sou sabetan compo</th><th>I concernent
location and an and a second
location and an and an and
frave and an and an and
frave and an and an and
frave and an and and
frave and an and and
frave and and and and
frave and and and and
frave and and and and
frave and and and and and
frave and and and and and
frave and and and and and and
frave and and and and and and
frave and and and and and and and
frave and and and and and and and
frave and and and and and and and and
frave and and and and and and and and and
frave and and and and and and and and and and</th><th>f (networker
the set of the set</th><th>Level Kin referendenne
Socialisation
(Throw difficulture
(Throw difficulture)
(Decrementation)
(Throw difficulture)
(Throw difficulture)
(Throw difficulture)
(Throw difficulture)
(Throw difficulture)
(Throw difficulture)
(Throw difficulture)</th><th>The result of the result of th</th><th>L each/CII on providence
(Provide call headth)
(Provide call headth)
(Providentify call headth of the
provident of the call headth of the
formatic call headth of the
formatic call headth of the</th><th>Phonoi advanta
Poscito ladvanta
Poscito ladvata
Poscito ladvata
Poscito ladvata
Poscito ladvata
Provide advata
Ladvati reducenderzo
Ladvati reducenderzo</th><th>Instructure
Instructure
Instructure
Leadin Adversion
Leadin Adversion
Provide Adversion
Provide Adversion
Provide Adversion
Social Material
Control Manachine</th></li<> | Thosis alkeath.
Tevelitin networksno
Practic section of the section
of the section of the section
Social section of the section
the section of the section of the section
to section of the section of the section
of the section of the section of the section of the section
of the section of th | Parch bagraf Headh
(Sacrto bagraf Headh)
(Sacrto bagraf Headh)
(Tacrist Headh)
(Trosis Aleadh)
(Trosis aleadh)
(Trosis aleadh)
(Trosis aleadh)
(Trosis aleadh)
(Trosis aleadh) | Levelor Independence
Processit Jakon Carlo
Parcito Ingel Literahh
Parcito Ingel Literahh
Processit Jakon Carlo
Level Citi ndependence
Sou sabetan compo | I concernent
location and an and a second
location and an and an and
frave and an and an and
frave and an and an and
frave and an and and
frave and an and and
frave and and and and
frave and and and and
frave and and and and
frave and and and and and
frave and and and and and
frave and and and and and and
frave and and and and and and
frave and and and and and and and
frave and and and and and and and
frave and and and and and and and and
frave and and and and and and and and and
frave and | f (networker
the set of the set | Level Kin referendenne
Socialisation
(Throw difficulture
(Throw difficulture)
(Decrementation)
(Throw difficulture)
(Throw difficulture)
(Throw difficulture)
(Throw difficulture)
(Throw difficulture)
(Throw difficulture)
(Throw difficulture) | The result of th | L each/CII on providence
(Provide call headth)
(Provide call headth)
(Providentify call headth of the
provident of the call headth of the
formatic call headth of the
formatic call headth of the | Phonoi advanta
Poscito ladvanta
Poscito ladvata
Poscito ladvata
Poscito ladvata
Poscito ladvata
Provide advata
Ladvati reducenderzo
Ladvati reducenderzo | Instructure
Instructure
Instructure
Leadin Adversion
Leadin Adversion
Provide Adversion
Provide Adversion
Provide Adversion
Social Material
Control Manachine |
|--|--
--
---|---

--	--	--
---	---	---
Tras address	find advintions find advintions first advintees	SLAN OF ALLOW

 | 71 constant |
 | | |
 | Hvsi caliser ptoces
Behaviour
Phvsi caliser ptoces | | Physical Surgerout | ffbyi albyrators.
 | Physica85yreecons
Physica85yreecons | Physical Synetcone | Prvi alžyratove | | | | | | | | | | | | |
| Proceeding a labority
home of each of the
probability of the second of the
probability | Pain
Brancons
Behaviorations
Benaviorations
Benaviorations
Lead sum
Lead sum
Lead sum | Promotes for un
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation
formation | Mont Capacity
Date
Dahl Mag
Dahl Mag
Dash Market
Fregat sare of ke
Processing of active
Processing of active
Processing of active | protect of the second s | Brist
Legal soue
Insuit of at re-
solution
Solution
Cognitization
Behold tation
Frunci dissee | Negati sef ret he
Negati sef ret he
Dros Elver seltre
Regati sef ret he
Regi not on Allerion
Treatment
Ar ou Arregerformatio | Treatment
Spirtuality
Cogniti vef tre tion
Cogniti vef tre tion
Spirtuality
Model by
Model by
Model by | Physical Internet
Provided Internet
Provided a allocation
Debidion
Debidion
Physical Internet
Physical Internet
Debidion
Contenting and
Contenting and
Contenting
Lead allocation
Debidion
Lead allocation
Debidion
Lead allocation
Debidion
Lead allocation
Debidion
Lead allocation | The statement
was constructed
as constructed
as the state of here
the state of here
of the state of here
and state of here
as a | Work Accessified at A | Physical Environment
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Interface
Inter | Frenchellerue
Beschraten vallenes Sve
Paan
Spersualery
Prenceale aktionelig
Frenche | Restration of head So
Reserved and the
Reserved and the
Cognitised and head So
Restration of head So
Restration of head So
Restration of the So
Provide of Environment
Physical Environment | Arc airraich formation
Phose all fravour
Phose all fravour
Phose all fravour
Best arran valiences
Best arran valience far
Best arran valience far
Devended a aborefar
Phose arran valience far
Phose |
| Dad
Dages see
Atternation
Dearma
Dates of
Dates of
Dates of
Dates of
Dates of
Dates of States of | Discorts in
Discorts in
Discort in
Distances
Distances
Environce | Conterrentico M
Conterrentico M
Estatione
Estatue
Feature
Feature
Feature
Feature
Feature
Feature
Feature
Feature
Feature
Feature
Feature
Feature
Feature | File administration
File administration
Frond Properties
Frond Ran
Function
Grant had
Grant had
Grant had | Gun
Gun
Heradis pilen
Heradores
Humocoa
Humocoa
Humocoa
Humocoa
Liberitare
Liberitare
Liberitare | Irroorria
Irrourano
Jean John
Jean John John John John John John John Joh | Loss Of research
Loss Officeaure
Lostin
Losting
Mode and
Mode and
Mode and
Mode and
Mode and | Model internation
Methodistics
Methodistics
Methodistics
Methodistical
Methodistical
Methodistics | Mond
Mond
Munt
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
Nation
N | Ports
Postitine
Postitine
Postanoi diset
Postanoi di
Postinon di posti
Postinon di posti
Di di control Posti
Di di control Posti
Di di control Posti
Di control | Bet at two
Bet at two
Bet at two
Bet a at two
Ber crobicing
Ber crobicing
Safest
Safest | Security
Security
Security and a lation
Security about
Security and a lation
Security an | Soc Inf. connected ADIane
SoceThroad
SoceThroad
Sectoration
Sectoration
Sportual
Sportual | Sentierd, versehte da
Tearthens
Ternion
Ternion
Tearine
Teare
Tearine
Tearine
Tearine
Tearine | Training
Transmost
Transmost
Transmost
Maria
Wheesing
Mile
Wite Mile
Workbad |
| | |

 | |
 | | | |
 | | | |
 | | | |
| | |

 | |
 | | | |
 | | | |
 | | | |
| | | (n/connect

 | | too in the second se
 | | Porto logical Health | ffnok atteath
Phyto kajoal Heath
 | di un di | | |
 | | | |
| | | Trainine

 | | A Local
 | | Third of | Pein
Freix, Hope
 | a bu | | |
 | | | |
| | Socialities for contrip
Environment & | toc understation of the set h

 | Physic already
Physic already
Chrysic advector | co servero corres
Socialitadas
Parcho Japa di Health
Leedo Orindependence
 | Chock althrauth. | Socialities contraction
Physical execution
control independence | reveloping a second
 | Practiculogic differenth
Leveloff independence
fitnet alfebrah
Socialization of the | Post to back directo | |
 | | | |
| | Child
Defaulded turi con | There is a construction of the construction of
 | undGroot
 | friend
Montal Health
Montal Health
 | \$reeth | Mon
LingGinor | Irotreet
LugGeoor
Irotreet | Diritica
Modelty
Doct
Mee
Zerochtul
Decrise Tota
 | to pe | | | | |
 |
Proceedings	fitystationth fitystationth fitystationth fitystationth fitystationth fitystationth fitystationth fitystationth fitystationth	Prvs catroath Ervicenteath Ervicenteath Sev attraction Sev attraction Prvs catroath Prvs catroath Prvs catroath	Sociality deficiency Physicial deficiency Physicial deficiency Physicial deficiency Social deficiency	La transmission Sociality factorerity Sociality factorerity factorerity Sociality of efforming Sociality of efforming Environment Sociality of efforming Environment Sociality of efforming Environment Sociality of efforming Sociality o	Drystathreath social reviewenting providing at reviewenting Providing at reviewenting from the attraction reviewenting of the attraction providing at reviewenting of the attraction of the attraction providing at reviewenting of the attraction of the	sectar teleformin Post indexection Proceeded the Proceeded teleformin Proceeded teleformin Proceeded	so card chalcoching Poor hologic articulus Poor hologic articulus Poor hologic articulus Larves Schemer articulus Physic articulus Physic articulus	LovelChristenderce (controrted) (controrted) Mascated (f) Mascated (f) Controrted Scutted (f) Controrted Control (f) Control (Pers contrata activities in Prace at search Prace at search	Socials detection				
 Precoverta And Benerowska Marcoverta And Marcoverta And	els Strength 28 Part 28 Part 26 Day 26 Day 26 Day 27 Freisrert 21 Strengt	al Data 216 Everke 216 Everke 45 Everk 46 Everk 31 Courth 33 Courth 33 Courth 55 Ford Kinner	20 Mont Priced 20 Mont Priced 20 Mont Priced 20 Pert Stortman Admin 20 Lingtheod 21 Lingtheod 21 Lingtheod 21 Lingtheod 22 Lingtheod 23 Lingtheod 24 Lingtheod 25 Lingtheod 26 Lingtheod 27 Lingtheod 28 Lingtheod 28 Lingtheod 29 Lingtheod 20 Lingtheod	25 Entry Rind 26 Entry Rind 26 Encontract 21 Encontract 21 Encontract 21 Encontract 21 Encontract 21 Encontract 21 Encontract 22 Encontract 23 Encontract 24 Encontract 25	M Lundincer 50 Treatment 11 Concertration 50 Treatment 10 Treatment 11 Treatment 11 Treatment	42 Factor Knool 9 Thraker 52 Loconor 53 Houser 54 Houser 56 Houser 56 Houser 56 Loconor 56 Loconor 57 Loconor 58 Loconor	de Fronte 21 Hope 21 Hope 28 Hope 28 Hope 29 Lonationet 20	 (a) Transmer (b) Transmer (c) T	20 Bellef 20 Stellef 21 Stronking 21 Londing 21 Londing Reh 30 Londing Reh 31 Den 31 Den 31 Den 31 Den 42 Anteonoffmoti	66 66 Friend 34 LumD besse				
21 65 21 73 21 73 21 75 21 75 21 75 21 75 21 75 21 75 21 75 21 75	21 25 21 25 01 04 12 22 26 12 26 41 26 41 26 41	21 25 0 13 1 2 51 2 0 23 3 2 23 3 2 23 3 2 23 3 2 23 3 2 23 1 3 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 2 1 1 2 1 2	81 41 0 04 0 14 26 35 27 25 12 25 12 21 12 21 27 25 25 25 25 25 25 25 25 25 25 25 25 25 2	01 21 02 23 01 21 01 21 02 13 05 13 05 13 05 13 05 0	13 2 24 33 54 53 54 63 54 63 04 63 05 23 05 23 23 41	41 45 53 54 01 06 1 15 16 25 05 12 22 3 22 8	22 3 04 13 21 26 01 04 04 12 13 25 25 25 25 25	10 24 26 55 26 55 26 55 21 26 21 26 21 26 82 45 82 45 82 45 82 45 81 20 11 21 21 20 10 05	05 1.8 0.0 1.6 0.0 1.6 0.0 1.6 0.0 2.3 2.3 2.3 2.1 2.3 2.1 2.3 2.1 2.3 2.1 2.3 2.3 2.3 2.5 2.3 2.5 2.3 2.5 2.3 2.5 2.3 2.5	00 12 14 22 32 35				
11 III IIII IIIIIIIIIIIIIIIIIIIIIIIIII	10 cov.or.1010 11 cov.or.1010	 reshift(20200) Bizzlifican (GLIA) Dizzlifican (GLIA)<td>00 (0.1000-0000) 00 (0.1000-0000-0000) 00 (0.1000-0000-0000) 00 (0.1000-0000-0000) 00 (0.1000-0000-0000-0000) 00 (0.1000-0000-0000-0000-0000-0000-0000-00</td><td> Contract and the second second</td><td>110 WAT 10m056 111 WAT 10m058 110 WAT 10m0581 110 WAT 10m0581 111 WAT 10m0581 100 WAT 100000000000000000000000</td><td>20 314-02-02-02-02-02-02-02-02-02-02-02-02-02-</td><td>10 aff antworks aff antworks anthreactes 1 yours and 1 yours and 1</td><td>at according to the second sec</td><td>10 International International</td><td>00 2014/10/2017</td><td></td><td></td><td></td><td></td>	00 (0.1000-0000) 00 (0.1000-0000-0000) 00 (0.1000-0000-0000) 00 (0.1000-0000-0000) 00 (0.1000-0000-0000-0000) 00 (0.1000-0000-0000-0000-0000-0000-0000-00	 Contract and the second second	110 WAT 10m056 111 WAT 10m058 110 WAT 10m0581 110 WAT 10m0581 111 WAT 10m0581 100 WAT 100000000000000000000000	20 314-02-02-02-02-02-02-02-02-02-02-02-02-02-	10 aff antworks aff antworks anthreactes 1 yours and 1	at according to the second sec	10 International	00 2014/10/2017				

Appendix G

Combinational Algorithm tracing



LevelOffindspend an or P sychodogi calreadh P er son si Valu eskindliel ek Socialit d aloinn hp P vichodai calreadh	Environment	Psychologi calificath	Ps ych ologi calificath	Physicalification	P hysicalitealth LevelTh reference an co	Environment	LevelOff independien op	P or son al Volu es And Beli ets	P sychologi calificath	P or son al Volu es And Beli ets LevelOff notecered on on	SocialR diations hip	Environment	Socialitie aboors rep Per son al Volu es Andibeli ets	Physical-tealth Discriticistics	Environment	Physicameratin Environment	P hysicalHealth	LevelOff independien op	SocialR diations hip Environmente	Level Of independien on	Provincian control Provincial Provincia Provinci Provincia Provin	PhysicalHealth	Environment, Per son stroku erkindheli ett	Physic all tealth	PhysicalHealth Level Chindecend an or	Environment	Environment.	SocialR diations hip	Ph yrsk alfriealth P ar son si Yoldu eeld ndfeel eft	Psychologi calificath	Physical-tealth Deviced-sized-th	Environment	Psychologi calriceath Psychologi calriceath	Environment	LevelOff independ an ce P sychologic diffecith	Psychologi calreealth	Socialiti di accorsingi Pitrysicalitieshth	Level Off independien on	Physical-teath	PhysicalHealth Louid Of orboard an on	P hysical-tealth	P hysic alt-lealth P or son 31/ohu eol rotheli eft	P or son d Volu es And Beli ets	SocialR diations hip Disacter abliculation	P hysicalHealth	P sychologi calrieoth	P sychologi calificath	Psychologi calrealth	PhysicalHealth Level Cfl released an or	Environment	Environment.	Environment,	LevelOff independ an or LevelOff independ an or	PhysicalHealth	Per son al Valu es Ancelea es PhysicalHealth	P hysicalHealth	Socialet el actoris rigi LevelOff indépend en ce	Level Off independien on
						8					b	*	a	PhysicalSymptoms	*	Ben av our	PhysicalSymptoms		۵,				×					4			im PhysicalSymptoms	*		*			im PhysicalSymptoms			rrr. PhysicalSymptoms	im PhysicalSymptoms	PhysicalSymptoms		a.	im PhysicalSymptoms				im PhysicalSymptoms	*	* 5	5 ×				im PhysicalSymptoms		
Legal is cue Negativel' ed ing Retigion Sociada di vity Combinel'unity	Reh db ill tetion	NegativeFeeling	PostwFeelrg	Perspiratory and reg	Respiratory/liness Treatment	Aquiringriformati	Treatment	Spirituality	Co gritive function	Spirituality Mobility	Per son a Relation sh	PhysicalEnvir onne	P of son a reason on Religion	Other Syngtoms	PhysicalEnvironne	Economic Status	Other Synptoms	Legitsue	P er son al Relation sh Dhuirin Brain correct	WorkCapacity	Post tvofeeline	Respiratory filness	PhysicalEnvir conne Petiteion	Respiratory lines s	Pespiratory liness Trootment	SocialC are	Rehabilitation	Per son a Relation sh	Pres t Pred inform	C ognitiveFunction	Ridger alb ryflines s5 Prinnet referrine	PhysicalEnvir onme	Postvefeeling NegativeFeding	PhysicalEnvir omre	MedicaSelf.are 8 ob/imae	NegativeFeding	Sociated of Vity Ridgination of lines (S	DalyJAing	Behaviour	Received reviews S	Ridgin alth Millines sS	Pain Soldhualtv	Spi ritsuali ty	Per son diRelation sh Energy	Riscoir abon Mines s5	NegativeFooling	hisstivef edire	CognitiveFunction	Rid pin alti ni fines s5 Dallu kinine	PhysicalEnvir omne	PhysicalEnvir omne Amulational efformation	PhysicalEnvir onme	Finandalissue Finandalissue	Respirato rythness	Spiribuality Behaviour	Rid pir alb rollines sS	P or son a Meadown yn Llegail ss ue	W orkCapacity
surance sublity with anguagoOffitu ity annin s	eiture .	os sOft reteres t	avive	ower rejurings of	ungDisease and rul airt	fed al Information	ted to net ntake	fed tation	fertal Health	4 nd/uh ess to bili trakid	40 m	50 up	20 M	laus ea	8.0	kur kerti muake Ki cupation	ed arra	r gm Do nation	arents z k	atTre	er som oves t	neumonia	ollution raw	u hnoneny fib rosis	u trionary Hypertension u trionary Rehabi	au allty Of SocialCare	et reation	datve	el axetion disious Celebrration	arrenbern g	unnyñios e	afety	atisfied cared	ecurty	eekingEmergencyCare effisteen	enseOff sol ation	exualmenty ho rtness Offlineath	ho werin g	moking	neezing oo inferonoori onli Mand	or effirroat	or en ess of rith uell exour cos	piritualServices	po use	w effect, yrrp hrkodes	early ness	ension	Nriking	igh Chest. ciletine	aire	raffic estation	ransport	ravelExpensies motimentCost	unor	adue active	Prestre	e e	forkload
					22	2	2 2	2	2	2 2	2	2	2 2	2 2	2				6.0			ď	<u>a</u> a	. «	44			æ	đ a	8	æ ü						2.0							0.0	80			F			Ī				Ì	5	5.5	~
						Psychol opic alf led th		Physical Head th			Psychol opic alf led th								SocialPeterio neh jo																																							
						Thirking		Pan			Fear, Hope								Farray																																							
hysi caloodhh			to del Rebelors hip	Physical (each)		evelOf integerd ence	rive contecto	evelOR independ ence	Physical-teody	evelOf referend ence	Social Relations hip		Psychologi calificath	evelOf independience		Physical-tealth	for the Red Address May	to dal Relations hip	Environment, 2 autological militarithi	Canada and Diana and Di	evel20 released even																																					1
Szength			Mom	LungCancer 1		Treatment	Uniform of	Treatment,	LungCancer	Treatment	Family		Thirking	Mobility		Pest		Grand this	Exercise, Trainin I		Treatment																																		T			
So d d Relations N p LavelOff ndap an denoe Psychodo gool Headth LavelOff ndap an denoe Physic califerably	LevelOff roleg an dense	So del Reletions hip	Physic calification	So dal Relations hi p	Physical feedby Developments	Bi vicorment	So dal Relations hip	Paycholo gical Health	LevelOff ridep in dense	Paycholo gical Health PhysicalHealth	Physicableokh	LevelOff rideo en dence	So dal Relations hip	Physical-teach	So dal Relations hip	So dal Relations hip	Physical-tealth	Physical-teath	Physical-teath Doorn nutrition and orbital	Physical-teach	Physical-coath	Physical-tealth	Physicatriesth Level20 released denor	Physical-teach	Physicatriesth Physicatriesth	Physical-teach	So dal Relations N.p.	So dal Relations N.p.	Physical-tealth																													
36 Wrle 50 Treatment 31 Contine allon 50 Treatment 38 Pain	10.2 Treatment	42 Famby, Friend	52 LungGincer, Pain	55 Hustand	30 Copd 5.0 LoveCarrier Dove	42 Gym	ad Famby	51 Hope	33 Treatment	36 Hope 84 Free av	59 LuneCancer	48 Treatment	5.0 Exercise 8.3 Child	91 Putronen/Ebrosis, Col 20 Exerctes	49 WYe	2.1 Family	73 LungCancer, Smoking	33 LungCancer	47 LungDisease 3.6 Dated	35 Smoking	12 Lundoncer	147 LungCancer	36 Turnor, Drug, Pain 14 Insurance	38 Pán	34 Drug 31 Pain	42 PuknovanyFibrosis	d d Family	46 Friend	3.4 LungDis ease																									_			_	
2.04 2.4 2.46 3.3 5.05 5.4 5.39 6.3	0.47 2.3	4.11 45	0.05 0.6	1.56 2.5	0.47 1.2	22 3	22 3	0.39 1.3	0.05 0.4	1.25 2.5	2.52 3.5	1.47 2.4	331 45	0.03 1.3	2.08 2.6	8.50 A.2 A.2M A.5	0.06 1.2	2.07 2.4	0.03 0.5	0.02 0.4	2.19 2.3	0 2.3	2.82 8.1	0.03 0.4	2.47 3.3	0 0.4	0.47 1.3	1.38 2.2	3.19 3.5																									Ħ				
117 WShT-I0w00 48 118 WShT-I0w00 480 119 WShT-I0w00 488 120 WShT-I0w00 420 121 Dutword/uk1	122 Dutrippid ucl	124 Autoppeducte	1080304774d 821	128 preuving 52407	129 0v8m0v807 84	131 0/RmGVEG 89	133 afficience via	134 9n8hetWo033	100 01/02/07 00:1	1.37 Tyo27540 Y02 1.301 Tyo27540 Y02	139 Tyu27540 Y05	141 ZoffgLigLmB 5	140 Z0655490-m89	1.44 EH7aHDHHAU1	140 51724/0142/0	140 FH7aHDH46U14	149 Cf\0395 C2w1	151 Ch4395 C2w7	102 m5Y846VV401	154 32 34409500 MI	100 52 Ph/0600109	157 LYNO31A-20641	159 LYNCHM-20649	100 top/or/028A1	101 by/or/05/2010	163 grAnt/804X01	164 QLANV8DLK OF	100 2p.845vr45V486	167 2p.84kvm65W812																													

Appendix H

Expert's Evaluation Results of the Semantic Tagging Algorithm (First Domain)

The evaluation result of the Semantic Tagging Algorithm with an expert after implementing it in the first domain (Giving Pitch Presentation)

×	8 C	0	в	2	U	Ŧ	-		×	L I	N O P Q R S	F
1 Segmentid from a v	a vi starti tri.	the rearsering t	Loo nothing it you agree with the contept and the topic Or highlight the co.	reapt or the topic you mains or feet with real or highlight it w	m Orange in you are not sure				Can made	The main to	DK OI THE	
3 TmbOFW8vTV1	0 03	12 When us stand and messed in unit to some summer a confident and and some control	Sterici	danio pasau	my now on the evaluation result	BothPosition	Bod of another amother and another amother another amother amo	MonVietalCommunication	colinebro	Concepto		
		And one of the great way to do that by working access a cumulant and cuptoed. For the first prest ways to do that by working on our body language. Let's	Confident Bortel annumeSternal			SpeakerEmotion Bodia annuneramehodilanninane	Non Method/Communication			Delive		
2 8		Conciliant rices way comprove you narrenda oooy rangaagen i presentanon . Concilia up Helio again friend . If you never seen one of my video i'm Alex Lyon .	Body, and used signal Dominiational Communication			Body, ang ug enames of fan gu age Int eract Mity	NonVerba/Communication			Delive Presentation	VV V VV Atribute	
a 9	ŧ	And this channel Communication Coach is here to help you increase										
11 TmbQFWBvTtY2 12	0.25 0.4	0.4 your impact so you can lead your team to higher level of eccellence. And I bolleve that presentation still are a really important part of leadership.	Impect Sarding			BodyPostion	BodyLangugenamebodylanguage	NonVerbalCommunication		Presentation	(Ktribute 1)	
2 1 1		any kind of leadership activity and presenting your includency in your remi- any kind of leadership paction. So we want to get just a fitte before the offer of the his of any fitter way way way more want more and a hindri particity and an	courte and unagroup rate Confident			SpeakerEmotion				Detive		
16		the text a row a rear ray just we much one to the row of the row of an induced and and the row of t		Composure	To fite- Domain ontology (composed)						Delivery	
17 TmbQFWBvTtY3	0.47 1.1	11	Eye		ignore - It was picked for the	EyeMoton	BodyMotion B	BodyLangugenameb odylanguage	NonVerbalCommunication	Delive		Ľ
19		And these tip by the wery go together. So each of the three tip have a couple of piece a you 'I are. So the first tip is eye contact and article. So eye contact .	Smle		segment	FacialExpression	Gesture	BodyMotion	Bodyl, angugenamebodylanguage	NonVerbalCommunication Delive		
8 12 8		You have bolook diredly into somebody's eye that's listening about 99 % of the time. You do n't want to look too much at your note. You do n't want to	Eye Eye Bisye			EyeMoton	BodyMotion B BodyMotion B	Bod yLang ug ena meb odyfan gu age Bod yLang ug ena meb odyfan gu age	NonVerbalCommunication NonVerbalCommunication	Defive		
7 22 72	ŧ	took back as your store really. You contraining out it warm to took as the floor over their head. A bit of bad habit out there in term of eye contact.	Silog Eye			EyeMotion	BodyMotion B	Bod yl, ang ug ena meb odylan gu age	NonVerbalCommunication	Detive		
25 TmbQFWBvTtY4 26	1.13 1.3	1.3 You want to book directly in some one is eye almost the whole time and that really take quite a bit of practice. The other thing you want to do is look.	Eye Pradice			EyeMotion Preparation	BodyMotion B	BodyLangugenamebodylanguage	NonVerbelCommunication	Detive		
27	╞	drectly into somebody's eye long enough to finish a thought and then move on to the next prestor. So you do n't want to lust stare at once person the whole	Eye Communication			EyeMoton	BodyMotion B	BodyLangugenamebodylanguage	NonVerbalCommunication	Delive	IV Attibute	
28		the and you don't want to scan so that you he news looking in . A communication coach actually once told me One thought one look So long errough to finish about										
31 32 1VCaJ4dSHek1	0.01 0.3	3.3 Using gesture when give a public speech is a very important part of your presentation.	Gesture			BodyMotion	Bodytangugenamebodylanguage	NonVerbalCommunication		Delive		
88		If you fail to use pesture properly and you do awkward pesture throughout your presentat you are probably going to distract your audience and you are not going to get the impact (Speech Gesture			L ang uag eElement BodyMotion	VerbalCommunication BodyLangugenamebodylanguage	Nor/VerbalCommunication		Detive		
88		you want when you presenting . So what are four gesture that we use and how can we use them and use them effectively	Gesture Impact			BodyMotion	BodyLangugenamebodylanguage	NonVerbalCommunication		Presentation	AV Attbute	
10		during our presentation ?	Gesture			BodyMotion	BodyLangugenamebodylanguage	NonVerbalCommunication		Detiv		
38 1VCaJ4dSHtK2 40	80	0.5 Hi I am Ryan McLean and I am rom publicapeakingpower.com where we leach people like powerful public speaker.	Gesture		To fix- Domain on totogy (walk)	BodyAkiton	BodyLangugenamebodylanguage BodyLangugenamebodylanguage	NorVerbalCommunication NorVerbalCommunication		Defive		
42			Gesture Gesture Contro			BodyMotion BodyMotion	Bod yLangug en améb odylanguage Bod vLangug en améb odylanguage Bod vi angi venaméh odylanguage	NoriVerbalCommunication NoriVerbalCommunication		Delive		
44		describe something or shuadon. We might draw comparison between something that 85 ms	0.01600	Speaker Aura 7	Ignore- Suggested new concept is irrelevant	and the second second					Detivery?	
45 1VCaJ4dSHak3	1 12	12	Metaphor	Speaker Aura 7	Ignore- Suggested new concept	RhetoricalDevice	VerbalCommunication			Delive		ľ
46 47	+	really big and something that 6 th s really small or we might use it to contrast certain fem or to depict the size .	Stand		Is indevant To fix- Domain ontology (stand)	BodyPosition	BodyLangugenamebodylanguage	Nor/VerbalCommunication		Delive		
48		A are at example that I am taking from tow master is if you are using a metaphor something is like a tiny little bird and you are holding like cup stand stating the metaphor of the	Metaphor Gesture			RhetoricalDevice BodyMotion	VerbalCommunication BodyLangugenamebodylanguage	NonVerbalCommunication		Delive		
85	ŧ	bird and also implying size being amail . You cam also use descriptive greature to any shape say it wa the box this big back										
53 h6sm47j-Am42	02 03	0.3 and there are a lotta reason why .	Reason		To fix- Domain ontology	Contentinitroduction	OpeningComponent	StructureComponent		Stuch	ue NA	
5 88 1	╞	of using lighthearted humor in a presentation				NIMUTORIANOVO	Verseinverinkritionen					
8 13 3		and when leach in college to the presentation										
50 60 h6em47.Am43	0.35 0.5	15 hur Ican die vou a melik straicht alheed wav	Humor			RhetoricalDevice	VerbalCommunication			Delve		l
61 61		to add humor without the big risk of a punchline joke. So let's take a look.	Communication			Interactivity				Presentation	Attribute Detivery both topics are covered here.	
88		(light music) Hi I'm Alex Lyon and this is Communication Coach .										
8		We here to help you increase your impact		ling act	ignore- It was picked for the segment						Presentation Attribute	1
67 dEDcc0aQaA1 68	0.01 0.1	0.1 " H I'm Jeson Tetesk. 1.1 help seeded overcome their fear of cubilo speaking and deliver amazing presentation	Opener			StructureComponent				Stuck	Te -	
60 20		and bolay I in going to show you how to captivate your audience within the first 5 minute your presentation.	8									
12	ŧ	Here 's how I'm going to doit." '' I'm going to show you 5 sep to defiver a killer opener."										
74 dEDcc0aQaA3	0.43 0.5	0.5 What am I going to do ?	Opener		ferrors. It uses addred for this	StruchureComponent				Shuck		
75	+	Hi Jim. How is it notion 2	LT ACT BOO		anue n'es prova la rue	Linpastor.						
07 72		I mow singurs r I'm freaking out about this big presentation I have to give tomorrow . Wali have unuted on a voir some and receive a financial financial									Distance	
9 R 8	ŧ	No.										
81 dEDcc0aCjaA5 82	1.1 1.3	1.3 Let's start off with skep number one which is " Give a Confident introduction." One of the biggest mistake people make when they start their presentation is showing neil	Start Confident			StructureComponent SpeakerEmotion				Stuck	14.	
88		Here 's the key : it is not whether you're menous it's whether you show it. The first thing you need to do is stand STILL.	Introduction Start			Opering Component Structure Component	StuctureComponent			Stuck	re r	
888		Don't move. Don't pace and keep your hand at your side.	Nervous Nervous Stanrot			Speakerfermotion Speakerfermotion BortvDn stiton	Rotol and unen emotion/damarane	Non/achalCommunication		Delive Delive		
88	Ę		Pace	Guesture	To fix-Domain ontology (hand)	Paratanguage	NonVerbalCommunication			Delive		
90 MhiPpUrTcRcd5 91 00	89	6 confidence add an authority to everything you say during your	Confidence Confidence			SpeakerEmotion				Defive		
383		presentation but you re not going to develop the confidence you need if you do n't moeting and monetive at	Practice			Preparation				Delive		
1 3 8	ŧ	whatever Dama give a talk or										
97 MnPpUTcRc36 98	6.05 6.2	6.2 presentation he practice he say for hour they do n't want to make sure he	Practice Slide			Preparation Visual Att Artefact				Visual		
80	+	hathe timing down that he know every slide so he doe n't have so read off	2010	Lenger or Arganesion	ignore- suggested new concept is indevant	A BULGENO M DEL COL				IN I	Presentation Attribute	
101		them he know what the next slide is so		TransitionComponent	Ignore- Suggested new concept							
103 103 104 Demt XFili Buo20	5.46 6	To the first structury surveyor pervects areas To fastion 3 - to second, executions that fail under the readon of socializing and personal	Chailstrine		S Ittoren	ScreeterAura				Delive	Structure	T
105	-	development ; has sufficient or the second in and want to help you with .	ALL REVEALED.	inter activity	Ignore- Suggested new concept is indexant	Appendix tax u					PresentationAttribute	
105	+	So go alread and subscribe. You need to get our newest whoo including Charlisma Breakdowns and again these talking										Π
100	ŧ	head style video . A question for you guy and for a request is that you write in the comment thing that										
111 gaox/Fju@uo22 112	6.24 6.4	6.4 up the channel ? What make a good video and what go into it ?	Video Video			Fim	Moving Image Moving Image	Illustration	Visual Ad Attel act Visual Ad Attel act	Vensiv	944	
113	#	And I've learned a little bit about not only you know the psychology behind it which is my bag but also how to do it all a kochnella way.										
116	ŧ	curring in the source model and the part includes a source processing and the source and the source of it is a water of its and the source of										
118 d812a7qG9Kw27 119	6.12 6.2	6.2 you 're gring to be boosing your overall performance up gute a bit	Practice Practice			Preparation Preparation				Delive	N	
120	⋕	even if you do n'tiget to practice it a few extra session . So those are the five by I recommend	Sprech			L ang uag ettiem ent	VerbarCommunication			D001M		Π
122	-	In term of how to practice for a speech .				-	-	-				1

			Protection and a second and a s	Domany Library Control Dependent
	Ruckar Kongeneration	Verbicities of the contraction o	Determinant Determinant Markatorian Providentian Providentian Ma	Image: control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control
table table Screenent Commont	are bion are bion contractions Contractio	memory Build in Contraction Month in Contraction <td>method Statistication method Statistication method</td> <td>mono Mono Mono</td>	method Statistication method	mono
Provide the second seco	Tegenson. Progenson. Progenson. Progenson. Progenson. Progenson. Progenson. Blocker/Componentin Blocker/Componentin Blocker/Componentin </td <td>Properties Properties Properies Properies Properies</td> <td>Figure for Properties Properties Montendencies Montendencies Figure for Properties Properties Montendencies Montendencies Properties Standardences Standardences Montendencies Standardences Standardences Montendencies Montendencies Montendencies Montendencies Montendencies Montendencies Montendencies Montendencies</td> <td>Programmini Freemationi. Programmini Freemationi. Programmini Freemation. Prog</td>	Properties Properies Properies Properies	Figure for Properties Properties Montendencies Montendencies Figure for Properties Properties Montendencies Montendencies Properties Standardences Standardences Montendencies Standardences Standardences Montendencies Montendencies Montendencies Montendencies	Programmini Freemationi. Programmini Freemation. Prog
Autochamment Au	Ardiforment Ardiforment etablication etablication etablication etablication etablication etablication biologication b	Andocratorioni and Science on a Machine on a Ma	Andochonomi Pasamon Pasamon Andochonomi Pasamon Pasamon P	Distribution Distribution<
Botto Accimicante Botto Accimicante Botto Accimicante Montal Brancing Montal Brancing Montal Brancing Botto Accimicante Botto Accimicante	Book and function of the	 Biolochicomic I Biolochicomic I Biolochi	 Balantarelinionenti di alla di al	 Brandomonti, I. Martini, Santa Simonti, J. Martini, Santa Simonti, J. Martini, Santa Simonti, J. Martini, Santa Simonti, Santa S
Manual Manua Manual Manual Manua Manual Manual Manu	1 20000 1 20000 1 20000 1 20000 1 20000 1 20000 1 20000 1 20000 1 20000 1 20000 1 20000 1 20000 2 20000 2 20000 2 20000 2 20000 2 200000 2 20000 2 20000 2 20000 2 20000 2 20000 2 20000 2 20000 2 20000 2 20000 2 20000 2 20000 2 20000 2 20000 2 20000 2 20000 2 20000 2	The Documents of the Document of t	Image: Section of the processing of the processin	In the interaction of the interact
Audiecodi módon Ukrientanudality?	Audie configuration Uncontract dating / Uncontract dating / Uncont	model model and a model mode		
		Nurde Nurde Image: State of the s	Nutrate nation of Aurophy of Auro	Murden cell motion Marten and Ming? Parten and Ming? Parten and Ming? Mind Cell and Mind Cell Mind Cell and Mind Cell and Mind Cell Mind Cell and Mind
Merrindion Morrindion Morrindion Morrindion	Montakon Montakon Montakon Montako Monta Monta Monta Monta Monta Monta Monta Monta Monta Monta Monta Montako M	Utda	Understandslity/ Understandslity/ Cutary, Understandslity/ Cutary, Understandslity/ Page 2004 Page 2004	Understandsbilly/ Understandsbilly/ Clarky, Unsessandsbilly/ Clarky, Unsessandsbilly/ Clarky, Unsessandsbilly/ Makeroni
		Memory Memory Reson Banon	Resent International Participation Undernation// International Participation International Participation Construction// Participation International Participation Construction// Participation Enternational Participation Construction// Participation Enternational Participation Construction// Participation Enternational Participation Construction// Participation Enternational Participation Construction// Participation Enternational Participation Construction// Participation Enternation Construction// Participation Enternation Construction// Participation Enternation Construction// Participation Enternation Participation	Result Constrained Note Underworksing Note Note Note Note Note Underworksing
66 600 800	son and a second and a second	Bit (a) (b) (b) (b) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c	Result Result<	Instant Instant Undertweetingty/ Provide Provide Undertweetingty/ Provide Provide Provide Provide Provid
xt piece of information and allow to absorb it that moment of	La reason why the staff happens . Lister source why the staff happens . Lister source activity attractions . Lister source activity attractions . Lister source activity attractions . Lister source activity attractions . Lister source . L	1. New Wy fit and flagens. In month 1. Street wy fit and flagens. In month 1. Street with append. Bart 1. Street with a point Bart 1. Street	Tenent of the Afriction of the Africion of the Africion of the Afriction of the Afriction of the Afric	Transmission Transmission<
the next p then to all them to all	The short of the more and the second	Bit of the phone Financial State (Control of the phone Financial State (Control of the phone Financial State (Control of the phone Bit of the phone Financial State (Control of the phone Financial State (Control of the phone Financial State (Control of the phone Bit of the phone Financial State (Control of the phone Bit of the phone Financial State (Control of the phone Bit of the phone Financial State (Control of the phone Financial State (Control of the phone Financial State (Control of the phone Bit of the phone Financial State (Control of the phone Financial State (Control of the phone Financial State (Control of the phone Bit of the phone Financial State (Control of the phone Financial State (Control of the phone Financial State (Control of the phone Bit of the phone Financial State (Control of the phone Financial State (Control of the phone Financial State (Control of the phone Bit of the phone Financial State (Control of the phone Financial State (Control of the phone Financial State (Control of the phone Bit of the phone Financial State (Control of the phone Financial State (Control of the phone Financial Sta	Number Control Control Rev Rev Mark Strates Rev Rev Mark Strates Rev Rev Mark Strates Rev Rev Mark Strates Rev Rev Mark Strates Rev Rev Mark Strates Rev Rev Mark Strates Rev Rev Mark Strates Rev Rev Mark Strates Rev Rev Mark Strates Rev Rev Mark Strates Rev Rev Mark Strates Rev Rev Mark Strates Rev Rev Mark Strates Rev Rev Rev Mark Strates Rev	The start of the star
380	terror use subsort is necessarily and the second	Inclusion to interact or a many of the difference	Increase	International control of the contro of the control of the control of the control of the
	10 0.01 1.02 2	In the second	Image: Constraint of the constr	International control of the contton control of the control of the control of the contro

A	D D	E	F	Ŧ	-	J K	 M N O P Q R S T
241 I-00UTUMVINID 241	3.1.3 3.3 You region to externing accurt so even (using preadice that had your promition and had not contrain hear use	Conclusion Conclusion Standa		ClosingComponent Roch/Position	StructureComponent Roctvi annuneramehorbiannunen	Northerball/Communication	Studiery Studiere Defense
9	then were to anthrough the research after	21862	Transition To fite Domain on lotogy Transition	100000 100000			Statribut
245	now people any well 1 to got to tak for an hour 1 mnot going to stand up and do						
247 El2bHe0GuZQ5 248	1 1.1 threat you are a bit nervous they'll approvale and respect the fact that	Nervous Visu a Andra		SpeakerEmotion			Detrierv
249	you're dong your best for them poincide number fwo use risual acts	VisualAdds					Visuality
222	make sure that your use of visual aids is skilled and smooth and not excessive						
254 El2bHe0GuZQ6	1.15 1.3 remember that a picture speaks a	Picture		Figure	Illustration	Visu at Aid Artef act	VisuaNd
58	mouseric words using mause and swill make things clearer and will also fine e	SDM BINGA					
255	do i have be spend explaining this time is before spend intereacting with your						
260 261 9/Eys7g2YFc24	5.13 5.3 like just do it until you're comfortable	Voice		Paralanguage	NonVerbalCommunication		Delivery
<u> 20</u>	their something in your voice has changed I don't want you to do this who ut	Voice		Paratanguage	NonVerbalCommunication		Delivery
	using a warning to under wro or pripacua exercises that get your body and your wrom amady to reader new your confid this						
267 268 9IEvs7q2YFc27	5.51 6 to through all my whoke it find the dwam with high	Practice		Preparation			Delivery
200	you can practice daily make scenetring up maybe just do one	Practice		Preparation			Delivery
271	video a day failing that if you really don't work that hour sources and work						
273 714	don't want to practice daily you're not						
275 Js7-ON-d19PE 160 276	42.3 4.2 an impression of you your pitch and your how uttrin the first floor to loor	Pitch		Tone	Paralanguage Mwinoimane	NorVerbaCommunication VendArt Andrea	Delivery
217	seconds to neconds I almarity have an		Impresseion Ignore- Suggested new concept is indexeed				Presentation attribute
276	interestion of votal scenario of and		Begiming Ignore-Suggested new concept is indexecting				Statching
202	There is actually a video contraction of the second se						
281 LT CALIFORD 101	10.1 1.4 1.1 interest of the state interest of the state	TITE OF T		Tene	Duralan ar care	All and Anthrop (Constraints in the field for as	
202 Js7-OWd19PE 164	43.4 44 I was boking at in this application was	High	Opering, begiming, introduction types ted new concept	Ione	Paratanguage	NorVerbalCommunication	Delivery
202	Liou Mu de 1988 Alan Ducces e le 1989		impression Ignore-Suggested new concept				Structure
285	you're giving your pitch make sure you don't waste those opening moments and		Is indexant				Presentation attrbute
285	give people time to leit their mind wander or make an impression about you						
289 Js7-OWd19PE 165	5 43.5 44 and put you in a box that has nothing to	Nervous		SoealerEmotion			
8	do with who you are but rather that box mich has a mileration of the fact that						
200	you are a little nervous about starting						
22	and you don't water to seem tude come right out there with the hook now I want						
296 3ozZnT72X0M105	28.1 28 chalenges that you've spoken about	Summarize		ClosingComponent	StructureComponent		Stucture
208	moughout the presentation nowifs time to summarize that	Summarzo		coangcomponent	Structured.comp.comm		Sourceare
300	to recap so you just go over it again very outickly divery or you can say let's						
100	summarize briefly what we we boked at						
303 3ozZnT72X0M106	1 28.3 28 let's summarize briefly what we've	Summarize		ClosingComponent	StuctureComponent		Stuchtre
305	right or you can say in conclusion	Conclusion		ClosingCompanent	StuckneComponent		Stuchus
307	there are three big challenges						
		10 - 10					
310 XyRJKCT wrk11 311	2.35 2.5 so definer a performance by standing up now if you're going to stand up you need	Stand		BodyPoston BodyPoston	BodyLangugenamebodyfanguage BodyLangugenamebodyfanguage	NorVerbalCommunication NorVerbalCommunication	Delivery
312	to raise your laptop up higher so that	Lap top	Ignore - It was picked for the segment	Computer	VisualAldDevice		VisuatAd
313	the lens of the camera is in your direct		Eyeline, Eye,Non/teitas/Communication tynore- Suggested new concept is indexant				Delytery
315	eyeline you do not warm to be looking down in your do not warm to be looking						
316		An anna the an		Bank Do Allon		A look to the difference of booties of	
312 XyMjKCI_WK12 318	2.522 3.11 all your audience can see is the inside of your noise carly so standing up will	Stand		BodyPoston	Bod yLangugenameb odylanguage Bod yLangugenameb odylanguage	NonVerbalCommunication NonVerbalCommunication	Delivery
319 320	just maturally make you definer a better presentation and that is key when						
700	agreeming a presentation online you are still performing so stand up and perform						
324 WuffM1xHX3027	5.49 5.6 with your audience.	Speech		L ang ugg eEl em ent	VerbalCommunication		Delivery
325	So the final point is that most of the time when people deliver speech	Projection		Paralanguage	Non/tetbalCommunication		Detivery
328	they speak really quietly and no one can hear them.						
88	Rght so vocat projection really really important .						
331 Wuffing 1xHX3030	6.23 6.3 before going out to deliver a speech	Speech	Ignore- Suggested new concept	LanguageElement	VerbalCommunication		Delivery
22 22	is actually go to the tolet go to ike the back oral somewhere at achool		Projection is interested				
335 335 335	and Wanti you to just should . The which point of this is to just an owner of an another point of the point o						
337 337 200 Diversion Vanielo		Enter Ante Anterna II		tal annuality in .			
000 000	2.10 AD International and a series a series and a series a series and a series a	Chaisma		SpeakerAura			
341	and speaker chaitsman and speaker that and speaker that and speaker that and speaker that the speaker that that the speaker that that the speaker that that the speaker that that that the speaker that that that the speaker that that that that that that that tha						
342	if you can't keep your audence engaged						
Rua5eBaX4gw16 345	2.55 3 you need to start achig them	Start	Ignore - If was picked for the seament	StructureComponent			Stucture
346	and if you are someone who doe rithave charisma	Chaisma		SpeakerAura			Delivery
349	or who doe n't have the ability to keep your audience engaged						
351	consider doing this course orial						
352 cpaSWeJF 42 353	0.24 0.4 first thing to look more confident is believe in yourself but that do n't	Confident		SpeakerEmotion SpeakerEmotion			Definery Definery
355 355	audentritien darg entorier examinque to look more confidient lubien you i're speakking is to broede acard allu bu sour sood						
150	to the measure events and you need						
350 c[5aSWeJF]43 360	0.43 0.6 hell me to breathe but here 's something to keese in mind	Nervous Breath		SpeakerEmotion MentalPrecorration	Preoaston		Delivery Delivery
150 XX	number on when we get nervous we tense up and we tend to hold our breath it's a						
***	Up and were very unit you up to exercise a a normal reaction we have note here vou						

μ	<	8 C	0	E	F G	H	-	Р		M N O P Q R S T
18			breathe we do the same thing up there on							
305 c/5aSW 367	oJF]44 1	-	stage how many get nervous so remember to remind celebrate and the second thing	Nervous Breath		SpeakerEmotion MentalPreparation	Preparation			Delivery
88			Is to read member or house that we read correctly do we have any singer in the							
370	+	Ŧ	room dkary so tell me the belly breasth you get							
372 373 OEI3rSY,	VPp.0174 49	49.2 49	where it is if it within the management of the	Communication		Interactivity				Presentation Mathula
374			Dut webhars are the hardest when they							
376			Or association would be reported use							
377 378	+	Ŧ	marketing team did a great immediate communication she did an awasome job of							
379 380 (JET2-SV	VBnO177 50	50.1 50	This field is been that is present 11 kind of	da ji ji		V(surred Ab) (brief and				
381		8	bring it back in by saying you know can	No. In Contract of						
88	t		we table that here can you tak to me afterwards about it curz I know where I 'm							
58	t	Ŧ	ar wan my decit can see my powerpoint I know what my recit slide is and they							
387 ryXOW1	OSOZM2 01	0.27 0.5	Now I tell you this because there is this huge misunderstanding	Nervous		SpeakerEmotion				Delivery
88		+	that experienced speaker they are not nervous. When you see that speaker on stage and you say like like : "Well they do it in a natural way."	Nervous		SpeakerEmotion				Delivery
390 191	Ħ	Ŧ	No they don't . They are nervous . But the difference to me between an experienced and ineccerismical speaker is that eccenterize pecole							
8	t	F	know how to handle the nerve . I want to give you this insight because it							
ryXOW1	QS0ZMM 1.	1.12 1.3		Ottor	Ignore - It was picked for the	EmphasisedText	Text	Visu al Atd Artefact		VisualAid
582 582	t		So a soom a the amema perceive a kind of danger a button is pushed . It's a button somewhere here		Seament					
306	+		In your head in your brain and it's called the amygdata. It ha to shape of an atmond the color of an atmond.							
308		₿	R is small a m almond R is a very important button		Tablet R to de ordene Manager					
300	_	_	because a soon a the button is pushed that's what we call the fight or flight reaction		I think it is about geting Nervous Ignore-Suggested new conset	10				Detivery
401 wd9reH4	a-Y109 32	32	else arrhoth want to intox ves	Freedback		AudienceDestre	AudienceEmotion			
402	5	8	absolutely involved in with you Card							
404			when would you suggless to the presenter about asking their audience to put in a							
405		+	short feedback form to their future improvement who I 'm wondering what do							
407	- 1111			Control managements			Ci trata			
400 WORKEN		26.02	TO BLE LINE REAL BAD LOW LOW BAD IGHT	extra a transferidue	Pace Pace To fix-Domain ontology (Pacing	afenfuero contrato for a la contrato de la contrato	Norverbandminuteration			
409	t	+	their pacing went and how they thought they were excreasive in body language							
411		F	I'm not sure if that answer your							
412			duestan		Critery (QuestionAnswer)					
413			sure ofter audience question with we 've							
415 wd9reH	'g-Y125 3%	36.4 37	character that 's referant to your topic	Topic		BodyComponent	StuctureComponent			Stucktre
415		F	and they want to know more they want to see what that person is like they want	Kerb asa		Preparation				Delivery
418			to know how they talk they want to							
419	T	ł	experience their behavior that is true from rehearsal something to listen to							
421										
423 MDMRH	40-1128 3 ¹	3/ 3/	so each each section each chepter ha It own color story image alongside	Secon		EmphasisedTeet	Stucknedoment	Visual Atd Artefact		Structure VisuatAid
424	+		It you here yeer welcome thank you	Slory Image		StructureApproach	Frite	Its rated inco		Stuchtre
1		F			Thanking ? To fit- Domain on totogy					
427			uence we gogreen earliered to hand there's anything else I'm going to hand							
428			back to Henry and guy rice thank you							
430 ISm4Fipt	teH810 3.	3.25 3.4	a visual aid even something on the screen could be a powerful way to start	Parat.		Providence Constraints				VisualAid
420			what you he about to say can be a really great way to start. And the last way is	Board		Visual Ad Device				VisualAd
53 53		+	quotation . Now you can start with it a well quotation but I don't think the	Start		StructureComponent StructureComponent				Stuchte
4.56			tool or other allower based		Coultation To file. Domain on tology					View of Abi
28			wei drostati are ameria pasi .							
437 ISn4Fip.	3aH811 3.	3.46 4.1	It is true the Abraham Lincoln	Impact	Orestation Proves. Surrented new concess					Presentation Attibule
438		+	and Enstein they all said some amazing thing but the problem with using a		in referent					VisualAd
440			common quosaton is that people have proceedly already near of it. And it mgin not have the impact that you 're trying to have . I think it's best if you look							
441	+	-	for a unique or off quotation . For example 1 saw a speaker recently give a quote from Buce Lee to beain his presentation . His presentation wan't							
443	Parties 0	0		Pirr.	Terrere Brown Block	0.110				
444	n interest	3	I'm Afex Lyon and you're about to start a seven mini on Concise	1980	anno e a was provou ar ann	ourcer excurbation				
445			Clear and Comdem Communication . I put this course together because I believe It's really the best first step for you in your professional development laying	Contration		Speare rtcm obon Interactivity			4	Detroit Abbute
447			that foundation to develop strong communication skill. Moving forward I encourant you to watch all seven leason.	Communication		Interactivity Pressention				Presentation Athbute Delivery
449	t	E	There is a little practice involved in each one so you can actually hear and							
451 20Bewr	rfgulk2 0.	0.24 0.5	then try the idea out. Let's jump right into the first lesson.	Communication		Interactivity				Presentation & thtp://www.marchine.com
453			In resource we regard area roug - rencourage you to wake this and gue through a many time a you need to to make this material a habit . Th one . Long	CONTRACTOR		I KGOYGAY				
455			wroeorreas is a way enemy or good communication . I m aways surpreed that really takethe people assume that they he good communication . But more							
455		-	talking doe not equal good communication . In fact a lot of time if							
458 a5K/NHI	VKQB85 1.	1.41 2	White space help you define and separate different section ; it give your content	Breath	Inneed - B sum nicked for the	MentialPreparation	Preparation			Delivery
459			room to breath.	1 MAN	seament a second	Contrast Language and A				
460	t	ŀ	If your work ever start to feet duffered or uncomfortable a liftle white space might		Section Ignore- Suggested new concept					
461	+	-	be just what the doctor ordered . Allowment is compliant way doal with all the time each if way do n't realize it		is indexant					structure
463	t	E	Whenever you type an email or create a document the text is aligned automatically.							
465 aSK/NHI	4KQB87 2.	23 26	Each grouping is also every spaced and aligned with equal margin .	Atention		Ctar of motor and				Presentation Athbute
467	Ħ	Ħ	we a trial another to use and the mark that will be a set to feel disorganized . Without consistent alignment your work could start to feel disorganized .	Larout	Increase Research Research	Clarity	Post Alater	South Comments and Advances of Adv		Presentation Attitute
468	+	+	Contrast simply mean that one item is different from another .	tye	SJIRCE - R Was purchased	Eyewonun	Bodywaron	BodyLangug enames ooylanguage worver oar ontinue work		Delivery
403	t	ŧ	In layout and composition contrast can help you do many anng : sor example usuu are	Alenson	Space, Megin, Alignment Ignore- Suggested new concept					Presentation & title use
470			reader 's eye create emphasis or call attention to something important .		is protocolarit					VisualAdds
472 a7nz551 473	9MUY2 0.	0.18 0.3	to make outline with you if you do 1 got some news for you that 's probably not	Outline		Contentintroduction	OpeningComponent	StructureComponent		Stuchtre
474	+	÷	going to be allowed instead you 're gon na		note, prepration lance- Suggested new concept					
475	t	+	need to take your speaking note with you something like this now you mank		is indexant					delivery/stau chure?
477	F	E	ask wity in the world are speaking note							
479 a7rz551	WUY3 0.	0.31 0.4	speaking note are an abbreviated	Outline		Contentin troduction	OpeningComponent Commonweat	Structure/Omponent		Stuchten
481	Ħ	Ħ	Version of your runner using and our	Outline	Assessmention - store assests	Contentintroduction	OpeningComponent	StructureComponent		Stuckre
482	_	_	podum now you math be wondering well		note, prepriation, cue cards is intervent	10				detheevbouchare?

H	¥	B C		Ш	F	Ŧ	_	Y I r	_	M N O P Q R S I	-
484		ŧ	wrty can't i luust take my outline my format outline with me welt the primary								
45LL-X	XB56UU14	4.38 5	The state of the s	Story	body lignore-Suggested new concept	StructureApproach				Shuchre	
487 488			Test internation should read and under a transmission and under an international and under the conclusion should be and under the international und	Conclusion Mosesone		ClosingComponent Conctration	StructureComponent	Statistics advisory		Sturching Startwing	
489			So I advance and this like a namative and tell the skory logically.	Narrative Story		StruchureApproach StruchureApproach				Stuchure Stuchure	
491			There ire two type of language essentially that you ire using in your writing . It is the same	e when	sequence transition lignore-Suggested new concept					structure	
494 494	XB56UU15	5.04 5.2	2 you 'ne speaking in this sort of semi stuation. There's content language: that's what you're taking about the topic ;	Topic FunctionalLanguage		BodyComponent StructureComponent	StructureComponent			Structure Structure	
405 406			and there is functional language which give u the listence the autience signal about what you're saying .								
408		ŧ	R might be thing the sectioning . You know "First I want to talk about* " The next thing I want to move on to"								
500 qFLL-X	XB56UU17	524 55	5 It could be transition: "Let me move on "or "Having said that let me move on to the ne. It might be rephrasing or clarifying.	Summary Section		ClosingComponent BodyComponent	StruchureComponent StruchureComponent			Stuckre Stuckre	
205			". Let me rephrase that "". Let me point out that "	Function allanguage	transition To fee. Domain on totogy	StructureComponent				Stucture	
305			Tress trips and the super super super treat sometring inportant is about to be said . Emphasis : pummery: coulding : moring between addim : giving example		(10000000)					ainense	
905 205			All of that type of functional language is extremely important.								
508 wp4ho 509 610	09raVjA1	0	2 not a long time app there was this famous survey which concluded that most neorons force nut in an evaluation that most	Statistic		Analysis	B ody Comp oment	StructureGomponent		Stuchure	
511 512			They fear and a service spreamenty in the service of the service o								
513 514		ŧ	billiardy when he said so for the								
615 Wp 4hot	-9mV/A4	0.65 1.1	1 atong now pay attention and take your	Attention	ignore - R was picked for the seament					Presentation Attbute	
516 517 840			first step to becoming a prosentation nhja								
0 10 0			you like with any project you can divide		dariek bes						
52			your presentation into three distinct		devide into Ignore- Suggested new concept					I think it is taking about the Structure but there is no word to explicitly indicate that.	
522 wp4hot	9 aV/A6	1.34 1.5	5 about your listener their bias and their expectation address your topic	Topic Objective		BodyComponent Contentin troduction	StructureComponent OpeningComponent	StuchureComponent		Stuchure	
524 525			from the audience is perspective bientify the presentation objective the next	Outline		Contentin troduction	OpeningComponent	StruchureComponent		Stuchre	
525 527		+	step is to outline your presentation so that you get a clear overview of it								
528 taiMOv	v.JmZk5	1.32 1.5	5 think about that and apply that to your	Message		Conclusion	ClosingComp oner/t	StructureComponent		Stucture	
123			presentation your public spreaming and even go bit deeper think about the time								
200			you 'w been in an autoen oo where you have pushed back against the presenter on the memory and the presenter								
505	1 martine		ou area rinessago izon artik or wiry area	An anna ratantina.		ta matta.					
537 537	POILER.	41.7	 speakering and from provide and from a work 	CIIIIIIII		Lancard Lancard					
540			anytime we need to presuade some one to take and/on think of this exercise you								1.1
542			just went through and again when you								
543 XqOQ	58-strg2	0.13 0.3	.3 to the computer and drill down on this wetcome back friend in this video we he	Computer Video		VsualAdDerice Film	Movingimage	Illustration Visual Add Arted act		Visual Aid Visual Aid	
545			going to talk about how to organize the	Approach	organize, prepration Ignore- Suggested new concept	BodyComponent	StructureComp onent			Shuckire	
546			main point of a presentiation we 'II talk about several approach and we 'II drill		is referant					delivery, presentation Ratio Lie?	
548			down a little deeper on the topical								-
561 XqOC	084603	03	4 approach which is often the most tricky for people we're assuming here you have	Aproach Introduction		BodyComponent OperingComponent	StuctureComponent StructureComponent			Stuchre Stuchre enternet	
			at its output tooly allo contributi to		Main point To file- Domain ontology	100000000	lisindingamone			Condition of Condition	
200 1930			your presentation and we recently secure to a contraction to the present point of the contraction much most patience about							ancox	
225 556	11 mar	0.04	1 the presentation we remove and groups							Strukto	
558 569		5	International and the second structure of the second s								
260			structure vour notes that make a second state and the second state of the second state		structure lgmore-Suggested new concept						
158			get to it.								1.1
563 564 bMNN-ro	nczo/12	0.16 0.3	3 about visuals and book language we can	Body, and use oSignal		Bod M, and un enameb odvian guage	Non Verbal Communication			Definence	
565 505			now get into the presentation itself where do you start regardless of	Start PresentationScel		StructureComponent ContentIntroduction	OpeningComponent	StructureComponent		Stuchre	
567 568			different presentation goal there 's one thing everyone love a good story I 'm	Story		StructureApproach				Stuckine Stuckine	
888 220			going to show you how you can structure								-
571 RVPW 572	V9MMPO3Y2	2 0.15 0.3	3 some of the secret here with the three core quote unpuote secret 11 leach	ଥାପନ		Visu al Aid Artefact				VisualAid	
574 574 676			you have and the whole goal is to anoid presentation like this if you 've ever how any and the second provide the second provide the								
577		ŧ	presenting a slide like this you know								-
578 RVPM	9MMPO3Y3	3 0.3 0.5	5 how painful it is to be in the audience left's, avoid that so what moke M-Kinsov	Information		BodyComponent	StuctureComponent			Stuchre	
580			presentation of the or the contract model of the presentation of the ordination of the ordinatio of th								
582			that the presentation is easy to digest		undrestandability To fits- Domain ontology (undrestandability)					presentation Attribute	
585 765			even though it ha complex information								
585 4prTh2 596	3SnLyi2	0.15 0.3	3 however one of the key fast of an effective presentation is it	a sea		,	0			Structure Structure	
888			servicence algoco servicence respirator referentativo referentativo fencilo and avvid anvia autoreard	Silence		Paralanguage	NonVerbalCommunication			Selvery Delivery	
500		ŧ	sierce								1 100
592 4prTh2 593	3SnLyl3	0.28 0.4	4 during the presentation atypical presentation atruature core:	Sector		BodyComponent	StructureComponent			Stuckure Stuckure	
505		+	all the Well Section and is a groot		start To file. Domain ontology (Starting Point)					ditruck (6	_
200 200		+	I allows your audience to easily follow you during the presentation								
<u>508</u> 599 d2 Xcm	rodE1A1	0 0.1	1 hello i am Ram Gopal from presentation	Dagam		Figure	Illustration	Visual Attl Antel act		Visuativid	
80 100 100			process calm in this edition of PowerPoint dagram turing series you								
803 203		⋕	will learn how to create brick wait teature in PowerPoint the brick wait								
604 605			teriture you will learn to create look					+			-

V	B C D	в	5	Ξ	-	~	×	L M	N O P Q R S T
607 012 X000E1A3 U.	24 U2 to a new stde the mist stop is to go to auto ship menu and under rectangle	2008		VISU BY AND MORE BOX				Divensio	
100	pick up this barac rectangle shape and draw the size of a brick so this tis								
610	the brick that you are going to replicate now select that one and while								
613 7sLXTR9AvKg3 01	34 0.6 presentation in this video we 11 talk about the ord measuration common and of	Video		Fim	Moring Imag e	Illustration	Visu al Ald A tefact	WsualAid	
615	the PG online you can expect to have		CinActosing component Ignore-Suggested new concept						
616 617	about 40 to 60 minute for your PG online presentation including time for		8 freierant					65	Ruchure
618 619 820 7-1 VTD0A.4/-0	question and answer you can use visual	Cont		Dia tah no ta				Obsident	
620 7sLX IRSAM606 1.	19 1.3 all the detail in your written research proposal insis ad of giving a close view	Close Focus		StuctureComponent Paralanguage	Non Verbal Communication			Delvery	
223	of all the detail focus on the big picture the heading in the written PG	Heading	oggerpoure, Keytoea	Text	Visual Ald Artertect	De Jour Dry en SA		S proteinsiv	Auchure
102	contine proposal can guide the structure of vour presentation of these heading	Guide		Contentin troduction	OpeningComponent	StructureComponent		Stuchure	
627	Busered action in a second and and	Heading		Text	Visual Aid Artefact			Distance	
628 U/w_21AuU5k1 0 629	0.1 Everybody torve a good story and story are a great way to open a presentation.	Story Story		StructureApproach StructureApproach				Stuchure Stuchure	
00	That attention is a key moment	Atention	Open, Beginning To file. Domain ontology					Presentation Attbute	
122	in every presentation . Alot of people tell me though I'm just not		(Starsing Point)						Auchre
634 11w 214x415v2 0.1	. a good sam yearer .	Communication	Innere - it was noticed for the	Interactivity.				Proceed tables of the line	
	Well 1'm going to show you a very simple recipe that 11 help you morever.		10000 to a the a set of a	f waterwater a				Presentation & the last	communication Coach is different from Communication concopt in presentation skills ontology?
637	Helio I'm Aex Lyon and this is Communication Coach . Wai'm here to help you increases your increase.								
639	to lead your to am to highly favel of excellence								
641 dXMRPMcxxGU1 0 642	0.2 my next speaker need no introduction unless you are david letterman yes your guest need that introduction more often than not guest speaker	Introduction Introduction		Opering Component Opering Component	StruchureComponent StruchureComponent			Stuchure Stuchure	
643	infroduction mas the mark a they are riddled with basic blo	Introduction	SpeakerIntroduction Ignore-Suggrated new concept	Opening Component	StuctureComponent			Structure	
044 045 046	point lack emplored intensity and sometimes people just say the speaker's name and fait to care leary momentum withorever interview and an univer-		B ITOGOVATI					\$	tucture
647 648 dXMRPWcxvGU2 0.2	34 [0.4] four formula to deliver a memorable speaker introduction that will	Memorable		hpad				Presentation Attbute	
649 650	captivate your audience hey i'm Raz Maghji founder of the every conversation count platform	Speaker http://www.com/sterce Confidence		OperingComponent SpeakerEmotion	StructureComponent			Structure Delivery	
651	this is the channel that help you present with confidence materize your impact and connect authentically with your audience	Impact						Presentation Attitute	
653	on camera and on stage if you he new here subscribe to the channel and hit								
655 KVbRQ0cE00k1 0.1 656	03 0.2 you have to also use a visual aid do n't panici l'm here to help but first thing							VisualAid	
259	fret where is a visual and a visual and is a first where is a distribution of the pair would be distri	Speech		LanguageElement	VerbalCommunication			Defivery	
050	audience to understand yours peer your identifies the second second your								
661 662 KV/RC00-F00/k2 0.2	5 10.4 whet we are serving it should enhance	Messane		Conclusion	ClosinoCommonent	StruchureComponent		Shuchire	
663 964	and help your audience to understand			-	Abricelmena	Illustration	View of Airl Articlesce	Visual Aid	
500	PowerPoint presentation a video cito a	Handout		VisualAtdAted	o Pour a Pour o Auto			Distribution	
200	prop that you bring to class for most							32000	
600 WHH8Gx0u9ls1 0.	17 0.3 welcome to my you/ube channel PowerPoint	Diagram	Proved Point To fer Demain ontotoory	Figure	Illustration	Visu al Aid Artefact		VisuaNid	
670	University in this tutorial I am going to show you a process flow diagram in		(PowePohl)					~	tauat Act
672	PowerPoint which you can do it in le than five minute you can see here 1								
674 675	have just created the symbol for the								
676 WHH8Gx049bs2 0.	31 0.4 start then the decision and then after ated process three process process 1	Start End		StructureComponent StructureComponent				Stuchre Stuchre	
678	process to a process 3 and based on the	Dagram	PowerPaht To fike Domain ontology	Figure	Illustation	Visu al Aid Attel act		Pitransia	
679	process the action will be taken and the end ao let's see how to create this		(PowerPoint)					~	1sust-Add
681	process flow diagram in PowerPoint in								
683 k4dEKe6OMMw1 0 684	0.1 ready to take your infographics to a completely different level	Infographics Animation		Diagram Movingimage	Figure Illustration	Illustration Visual Ald Artefact	Visual Atd Attel act	VisualAid VisualAid	
685	animation can take you there and visme can help	Infographics		Diagram	Figure	Illustration	Visu al Atd Attefact	MsusMdd	
2880	on this very youtube channel we 've discussed building infographics for								
680 k4dEKe60MMw2 0.1	16 0.3 quite some time now i could read off	Infographics Construction	Total Connect Total	Diagram	Figure	Illustration	Visu al Aid Artefact	Naura	
601	reason after reason after maxim a to upty unit should be	900	Agortem	on possible system	coupr costor	afenfueringenersteftening		finance	
603	building infographies to display your information if you 've								
605	been listering and following suit								
697 Q4SCn6J4ako1 0	0.3 to day we're gong to tak about how you can most effectiwny fulld visual noe	Video		Fim	Morking Imagie	Illustration	VsuaAdAtefact	Wara Ad	
700	Intro a presentation trans you re gwing lef 's sketch it out welcome to purple to								
202	ysuaryoo i am your rost upon year and this question of how to build in								
704 O4SCn6J4ako2 0.	22 1 more visuals into a talk that you 're ghing is one of the queetion that !	Visuals		Visu al Aid Artefact StructureComponent				Visue/Aid Struchre	
706	most frequently get asked so let 's jump		take away keykilea To fite. Domain on talogy					07	Ruchre
705	right into it first up 1 think it is important to start with the takesway in								
208	mind and then build your presentation								
17alhDQ1k8s2 0: 211	22 0.3 of presentation	Objective	Ignore - R was picked for the seamont	Contentintroduction	OpeningComponent	StructureComponent		Shuchure	
712	to help uto that for totality we the going so go through some bibliches	F eeo tack		Audienceuesre	Audienceemoson			Detryoly	
715	the rate of the second on the rate of the second on the second of the se								
717 718 17ahDQ1k8s3 0.3	35 0.4 we've going to also learn about how we	Story		StructureApproach				Stuchre	
22	Use presentance a an act to tell our Own story								
722	and teapy we it go orver some key technical presentation all								
726 dShvhZ9oZ02 0.2	38 Q.66 Visual aid but Oh tell me carry can							DIA RULEI	
727	about the first visual ad we tre not going to be talking about the first visual ad we tre going to be							DIARUEV DARUEV	
728	about even meuer au ver regung vuuv							VisualAid	

Store.	<	B C D	Ш	F	т	-	ſ	K	M N O P Q R S T
82	t	according to research with visual aid							
ulsp of	yhZ9pZ03 1	12	Video	To the Semantic Tagging	Flm	MovingImage	Illustration	Visual Ald Artefact	VisuaRaid
202		you can say someting without even seeing you know so today in this video l	and a second	MORE AND A M	1. second fill second	Math 40 months in the factor			Naura Al
52		am gong to fell you about veuer ad you can use in your speech firsty	639.92%		Languagestemen	VerbalCommunication			VisualAid
737		know what a visual aid is now visual aid are thing or object that you use							
736	MURACA1 0		Viteo	To fie. Service Description	Ellin	Mexicolmene	Illustration	Visual Airl Arteliant	Variative
730		helio and congratulation on getting	1000	Algorithm		o Berri Brinom			
/40		access to this video course	Engaging Video	To fix- Semantic Tagging	Fim	Moving Image	Illustration	Visu at Atd Arter act	VisuaAid
741 742		on how to create engaging presentation so this video course is all about preci	Prezi	Agentitum	Side	Visual Aid Artefact			VsueAid
743		which is (the software pisiform that enables you							
745 746 EQT3	10 000	10 10.0	Encenter		Indiana-thild v				Descentibility its
747		engaging presentation now why did we	Engeding		Interactivity				Presentation Attibute
748		choose this over powerpoint well i'll talk about that in just a minute a	Video		Fim	Moving Image	litustration	Visu al Atd Artefact	VisualAid
761		you know							
752									
754 KW -	Yn Lowcznion	U.U. U.S. If you really wark to accession exclorement to your presentation if y access an marcon PowerPoint has a wide variety of animated effect that can be applied to text and other years.	Avimation		Spleane nam of on Mo ving image	Illustration	Visual Atd Attel act		Distant A
755		object. In this exempted the creation areas basis ouis that have current on each stilde	Teat		Visu al Aid Artef act Visu of Airl Artef act				VisiterAdd
757		I'd like to add armston to these two text box and also the man on top of the moon.	America		MoringImage	Illustration	Visual Atd Attellact		Vauadarea
8		I U BRI SARING SREACH WIR IIISH U UNOCH JOU WARK JOU WARK 10 BRITHING WRITIG U U UNO	10001		total and an inclusion				
760 kw	ud/D2 wvo2 0.2	0.37 0.6 Animations tab .	Side	An imation To file-Domain on totopy	Visu al Ald Artel act				Visuation
761	+	Next click the drop down arrow to see all the available effect. There are four different or borows - Extension which will be more when the object		(Animators)					VisualAdds
282		first appears							
765		Emphasis which happens while the object is on the slide Exit which will cause the object to disappear							
705	A About Att	111 [0.1] mode non-ten transfer and the model tests of	MedDee		View of Abel Developed				
768		highly realistic presentation drawing	Drawing		Dagram	Figure	Illustration	Visu al Aid Artefact	VisualAd
1001		of product but here we re using mem for quickly adding information to our	Product Information		BodyComponent	Visua Ad Attelact StructureComponent			Stuchte
771		design idea to help convey our thought							
1									
776 MdL	MLMzudbM2 0.1	0.19 0.4 marker pen come in a big range of color but they also come in a ranow of	Marker Oder		Visual Aid Device From breated Text	Text	Visu of Airl Artic fact		VisualAvd
776		tone here we're looking at a range of	Tone		Paralanguage	Non Verb al Communication			Delivery
7778		gray tone and you can see that with each pen the crav is a tone darker	Tone		Paralanguage Paralanguage	NonVerbalCommunication NonVerbalCommunication			Delivery
URL L		A hard a second s	Marker	MarkerPen Ignore it was picked for the	VsualAdDevice				VisualAid
780				N I SALIN POR					
TB1 KBX	Ozq00pS01 0	0 0.2 default and with a hiltory num an electron	Start	Ignore - It was picked for the	StructureComponent				Stuchine
0		UNARY SKAN WALL & DIAR N. PIC SCHRAUM	Start	Ignore - it was picked for the	StructureComponent				Stuchne
782	+	that 's going to open up and give you		Contraction and loss and los					
783		this automatic setup you do n't want this		CONTRATTOR DISTORT					delivery
784		so dick on each of these box and							
286	t	acreact ment you want to spart morn screach most up insert							
787 788 KBX	Ozo00aS02 0.2	122 0.4 vour video deaviso vou anactiv	Victeo		Fim	Moving Image	Illustation	Visual Att Arteriact	Progensy
692		recorded your video you 're gaing to go	Video		Fim	MovingImage	Illustration	Visual Atd Attef act	VisualAid
791		and get your voteo irom your computer here 's mine I'm going to insert it you	Video Computer		VisualAdDevice	MOMP3 Image	llustration	VeueAdAteed	Division of the second se
792		can put it anywhere you want most people							
704									
795 [201)	4 ZnqS81 0.0	3.05 0.2 Hello I'm Calife. What is the fastest wer to fell the difference between a speech and a preservation ?	Speech Audo		L ang uag eElem ent Visu at Aid Artefact	VerbalCommunication			Delivery
262		a kaudo			Construction of the second second				VisualAid
802		Thanks for visiting Toastmasters international .	Contacto		Speakertmotion				CONOL
800		Since 1924 we 've helped more than 4 million people gain the confidence to communicate							
20ft	L_ZnqS82 0.2	127 0.5	Video	To fix- Semantic Tagging	Fim	MovingImage	Illustration	Visual Ald Artefact	VisualAid
803		In all circumstance . One of the most demanding situation you'll ever face is speaking to an audience .	Confident	Alocritism	SpeakerEmotion				Delient
i			Video	To fix- Semantic Tagging	Flm	MoringImage	Illustration	VsualAtdAtefact	Visuation
805	-	Intel video is part o tal agrices of iloastimasters - time (b) triat will neep you become a more confidenti speaker -		MODULE					VisualAid
805		In this video we'll explore simple and effective way to make the most of visual aid . I wi's benin with a few stimle way to and face to your neet exercit by incomposition	Speech		LanguageElement	VerbalCommunication			Delivery
908									
810 810	50 555 but 4	U.00 1.1.2 Visua ad . When presenting for business the most commonly tool are a computer and projector and	Computer		VsualAdDevice				Defensiv Defensiv
811		the help of presentation software image can be projected in sequence on a large screen	Project or	Screen lignore- Suggested new concept is intervant	VisualAdDevice				VisualAid
812	+	that even people in the back of the room can easily see . Additionally preceded in the back of the room can easily see .	Image		Picture	Figure	Illustration	Visual Ald Artefact	Visual Aid
814		however keep in mind that when you play a video that ha audio you 'Il need to have a	Video		Fim	Moving Image	Illustration	Visua Att Artefact	Astrony of the second s
816	+		0000		the second provide instal				DATES
818 4cr/m	ofPHO6SA3 0.3	1.34 0.5 million text all of this data is	Information		BodyComponent	StuctureComponent			Stuckre
819		created no not just in a day in 60							
821		second which is writy we want to let une data scientist decipher this							
823 823	+	never feed of information but it is important for all of u to know how							
824 ent	a O a Magnetice	17 0.0 to reconstrain the indemonstrate is a	India ana odia a		Bo di Como nome 1	C fas s of wear energy energy			Ohishaa
202 202 202	mP*K26544 0.4	U.4. U.6. to present this information in a visually appreading way that's why I hm	Lifo mation		BodyComponent	StuctureComponent			Stuckue
828	+	here hello world mry name is Mike pluziger here							
828		with viz me and I im here to show you how							
830	_	you can create data that appeal to your		VeualAds?					
831 620	Theorem 100	.01 [0.2] Destroyed off-softward and rise to the model to mode according to the softward and to							
833	-	enhance your presentation	Prove Control of Contr		- 1900	Alter of Alter Aster Sect.			District of the second s
834		after you have planned your content it is time to think about how you will present	Prezi	PowerPoint? To fite-Domain ontology	Side	Visu al Ald Arte fact			VisuaRid
835 836	+	It first decide which software to use		(PoweePoint)					
837		most student use PowerPoint but there are other option such a preat							
S3h	T0yimfM2 0.5	125 0.4 and writener consider that	Video	To file: Semantic Tagging	Fim	MoringImage	Illustration	Visu al Atd Arteriact	VisuelAid
840		this video us PowerPoint but the same idea can be used in prezis or other	Start		StructureComponent	Tak	View of Air Annual and		Stutchre Vienderie
8		1 do o comerce	45L	Prezi To fit- Domain ontology (Prezis)	In the second second	NDI .	VBUGWARVOIGU		
842	+	OK when it come to designing your Power-oun		PowerPoint To fix- Domain ontidogy					Visit at Adds
843	+	there are some main point to keep in mind you should start with choosing a background and fort		(Howebourt)					VisuaAtts
845		The second many second is foreigned as the second many many second in the second							

ALC: LEVE	A 1	B C	D	E Kuuduta a	1	9	H Boch@mmonuet	S for other and conservations	×	_	M N O P Q R S T
4 according 2 P		5	impactful is to use some key idea and	Graphou		foreca. It uses without for the	Diagram	Figure	If Lust ration Visu al Act Arted act		Visuation of Visua
348	+	-	principle from the realm of graphic desirin free line's earer with a hintradi	Slich		seament and seament	Visual AM Andreas				Verstähl
22	+	F	slide what do you think and pethaps more importantly what will your audience take								
852 RJOp4FPA	pd42 0.1	0.19 0.3	yes that 's probably a pretty common form	Teet	e L	To be breach and the	Visual Aid Atefact				VisualAid
854 845			but not specifically what 's wrong with is othere than the otherwork		purpo doctera	10 file- Lotmain on totogy (Background) To file, Domain orderoory (Buillian)					Visual Adds New of Advis
958			well the solid while background can be really hash most of the three here 's		5						
858			copious amount of text in the builtet								
960 VO/7ZON	IKz6A3 0.4	0.55 1.1	of expertise any other specialist knowledge do they know you or each other	Makeral Key/Message			Prop Conclusion	Visual Aid Arte fact Closing Component	Structure/Omponent		Visuelad Structure
202 503			this will help determine the material can apply and the amount of detail you can apply hypersemiation number two								
500			one key message there should only be one								
967 VO/7Z/NP	IK26A4 1.1	1.16 1.3	key message in your presentation which act like the spinal cord to the	KeyMessage Summarize			Conclusion ClosingComponent	ClosingComponent StuckreComponent	StructureComponent		Struchure Struchure
800			presentation and hold everything else together summarize this in one powerful								
128	╞	+	sentence and keep it foremost in your mind throughout the presentation number								
873 874 P4z4VqC	: KCY2 0.1	0.13 0.3	they believe	Information			BodyComponent	StuctureComponent			Stucture
876			there ha to be a power point if you		PoweePoint	Ignore- Suggested new concept is indexant					VisuatAdis
2/2	╢		Unrik that is presentation you are wrong presentation mean								
828	t	ŧ	any idea any intomation that is there in your mind when you want to share it								
881 P4zdVqC	: KCY6 1.0	1.08 1.2	attact the audience in order to grab	Atertion							Presentation Attbute
200			the attention of the audience through different aid are just								
88			acting a assistance to your presentation								
500			so if somebody say that you should be								
888 yGQrYH90 889	axose1 0	0	helio and welcome to this video on presentation skill from the Department	Story			Structure/kproach				Stucture
891			of Engineering Ubrary University of Cambridge in this video we will cover								
255			the following point telling your story		prepration/organisation	Ignore- Suggested new concept					
100	1	-	managing your time planning presenting			is indevant					Delivery present all on Attribute
205 yGQrYH93	3XOSE2 0.1	0.15 0.3	with style presenting with confidence dealers with or assists work to leave to	Confidence			SpeakerEmotion EnvolvesisedText	Tax	View of the fact and		Delivery
768			avoid this how many of you have seen something the main of you have seen								
008	H		problem is that it's too cluttered it's		Childrand Condenard Carety	Innora. Surranted new concent					
8		Ŧ	difficult to read and too many ordor		Curvered, Crendar, Carry	gine ouges and ment					Presentation/wrbutes
902 Uk FiRshy	yp11 0	0 02	In this lecture we 're looking at								VisualAid
30	t		regregating installer and and the multimetric two thing : first focusing the audience 's	A ten ton							Presentation Attibute
8	Ħ	Ħ	stating the visual 's general meaning. When	Mandana		former former altered for the		MarketOrmania			
200		+	you are presenting a visual aid to the	Parameter		segment	rain and acrossing to				(Amarking)
				And the second se							
911 W TRNS1	7007	3	automote and you want untern to look as a to pay attitemtion to it there are some	Expension Expression			SpeakerAura				
913			way expression train, you can use . Probably the most common is this :	Chart			Figure	Illustration	Visu al Atd Artef act		Nsurery Visual Visua
914			Ok tet 's take a took at this slide ; " or " Ok		FunctionalLanguate	ignore- Suggested new concept is indevant					Delivery
010			let's take a look at this chart.								
912 up 3PKm/	UNUSIT 0	5	In quash Milington here is speaker and speech coach and today we 're going to	Speech			L ang uag eElement	VerbalCommunication			Delivery Visual Aid
800			talk about how to use visual aid effectively in the presentation when it								VisualAdd
12 (2)	t	ŧ	come to using visual aid in a								
223 024 up 2PK mM	MU812 0.1:	0.15 0.3	ance heard and said that audience will								VisualAid
505	+	+	forgive you for anything during a		b orting (Sp eaker Emotion	Ignore- Suggested new concept					
200			presentation except being boing and you 've probably seen it a boing			is indexant					detheey
800	+	+	presentation that had visual aid and there are so many mistake that are made								
930 931 JF m8ANc	kouM1 0.1:	0.13 0.5	hello everyone welcome back to n pt el mock course on developing solt skill and person	Visuals			Visual Aid Artefact				VisuaRAd
225			this is week eight module number four lecture number forty six i am doctor rakchandran from the department of humanity and social science of I I i Kanpur giving you this course	VerbalCommunication			Visual Aid Atel act				VisualAd Dativery
10			now for the past eight week this is the concluding week and then i started with presentation skill and this module this lecture will conclude our discussion on presentation skill giving	-							
200			you more input on now to use visuals valuate is an aspect of non verbal communication as								
909 JF m84/4	KooM2 05	028 13	with body language we also need to know how you can use this effectively before i start let take a quick review of what I do in the previous led ure in the previous lecture i	Bod y, ang uag eSign at Start			Body, angugenameb odylanguage StructureComponent	NonVerbalCommunication			Delivery Delivery
941			an addressed the issue of body language in public speaking I tod you that the role of body language is so crucial that some people use very effective body language and get at	booy, ang ung esign a Body, ang ung esign a			Body, angugenameb odylanguage Body, angugenameb odylanguage	NonVerbalCommunication NonVerbalCommunication			Delivery
943	t	ŧ	with a content when a source and the powerful body language to convince the audience but	Body, ang uage Signal			Body, angugenamebodylan guage	NonVerbalCommunication			Delivery
945 Ineja7-Mon	r02 0.1	0.12 0.2	advantage of using presenter view the	Slide			Visual Atd Attel act				VisuaRid
247			you get now alongside your slide so lef's sav that you have a bid		note	To fix- Damain antology (Note)					detweevisituichure?
949			presentation coming up at work or maybe you 're a teacher presenting to								
951 952 hraia7-Mord	103 0.2	0.23 0.3	vour dass and vou want to make sure	8 8			Visual Atd Attellact				VisiteAd
363 363			that you hit certain point with the note you can make sure you								
956 956			correr all the point that you want to correr								
355			on that slide also you see what's coming								
000 MISO001	0	0	so let's do those and then after you	Excitement		Ignore - It was picked for the seament	SpeakerEmotion				Definery
	+	Ħ	know you can just betreve mer re kind of with your name duck in exclement								
200			meers & what is going to respect to any i will talk to you about presentation aid								
905 006		0.04	and aren't will take to you oneny								Christian
100 Lange	3	70 400	You have some question area arrow.								anone
889	+	+	audence about any other ming matt						-		

e A	G and the second se		2	- T	7	×	z	2 2	0
	presentation please ask me and of								_
H									_
		345	71						
			Summary.						
			Culturer securits the raid						
			7 50/ 136 concents) of the 1346 concents) are uncondized						
			picked						
			2% (7 concepts) of the concepts wasn't sure about						
			71 new concept suggested						
			New concepts suggested by Annotator						
		To five secondity Transies A leadition	±0 ⊔						
		10 TK - Semantic Lagging Agorium	0						
		ignore - Jug esteutiew concept is interevant. Innora - it was nicked for the commant	0+						
		Ignore - It was preveation the segment	100						
		-0181	700						
									-

Appendix I

Google Forms for VISC-L Evaluation with experts (First Domain)

Two Google Forms designed and shared with domain expert to evaluate VISC-L by comparing it with the Google Segmentation and Characterisation within the first domain (Giving Pitch Presentation) **First Domain- Giving Pitch Presentation**

Evaluate VISC-L in comparable with Google video segmentation and characterisation.

Here, we have included two forms. In the first form we have include 4 videos characterised via Google. Then, we include other 4 videos characterised via VISC-L.



Section 2 of 12

Consent Form

INFORMATION

You are invited to take part in a study on using videos for learning soft skills. The aim of the study is to assess the learning effect and the usefulness of video-segment characterisation and linking to support video-based learning. We have chosen a set of video segments aimed at improving presentation skills.

Your involvement in this study will include the following steps:

1. Read and accept the consent form.

2. Complete a short pre-study section about your profile.

 Watch several suggested videos with recommended video-segments and their short description related to presentation skills generated using the first computer algorithm.

4. Give your feedback on the video-segments you have watched and their description.

5. Give short video summary.

 Answer questions providing feedback on the usability and usefulness of recommended video-segments for learning about giving presentations.

7. Repeat the steps from 3-6 with segments from the second computer algorithm.

On completion, you will be entered into a prize draw for one of six £50 Amazon vouchers.

Participation in this study is voluntary and fully anonymous. The study will be conducted using the Google Forms Survey which is fully compliant with all current UK data protection laws. All data that we collect about you during the course of the research will be anonymous (no names, no email addresses, no personal data will be collected). All responses will be saved on the University of Leeds secured OneDrive, which will be accessed only by the members of the research team. The results will be published in conference papers and will be part of PhD research. Your personal response will not be recognised as they will be part of the group of responses which will be analysed together. However, the name of the organisations that we ask to help us conduct the survey might be mentioned.

:

X

You are under no obligation to take part in the study. Once you start the online study you may withdraw at any point provided you have not clicked on the 'Submit' button at the end. If you withdraw, your responses will not be used for analysis or for any other purposes. You do not need to give a reason for discontinuing with the survey. Once you click on 'Submit' it will not be possible to withdraw your responses because they are anonymised.

There will be no risks to you regarding your participation in this survey as no personal data will be collected and all your data will be fully anonymised and accessed only by the research group - PhD researcher Abrar Mohammed and the supervisor Prof. Vania Dimitrova. If you wish to enter the prize draw, you will have to leave your email. It will be deleted after choosing the winner and will not be part of your responses on the study.

This evaluation study is part of research carried out at the University of Leeds by PhD researcher Abrar Mohammed (<u>a.mohammed1@leeds.ac.uk</u>) supervised by Prof. Vania Dimitrova (<u>v.g.dimitrova@leeds.ac.uk</u>). They will be pleased to discuss any concerns or feedback you may have about participation in the study.

Thank you for your time. We hope you will join the survey and will enjoy doing it.

Best wishes, Abrar Mohammed and Vania Dimitrova School of Computing University of Leeds

CONSENT

I have read and understood the information about the evaluation of research on characterising video-segments for learning based on user interaction and video content, conducted at the School of Computing, University of Leeds, UK. On this basis I agree to participate in this online survey, and I consent to publication of the results of the evaluation with the understanding that confidentiality will be preserved. I also understand that I may at any time withdraw from the online survey, including withdrawal of any information I have provided. I note that the survey has been reviewed and approved by the Research Ethics Committee of the Faculty of Engineering and Physical Sciences , University of Leeds.

Please tick Proceed to carry on with the survey *

Proceed

After section 2 Continue to next section

-

Section 3 of 12		
Your Profile	×	:
Note that all your answers are kept in an anonymous way.		
Occupation *		
O Student: Undergraduate		
O Student: Master		
O Student: PhD		
O University Staff		
O Industry		
Other		
Subject area or job sector *		
Short answer text		

What is your gen	der *							
Female								
Male								
Prefer not to s	say							
Other								
What is your age	?*							
	18-23	24-29	30-35	36-41	42-47	48-53	54+	
Your choice								
Is English your n	ative langua	ge? *						
O Yes								
○ No								
How often do you watch Youtube videos? *								
	Nev	er (Occasionally	Once a month	Every	week	Every day	
Your answer]						
	 What is your gen Female Male Prefer not to a Other What is your age Your choice Is English your n Yes No How often do yo Your answer 	What is your gender * Female Male Prefer not to say Other What is your age? * 18-23 Your choice Is English your native langua Yes No How often do you watch You New Your answer	What is your gender * Female Male Prefer not to say Other What is your age? * Your choice Never No	What is your gender * Female Male Prefer not to say Other What is your age? * Your choice Senglish your native language? * Yes No	What is your gender * Female Male Prefer not to say Other What is your age? * 18-23 24-29 30-35 36-41 Your choice 18-23 24-29 30-35 36-41 Your choice 18-23 24-29 30-35 36-41 Your choice 19-24-29 9 Your c	What is your gender * Female Male Prefer not to say Other What is your age? * 18-23 24-29 30-35 36-41 42-47 Your choice Is English your native language? * Yes No How often do you watch Youtube videos? * Never Occasionally Once a month Every Your answer	What is your gender * Female Male Prefer not to say Other What is your age? * 18-23 24-29 30-35 36-41 42-47 48-53 Your choice 1 1 1 1 1 Is English your native language? * Yes How often do you watch Youtube videos? * Your answer 	What is your gender * Female Male Prefer not to say Other What is your age? * 18-23 24-29 30-35 36-41 42-47 48-53 54+ Your choice Image:

How often do you watch videos for learning? *									
	Never	Occasionally	Once a month	Every week	Every day				
Your answer									
How much training	g have you had on	presentation sk	ills? *						
	Not at all	Some training	Quite a bit	A lot	Extensive traini				
Your answer									
Select the type(s)	of training about p	presenting you h	ave had: *						
Training at high	school								
Training at university									
Practice with feedback									
Professional development training									
Other									
Please indicate your experience in giving presentations: *									
	Not experienced	A little	Medium level	Experienced	Highly experie				
Your answer									

Please specify the type of presentations you have given: *
Project presentation
Course work presentation
Seminar
Conference presentation
Pitching an idea
Outreach presentation
Presentation for a general audience
Other
After section 3 Go to section 4 (List terms that reing presentations) 🚽

Section 4 of 12

List terms that relate to giving presentations

Description (optional)

Write all words/phrases (separated by commas) that you can think of which associate with STRUCTURE (as related to giving presentations):

Long answer text

Write all words/phrases (separated by commas) that you can think of which associate with DELIVERY (as related to giving presentations):

Long answer text

Write all words/phrases (separated by commas) that you can think of which associate with VISUAL AIDS (as related to giving presentations):

Long answer text

Write all words/phrases (separated by commas) that you can think of which associate with PRESENTATION CHARACTERISTICS (as related to giving presentations):

Long answer text

After section 4 Continue to next section

Section 5 of 12

In this Section you will watch 4 videos about presentation skills. Please pay attention to the terms related to giving presentations - Structure, Delivery, Visual Aids and Presentation Characteristics.

÷

Description (optional)

You will also be provided with a list of interesting segments in the video which were identified by a computer algorithm. For each segment, you will see the start as well as a short description what information is covered. The segments and the short description are aimed to assist with your video watching experience by showing main points related to presentation skills.

Description (optional)

Video 1: 5 Tips for delivering a great online presentation



Recommended segments to watch with a short description:

From 0.30 Know the setting you' re in From 1.13 How quiet is the area? From 1.29 LIGHTING From 1.53 Stand up and recite your speech with... From 2.40 Raise your laptop From 3.11 Rehearse From 3.30 Dress to impress From 4.26 CONSIDER YOUR SUPPORTI.... From 4.59 HAVE A BACKUP Please give your feedback on the recommended segments for learning about giving presentations (you can select several options).

	This video segment was USEFUL	This video segment was NOT USEFUL	The short DESCRIPTION of the video segment was HELPFUL	The short DESCRIPTION of the video segment was UNHELPFUL	I DID NOT WATCH this video segment
From 0.30 Know the setting you' re in					
From 1.13 How quiet is the area?					
From 1.29 LIGHTING					
From 1.53 Stand up and recite your speech with					
From 2.40 Raise your laptop					
From 3.11 Rehearse					
From 3.30 Dress to impress					
From 4.26 CONSIDER YOUR SUPPORTI					
From 4.59 HAVE A BACKUP					


From 0.28 Why are Speaking Notes... From 1.15 How to prepare speaking... From 1.43 What is a Quotation? From 2.18 Speaking Note Examples From 2.58 Note Card Size and Spacing From 3.29 Use a few note cards From 3.48 Why you should have only a few ... From 3.58 Note card layouts From 4.06 Use Large Font From 4.41 Using Speaking Notes with ...

	This video segment was USEFUL	This video segment was NOT USEFUL	The short DESCRIPTION of the video segment was HELPFUL	The short DESCRIPTION of the video segment was UNHELPFUL	I DID NOT WATCH this video segment
From 0.28 Why are Speaking Notes					
From 1.15 How to prepare speaking					
From 1.43 What is a Quotation?					
From 2.18 Speaking Note Examples					
From 2.58 Note Card Size and Spacing					
From 3.29 Use a few note cards					
From 3.48 Why you should have only a few					
From 3.58 Note card layouts					

Use Large Font					
From 4.41 Using Speaking Notes with					
SUMMARY: Lis video, focus on CHARACTERIS Your answer	t main point : DELIVERY, TICS.	s about giving STRUCTURE,	g presentations VISUAL AIDS, I	that you picke PRESENTATION	d in this N
Video 3: Acade	emic Skills -	Presenting Ef	fectively - Part	1	
Video 3: Acade	emic Skills - emic Skills	Presenting Ef - Presenting	fectively - Part Effectively s	1 Bation Sti Watch late duction:	ructure Fr Share
Video 3: Acade	emic Skills -	Presenting Ef - Presenting Tell the	fectively - Part Effectively s Intro	1 B*ion Sti Watch late duction: are going t	ructure share
Video 3: Acade	emic Skills -	Presenting Ef	fectively - Part Effectively s Intro m what you	1 B*ion Str Watch late duction: are going t	ructure share
Video 3: Acade	emic Skills -	Presenting Ef	fectively - Part Effectively s Intro m what you B Tell	1 B *ion Str Watch late duction: are going t ody: them	ructure share
Video 3: Acade	emic Skills -	Presenting Ef	fectively - Part Effectively is Intro em what you B Tell	1 B *ion Str Watch late duction: are going t ody: them	ructure share
Video 3: Acade	emic Skills -	Presenting Ef	fectively - Part Effectively is Intro em what you B Tell Conc	1 Btion Str Watch late duction: are going t ody: them them Lusion:	o tell them
Video 3: Acade	emic Skills -	Presenting Ef	fectively - Part Effectively s Intro m what you B Tell Conc ell them what	1 B*ion Str Watch late duction: are going t ody: them Liusion: at you told the	ructure share
Video 3: Acade	emic Skills -	Presenting Ef	fectively - Part Effectively is Intro- em what you B Tell Conc ell them what	1 B *ion Strukture duction: are going to ody: them Lusion: at you told the	nucture share
Video 3: Acade	emic Skills -	Presenting Ef	fectively - Part	1 B*ion Struwatch late duction: are going t ody: them them lusion: at you told the	n content them

From 0.13 Presentation Preparation From 0.45 Divide Your Talk into Sequenced... From 1.15 Examples of Presentation Formats From 1.31 Three – Minute Thesis From 1.52 Be Aware of Multiple Sensory... From 2.43 Call to Action From 3.00 Take – away Message From 3.34 Organization From 4.07 Using a Hook From 5.24 Transitions

Please give your feedback on the recommended segments for learning about giving presentations (you can select several options).						
	This video segment was USEFUL	This video segment was NOT USEFUL	The short DESCRIPTION of the video segment was HELPFUL	The short DESCRIPTION of the video segment was UNHELPFUL	I DID NOT WATCH this video segment	
From 0.13 Presentation Preparation						
From 0.45 Divide Your Talk into Sequenced						
From 1.15 Examples of Presentation Formats						
From 1.31 Three – Minute Thesis						
From 1.52 Be Aware of Multiple Sensory						
From 2.43 Call to Action						
From 3.00 Take – away Message						
From 3.34 Organization						
From 4.07 Using a Hook						
From 5.24 Transitions						

SUMMARY: List main points about giving presentations that you picked in this video, focus on: DELIVERY, STRUCTURE, VISUAL AIDS, PRESENTATION CHARACTERISTICS.

Your answer

Video 4: Powerful Presentation Skills: Practice Your Delivery



Recommended segments to watch:

From 0.02 Why you need to practice your delivery From 0.40 How to practice your presentation From 0.48 Dress rehearsal – get familiar with ... From 1.20 Dress rehearsal – practice bei... From 2.28 Where do we get tripped up? From 2.47 What is the best place to practice? From 3.26 Dress rehearsal for a full hour? From 3.59 Dress rehearsals for new content

	This video segment was USEFUL	This video segment was NOT USEFUL	The short DESCRIPTION of the video segment was HELPFUL	The short DESCRIPTION of the video segment was UNHELPFUL	I DID NOT WATCH this video segment
From 0.02 Why you need to practice your delivery					
From 0.40 How to practice your presentation					
From 0.48 Dress rehearsal – get familiar with					
From 1.20 Dress rehearsal – practice bei					
From 2.28 Where do we get tripped up?					
From 2.47 What is the best place to practice?					
From 3.26 Dress rehearsal for a full hour?					
From 3.59 Dress rehearsals for new content					

SUMMARY: List main points about giving presentations that you picked in this video, focus on: DELIVERY, STRUCTURE, VISUAL AIDS, PRESENTATION CHARACTERISTICS.

Your answer			
Back	Next	 Page 5 of 12	Clear form

The questions below ask how you perceive the usefulness of recommended video- segments for learning about giving presentations.									
short description	* The recommended video-segments with the short description enabled me to:								
	Extremely likely	Quite likely	Slightly likely	Neutral	Slightly unlikely	Quite unlikely	Extremely unlikely		
Identify key points related to giving presentations									
Focus on key points at a time									
Identify key topics/concepts to learn									
Link main points mentioned in the video									
What do you find description for le Your answer	What do you find positive about the recommended video-segments with short description for learning? Your answer								
What do you find negative about the recommended video-segments with short description for learning? Your answer									
Back Nex	Back Next Page 6 of 12 Clear form								

The following questions gather further feedback on the effect of recommended video-segments for learning

MENTAL DEMAND (1-low 20-high)

How mentally demanding was it to watch recommended video-segments to support your learning? For example, how much mental and perceptual activity was required thinking, deciding, remembering, looking, searching? Your score should be within the range of low(1) to high(20).

Your score on MENTAL DEMAND (1-low 20-high) *

Your answer

Please explain your score.

Your answer

EFFORT (1-low 20-high)

How hard did you have to work (mentally and physically) to watch recommended video-segments to support your learning? Your score should be within the range of low(1) to high(20).

Your score on EFFORT (1-low 20-high) *

Your answer

Please explain your score

Your answer

FRUSTRATION (1-low 20-high)

How discouraged, irritated, stressed and annoyed did you feel while watching recommended video-segments to support your learning? Your score should be within the range of low(1) to high(20). Your score on FRUSTRATION (1-low 20-high) * Your answer Please explain your score. Your answer PERFORMANCE (1-low 20-high) How successful do you think you were to identify new useful terms from watching the recommended video-segments to support your learning? Your score should be within the range of low(1) to high(20). Your score on PERFORMANCE (1-low 20-high) Your answer Please explain your score. Your answer Back Next Page 7 of 12 Clear form In this Section you will watch another 4 videos about presentation skills. Please pay attention to the terms related to giving presentations - Structure, Delivery, Visual Aids and Presentation Characteristics.

You will also be provided with a list of interesting segments in the video which were identified by another computer algorithm.

For each segment, you will see the start and end, as well as a short description what concepts about presentation skills are covered. The segments and the short description are aimed to assist with your video watching experience by showing main points related to presentation skills.



Video 5: How To Make an Interview Presentation

0.09 - 0.52 Focus on the topic Delivery by mentioning Audience

0.52 - 1.29 Focus on the topic Structure by mentioning Key Message (Conclusion)

- 1.29 2.08 Focus on the topic Structure by mentioning Section (Body Component)
- 2.08 2.36 Focus on the topic Structure by mentioning Opening (Structure Component)
- 2.36 2.52 Focus on the topic Delivery by mentioning Confidence, Nervous (Speaker Emotion), Practice (Preparation)

Nervous (speaker Enrouon), Fractice (Freparation)

- 2.52 3.25 Focus on the topic Visual Aid by mentioning PowerPoint (Slide)
- 3.25 3.42 Focus on the topic Visual Aid by mentioning Slide (Visual Aid Artefact) and Focus on the topic Delivery by mentioning Pause (Paralanguage)
- 3.42 4.00 Focus on the topic Delivery by mentioning Voice (Paralanguage)

4.00 - 4.58 Focus on the topic Delivery by mentioning Audience

Please give your feedback on the recommended segments for learning about giving presentations (you can select several options).						
	This video segment was USEFUL	This video segment was NOT USEFUL	The short DESCRIPTION of the video segment was HELPFUL	The short DESCRIPTION of the video segment was UNHELPFUL	I DID NOT WATCH this video segment	
0.09 - 0.52 Focus on the topic Delivery by mentioning Audience						
0.52 - 1.29 Focus on the topic Structure by mentioning Key Message (Conclusion)						
1.29 - 2.08 Focus on the topic Structure by mentioning Section (Body Component)						
2.08 - 2.36 Focus on the topic Structure by mentioning Opening (Structure Component)						
2.36 - 2.52 Focus on the topic Delivery by mentioning Confidence, Nervous (Speaker Emotion), Practice (Preparation)						

2.52 - 3.25 Focus on the topic Visual Aid by mentioning PowerPoint (Slide)							
3.25 - 3.42 Focus on the topic Visual Aid by mentioning Slide (Visual Aid Artefact) and Focus on the topic Delivery by mentioning Pause (Paralanguage)							
3.42 - 4.00 Focus on the topic Delivery by mentioning Voice (Paralanguage)							
4.00 - 4.58 Focus on the topic Delivery by mentioning Audience							
	in naiste	ala autoriuine e		*h	in this		
SUMMARY: List main points about giving presentations that you picked in this video, focus on: DELIVERY, STRUCTURE, VISUAL AIDS, PRESENTATION CHARACTERISTICS.							
Your answer							



Video 6: How to Use Humor in a Speech Opening

Recommended segments to watch:

0.0 - 0.17 Focus on the topic STRUCTURE by mentioning START (STRUCTURE COMPONENT) and Focus on the topic PRESENTATION CHARACTERISTICS by mentioning ATTENTION and Focus on the topic DELIVERY by mentioning STAND
0.17 - 0.50 Focus on the topic DELIVERY by mentioning HUMDR (RHETORICAL DEVICE)
0.50 - 1.46 Focus on the topic STRUCTURE by mentioning START (STRUCTURE COMPONENT)
1.46 - 2.56 Focus on the topic PRESENTATION CHARACTERISTICS by mentioning ATTENTION
3.07 - 3.47 Focus on the topic PRESENTATION CHARACTERISTICS by mentioning ATTENTION
3.47 - 4.27 Focus on the topic STRUCTURE by mentioning STORY and
Focus on the topic DELIVERY by mentioning HUMDROUS (RHETORICAL DEVICE)
4.27 - 4.39 Focus on the topic DELIVERY by mentioning SMILE (FACIAL EXPRESSION) , AUDIENCE
4.39 - 4.52 Focus on the topic VISUAL AIDS by mentioning PICTURE (FIGURE),
SLIDE (VISUAL AID ARTEFACT) , IMAGE (PICTURE)
4.52 - 5.54 Focus on the topic DBLIVERY by mentioning HUMDR (RHETORICAL DEVICE)

	This video segment was USEFUL	This video segment was NOT USEFUL	The short DESCRIPTION of the video segment was HELPFUL	The short DESCRIPTION of the video segment was UNHELPFUL	I DID NOT WATCH this video segment
0.00 - 0.17 Focus on the topic Structure by mentioning Start (Structure Component) and Focus on the topic Presentation Characteristics by mentioning Attention and Focus on the topic Delivery by mentioning Stand					
0.17 - 0.50 Focus on the topic Delivery by mentioning Humor (Rhetorical Device)					
0.50 - 1.46 Focus on the topic Structure by mentioning Start (Structure Component)					
1.46 - 2.56 Focus on the topic Presentation Characteristics by mentioning Attention					

3.07 - 3.47 Focus on the topic Presentation Characteristics by mentioning Attention			
3.47 - 4.27 Focus on the topic Structure by mentioning Story and Focus on the topic Delivery by mentioning Humorous (Rhetorical Device)			
4.27 - 4.39 Focus on the topic Delivery by mentioning Smile (Facial Expression), Audience			
4.39 - 4.52 Focus on the topic Visual Aids by mentioning Picture (Figure), Slide (Visual Aid Artefact), Image (Picture)			
4.52 - 5.54 Focus on the topic Delivery by mentioning Humor (Rhetorical Device)			



0.00 - 1.20 Focus on the topic STRUCTURE by mentioning START (STRUCTURE COMPONENT)
1.20 - 1.55 Focus on the topic DELIVERY by mentioning AUDIENCE
1.55 - 2.17 Focus on the topic STRUCTURE by mentioning TOPIC (BODY COMPONENT) and Focus on the topic DELIVERY by mentioning EXPRESSION (SPEAKER AURA)
2.17 - 2.27 Focus on the topic DELIVERY by mentioning EYE (EYE MOTION)
2.27 - 3.17 Focus on the topic DELIVERY by mentioning AUDIENCE
3.17 - 3.32 Focus on the topic DELIVERY by mentioning CONFIDENCE (SPEAKER EMOTION)
3.32 - 3.55 Focus on the topic STRUCTURE by mentioning REASON (CONTENT INTRODUCTION), START (STRUCTURE COMPONENT)
3.55 - 4.43 Focus on the topic DELIVERY by mentioning TOPIC (BODY COMPONENT)
4.43 - 5.16 Focus on the topic DELIVERY by mentioning CONFIDENCE (SPEAKER EMOTION), TONE (PARALANGUAGE), EYE (EYE MOTION)

	This video segment was USEFUL	This video segment was NOT USEFUL	The short DESCRIPTION of the video segment was HELPFUL	The short DESCRIPTION of the video segment was UNHELPFUL	I DID NOT WATCH this video segment
0.00 - 1.20 Focus on the topic Structure by mentioning Start (Structure Component)					
1.20 - 1.55 Focus on the topic Delivery by mentioning Audience					
1.55 - 2.17 Focus on the topic Structure by mentioning Topic (Body Component) and Focus on the topic Delivery by mentioning Expression (Speaker Aura)					

2.17 - 2.27 Focus on the topic Delivery by mentioning Eye (Eye Motion)			
2.27 - 3.17 Focus on the topic Delivery by mentioning Audience			
3.17 - 3.32 Focus on the topic Delivery by mentioning Confidence (Speaker Emotion)			
3.32 - 3.55 Focus on the topic Structure by mentioning Reason (Content Introduction), Start (Structure Component)			
3.55 - 4.43 Focus on the topic Structure by mentioning Topic (Body Component)			
4.43 - 5.16 Focus on the topic Delivery by mentioning Confidence (Speaker Emotion), Tone (Paralanguage), Eye (Eye Motion)			

SUMMARY: List main points about giving presentations that you picked in this video, focus on: DELIVERY, STRUCTURE, VISUAL AIDS, PRESENTATION CHARACTERISTICS.

Your answer

Video 8: Beginning Graphic Design: Layout & Composition

Beginning Graphic Design: Layout & Comp	S → Watch later Share
Layout &	
Composi	tion
Watch on 🕞 YouTube	

Recommended segments to watch:

- 0.0 0.44 Focus on the topic PRESENTATION CHARACTERISTICS by mentioning LAYOUT (CLARITY) and Focus on the topic VISUAL AIDS by mentioning GRAPHICS (DIAGRAM), SOUND (AUDIO)
- 0.44 2.26 Focus on the topic VISUAL AIDS by mentioning TEXT (VISUAL AID ARTEFACT)
- 2.26 3.49 Focus on the topic DELIVERY by mentioning EYE (EYE MOTION)
- 3.49 4.55 Focus on the topic VISUAL AIDS by mentioning COLOR (EMPHASISED TEXT)

	This video segment was USEFUL	This video segment was NOT USEFUL	The short DESCRIPTION of the video segment was HELPFUL	The short DESCRIPTION of the video segment was UNHELPFUL	I DID NOT WATCH this video segment
0.00 - 0.44 Focus on the topic Presentation Characteristics by mentioning Layout (Clarity) and Focus on the topic Visual Aids by mentioning Graphics (Diagram), Sound (Audio)					
0.44 - 2.26 Focus on the topic Visual Aids by mentioning Text (Visual Aid Artefact)					
2.26 - 3.49 Focus on the topic Delivery by mentioning Eye (Eye Motion)					
3.49 - 4.55 Focus on the topic Visual Aids by mentioning Color (Emphasised					

SUMMARY: List main points about giving presentations that you picked in this video, focus on: DELIVERY, STRUCTURE, VISUAL AIDS, PRESENTATION CHARACTERISTICS.									
Your answer									
Back Next Page	e 8 of 12 Clear form								

The questions below ask how you perceive the usefulness of recommended videosegments for learning about giving presentations.

The recommended video-segments with the

*

short description enabled me to:

	Extremely likely	Quite likely	Slightly likely	Neutral	Slightly unlikely	Quite unlikely	Extremely unlikely	
Identify key points related to giving presentations								
Focus on key points at a time								
Identify key topics/concepts to learn								
Link main points mentioned in the video								
What do you find positive about the recommended video-segments with short description for learning? Your answer								
What do you find negative about the recommended video-segments with short description for learning? Your answer								
Back Nex	t		_	Pag	e 9 of 12		Clear for	m

The following questions gather further feedback on the effect of recommended video-segments for learning

MENTAL DEMAND (1-low 20-high)

How mentally demanding was it to watch recommended video-segments to support your learning? For example, how much mental and perceptual activity was required thinking, deciding, remembering, looking, searching? Your score should be within the range of low(1) to high(20).

Your score on MENTAL DEMAND (1-low 20-high) *

Your answer

Please explain your score.

Your answer

EFFORT (1-low 20-high)

How hard did you have to work (mentally and physically) to watch recommended video-segments to support your learning? Your score should be within the range of low(1) to high(20).

Your score on EFFORT (1-low 20-high)

Your answer

Please explain your score.

Your answer

FRUSTRATION (1-low 20-high)

How discouraged, irritated, stressed and annoyed did you feel while watching recommended video-segments to support your learning?

Your score

should be within the range of low(1) to high(20).

Your score on FRUSTRATION (1-low 20-high) *

Your answer

Please explain your score.

Your answer

PERFORMANCE (1-low 20-high)

How

successful do you think you were to identify new useful terms from watching the recommended video-segments to support your learning?

Your score should be within the range of low(1) to high(20).

Your score on PERFORMANCE (1-low 20-high) *

Your answer

Please explain your score.

Your answer

Back

Next

Clear form

Thank you very much for completing the study.	
Please provide your email address below if you want to enter a draw of Si Amazon vouchers.	x £50
If you provide your email address, it will be used only to pay you the voucher and deleted afterwards. It will not be part of the data used in this study.	will be
Your answer	
Back Next Page 11 of 12	Clear form

Below is the second form, where we have included 4 videos characterised by VISC-L first. Then, we included other 4 videos characterised by Google. In this case, we will have evaluated VISC-L and Google on the same number of videos. The consent form and the user profile is the same as above. Here, we will include the videos, their characterisation, the Quality categories, the Perceived Usefulness scales and the Cognitive Work Load scales.

:

Section 5 of 12

In this Section you will watch 4 videos about presentation skills. Please pay attention to the terms related to giving presentations - Structure, Delivery, Visual Aids and Presentation Characteristics.

Description (optional)

You will also be provided with a list of interesting segments in the video which were identified by a computer algorithm. For each segment, you will see the start and end, as well as a short description what concepts about presentation skills are covered. The segments and the short description are aimed to assist with your video watching experience by showing main points related to presentation skills.

Description (optional)

Video 1: 5 Tips for delivering a great online presentation



0.00 - 0.30 Focus on the topic VISUAL AIDS by mentioning VIDEO (FILM), LAPTOP, PHONE (COMPUTER)
0.43 - 1.16 Focus on the topic DELIVERY by mentioning STAND (BODY POSITION)
1.43 - 3.06 Focus on the topic DELIVERY by mentioning STAND (BODY POSITION)
3.06 - 3.27 Focus on the topic DELIVERY by mentioning PRACTICE (PREPARATION)
3.27 - 3.58 Focus on the topic DELIVERY by mentioning DRESS (SPEAKER APPEARANCE)
4.23 - 5.14 Focus on the topic VISUAL AIDS by mentioning VIDEO (FILM)
5.14 - 5.48 Focus on the topic DELIVERY by mentioning AUDIENCE

	This video segment was USEFUL	This video segment was NOT USEFUL	The short DESCRIPTION of the video segment was HELPFUL	The short DESCRIPTION of the video segment was UNHELPFUL	I DID NOT WATCH this video segment
0.00 - 0.30 Focus on the topic Visual aids by mentioning Video (Film), Laptop, Phone (Computer)					
0.43 - 1.16 Focus on the topic Delivery by mentioning Stand (Body position)					
1.43 - 3.06 Focus on the topic Delivery by mentioning Stand (Body position)					

3.06 - 3.27 Focus on the topic Delivery by mentioning Practice (Preparation)			
3.27 - 3.58 Focus on the topic Delivery by mentioning Dress (Speaker Appearance)			
4.23 - 5.14 Focus on the topic Visual Aids by mentioning Video (Film)			
5.14 - 5.48 Focus on the topic Delivery by mentioning Audience			

SUMMARY: List main points about giving presentations that you picked in this video, focus on: DELIVERY, STRUCTURE, VISUAL AIDS, PRESENTATION CHARACTERISTICS.

Your answer



Video 2: How to prepare Speaking Notes From Thinkwell's Public Speaking

Recommended segments to watch:

- 0.00 0.16 Focus on the topic DELIVERY by mentioning SPEECH (LANGUAGE ELEMENT)
- 0.16 1.33 Focus on the topic STRUCTURE by mentioning OUTLINE (CONTENT INTRODUCTION)
- 1:33 2.07 Focus on the topic STRUCTURE by mentioning QUOTATION (CONTENT INTRODUCTION)
- 2.07 2.18 Focus on the topic DELIVERY by mentioning AUDIENCE and Focus on the topic STRUCTURE by mentioning REFERENCE (CLOSING COMPONENT)
- 2.52 4.31 Focus on the topic VISUAL AID by mentioning NOTE (VISUAL AID ARTEFACT)
- 4.31 5.24 Focus on the topic DELIVERY by mentioning SPEECH (LANGUAGE ELEMENT)

	This video segment was USEFUL	This video segment was NOT USEFUL	The short DESCRIPTION of the video segment was HELPFUL	The short DESCRIPTION of the video segment was UNHELPFUL	I DID NOT WATCH this video segment
0.00 - 0.16 Focus on the topic Delivery by mentioning Speech (Language Element)					
0.16 - 1.33 Focus on the topic Structure by mentioning Outline (Content Introduction)					
1:33 - 2.07 Focus on the topic Structure by mentioning Quotation (Content Introduction)					
2.07 - 2.18 Focus on the topic Delivery by mentioning Audience and Focus on the topic Structure by mentioning Reference (Closing Component)					

2.52 - 4.31 Focus on the topic Visual Aid by mentioning Note (Visual Aid Artefact)							
4.31 - 5.24 Focus on the topic Delivery by mentioning Speech (Language Element)							
SUMMARY: List main points about giving presentations that you picked in this video, focus on: DELIVERY, STRUCTURE, VISUAL AIDS, PRESENTATION CHARACTERISTICS.							


Please give your feedback on the recommended segments for learning about giving presentations (you can select several options).

	This video segment was USEFUL	This video segment was NOT USEFUL	The short DESCRIPTION of the video segment was HELPFUL	The short DESCRIPTION of the video segment was UNHELPFUL	I DID NOT WATCH this video segment
0.00 - 0.23 Focus on the topic Structure by mentioning Strategy (Structure Component) and Focus on the topic Visual Aids by mentioning PowerPoint (Slide)					
0.23 - 0.41 Focus on the topic Structure by mentioning Start (Structure Component)					
0.41 - 1.47 Focus on the topic Structure by mentioning Information (Body Component)					
1.47 - 2.27 Focus on the topic Visual Aids by mentioning Audio (Visual Aid Artefact)					

2.27 - 3.29 Focus on the topic Delivery by mentioning Audience			
3.29 - 4.08 Focus on the topic Structure by mentioning Introduction (Opening Component)			
4.08 - 5.00 Focus on the topic Structure by mentioning Story (Structure Approach)			
5.00 - 5.49 Focus on the topic Structure by mentioning Functional Language (Structure Component) and Focus on the topic Delivery by mentioning Audience			

SUMMARY: List main points about giving presentations that you picked in this video, focus on: DELIVERY, STRUCTURE, VISUAL AIDS, PRESENTATION CHARACTERISTICS.

Your answer



- DEVICE), SLIDE (VISUAL AID SUPPORTS)
- 2.46 3.25 Focus on the topic DELIVERY by mentioning PRACTICE (PREPARATION), STAND (BODY POSITION)
- 3.26 4.07 Focus on the topic DELIVERY by mentioning DRESS (SPEAKER APPEARANCE) , REHEARSAL (PREPARATION)

Please give your feedback on the recommended segments for learning about giving presentations (you can select several options).

	This video segment was USEFUL	This video segment was NOT USEFUL	The short DESCRIPTION of the video segment was HELPFUL	The short DESCRIPTION of the video segment was UNHELPFUL	I DID NOT WATCH this video segment
0.00 - 0.29 Focus on the topic Delivery by mentioning Practice (Preparation) and Focus on the topic Structure by mentioning Message (Conclusion)					
0.29 - 0.49 Focus on the topic Delivery by mentioning Dress (Speaker Appearance), Rehearsal (Preparation), Audience					
0.49 - 1.28 Focus on the topic Delivery by mentioning Walk (Body Position)					
1.28 - 1.51 Focus on the topic Delivery by mentioning Suit (Speaker Appearance)					

1.51 - 2.46 Focus on the topic Visual Aids by mentioning Computer (Visual Aid Device), Slide (Visual Aid Artefact)					
2.46 - 3.25 Focus on the topic Delivery by mentioning Practice (Preparation), Stand (Body Position)					
3.26 - 4.07 Focus on the topic Delivery by mentioning Dress (Speaker Appearance), Rehearsal (Preparation)					
SUMMARY: List m video, focus on: D CHARACTERISTIC Your answer	iain points ELIVERY, S S.	about giving TRUCTURE, \	presentations /ISUAL AIDS, P	that you picked RESENTATION	d in this
Back Next	_		Page	5 of 12	Clear form

The questions below ask how you perceive the usefulness of recommended video-
segments for learning about giving presentations.

*

		The	recomme	ended vid	leo-segm	ents wit	h the
short descriptio	n enabled me	e to:					
	Extremely	Quite	Slightly		Slightly	Quite	Extremely

	likely	likely	likely	Neutral	unlikely	unlikely	unlikely
Identify key points related to giving presentations							
Focus on key points at a time							
Identify key topics/concepts to learn							
Link main points mentioned in the video							
What do you find p description for lear Your answer	ositive ab rning?	out the re	ecomme	nded vide	eo-segme	ents with s	short
What do you find negative about the recommended video-segments with short description for learning? Your answer							

The following questions gather further feedback on the effect of recommended video-segments for learning

MENTAL DEMAND (1-low 20-high)

How mentally demanding was it to watch recommended video-segments to support your learning? For example, how much mental and perceptual activity was required thinking, deciding, remembering, looking, searching? Your score should be within the range of low(1) to high(20).

Your score on MENTAL DEMAND (1-low 20-high) *

Your answer

Please explain your score.

Your answer

EFFORT (1-low 20-high)

How hard did you have to work (mentally and physically) to watch recommended video-segments to support your learning? Your score should be within the range of low(1) to high(20).

Your score on EFFORT (1-low 20-high) *

Your answer

Please explain your score

Your answer

FRUSTRATION (1-low 20-high)

How discouraged, irritated, stressed and annoyed did you feel while watching recommended video-segments to support

Your score

should be within the range of low(1) to high(20).

Your score on FRUSTRATION (1-low 20-high) *

1

Please explain your score.

Your answer

your learning?

PERFORMANCE (1-low 20-high)

How

successful do you think you were to identify new useful terms from watching the recommended video-segments to support your learning?

Your score should be within the range of low(1) to high(20).

Your score on PERFORMANCE (1-low 20-high)

Your answer

Please explain your score.

Your answer

Back

Next

In this Section you will watch another 4 videos about presentation skills. Please pay attention to the terms related to giving presentations - Structure, Delivery, Visual Aids and Presentation Characteristics.

You will also be provided with a list of interesting segments in the video which were identified by another computer algorithm.

For each segment, you will see the start as well as a short description what information is covered. The segments and the short description are aimed to assist with your video watching experience by showing main points related to presentation skills.



Video 5: How To Make an Interview Presentation

Recommended segments to watch:

From 0.43 Create a profile on the interviewer ... From 1.35 Restrict your presentation to three mai... From 2.29 Set the tone for the rest of the ... From 3.09 Address the interview panel at the ... From 3.46 Gather your thoughts From 4.34 Leave the audience with a final thought

Please give your feedback on the recommended segments for learning about giving presentations (you can select several options).

	This video segment was USEFUL	This video segment was NOT USEFUL	The short DESCRIPTION of the video segment was HELPFUL	The short DESCRIPTION of the video segment was UNHELPFUL	I DID NOT WATCH this video segment
From 0.43 Create a profile on the interviewer					
From 1.35 Restrict your presentation to three mai					
From 2.29 Set the tone for the rest of the 					
From 3.09 Address the interview panel at the					
From 3.46 Gather your thoughts					
From 4.34 Leave the audience with a final thought					

SUMMARY: List main points about giving presentations that you picked in this video, focus on: DELIVERY, STRUCTURE, VISUAL AIDS, PRESENTATION CHARACTERISTICS.

Your answer

Video 6: How to Use Humor in a Speech Opening



Recommended segments to watch:

From 1.00 Attention Grabbers From 2.07 Deliberately Quote Someone From 3.35 Telling a Light-Hearted Story From 4.39 Show Something Visual on th... From 5.39 Question of the Day

Please give your feedback on the recommended segments for learning about giving presentations (you can select several options).					
	This video segment was USEFUL	This video segment was NOT USEFUL	The short DESCRIPTION of the video segment was HELPFUL	The short DESCRIPTION of the video segment was UNHELPFUL	I DID NOT WATCH this video segment
From 1.00 Attention Grabbers					
From 2.07 Deliberately Quote Someone					
From 3.35 Telling a Light-Hearted Story					
From 4.39 Show Something Visual on th					
From 5.39 Question of the Day					
SUMMARY: List main points about giving presentations that you picked in this video, focus on: DELIVERY, STRUCTURE, VISUAL AIDS, PRESENTATION CHARACTERISTICS.					
Your answer					



From 2.48 Practice your oral memorising ...

Please give your feedback on the recommended segments for learning about giving presentations (you can select several options).

	This video segment was USEFUL	This video segment was NOT USEFUL	The short DESCRIPTION of the video segment was HELPFUL	The short DESCRIPTION of the video segment was UNHELPFUL	I DID NOT WATCH this video segment
From 0.35 Start off with a bang (rhetorical					
From 1.46 Body language (arms, facial)					
From 2.23 No awkward breaks (and how to deal					
From 2.48 Practice your oral memorising					

SUMMARY: List main points about giving presentations that you picked in this video, focus on: DELIVERY, STRUCTURE, VISUAL AIDS, PRESENTATION CHARACTERISTICS.

Your answer



Recommended segments to watch:

From 0.49 Proximity From 1.20 Whitespace From 1.55 Alignment From 2.16 Be consistent From 2.46 Contrast From 3.48 Repetition Please give your feedback on the recommended segments for learning about giving presentations (you can select several options).

	This video segment was USEFUL	This video segment was NOT USEFUL	The short DESCRIPTION of the video segment was HELPFUL	The short DESCRIPTION of the video segment was UNHELPFUL	I DID NOT WATCH this video segment
From 0.49 Proximity					
From 1.20 Whitespace					
From 1.55 Alignment					
From 2.16 Be consistent					
From 2.46 Contrast					
From 3.48 Repetition					

SUMMARY: List main points about giving presentations that you picked in this video, focus on: DELIVERY, STRUCTURE, VISUAL AIDS, PRESENTATION CHARACTERISTICS.

Your ans	ver		
Back	Next	 Page 8 of 12	Clear form

The questions below ask how you perceive the usefulness of recommended video- segments for learning about giving presentations.								
short description	* The recommended video-segments with the short description enabled me to:							
	Extremely likely	Quite likely	Slightly likely	Neutral	Slightly unlikely	Quite unlikely	Extremely unlikely	
Identify key points related to giving presentations								
Focus on key points at a time								
Identify key topics/concepts to learn								
Link main points mentioned in the video								
What do you find positive about the recommended video-segments with short description for learning?								
What do you find negative about the recommended video-segments with short description for learning? Your answer								
Back Nex				Page	e 9 of 12		Clear for	m

The following questions gather further feedback on the effect of recommended video-segments for learning

MENTAL DEMAND (1-low 20-high)

How mentally demanding was it to watch recommended video-segments to support your learning? For example, how much mental and perceptual activity was required thinking, deciding, remembering, looking, searching? Your score should be within the range of low(1) to high(20).

Your score on MENTAL DEMAND (1-low 20-high) *

Your answer

Please explain your score.

Your answer

EFFORT (1-low 20-high)

How hard did you have to work (mentally and physically) to watch recommended video-segments to support your learning? Your score should be within the range of low(1) to high(20).

Your score on EFFORT (1-low 20-high) *

Your answer

Please explain your score.

Your answer

FRUSTRATION (1-low 20-high)

How discouraged, irritated, stressed and

annoyed did you feel while watching recommended video-segments to support your learning?

Your score

should be within the range of low(1) to high(20).

Your score on FRUSTRATION (1-low 20-high) *

Your answer

Please explain your score.

Your answer

PERFORMANCE (1-low 20-high)

How

successful do you think you were to identify new useful terms from watching the recommended video-segments to support your learning?

Your score should be within the range of low(1) to high(20).

Your score on PERFORMANCE (1-low 20-high) *

Your answer

Please explain your score.

Next

Your answer

Back

Page 10 of 12

Clear form

Thank you very much for completing the study.								
Please provide your email address below if you want to enter a draw of Amazon vouchers.	of Six £50							
If you provide your email address, it will be used only to pay you the voucher deleted afterwards. It will not be part of the data used in this study.	If you provide your email address, it will be used only to pay you the voucher and will be deleted afterwards. It will not be part of the data used in this study.							
Your answer								
Back Next Page 11 of 12	Clear form							

Appendix J

Google Form for VIN-L Evaluation (First Domain)

The Google Form designed and shared with the domain experts to evaluate VIN-L output in the first domain. Another four forms with the same structure but each form for a specific video narratives have been shared with Amazon Mechanical Turk workers.

Giving Pitch Presentation domain: The Video narratives evaluation form using Google Forms

Evaluation of video narratives to support presentation skills learning

Thank you for helping with evaluating the video narratives that have been generated from Abrar Mohammed's PhD project conducted at the University of Leeds, UK.

In this evaluation, you will assess **several video narratives** to help with learning presentation skills, including delivery, structure, visual aids, presentation attributes. <u>A video narrative is a combination of video segments which have been automatically characterised and linked together by computer algorithms.</u> You will be given a scoring sheet for each video narrative.

You will watch **19 video narratives**, and the evaluation takes **around 8-10 minutes per narrative**. You will have to log on with your Google account. When you fill out a Google Form in your Google account, **your progress is automatically saved**. This means if you can't complete the form or need to switch devices, your earlier responses will be saved, if you do not press SUBMIT. When you come back to the form, you will start from where you stopped last time. If you press SUBMIT and would like to come back to the form, you will start from the beginning, just be careful, please!

Your evaluation is essential to assess the potential benefits of the generated video narratives to support learning. Based on your evaluation, the best narratives will be selected for evaluation with learners.

The evaluation will take all together approximately 3 hours. You will be reimbursed for your time with £50 Amazon voucher.

Thank you very much for your help.

If you have any questions please contact:

- Abrar Mohammad <a.mohammed1@leeds.ac.uk>
- Vania Dimitrova <v.g.dimitrova@leeds.ac.uk>

a.mohammed1@leeds.ac.uk Switch account

Not shared

Next

Clear form

 \odot

The evaluation will include the following steps:

1. Click on the video narrative link, which will open in a new tab. You can also copy the link and open it in a browser yourself.

2. Go through the video narrative page by following the descriptions and watch the suggested video segments (you will watch between 3 to 5 minutes of video footage for each video narrative).

3. Go back to the screen with the survey form (this form).

4. Complete the scoring form for each video narrative and give suggestions, if applicable, for improving the narrative.

5. Repeat steps 1-4 for each video narrative link included in the survey.

After each narrative type (Derivative, Super-ordinate, Correlative and Combinational), complete the:

6. Perceived usefulness section.

7. Further feedback section.

Back

Next

Clear form

Your profile:	
Note that all your answers are kept in an anonymous way.	
Your occupation: *	
Your answer	
Please indicate your experience in giving presentations: * Your answer	
Back Next	Clear form

Derivative Narratives: learning a concept by linking it to a more generic concept

Video narrative: Learning SMILE by linking it to GESTURE (Click on the link, it will * open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.					
All video segments clearly link to the presentation skills areas mentioned in the introduction.					
The descriptions which introduce each video segment provide a useful summary.					
All video segments provide relevant content for presentation skills learning.					

text at the end of the video narrative provides appropriate summary.					
This video narrative is useful for learning presentation skills.					
Can you give some narrative can be in	e feedback t nproved.	o explain you	Ir scores or st	uggest how th	ie video
Your answer					

<u>Video narrative: Learning DISTRACTION by linking it to ATTENTION</u> (Click on the * link, it will open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.					
All video segments clearly link to the presentation skills areas mentioned in the introduction.					
The descriptions which introduce each video segment provide a useful summary.					
All video segments provide relevant content for presentation skills learning.					

our answer/					
Can you give som arrative can be ir	e feedback t mproved.	o explain you	Ir scores or si	uggest how th	ie video
useful for learning presentation skills.					
This video narrative is					
text at the end of the video narrative provides appropriate summary.					

<u>Video narrative: Learning CHART by linking it to FIGURE</u> (Click on the link, it will * open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.					
All video segments clearly link to the presentation skills areas mentioned in the introduction.					
The descriptions which introduce each video segment provide a useful summary.					
All video segments provide relevant content for presentation skills learning.					

The concluding text at the end of the video narrative provides appropriate summary.						
This video narrative is useful for learning presentation skills.						
Can you give some feedback to explain your scores or suggest how the video narrative can be improved. Your answer						
Back Next					Clear form	

_

The questions below ask how you perceive the usefulness of the Derivative Narratives (learning a concept by linking it to a more generic concept).								
The video narrativ	es with the	short d	escriptior	n are use	ful to *			
	Extremely likely	Quite likely	Slightly likely	Neutral	Slightly unlikely	Quite unlikely	Extremely unlikely	
Identify key points related to giving presentations								
Focus on key points at a time								
Identify key topics/concepts to learn								
Link main points mentioned in the video								
Back Next							Clear for	rm

Further feedback on Derivative Narratives (learning a concept by linking it to a more generic concept).	
What do you find positive about the generated Derivatives Narratives to support ^a learning?	*
Your answer	
What do you find negative about the generated Derivatives Narratives to support learning? Your answer	*
Back Next Clear for	orm

Super-ordinate Narratives: learning a concept by linking it to more specific concepts

<u>Video narrative: Learning STRUCTURE by linking it to SUMMARY, STORY and</u> <u>INFORMATION</u> (Click on the link, it will open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

*

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.					
All video segments clearly link to the presentation skills areas mentioned in the introduction.					
The descriptions which introduce each video segment provide a useful summary.					
All video segments provide relevant content for presentation skills learning.					

arrative can be i	mproved.				
Can you give som	e feedback 1	o explain you	Ir scores or si	uggest how th	e video
learning presentation skills.					
This video narrative is					
of the video narrative provides appropriate summary.					

<u>Video narrative: Learning ATTENTION by linking it to DISTRACTION</u> (Click on the * link, it will open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.					
All video segments clearly link to the presentation skills areas mentioned in the introduction.					
The descriptions which introduce each video segment provide a useful summary.					
All video segments provide relevant content for presentation skills learning.					

The concluding text at the end of the video narrative provides appropriate summary.						
This video narrative is useful for learning presentation skills.						
Can you give some feedback to explain your scores or suggest how the video narrative can be improved.						
Your answer						
<u>Video narrative: Learning CONCLUSION by linking it to MESSAGE</u> (Click on the link, it will open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.					
All video segments clearly link to the presentation skills areas mentioned in the introduction.					
The descriptions which introduce each video segment provide a useful summary.					
All video segments provide relevant content for presentation skills learning.					

The concluding text at the end of the video narrative provides appropriate summary.								
This video narrative is useful for learning presentation skills.								
Can you give some feedback to explain your scores or suggest how the video narrative can be improved. Your answer								
Back Next					Clear form			

The questions below ask how you perceive the usefulness of the Super-ordinate Narratives (learning a concept by linking it to more specific concepts).

The video narratives with the short description are useful to *

	Extremely likely	Quite likely	Slightly likely	Neutral	Slightly unlikely	Quite unlikely	Extremely unlikely
Identify key points related to giving presentations							
Focus on key points at a time							
Identify key topics/concepts to learn							
Link main points mentioned in the video							
Back	t						Clear for

Further feedback on Super-ordinate Narratives (learning a concept by linking it to more specific concepts).	0
What do you find positive about the generated Super-ordinate Narratives to support learning?	*
Your answer	
What do you find negative about the generated Super-ordinate Narratives to support learning? Your answer	*
Back Next Clea	ar form

Correlative Narratives: learning a concept by linking it to similar concepts from the same class

<u>Video narrative: Learning IMPACT by linking it to ORGANIZATION and</u> <u>ATTENTION</u> (Click on the link, it will open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.					
All video segments clearly link to the presentation skills areas mentioned in the introduction.					
The descriptions which introduce each video segment provide a useful summary.					
All video segments provide relevant content for presentation skills learning.					

The concluding text at the end of the video narrative provides appropriate summary.			
This video narrative is useful for learning presentation skills.			

Can you give some feedback to explain your scores or suggest how the video narrative can be improved.

Your answer

Video narrative: Learning MEMORABLE by linking it to INFLUENCE and

<u>PERSUASIVE</u> (Click on the link, it will open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.					
All video segments clearly link to the presentation skills areas mentioned in the introduction.					
The descriptions which introduce each video segment provide a useful summary.					
All video segments provide relevant content for presentation skills learning.					

The concluding text at the end of the video narrative provides appropriate summary.								
This video narrative is useful for learning presentation skills.								
Can you give some feedback to explain your scores or suggest how the video narrative can be improved. Your answer								

Video narrative: Learning PAUSE by linking it to FOCUS, FORGETTING and TEMPO *

(Click on the link, it will open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.					
All video segments clearly link to the presentation skills areas mentioned in the introduction.					
The descriptions which introduce each video segment provide a useful summary.					
All video segments provide relevant content for presentation skills learning.					

Can you give som narrative can be i	e feedback t mproved.	o explain you	Ir scores or su	uggest how th	e video
narrative is useful for learning presentation skills.					
summary. This video					
of the video narrative provides appropriate					

<u>Video narrative: Learning SLIDE by linking it to NOTE, TEXT and HANDOUT</u> (Click * on the link, it will open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.					
All video segments clearly link to the presentation skills areas mentioned in the introduction.					
The descriptions which introduce each video segment provide a useful summary.					
All video segments provide relevant content for presentation skills learning.					

The concluding text at the end of the video narrative provides appropriate summary.								
This video narrative is useful for learning presentation skills.								
Can you give some feedback to explain your scores or suggest how the video narrative can be improved. Your answer								

<u>Video narrative: Learning INFORMATION by linking it to MAINTOPIC and TOPIC</u> * (Click on the link, it will open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

Strongly Strongly Agree Neutral Disagree agree disagree The introduction text in the video narrative gives a clear idea of the aim of the narrative. All video segments clearly link to the presentation skills areas mentioned in the introduction. The descriptions which introduce each video segment provide a useful summary. All video segments provide relevant Π П П content for presentation skills learning.

The concluding text at the end of the video narrative provides appropriate summary.					
This video narrative is useful for learning presentation skills.					
Can you give som narrative can be ir	e feedback t nproved.	o explain you	r scores or si	uggest how th	e video
Your answer					

<u>Video narrative: Learning START by linking it to BEGINNING and OPENNING</u> (Click * on the link, it will open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.					
All video segments clearly link to the presentation skills areas mentioned in the introduction.					
The descriptions which introduce each video segment provide a useful summary.					
All video segments provide relevant content for presentation skills learning.					

The concluding text at the end of the video narrative provides appropriate summary.					
This video narrative is useful for learning presentation skills.					
Can you give som narrative can be in	e feedback t nproved.	o explain you	Ir scores or s	uggest how th	ne video
Your answer					

<u>Video narrative: Learning COMMUNICATION by linking it to ENGAGEMENT</u> (Click * on the link, it will open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.					
All video segments clearly link to the presentation skills areas mentioned in the introduction.					
The descriptions which introduce each video segment provide a useful summary.					
All video segments provide relevant content for presentation skills learning.					

text at the end of the video narrative provides appropriate summary.					
This video narrative is useful for learning presentation skills.					
Can you give som narrative can be ir Your answer	e feedback t nproved.	o explain you	Ir scores or si	uggest how th	e video

<u>Video narrative: Learning ATTENTION by linking it to ORGANIZATION and</u> <u>LENGTH</u> (Click on the link, it will open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.					
All video segments clearly link to the presentation skills areas mentioned in the introduction.					
The descriptions which introduce each video segment provide a useful summary.					
All video segments provide relevant content for presentation skills learning.					

our answer					
an you give som arrative can be ir	e feedback t nproved.	to explain you	Ir scores or si	uggest how th	e video
learning presentation skills.					
This video narrative is useful for					
of the video narrative provides appropriate summary.					

<u>Video narrative: Learning CONCLUSION by linking it to SUMMARY</u> (Click on the link, it will open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.					
All video segments clearly link to the presentation skills areas mentioned in the introduction.					
The descriptions which introduce each video segment provide a useful summary.					
All video segments provide relevant content for presentation skills learning.					

The concluding text at the end of the video narrative provides appropriate summary.					
This video narrative is useful for learning presentation skills.					
Can you give some narrative can be in Your answer	e feedback t nproved.	o explain you	r scores or su	uggest how th	ne video

<u>Video narrative: Learning STRESS by linking it to CONFIDENCE, NERVOUS and</u> <u>ENTHUSIASM</u> (Click on the link, it will open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.					
All video segments clearly link to the presentation skills areas mentioned in the introduction.					
The descriptions which introduce each video segment provide a useful summary.					
All video segments provide relevant content for presentation skills learning.					

The concluding text at the end of the video narrative provides appropriate summary.			
This video narrative is useful for learning presentation skills.			

Can you give some feedback to explain your scores or suggest how the video narrative can be improved.

Your answ	/er	
Back	Next	Clear form

The questions below ask how you perceive the usefulness of the Correlative Narratives (learning a concept by linking it to similar concepts from the same class).

The video narratives with the short description are useful to *

	Extremely likely	Quite likely	Slightly likely	Neutral	Slightly unlikely	Quite unlikely	Extremely unlikely	
Identify key points related to giving presentations								
Focus on key points at a time								
Identify key topics/concepts to learn								
Link main points mentioned in the video								
Back	t						Clear for	m

Further feedback on Correlative Narratives (learning a concept by linking it to similar concepts from the same class).	
What do you find positive about the generated Correlative Narratives to support * learning?	
Your answer	
What do you find negative about the generated Correlative Narratives to support * learning?	
Your answer	
Back Next Clear for	rm

Combinational Narratives: linking concepts from two presentation skills areas

<u>Video narrative: Learning STRUCTURE by linking it to DELIVERY</u> (Click on the link, * it will open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.					
All video segments clearly link to the presentation skills areas mentioned in the introduction.					
The descriptions which introduce each video segment provide a useful summary.					
All video segments provide relevant content for presentation skills learning.					

The concluding text at the end of the video narrative provides appropriate summary.						
This video narrative is useful for learning presentation skills.						
Can you give some feedback to explain your scores or suggest how the video narrative can be improved. Your answer						

Video narrative: Learning PRESENTATION ATTRIBUTE by linking it to VISUAL AID *

(Click on the link, it will open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.					
All video segments clearly link to the presentation skills areas mentioned in the introduction.					
The descriptions which introduce each video segment provide a useful summary.					
All video segments provide relevant content for presentation skills learning.					

The concluding text at the end of the video narrative provides appropriate summary.						
This video narrative is useful for learning presentation skills.						
Can you give some feedback to explain your scores or suggest how the video narrative can be improved. Your answer						

<u>Video narrative: Learning DELIVERY by linking it to VISUAL AID</u> (Click on the link, * it will open in a new tab. You can also copy the link and open it in a new tab/window in your browser.)

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.					
All video segments clearly link to the presentation skills areas mentioned in the introduction.					
The descriptions which introduce each video segment provide a useful summary.					
All video segments provide relevant content for presentation skills learning.					

This video narrative is useful for learning presentation skills.	The concluding text at the end of the video narrative provides appropriate summary.			
	This video narrative is useful for learning presentation skills.			

Can you give some feedback to explain your scores or suggest how the video narrative can be improved.

Your ans	wer	
Back	Next	Clear form

The questions below ask how you perceive the usefulness of the Combinational Narratives (linking concepts from two presentation skills areas).

The video narratives with the short description are useful to *

	Extremely likely	Quite likely	Slightly likely	Neutral	Slightly unlikely	Quite unlikely	Extremely unlikely
Identify key points related to giving presentations							
Focus on key points at a time							
Identify key topics/concepts to learn							
Link main points mentioned in the video							

Further feedback on Combinational Narratives (linking concepts from two presentation skills areas).

What do you find **positive** about the generated Combinational Narratives to support learning?

*

*

Your answer

What do you find **negative** about the generated Combinational Narratives to support learning?

Your answer

Thank you very much for completing the evaluation of the video narratives	
Is there any thing else you would like to tell us about the generated video narratives?	
Your answer	
Please provide your email address below for the £50 Amazon voucher arrangements. Your answer	*
Back Submit	Clear form

Appendix K

Amazon Mechanical Turk Study Design

The steps we followed to design our Amazon Mechanical Turk study to invite workers (learners) to evaluate the output of VIN-L in the first domain.

The process of using Amazon Mechanical Turk marketplace to invite learners to participate in the evaluation study (Study-2) of assess the video narratives generated for the presentation skill domain.

We have used Amazon Mechanical Turk marketplace to invite potential learners to assess the use of the video narratives to support learning. For that purpose, we have created an account with Amazon Mechanical Turk where we created a project and included four batches within it.

The design of each batch as presented in Appendix E includes three stages. First, is the 'Enter Properties' stage where we include the name of our project, the title, and the description, where we describe the aim of the evaluation study. Then, we include key words related to our study to help Amazon workers to search for the survey. After that, we set up the survey by specifying the following: how much we pay each participant (the rewards per response), number of responses we want, the time allotted per participant to complete the study, the maximum time for the survey to be available for the participants, and the amount of time for us to decide who to pay or reject from the participants' responses. Also, in this stage we designed our selection criteria to invite only the participants who are classified by Amazon Mechanical Turk as 'Master Workers' who have hit approval rate >= 95.

The second stage includes the instructions we specify for the participants to follow to make sure they submit the study correctly. Also, we informed them that in our study we follow the General Data Protection Regulation and the study had been reviewed by the Research Ethics Committee of the Faculty of Engineering and Physical Sciences at the University of Leeds, UK. The link of the study is also presented to the participants, and we included a space for them to add their Worker Id to enable us to proceed their payments as we showed in the first stage.

We have designed four batches in our Amazon project, one for each narrative type. Only the Amazon participant, who match the selection criteria we specified earlier, can access to the link of the surveys. Each participant will go through the following steps: (1.) Read and accept the consent form; (2.) Complete a short pre-study section to collect their profiles; (3.) Click on the video narrative link which will open in a new tab. They can also copy the link and open it in a browser
themselves; (4.) Go through the video narrative page by following the descriptions and watch the suggested video narratives which have a duration of 3 to 6 minutes; (5.) Go back to the screen with the survey form and complete the Quality, Perceived Usefulness and Mental workload sections. After the worker submit the form, they need to enter their Worker ID on Amazon Mechanical Turk marketplace to be used to proceed their payment process.

Amazon Mechanical Turk project to evaluate the generated video narratives to support learning Giving Pitch Presentations with learners.

1 Enter Pro	perties 2 De	sign Layout ③ Preview and Finish
Project Name	Evaluation of Vide	30 Narratives (C This name is not displayed to Workers.
Describe you	r survey to Worker	S
Title	Evaluation of Vide	o Narratives (Combinational narrative type) about Pre-
	Describe the survey to Wo	rkers. Be as specific as possible, e.g. "answer a survey about movies", instead of "short survey", so Workers know what to expect.
Description	You will participate	e in an evaluation study using Google forms to assess 2 video narratives (sequei
	Give more detail about this	survey. This gives Workers a bit more information before they decide to view your survey.
Keywords	Giving presentation	ns, video-based learning, su
	Provide keywords that will	help Workers search for your tasks.
Setting up yo	ur survey	
Reward per re	esponse	\$ 4.0
		This is how much a Worker will be paid for completing your survey. Consider how long it will take a Worker to complete your survey.
Number of res	spondents	50
		How many unique Workers do you want to complete your survey?
Time allotted	per Worker	20 Minutes
		Maximum time a Worker has to complete the survey. Be generous so that Workers are not rushed.
e	a la	

Worker requirements

Require that Workers be Masters to do your tasks (Who are Mechanical Turk Masters?)

⊙ Yes 🔿 No

Specify any additional qualifications Workers must meet to work on your tasks:

HIT Approval Rate (%) for all Requesters' HITs 🔹 greater than or equal to 🔹 95 ✓ Remove ✓ has not been granted
 ✓ Remove

Complete my survey already

(+) Add another criterion (up to 3 more) (Premium Qualifications incur additional fees, see Pricing Details to learn more)

Project contains adult content (See details)

This project may contain potentially explicit or offensive content, for example, nudity.

Task Visibility (What is task visibility?) O Public - All Workers can see and preview my tasks

O Private - All Workers can see my tasks, but only Workers that meet all Qualification requirements can preview my tasks

Hidden - Only Workers that meet my Qualification requirements can see and preview my tasks

Save Design Layout

This is how your task will look to Mechanical Turk Workers.

(1) Enter Properties (2) Design Layout (3) Preview and Finish

. Abrar monumica			Reward: \$4.00 per task	Tasks available: 0	Duration: 20 Minutes
ons Required: HIT Appro	val Rate (%) for all Requesters' HITs greater than or equal	to 95 , Complete my survey already has not been granted , Masters	has been granted		
	Survey Link Instructions (Click t	o collapse)			
	Thank you for joining this survey.				
	Your evaluation is essential to as	sess the potential benefits of the generated video n	arratives to support learning.		
	In this evaluation, you will spend help with learning presentation si segments which have been auto asked for your feedback on usab	up to 15 minutes watching videos about present iills, including some of the topics: delivery, structure natically characterised and linked together by comp lity of the narratives.	tation skills. You will assess 2 video e, visual aids and presentation attribu outer algorithms. You will be given a s	narratives (totalling 8 minutes video co tes. A video narrative is a combination o coring sheet for each video narrative a	ntent) to of <i>video</i> nd will be
	You will have to log on with your can't complete the form or need to where you stopped last time. If you	Google account. When you fill out a Google Form o switch devices, your earlier responses will be sav ou press SUBMIT and would like to come back to th	in your Google account, your progre red, if you do not press SUBMIT. Whe re form, <u>you will start from the beginn</u>	ess is automatically saved. This mean on you come back to the form, you will s ing, just be careful, please!	is if you tart from
	The study is fully anonymous -	although you will logon with your Google account, t	he responses do not save the email	of the responders.	
	The survey is fully complaint with Engineering and Physical Science	General Data Protection Regulation (GDPR) and h es, University of Leeds, UK.	nas been reviewed and approved by t	he Research Ethics Committee of the F	aculty of
	Thank you very much for your he	ip.			
	The link of the survey is provided	below.			
	Make sure to leave this window ope worker ID into the box. Please, make s	n as you complete the survey. When you are finished, you w ure you are writing your answers using your own words and D	vill return to this page to paste your survey co O NOT copy from the Internet	onfirmation code which is your	
	Template note for Requesters - To HIT. Please, make sure you are writin	verify that Workers actually complete your survey, require eac g your answers using your own words and DO NOT copy from	h Worker to enter a survey confirmation code n the Internet	(Worker ID) to your	
	L	Survey https://docs.google.com/forms/d/e/1FA link: UNKWsbE2g/viewform?usp=sf_link	lpQLSdnkG44vOdWXfXoPNQPWvoVh	medFugsY6ewFMPZ-	
		Provide the survey confirmation code which is your	Worker ID here:		
		e.g. 123456			

The full batch details after publishing it:

Manage Batches > Batch Details

Evaluation of Video Narratives (Correlative type of video narratives) about Presentation Skills 1

View the latest status of this batch, make changes, or get results.

rou will participate in an	revaluation study using Google forms to assess 2 video ha	manyes (sequences of video segments), the videos ar	e about giving pitc	n presentations. Fou do not need to be experienced in Presentation Skills.	
					Delete
Status: Pending Review	100% submitted	100% published			
Assignments Completed:	55 / 55	Average Time per Assignmen	a 11 minutes		
Creation Time:	August 10, 2023 8:06 AM PDT	Completion Time:	August 12, 2023	6:45 AM PDT	
Settings				Results	Results
Evaluation of \ View Project Note: If you have edited the Project Description:	Video Narratives (Correlative type of w t after publishing this Batch, you will see the latest version. You will participate in an evaluation study using Google form are about giving pitch presentations. You do not need to be	video narratives) about Presentation ns to assess 2 video narratives (sequences of video segme experienced in Presentation Skills.	n Skills	Assignments pending review: 0 Assignments approved: 50 Assignments rejected: 5 Cost Summary	
Keywords: Qualification Requirement(Giving presentations, video-based learning, survey. HIT Approval Rate (%) for all Requesters' HITs greater than (s): Complete my survey already has not been granted Masters has been granted	or equal to 95		Estimated Total Reward: \$200.00 Estimated Fees to Mechanical Turk: \$90.00 (fee details) Estimated Total Cost: \$290.00 These costs are only an estimate until all of the assignments have been submitted and reviewed.	
Number of Assignments per Reward per Assignment: Batch expired on: Aug	er task: 50 \$4.00 gust 13, 2023 3:50 AM PDT				

Example task from this Batch

Survey link	https://docs.google.com/forms/d/e/1FAIpQLScdUn- xRVILIfIR34F-oVYST1X1GX1A- LX5KLNUOmVuRLOA/viewform?usp=sf_link	
Provide the	survey confirmation code which is your Worker ID here:	
e.g. 1234	56	
You m	nust ACCEPT the HIT before you can submit the results.	

Appendix L

Evaluation Results of VIN-L (First Domain)

These are the evaluation results from experts and learners of VIN-L in the first domain which includes the: Quality and the Perceived Usefulness results.

Quality of the video narratives as evaluated by experts.

Narrative Q2 Usefulness Response type Q1 Q3 Q4 Q5 type CO Strongly Agree Agree Neutral Disagree Strongly Disagree CR Strongly Agree Agree Neutral Disagree Strongly Disagree D Strongly Agree Agree Neutral Disagree Strongly Disagree S Strongly Agree Agree Neutral Disagree Strongly Disagree

The numbers are the aggregation of the experts' responses where each expert had evaluated more than one video narrative:

Perceived usefulness of the video narratives as evaluated by experts.

Narrative	Response type				
Type		PU1	PU2	PU3	PU4
	Extremely Likely	4	3	3	3
СО	Quiet Likely	4	4	5	4
	Slightly Likely	3	2	2	1
	Neutral	0	2	0	2
	Slightly Unlikely	0	0	1	0
	Quiet Unlikely	0	0	0	1
	Extremely Unlikely	0	0	0	0
	Extremely Likely	4	2	3	1
CR	Quiet Likely	5	6	2	5
	Slightly Likely	1	2	5	3
	Neutral	1	0	1	2
	Slightly Unlikely	0	1	0	0
	Quiet Unlikely	0	0	0	0
	Extremely Unlikely	0	0	0	0
	Extremely Likely	5	5	6	4
D	Quiet Likely	3	3	3	4
	Slightly Likely	2	2	2	2
	Neutral	1	1	0	1
	Slightly Unlikely	0	0	0	0
	Quiet Unlikely	0	0	0	0
	Extremely Unlikely	0	0	0	0
	Extremely Likely	4	3	2	6
S	Quiet Likely	3	4	7	1
	Slightly Likely	3	2	1	0
	Neutral	0	1	0	3
	Slightly Unlikely	1	1	1	1
	Quiet Unlikely	0	0	0	0
	Extremely Unlikely	0	0	0	0

The numbers are the aggregation of the experts' responses where each expert had evaluated more than one video narrative:

Quality of the video narratives as evaluated by learners.

Narrative type	Response type	Q1	Q2	Q3	Q4	Q5	Usefulness
СО	Strongly Agree	14	27	24	27	27	35
	Agree	69	55	51	55	46	48
	Neutral	10	9	13	10	14	10
	Disagree	3	4	7	3	7	2
	Strongly Disagree	0	1	1	1	2	1
CR	Strongly Agree	35	46	36	51	40	45
	Agree	69	60	60	55	55	59
	Neutral	9	6	17	6	15	10
	Disagree	0	2	1	2	2	0
	Strongly Disagree	1	0	0	0	2	0
D	Strongly Agree	22	30	28	26	27	26
	Agree	60	51	52	46	53	54
	Neutral	33	24	26	28	29	25
	Disagree	9	16	14	17	12	12
	Strongly Disagree	2	5	6	9	5	9
S	Strongly Agree	23	42	32	47	31	45
	Agree	56	41	44	34	34	37
	Neutral	6	8	11	11	21	8
	Disagree	11	5	6	5	9	3
	Strongly Disagree	2	2	5	1	3	5

The numbers are the aggregation of the learners' responses where each learner had evaluated more than one video narrative:

The perceived usefulness of the video narratives as ranked by the learners.

The numbers are the aggregation of the learners' responses where each learner had evaluated more than one video narrative:

Narrative type	Response type	PU1	PU2	PU3	PU4
СО	Extremely Likely	6	11	18	11
	Quiet Likely	28	22	23	25
	Slightly Likely	10	12	3	10
	Neutral	3	2	4	1
	Slightly Unlikely	0	1	0	1
	Quiet Unlikely	1	0	0	0
	Extremely Unlikely	0	0	0	0
CR	Extremely Likely	17	25	26	18
	Quiet Likely	31	22	22	29
	Slightly Likely	6	6	7	9
	Neutral	3	4	2	1
	Slightly Unlikely	0	0	0	0
	Quiet Unlikely	0	0	0	0
	Extremely Unlikely	0	0	0	0
D	Extremely Likely	10	15	7	7
	Quiet Likely	18	11	23	23
	Slightly Likely	11	14	8	8
	Neutral	6	6	10	8
	Slightly Unlikely	9	4	8	7
	Quiet Unlikely	6	9	3	9
	Extremely Unlikely	3	4	4	1
S	Extremely Likely	17	15	20	14
	Quiet Likely	17	24	17	25
	Slightly Likely	7	4	7	4
	Neutral	4	2	1	2
	Slightly Unlikely	3	2	1	1
	Quiet Unlikely	1	2	3	3
	Extremely Unlikely	0	0	0	0

Appendix M

Expert's Evaluation Results of the Semantic Tagging Algorithm (Second Domain)

The evaluation result of the Semantic Tagging Algorithm with an expert from the InAdvance European Project after implementing it in the second domain (Health Related Quality of Life Awareness).

		My note on the evaluation result							To la-Semantic Tagging Algorithm	
The main topic		Topic Missed Topic SocialRelationship SocialRelationship	Physical Health Physical Health	So da Relationship	PhysicalHealth PhysicalHealth	Psychological Health Physical Health	Environment	So dalRelationship Psychological Health Physical Health	Physical Health	LevelOfindependence Environment Environment
ge if you are not sure		Concept 2 Concept 3 Personalkelationship Personalkelationship	Respritorylliness Respritorylliness	Personalikelationship	Respritorylihess Respritorylihess	Cognitive function Physical Symptoms	Acquiring Information	Personalite/lationship CognitiveFunction Respritonylliness	Respritory/liness symp	
d incorrect with red or highlight it with Orang	ories	Missed concept							cough	
: Or highlight the concept or the topic you fin	Patients St	Concep1 Mum Mum	Tumor LungCancer	Dad	Pneumonia LungGancer	Thinking Pain	Training	Mum Learning Copd	ChestPain	Treatment
hing if you agree with the cocnept and the topic		 Transcript 0.27 cancer 0.27 cancer cancer <licancer< li=""> cancer cancer <lic< th=""><th>screaming, don't them eie and she mum was just going I won't you're, not die and i was going. Just don't eir me 0.56 and we had a turnor, i never had any symptoms and i was working in a private school where you needed a physical every year. Each year I'd go in and I'd say my fahre reid of fung carner and 30 years earlier I had smoked. Maybe i should have earlier i had smoked. Maybe i should have</th><th>a chest×-ray. She said okay and there it 0.49 he was on the couch he had that her oxygen mask. He wasn't able to do those things. It retaily was a shock to us. My dad was going downhill very quickly, and we knew He was in and out of local hospitals a couple of times. Nobody was really able to accurately diagnose him. I think that was probably the scarter things not</th><th>3.14 a lot of us that get one kid's think about it so time, went past about is months later i got side. again and igot permemonia again and so now you re like okay, im 34 ud.2. im halthy, and ud.2. unb.2. did.1 one halt one and a so in ud.2. unb.2.</th><th>winy: winy: dual ge mino units cancer, again um. I remember thinking just before um. They said you know the of pain that you're in deesn't o what we're seeing</th><th>0.33 get better You know. Marine carps is key to some of the things that I doi's a and once one challenge is done, always looking for the next Soffst, full manuton was the Confirst, full manuton was the</th><th>0.46 Favorite trining about being a muming I think jush having her learning ust watching ther grow every day. B, just beaufful, fm so fins oucky that i get to be at home with her and at the moment freas was six momins old. When i was diagnosed with 3.19 block, because you know you're, hurdrig your headst och have an onmeter you has COPD needs to have an onmeter you</th><th>are not limited any as you know what your saturation level is it helps mind keeps iny mind off cit my toubles. wort leave home without. 2.09 what you we're dealing with so in 2006, in spekember 'hed ches pain like a ough went to and i remember because it was. It day in september where we're a ging gawy priver have an the militaori (whe navo navo</th><th>And there's really no known treatment yet for it And there's really no known treatment yet for it 1.46 except for lung transplantation. The lung is really the only organ in the body that's exposed to the environment 24 hours a day</th></lic<></licancer<>	screaming, don't them eie and she mum was just going I won't you're, not die and i was going. Just don't eir me 0.56 and we had a turnor, i never had any symptoms and i was working in a private school where you needed a physical every year. Each year I'd go in and I'd say my fahre reid of fung carner and 30 years earlier I had smoked. Maybe i should have earlier i had smoked. Maybe i should have	a chest×-ray. She said okay and there it 0.49 he was on the couch he had that her oxygen mask. He wasn't able to do those things. It retaily was a shock to us. My dad was going downhill very quickly, and we knew He was in and out of local hospitals a couple of times. Nobody was really able to accurately diagnose him. I think that was probably the scarter things not	3.14 a lot of us that get one kid's think about it so time, went past about is months later i got side. again and igot permemonia again and so now you re like okay, im 34 ud.2. im halthy, and ud.2. unb.2. did.1 one halt one and a so in ud.2. unb.2.	winy: winy: dual ge mino units cancer, again um. I remember thinking just before um. They said you know the of pain that you're in deesn't o what we're seeing	0.33 get better You know. Marine carps is key to some of the things that I doi's a and once one challenge is done, always looking for the next Soffst, full manuton was the Confirst, full manuton was the	0.46 Favorite trining about being a muming I think jush having her learning ust watching ther grow every day. B, just beaufful, fm so fins oucky that i get to be at home with her and at the moment freas was six momins old. When i was diagnosed with 3.19 block, because you know you're, hurdrig your headst och have an onmeter you has COPD needs to have an onmeter you	are not limited any as you know what your saturation level is it helps mind keeps iny mind off cit my toubles. wort leave home without. 2.09 what you we're dealing with so in 2006, in spekember 'hed ches pain like a ough went to and i remember because it was. It day in september where we're a ging gawy priver have an the militaori (whe navo navo	And there's really no known treatment yet for it And there's really no known treatment yet for it 1.46 except for lung transplantation. The lung is really the only organ in the body that's exposed to the environment 24 hours a day
Do not		Segmentid Start En MxmrjLQvg7E1 0.13	rQutyErq8G02 0.39	mWo-G3UO; 2U2 0.28	hQQ5xQGZxdA4 3.01	xY3B9nLm1u83 2.23	yv833Fx41E2 0.20	w/w/w/Zw/d6o2 0.30 UJ0Y40MLXEV4 3.03	I9k9q6FHex44 1.54	j3t5nFnTULQ4 1.23

NegativeFeeling	Psy chological Health
NegativeFeeling	Psy chological Health
RespritoryIIIness	Physical Health
Describered lise	Discretion (L1 and (sh
кезргионунитезя	РПУSIСАІНЕАІТП
PersonalRelationship	SocialRelationship
NegativeFeeling NegativeFeeling	PsychologicalHealth PsychologicalHealth
2	
RespritoryIllness	Physical Health
Personal Relationship	SocialRelationship
RespritoryIllness	Physical Health
Dobobilitation	E au disso se se se t
Respritorvillness	Physical Health
RespritoryIllness	Phy sical Healt h

Scared	Scared		Tumor			Pulmonary Fibrosis					Nom Scared	Scared				LungCancer				Family				Cond						Exercise								Cond						Pneumonia
and a lot of what we seethe problems like Jen had	are related the environment. Was I scared? Yeah.	I was definitely scared. You face the reality of death.	You get depressed. And as I got sicker and sicker and I wasn't able to do scan for disease progression	anu unioruniatery my usease nau progressed, but it was also fortunate in another way, because	with that progression a tumor grew that could be accessed for molecular	testing. breath. It was noticeable that something	was radically the matter with me. It muickly started to set worse where	the pulmonary fibrosis just started to	take over the lungs. A pulmonologist in Williamsport said we can't do any more	for you here. All I knew is that I was	chemo . mv mom had breast cancer . I mean mv	mother went through radiation	in the 90s . She went through chemo .	to die . I was scared	uh i did . I was afraid it was gon na hurt .	share um your story : ashley about lung cancer , getting diagnosed with it	and living life uh . What that means for	you , so thank you for joining us today . Thank you	and thank you for being here	, yeah , i 'm . I know there 's so much to go suppose veah	and just acknowledging what the patient	say is also a form of reassurance . Vou know acknowledning what the nationt	say and acknowledging how their family	or and there is difficulty breathing with	some simple task by the 50. People who	perform physical labor, for instance, find	it hard to continue this kind of work and most COPD patient can't walk for	long distance and ca n't lift object by	the late 50 or early 60. Quality of diagnosed ha heen to slow down -1 ife is no longer at	100 and actually that 's	quite a nice thing . My health care practitioner have hear very summerive in listening to	me listening to my suggestion listening to the way that	I want to address the condition . And they 've been very supportive in allowing me to develop my own exercise regime my own	nutritional plan . They 've been so supportive they are	happy to allow me to climb Kilimanjaro	so that 's erm a bit of a big risk but they 're willing to help me dot hat 1 use a device to	help strengthen my lung. It cause resistance when	you breathe in and out . Basically it Cond can only plan daily	What this mean is you can wake up in the morning	and litst feel awful.	You really know your not going to achieve	anything that day . There are several thing that make	living my life rather simpler than it wa .	that 's what we call it .
			3 1.5			1 1.1:					0 4.58					1 2.5				7 4.17				2.2						9 1.38								1 20						3 4.09
			1.4			0.5					4.4					2.4				4.0				2.1						0.5								-	-					3.5
			tfxB32Vb5hE3			Ofi_01yFmME1					llk-qgir9Nk9					bmvlK4qhAnU4				WArV8Ho C1A4	1			ehh89UrZtZU2						VyGp12XbPzs2								i7ITcn3d1c44						NI55TpSjWp44

									П
		Res piratory filoss Symptoms respiratory filoss	Strength	Friend	Exercise	Copd Copd Copd	Guilt	RufmonaryHypertension Purmon aryHypertension	Family
And I would say you know the report vary but probably 5 % to 10 % of all bacterial pneumonia	are from Legionella . Now I think higher percentage of that exists in the ICU	because there 's group of patient who tend to get very site. Then they get Legionella . responsive. It's The place to be dealing with dr. Cuiver, it's a relationship he with dr. Cuiver, it's a relationship he non-your leaderal, jike you, are the non-your administration that 's	wonderful work have a chronic respirator lillness and have to take care of it. So , so that we could get her back into shape. , so that we could get her back into shape. The relative the result of the state care and allow her to maintain her transplant and allow the form and heaven with the shear of a var of heaven with the	up enough strength up enough strength 17 % of eighth grader swith at they 've done it to eighth grader. The good news done it to eighth grader. The good news friend after seeing what he wont	through immediately the lilusion that through immediately the lilusion that to see that video. But it's important, I improved and feit better after the first eight program and if ound a very stremousts the first day or two and a common very on a to more very on a solution.	from them. Well, J1 direver a really from them. Well, J1 direver really in the U.S. tobacco smoke is the most common exposure that cause COPD . About 15 percent of all COPD case can be attributed	to workplace exposure such a dust gas and chemical dume. This video Illustrates some of the effect and the impact. The process and of the effect and the that COPD potentially related to work ha had on the fife everal people living with the disease. annew to make the and said : 1 need to speak to you just a moment in private. But you know thing like this do n't happent to me'. I' m altitle bit in shoct. But you know thing like this for a that the hike merican and a said in the disease. The and a said in the disease of the anoment infection. Wy primary care doctor Dr. Blake Longs scheduled met to a see Dr. Chen in minetion with the did the bits. We and the reformed me : We were able to sweb inside the lung and there are definitely cancer cell .	Immediately for me it's guilt. I wa a smoker for 30 years you know what have I done to myself ? So I met with the TOPS clinic to have the , and then throughout the body. There are frifterent category of pulmonary hypertension. The fourth group is labeled , chronic. Hromoberholic, pulmonary prepertension. And that involves a group of patient who have had a blood clot in the lung creation. What 'S typically	Yeah.
		.16 2.36	.11 2.23	2.2 2.33	.24 2.46	.25 0.49	36 1.05	.22 0.38	2.2 2.36
		6U7OpgFoJaw4 2	еокухитју 4 2	im4bTtQ995Y3	GV3Rew061FA3 2	Nkrz6_e_gwo2	q9xtexspy3c2	POKTSR_341	3Fker8Nmd5Y5

Physical Health	PhysicalHealth	SocialRelationship	Environment	Physical Health Physical Health Physical Health	PsychologicalHealth	Physicalities in Physical Phys	SocialRelationship SocialRelationship
PhysicalSymptoms	Energy	PersonalRelationship	Rehabilitation	Respirtory/Illness Respirtory/Illness Respirtory/Ilness	NegativeFeeling	Respritoryl liness Respritoryl liness	PersonalRelationship PersonalRelationship

the segment the segment the segment
Dicked 1
년 · · · · · · · · · · · · · · · · · · ·

								To fix -Semantic Tagging Alg																
			_																					
	SocialRelationship			SocialRelationship			SocialRelationship		levelof in dependence		SocialRelationship		PhysicalHealth			SocialRelationship	PhysicalHealth	SocialRelationship	SocialRelationship	SocialRelationship	SocialRelationship	PsychologicalHealth		
	PersonalRelationship			PersonalRelationship			PersonalRelationship		legalissue		SocialActivity		RespritoryIllness			PersonalRelationship		PersonalRelationship	PersonalRelationship	PersonalRelationship	PersonalRelationship	CognitiveFunction		

	9	Grandchild	Friend	a Exercise	end of life care	sk	Community	LungCancer	or Child Family Child Grandchild Child	Thinking
Emily : All we really ever wanted wa to have kid	and you know start a family . And it's so nice that he's going to see our kid grow u	Child : Daddy is my lunch ready ? them play football and rugby , it just me and : Really goodwn to see them . You just make me smile all the fine because the grandchild love the bone off me .	They always you know, write me little letter and they would have been good . If anything in ten year funny . I told my friend I wallke in a dakt tunnel for about two or three year Craig wa gone and since I started . Craig wa gone and since is tarted . coming back . I'n back and and 'I'l work	Then over time sort of even walking up the stair in then over time sort of even walking up the stair in your house or or at a railway station you know you co you thouse or or at a railway station you know you co way up the stair. Slowly it becomes difficult even to walk on the flat . Eventually you become dependent on supplementar	oxygen on to for exercise initially and then 24/7 at home just to stay alive and you effectively become disabled in your own house . You have need care .	You need to have somebody help you with simple ta like washing and dressing. It's just it's absolutely hornible. And wennibly those friend of mine I've sadly seen passed away go through end of life care.	um the philosophy of teaching swhiting about 17 So i did that for probably about 17 year professionally and even in other aspect wh in the community , if there wa class to be rundri. Takled so i'm a	these scan um. How about the blopsy at these scan um. How about the blopsy at happen ? And can you describe that procedure ? Procedure ? The know based on where you know where the cancer is in your body.	people die from it. My understanding is in people die from it. My understanding is in the U.S. about 17 000 adult die from RSV every seass the U.S. about 17 000 adult die from RSV every seass . Once of the real concern lhave is that I see these young child who are at high risk and they their parent their grandparent some of their great grandparent. So I'm really concerned for their parent their grandparent some of who will have RS and perhags not be very sid. can give it to their young grandchild or great and sear be not be very	we can also protect our youngest child. So i have you -know a kind of an arthritic knee i mean. Baybe this i you know people get older and maybe this is going to be your weak spot. So
		4.41	4.56	1.44			1.44	9.16	2.01	7.2
		4.29	4.39	1.05			1.28	9.04	1.25	7.09
		OHeU 30)8nmc4	Окнез. 12086	4AWZD3Clq W82			r20Ak5ZxP_42	MJEJaR3zLjg13	KFBT8X3DPuM2	HCgXqdpQGhc9

ignore-i was picked for the sigment	

.

PhysicalHealth	Physical Health	Physical Health Physical Health Physical Health	SocialRelationship	Psychological Health Psychological Health	Physical Health Physical Health Physical Health
	Reprovins	PhysicalSymptoms PhysicalSymptoms Behaviour	PersonalRelationship	Positivefeeling Positivefeeling	Respritory lifess En ergy Respritory lifness

Rest		PutronaryHypertension	Pain Pain	Drug	Family	Hope	Hope	LungGancer Strength	Pulmo nary Fibrosis
and what i mean by that is, after my college baseball playing career wa over and i got drafted um. I i really transitioned in myself into kind or different keyl	rs sport. Vorting out titness with whatever you want to all it, and so , when my plane career ended , which wa in around 1013 and joined the rest of the real world in a normal job uh. Just sustained uh. Just assumed just not be able to do some of the titning that you maybe were used to being able to do . And so i started to notice um my very first	I went to see dr. pressburg. He knew right away that I had something called promotary. Nypersofon and a nurse. I knew what it was but I also knew it wa. A really narsy evil disease and the A really narsy evil disease and the the converse of and that began	severe abdominal pain that we diagnosed a diverticulitis a diverticulitis and the provide the provided and and and abdominal pain my scan stylor the one lung. In retrospect I should not have one lung. In retrospect I should not have charact boards but want to concrete direct bit off. but want to concrete direct bit off.	a PD score greater than 5%. The start an interesting about 0 sound is that whe had a PD score of think 80%. He wa extremely sick when he had come . I do if think he would tolerate a plathrum chemotherapy to having a first interimmunotherapy drug so having a first interimmunotherapy drug	radiation therepy. I'Ve had a lung scan radiation therapy. I'Ve had a lung scan since and it 's clear. It was a pretty amazing process. I i have there involved either myself or a family member in in care in offfeett hospital	The area and any way. By har good chance of being cured, and it really make a dramatic difference. Every single what's your hospital and in dr Glassman's 'offree, wa so kind and so caring and gave me hope to get up and caring and gave me hope to get up and controt thrushound - Lipst on the chord for the theorem way how chord fealing with second	cancer. The second second second second second second second that you have we're told it's such a blow to be told that you have we're told it's such a to deal with a life disease. The second	rain a mary test, a they needed until rain a mary test, a they needed until they had to teal ime about it, which wa pretty shocking to me. At the time you pretty shocking to me. At the time you strength to deal with it when they told me I had cancer. Of course	Andy : There are a lot of thing that you give up when you 're diagnosed with a disease like pulmonary fibrosis .
4.1		6.0	ò	2.1	2.3	2.5	ਜ ਦ	1.0	0.1
се: С		0.11	0.2	2.0:	2	2.4	0.4	0.3	0.0
FHXbDHakK1w3		2 utaepga5 uk 2	pmuUnQ5Z4Q1	dyRmGVgG7_84	aREqsWoX-EY1	9nRhehMpOK3	Tyo2YSA0Y02	ZoEEgU6L-m81	FH7aHDHnkjU1

	To Fieldomain taxonomy taminy	
Psy chological Health Psy chological Health	PhysicalHealth Environment PhysicalHealth Social Relationship PhysicalHealth PhysicalHealth PhysicalHealth PhysicalHealth PhysicalHealth PhysicalHealth PhysicalHealth PhysicalHealth	SocialRebitonship SocialRebitonship
PositiveFeeling NegativeFeeling	Respritoryliness Respritoryliness Personalite lationship Personalite lationship Personalite lationship Respritoryliness Respritoryliness Respritoryliness Respritoryliness Respritoryliness Respritoryliness Respritoryliness	Personalke lationship Personalke lationship

him hope just to the him hope just to able to get up able to get up thing. Thing. Thing. The t	Hope Started LungDisease Exercise family Mom LungCancer LungCancer family Pain Pain Pain Pain
	LungCancer
±	Mum Mum
n do.	

		I wa an avid motorcycle rider for almost 30
cfyb395_CZ w2	1.2	I rode a Harley Sportster and I hard to give It. That a drivin because I would get sleepy while I wa drivin L44 because I had lived I hough cancer but at the same time I wanted to give him h live ive and let Just Play this out and see what happ and let Just Play this out and see what happ
		So I wa really scared for Victor . Some of the Joy in my home is being able to
mSYXddFVvH o1	0.03	and be healthy enough to do certain thing. Around the house it the to do bundry. 0.18 Lung disease is an invisible disease per se . You know, nobody know you have it untess they see you excrete. So really nobody knew, i had Lung disease except my close family for a long time. First
323hkORG0OM3	2.19	symptom were van interwarten unter- symptom were van interwarten interwarten King Poole downtown training for the . 1 2.31 actress symber or previous smoker, you should ask to get screened. Lung cancer early detection can lead to more and better restement. Johon . Ask you docor if you might be eligible for lung cancer if you might be eligible for lung cancer
LYwQdM-20k42	0.29	screening 0.47 look like and erasing what some say is a sigma for the cancer meet Moly goal. How 5 school hew 5 school 3 year old married working mom. Well, with one big exception // think when they told meet 's lung.
O5D0re3MFlw4	0.59	cancer. Just thought there is no way it 115 You can Please do whatever. Make yourseff a home. What was the environment around polio like back in the 1950s ? Was everybody wou so is all the time. Ear Everybody wou so is carrand frolio.
bqkquNb5Z8A1	0.03	It's a horrible disease . 0.18 Tara You do what you have to do to stay ali car door slam engine rev My pump wa popping outta my body every
qsAnIV8DUXo1	0.29	I wa on 24 hour pain killer . To make this drive down here to hoppeding get transpainted wa the only object forme . 0.42 bronchial infection , type thing that he needed to get over . I wa coughing or 45 minute straight and rould ond rach my breach . Then ded could on other other
2pzHkvmRSW82	3.19	and that's When i observed that in ad ipti diopathical pulmonary fibrosis. 3.33 2000 can make such a big impact. I wish had done it much earlier than i did. I think that if in your lung disease you's at a point our when when the simulater of
4ehrd RPE gE1	0.27	 yer yer yer yer yer yer yer yer yer yer
MxmrjLQvg7E1	0.13	what we do with your disease . 0.27 cancer cancer and there's there's fit's spread everywhere and there's nothing we can do.

		To Fix-Domain Taxonomy(friend)		
Physical Health Physical Health Sonal Values And Beliefs rsonal Values And Beliefs	PhysicalH ealth	Physical Health Physical Health Physical Health Physical Health So claffe lationship	Physical Health Physical Health Physical Health Physical Health Physical Health	evel Of Independence By chological Health Physical Health

RespritoryIllness CognitiveFunction PhysicalSymptoms

^hhy sicalSymptoms

toryllness

Religion Religion onship

RespritoryIllness

PhysicalSymptoms PhysicalSymptoms

sspiratoryIllnessSymptoms sspiratoryIllnessSymptoms RespritoryIllness

RespritoryIllness

PositiveFeeling

and unfortunately my disease had		i	
them some hope. cran for disease progression	1.57	1.43	+f×R32Vb5hE3
to help the patient and the physician plan and develop the best treatment, but also to c			
It's important to have an accurate diagnosis			
"I've got lives in tubes all in my body, I hurt like hell. but I did it!"			
as nara as you can so rean puntie ruse oue. And going to myself,			
hush	3.05	2.41	j3t5nFnTULQ5
The chest pain. was. still, there went to a then he said. "I'm gonna count to three and then			
A lingering cough			
wasn't as bad, developed a cough that wouldn't			
a cough at the time I if it was. It			
developed. Actually i don't know if	i	1	
won't leave home without. went on about my business um. l	2.46	2.35	19k9a6FHeX45
mind keeps my mind off of my troubles.			
are not intrited as iong as you know what your saturation level is it helps			
has COPD needs to have an oximeter you			
hurting your heart. I think anybody			
hormones as well. It was a really tough block, because you know you're.	3.19	3.03	uloY40MLXFY4
My eyes, you've got those new			
christmas and When vou've iust had a little eirl			
mean we were an convinced, my raminy and friends that i wouldn't even see			
how advanced my cancer was at the time i	0.58	0.49	wwiwjZw7d6o3
thinking or confusion or in the remember and So Lartually didn't			
rour, when they have metastases, either the Brain causing difficulties			
disease is already advanced at			
uncommon for patients with lung cancer to be diagnosed at a moment when	1.45	1.31	yV893Fx41IE3
feel better. The pain. started to			
that			
two weeks of starting to it	4.43	4.25	xY3B9nLm1u86
so um. and that's when i got the call. I remember the call.			
for lung cancer, right you just			
lot of primary care docs. Don'T think about screening vou know.			
was the first time because again i think a	24.0	DC'C	CHINZADXCDDII
•			
From the doctors to the testing that was done the			
And that was the best decision we ever made. Immediately you could feel a difference at Temple			
So they recommended the Temple Lung Center.			
We knew we needed a place that had more			
We didn't know what was going to happen. Was he going to be OK? Was he going to make it?	1.15	0.54	mWo-G3UQ-zU3
I nad no signs or symptoms . I know nothing . I did n't feel anything . www.is.moditers.Yvcs.huka			
doctor , get it quickly enough .			
nappened to be an aggressive cancer. Smoking small cell , which made you a			
was.Ijust thought my life was over.It	1.15	0.59	rQutyErqBG03
was just going I won't you're, not die and i was going. just don't let me			
screaming don't let me die and she mum			
I remember I was holding on to my mum			

Physical Health	PsychologicalHealth PsychologicalHealth	PhysicalHealth	SocialRelationship	Physical Health	PhysicalHealth Environment LevelOfIndependence	Ervironment PhysicalHealth Ervironment	Physical Health Physical Health
Respritoryliness	NegativeFeeling NegativeFeeling	PhysicalSymptoms	PersonalRelationship	Resprit ory liness	Energy Rehabilitation	Rehabilitation Respritoryliness Rehabilitation	Respritorylliness Respritorylliness

no symptoms

		Pulmo nary Fibrosis					scared					Pain						Family				لمعط	coba						Strength		Exercise	Treatment							Evercice	Copd		Exercise				Line (Jacob)		2 2						RespiratoryIllnessSymptoms
progressed, but it was also fortunate in	another way, because with that progression a tumor grew	testing. breath. It was noticeable that something	was radically the matter with me. It quickly started to get worse where	the pulmonary fibrosis just started to	when we have the rungs. A purification of the run we were we williams port said we can't do any more	for you here. All I knew is that I was	I was arraid what was gon na happen . I did n't want to lose mv hair .	I did n't want . I mean my youngest was	about to graduate	from college . Lives really crared	l did n't need to be scared . l	if certain you know , if air got in there	, and so it was all the bad things they	courne that court happen and himne . Okay i i can do this	they 're , the experts , not me	, i wo n't know it whatever it is and so	waking up . I was in a lot of pain ifelt	suppose yeah and iust arknowledging what the natient	say is also a form of reassurance.	You know acknowledging what the patient	say and acknowledging how their family	Of a sod thorsa is difficults broathing with	and there is unificatly breathing with some simple task by the 50. People who	perform physical labor , for instance , find	it hard to continue this kind of work	and most COPD patient can't walk for	long distance and ca n't lift object by the late 50 or each 60. Ouelity of	make the lung work harder and improves the strength	of them and improves their capacity	a well . To health professional that want to provide	good support to their patient	case with a condition. Each individual	is different . Things work differently for different	people particularly when it come	to managing your condition and exercise is always	is nutrition. I think that patient should be involved in	their treatment plan . I think	that they should be able to make suggestion. I think	they should be able to ask question . all of them and at the right time	, because I take 12 or 15 .	One thing , that 's very clear is that exercise is	important .	If you have COPD I make miscelf on to mix eversion class once a week	יו וומור ווולסבוו פס נס ווול בצבו מסב ממסס סווכב מ אבבורי	Some people do n't want to know anything	about what 's wrong with themwant .	ou more wound be parient over ou or with prior hund isease like CODD	or have some sort of immunocompromised state.	When I woke up and Aaron said to me you know	you have Legionnaires ' disease I wa like	`` what I thought that wa something back in the	seventy . " And he 's like `` no '' he ao `` that 's what vou ant ''	AND THE STIKE THO THE SO THAT SWITCH DOU SOL.	responsive . It 'S the place to be dealing
		0.51 1.1:				10	10.c					3.45 14.03						4.07 4.1				36.6 61.6	57.7 71.7						1.44 2.3										1 49 2 06								· · · · · · · · · · · · · · · · · · ·							2.16 2.36
		Ofl_01yFmME1 (001100-1	IIK-dgir9NKIO					bmvlK4qhAnU15 1:						warv8Ho_C1A4				C1124241100446							VyGp12XbPzs3										i7ITcn3d1a45							NIC CT & COMPARE								6U7OpgFojaw4

	To Fix Domain taxonomy (Parent)				groored was picked for the segment groored was picked for the segment			
SocialRelationship	PhysicalHealth	Psychological Health Environment	SocialRelationship	Physical Health	Physical Health Physical Health	SocialRebtionship	SocialRelationship	SocialRelationship
PersonalRelationship	Behaviour	PostiveFeeling Rehabilitation	PersonalRelationship	Respritoryilihess	Respritory litress Respritory litress	PersonalRelationship	PersonalRelationship	PersonalRelationship

		with dr. Culver, it 's a relationship he make. You 'feel special, jike you, are the only patient that he had and that 's	
	2.44 2.5	wonderful. I know lhave actronic respiratory limes sand I have to take care of it. So well and well and ha returned to all the different thing har returned to all the different thing especially taking care of her kid. I can	Child
	2.35 2.4	Development of the second s	Vaping parent???
	4.06 4.2	should be and more beard toor tearned should be an increased too treatment of even found a slight improvement in going up hill. So i Ve very glad of that . I wa asked me sk month . I still do quite a asked me sk month . I still do quite a but we also have a support group that we two exercise once a week and it 's veraliant' So theme 's with theme's choint	Glad Exercise
	1.26 1.4	because they can the perform the essential duty of their 4 job because of shortness of breath . This ha devastating socioeconomic consequence	Family
	1.13 1.4	for the patient and for the family. I have stair in my house and by the time I start at the bottom of the stair lo get to the top 1 'm out of breat lo get availability to ge and to sit down with three doctors my cardiochoracic surgeon my oncologist and my radiation oncologist. And then the three got together and they spoke with each	LungCancer
		The province with a stephonic succession of the province with a stage IIIB lung cancer. And I knew that that wap retty solutions being a concer and I knew that that wap retty solutions being a stage IIIB lung cancer. And I knew that that wap retty solutions being a concert and the read of the solution of the that was the needed to go into the mediastral area and the these would not pick up on an MRI and he needed to check those unake sure.	
	0.22 0.3	8 , and then throughout the body. There are five different caregory of pulmonary hypertension. The fourth group is labeled , choinc, thromboembolic, pulmonary hypertension, and that involves a group of patient who have had a blood dot in	Pulmon ay Hypertension Pulmon ay Hypertension
	2.38 2.5	the funds crudiation: while a stypically 1 Just getting your milk. Child : Okay. I wa back here for maybe two day before is tarted walking my daughter before is tarted walking my daughter to school in the moning. Dropping her off at kindergarten	Child
4	4.29 4.4	picing the cup in the attennoon if yust killer. 1 them pilay football and rugby if just me world. Really go down to see them. You just make me smile all the time because the grandchild love the bone off me. They alway you know, write me little letter and	Grandchild
	1.51 2.3	they would have been good . If anything 4 , colleagues at the School of Health and	Family

To IkSeminarite Tragging	
Environment Environment	PhysicalHealth

RespritoryIllness

Hope	
Exercise	
Parks	
Income	
accesstogreenspaces	
Mobility	
Mali in	
value	
	Ι
	Ι
Child	
Asthma	

, but there are differences, even within countries. For example in astralia. The life experiancy of indigenous speale is about 10 years lower than that of contribution contained
andigenous people
It was 83 years a difference of 33 years
of life. This can also vary between PH4JYF4Ns3 0.40 1.00 expectancy of a child born in sierra leone was S0 years whereas in uastralia it was S3 years a difference of 33 years
thing, when they think about the quality of the scope with the offferent doss and value systems in helping them to determine which factors they think are important when looking at their quality of file. This can also vary between PH4VF4Ns3 0.40 1.00 expectancy of a child born in sierra ierow as 59 years, a difference of 33 years it was 33 years, a difference of 33 years
BHAD 0.40 Intel entich self or hold onto the clients self or hold onto the clients and on his weak side. If the clients and on his weak side. If the means g55uGqgH82 0.20 0.34 Let's first look at what quality of life means means Foolp percent on many different means Foolp percent on many different means Foolp percent on many different off inter proping them to and value systems in helping them to determine which factors they think are interast them to use the important when used the there and the to an the analy H4YF4Ns3 0.40 1.00 expectancy of a child born in sierra If ite. This can also vary between there and syles syles a difference of a yas sin the startait
parme up terremeter with a gate cet is parme up terremeter with a gate cet is not being used, the care assistant can use the dients beit or hold onto the clients and on his weak side. If the clients and on his weak side. If the clients and on his weak side. If the means. Foelp perceive many different things when they think about the quality of life perplexit hercost they think are determine which factors they think are entropy the systems in helping them to determine which hercost they think are important when looking at their quality of life. This can also vary between the means. So years, whereas in australia theore was Sy years, a difference of 3 years it was Sy years, a difference of a years
backside of the clients waist with your pairs up remember when a gaits the til pairs up or remember when a gaits the til pairs up or clients and on the clients and on the clients and on the system many filter mans. People with have different ideas and value system in helping them to thing when they think are thing are the clients about the quality of life. People will have different ideas and value systems in helping them to determine which factors they think are important when looking at their quality of life. This can also vary between the system of the orn in sient a set whereas in australia to say sense different of a yaests it was 83 years difference of a yaests it was 83 years difference of a yaests it was 83 years difference of a yaests
 walk onlik of a network sold a a little behind grapt the gat Beit on the backside of the clients walst with your pairs of the arrest walst with your pairs of the clients bet or hold onto the client bet of the means. Poole percent many different data and value systems in helping them to determine which factors they think are important when looking at their data the duality of the hold the clients at their duality of the hold the clients at their duality of the hold the duality of the hold the clients at their duality of the hold the client at the hold the clients at their duality of the hold the clients at their duality of the hold the clients at the hold the clients of a substant the clients of a substant the hold the clients of a substant at the clients of a substant at the hold the clients of a substant at the clients of a substant
 starting on the same foot as the client walk on into orter weak soft and a little behind graps the gat Beit on the backled of the client side to not be backled of the client side in ond onto the particular the clients beit ond onto the clients beit on the ond onto the clients are not soft and an the clients are not provide the clients are not been as a side. If the collection of a soft are strain on the soft are and a not being used, the care assistant can use the clients are not been any officent things when they think about the quality of file. This can also vary between any Affrection of a difficant when also any between the client when also a soft are important when looking a their quality of file. This can also vary between the client was 30 years. Actied born in sient a lieu was 30 years a difficente of a 3 years a tilter out and the was 30 years.
Ristand up stand signifythe behind the client on the client's weaker side a client on the client's weaker side and a starting on the same foot as the client walk on this of her weaker side and a little behind gas the gat Beit on the backsled of the client walk on the packsled of the client walk on the client set of the client walk on the backsled of the client walk on the backsled of the client walk on the client walk on the client set of the client walk on the backsled of the client walk on the backsled of the client walk on the clients bet on the agait bet is not being used. The clients and on the man clients bet on the clients and on the man clients walk walk side. If the clients and on the client walk on the client walk on the client walk of the mans. Sopera, where were many different of a client where so the client duality of file. This can also vary between the client were and value systems in heiging their usely of the mans. So years, where so hard value so there are and value so the client walk and the value solution the client was solution.
Placement would be in forout of the client to a stand signity behind the client. If necessary, help the dirent to stand up stand signity behind the client on the client swalt with your plane up remember when a gat bet to the backside of the clients waits with your plane up the clients and a little behind gase, the reastsimit can use the client send to the clients and on his or her weaker side if the clients and no his weak side. If the clients and no his weak to her think about the quality of the stroken any different things when they think about the quality of the stroken where different deast and value systems in helping them to determine which factors they think are import that as systems. A different and a life. This can also vary between heavy filter in the about the quality of the stroken and value systems. The clients are able and the stroken and where sing a short short his direct deast and value systems and value systems. The bing them to determine which factors they think are import.
Place more work is most commonly. The placement such the intent of the client, if increasenty, help the client to stand up stand slip ty behind the client, if increasenty, help the client to stand up stand slip ty behind the client work is on the same foot as the client work is on the same foot as the client work is on the same foot as the client work is on the same foot as the client work is one to be placement work is one to be placement with a same sistent can use the dirent such and and a little behind gase the me assistant can use the dirent shear to all would be set to hold onto the clients arm on his work side. If the means , repolse precision and will be streng and have a site monto the clients arm on his work the transformation. provide the strengt with the source of the client strengt with the source of a strengt strengt with strengt with strengt strengt strengt strengt strengt with strengt stren
PARATIONING Postconde projective rulents anounty device. In this cases, a walker should be placement would be in form of the client, in the cases, a walker should be placement would be in form of the client on the client on the client of stand up stand sightly behind the client on the client of the client walker side a stand up stand sightly behind the client on the client of the client walk on his or her wasker side and a little behind grasp the gait bekt on the blacked of the dense wask with your placement would or the the client on the client of and the client on the client set or hold or the client and walk side. If the placement we also for a struction little behind grasp the gait bekt on the placement we also for a struction of the placement with a dout in the client and walk side. If the mans. Feople priceive many different things when they think about the client and value systems in helping then to determine which factors they think are important when looking at their quality of the struction struction and value systems and values systems and was S0 years. A different of a sustal is was S0 years. A different of a sustal is was S0 years. A difference of a sustal is was S0 years.
Advice for walking, make surer its postcomed property true fenets mobility device. In this case, a walker strond be for the electer most commony. The placement would be in front of the client of it necessary, help the dient to stand up the modility behind the client of it necessary, help the dient to stand up the modility behind the client of the client of stand a little behind gas the gal beit on the backed of the client of the device of the client of stand a little behind gas the gal beit on the backed of the client of the client on the client of stand a little behind gas the gal beit on the backed of the client stand a little behind gas the gal beit on the backed of the client stand a little behind gas the gal beit on the backed of the client stand a little behind gas the gal beit on the backed of the client stand a little behind gas the gal beit on the backed of the client stand and little behind gas the gal beit on the backed of the client stand and little behind gas the gal beit on the defines and which factors they think and and value systems in helping them to determine which factors they think and and value systems in helping them to determine which factors they think and and value systems in helping them to determine which factors they think and and value systems in helping them to determine which factors they think and and value systems in helping them to determine which factors they think and and value systems in helping them to determine which factors they think and and value systems in helping them to determine which factors they think and and value systems in helping them to determine which factors they think and and value systems in helping them to determine which factors they think and and value systems in which factors they think and and value systems in which factors they think and and value systems in which factors they think and and value systems and value systems and and value systems and value systems and and value systems and value systems and value systems and value sys
HgpOOMcdTE11 4.59 5.39 if the left is using an assister device for validity, make sure it is positioned properly the client must combility device for validity, make sure it is positioned properly the client must commony. The placed wherever it is must comfortable placed wherever it is must comfortable to the dirent must commony. The placement would be in front of the client of it recessary, help the idint en- client of the strand a liftle behind grass the dirent of the dirent of the strand a liftle behind grass the dirent way un placement would be in front of the client of the strand a liftle behind grass the dirent of the dirent of the dirent of the dirent when the dirent of the spate in holo dirent the dirent when the dirent the dirent when the dirent the dirent when the dirent of dirent when the dirent of dirent the dirent when the dirent of dirent the dirent when short here dirent of dirent when short here sites difference of al shars the was SD years difference of al shars the was SD years difference of al shars
HBpOOMGTE11 4.59 539 : 1; the clients burst and likely to die before the age of 65 than highrone Americans. Alle values and saststree device for walking, nases une its applicatione forgeting assisting and device in this scale, a walker should be device in this scale, a walker should be device in this scale, a walker should be placement would be in foot of the client in the client sonts comfortable for the client most commonly. The placement would be in foot at the client walk on this or her weaker side , stand up stand gliphty behind the client in this orker weaker side , stand up stand gliphty behind the client in the client swalt with your placement would be in foot at he client walk on this or her weaker side , stand up stand gliphty behind the client in the client swalt with your placement would be in foot at he client of the bliphty behind the client on the client swalt with your placement would be in foot at he client of the blipht guest, the care assistant can use the clients due and a little behind grass the gait bekt in placement when a gait bekt in or beging uset, the care assistant can use the clients due or hold onto the clients ann on its wask side. If the mass 's people priceive many different things when they think about the quality of life. This can abo vary between heat was 's Or as' was side in freetra in the system or hold onto in serra in the system side or hold onto in serra in the side and so wary between the or was S yeas; a difference of a shard a it was S yeas; a difference of a shard a it was S yeas; a difference of a shard a it was S yeas; a difference of a shard a it was S yeas; a difference of a shard a it was S yeas; a difference of a shard it was S yeas; a difference of a shard a it was S yeas; a difference of a shard a it was S yeas; was shard a it was s was was y difference of a shard it was S yeas; was was was was
In the dary ? Des: syntre genomood nave access to green spaces and to parks and to safe places to walk low income ? American a ethere terms more likely to die before the age of 65 than highncome Americans. My earliest work access to the second properity than the device for walking, make sure it is post to many and the inform of the placed wherever it is most comfortable for the derim 's washer's idea to a little behind grant light behind the client of the contact her left placed wherever it is most comfortable placed wherever it is most comfortable to the derim 's washer's idea that client on the client's washer's ide to the derim's washer's idea that in the behind grant spatial grant behind parts up the many different things when they think about the quality of life : his can abo vary between they think and the systems in helping them to determine which factors they think and there miss and value systems in helping them to determine which factors they think and the the systems in helping them to determine which factors they think and theore wiss 25 yeas, a difference of a yashers it was 83 yeas a difference of a yashers it was 83 yeas a difference of a yashers it was 83 yeas a difference of a yashers
QConfilhNmV45 2.03 2.26 of clime and sit steller wark uside in the day 2 Does your neighborhood have access to green spaces and to parks and to safe blacks or wark low income? HBpDOMc1TE11 4.59 5.39 : the lefter the sit space so the parks and to safe blacks or more itekty to die before the age of 65 than highincome Americans. My earliest work device for this case, a walker should be placed where erit is most conditrable placed where erit is most conditrable placed where erit is most conditable to the client most commony. The placed where erit is most conditable to the client or the client of stand up the inditable client, if necessary, help the client to stand up the inditable to the client or the client of the client or the client of so to as the client wark on his or the reader side at stand up the or the client or the client of the or the placed where erit is most conditable to the client of the client to client or the client of the or the placed where erits is most conditable to the client of the or the client or the client of the or the client or the client of the or the client or the client of the or the differ the stand at lifte behind gas the file or the dirent stand at lifte erits before where and if the to the dirent stand at lifte erits before the additerent of the dirent or the dirent of the or the dirent or the stand at use the quality of the stand when they think about the quality of the stand short in sterra hud value systems in helping them to determine which factors they think and the most SD vess, a difference of a stand to use steller many different to was SD vess, a difference of a stand to use the different disas a difference of a stand to use the different disas a difference of a stand the or the different disas a difference of a stand to use the different disas a difference of a stand to use the different disas a difference of a stand stand to use the the or the difference of a stand stand to
ACMFIHINMV45 2.03 2.26 of clime and/sit active walk worksofe in the day. Does your neighborhood have access to green space and to parids and to safe places to walk low income? HBpOOMcITE11 4.59 5.39 if the lefter the safe of the manore likely to die before the age of 65 than or safe places to walk low income? Americans are three inclines using an assister access to green space and to parids and to safe places to walk low income? Americans. My earliest work device for valids, make sure it is positioned properly the clients monoliny device for valids, make sure it is positioned properly the client on the place of the client mass commony. The placement would be in front of the client of it neas assister in this case. Americans are the monoly. The place of the client is substrationed and all client on the client of stand a liftle behind grave the plate infertor of the client was unown by the client on the client is washer side in the client stand and a liftle behind grave the are assister. If the client on the client should the quality our plates and on the same food as the client with on the same work may with your plates and on the same the quality of life. This can allow a plate the quality of life. This can also vary behind are in the same and sale systeme and with a stand and value systems in helping there in quality of life. This can also vary behind are and value systems in helping there in quality with a stand and value systems in helping there in quality where the plane work it was 83 years a difference of a shards it was 83 years a difference of a shards
university of essex. Her mature resurvation means and means and moment. The end of the and sist activity, physical activity, and the and sist safe to welk uside and the and sist safe to welk uside in the day? Desryour neglybohood have access to green space and to park and to safe blaces the welk low income? Americans. My sed instances to calle blaces the and sist safe access to park and to safe places to walk up income? Americans. My sed instances to calle blaces the and sist safe access to the access the ansist is postioned property the client mobility device in this case, a walker should be placed wherever its most confid table place and a stativity of the place and and a stativity our place and a initial behind graps the gait Beh to the place and and the same foot as the client walk on the client on state and a state side starting on the same foot as the client walk on the client on the client's walks whould place and a initial behind graps the gait Beh to the place and and the same foot as the client walk on the client on the client's walks whould place and a initial behind graps the gait Beh to the place and and the same foot as the client walk on the client as and whour we state. If the client on the client swals with your place and where systems in helping there in das and value systems in helping there in das and there systems in helping there in das and there s
 Parter and exercise science degree at the university of essex. Bort and exercise science degree at the university of essex. Hermain research interests include physical activity. And the neural environment. The environment. The activity of essex. Bort and exercise and the parts and the parts and to safe places to walk tow income? Bort and activity of essex. Sag the places to walk tow income? Antericans as three more likely to discrete age of 65 than Bort and a static activity to a safe places to walk tow income? Sag the before the age of 65 than Bort and to safe places towalk tow income? Antericans. My earlies work detection the electron of the places towalk work of the places of the direct to the same foot at the client of the client of
 Inist, 1 Cline to welce relation evaluate parked good moming, valence valence interact for the sport and exercise science degree at the university of essex. JConthNmW45 JConthNmW45<!--</td-->
WSPXX07333 0.55 1.10 speakers first, 1 liel kto vuocome valerie gladvell good moning, valerie valerie is the course director for the sport and exercise science degree at the university of sess. CmfHNINV45 2.03 2.04 liel kto vuocome valerie gladvell good moning, valerie valerie is the course director for the sport and exercise science degree at the university of sess. 2.10 2.25 of circme and is table to whole used the exercise to green spaces and to parks and to and the activity and the activity 2.25 of circme and is table work who income 7 Americans are three fines: more likely to de before the geo of 5 than to and the before the geo of 5 than to a file of the direct of the device in this case, a walker should be place of the direct of the controtable for the place term of work the the client on the client work defend a little behind grasp the galt behind the client on the client work of a stard sight behind the system in behing them to our being used, the client on the client on the client work of the to a stard threas stard went work of a stard sight the client on the client work of a stard sight the quality of the system in whole galt behind gass the was 83 years a different for the system in whole galt behind gass the was 83 years a different for the system in whole galt behind gass the was 83 years a different of a stard threas a different of a stard threas a different deverse tho was 83 years a difference a stard threas a difference
a mc an overvlew. This means that i'm first. 1 like to welcome valente first. 1 like to welcome valente protomoring, valente valente is the course director for the valente is the course director the valente is the course director for the valente is the course director for the valente is the course director for the valente is the course director the valente is the course director for the valente is the course director device. In this case, a valent should be placement would be in front of the course director the care assistant can device the mass commony. The placement would be in front of the client on the client of the context bis device. The case saids the client valente placed the director for a static device. The valente state the valent of the client on the client of the context bis device. The case saids the client valente and valente system the valent of the client of the client of the client of the client of the client of the client of the client of the client of the client of the client of the context bis device. The case saids the the valent valente and valente system the out valente system in the placetor of clients and valente system the quality of the client wale valente system the valente system in the placetor of a device system in the valent director of clients and valente system in the quality of the valente system in the valente system in the valente system in the valente system in the out valente system in the valente system in the out valente system state valeter the out vas as 3 vas as different of a state it
 months, with the new context we have um when i vas invited, they asted me to make a more an overview. This means that i m with the new context we have unwersity 0.55 1.10 packets good moning, valerie valer is the course director for the sport and exercise science dagre at the main research interests include physical activity. 2.03 1.20 for the and sit safe to walk working? 4.59 1.20 and the main research interests include physical activity. 3.216 of crime and is take to walk working? 4.59 and the main research interests include physical activity. a.3.2.3 2.03 2.04 of crime and is take to walk working? A.59 and the main research interests include physical activity. A.59 and the main research interests include physical activity. A.59 and the main research interests include to calle place to walk working? A.59 as the main research interest include physical activity of the interest of the context of a statical environment. The interest of the interest of a statical environment is a solution by the interest of the place divert is age of 65 than in the direct of the place divert is most condictable for the evice in this case, a walke sure it is postioned powerly the bind from of the placed wherever it is most condictable in the direct of the placed where were its most context at a solution bind device for a statical environment. The placed where were its most condictable in the direct of the client, if necessary, help the client to indicable in the direct of the client, when they then the client, if necessary is the taxing when they the direct of the client, if necessary is the taxing when they the direct of the client, when they the direct of the client, if necessary is the taxing when they the direct of the client, when they they they they they they they they
 MCMMIAL MCMMIAL MCMMALTELI MCSEXXD/333 MCSEXXD/334 MCSEXXXD/34 MCSEXXXD/
VEHIORTICI3 0.46 0.58 be very important in the next. In Dope in the next, versits, but especially in the next. Inter next. Inter next. WSEXXD/73Y3 0.55 11.01 be version invited, they asked me to make a more an overview. This means that im when i was: invited, they asked me to make a more an overview. This means that im when i was: invited, they asked me to make a more an overview. This means that im when i was: invited, they asked me to make a more an overview. This means that im when i was: invited, they asked me to make a more an overview. This means that im when i was: in the invited, version is and the sport and excites science degree at the variant encording the sport and excites science degree at the variant encord interests include physical activity. BOOM clTE11 4.59 2.256 for time and is it aske to walk controld be physical activity or dise activity or dise in the day? Does your neighborhood have access to green space and parts and to aske and parts and to aske are its physical activity to different in the day? Does your neighborhood have access to green space activity the physical activity or direct physical activity to a solution and its taske to walk controld be physical activity or different in the day? Does your neighborhood have access and to parts and a solution activity or different in the day? Does your neighborhood have access to green space and optical activity to different in the day? Does your neighborhood have access to green space activity our delevent is most control to the physical activity to different in the day? Does your neighborhood the physical activity out of physical activity to different is physical activity to different in the access and to parts and physical activity to different in this accs. A walker short to de
WEHKINTG3 0.46 0.038 Be very important in the next. WEHKINTG3 0.46 0.038 Be very important in the next. WESXXD/73Y3 0.55 Upope in the next. Bust spreaded in the next. WESXXD/73Y3 0.55 Upile in the next. Bust spreaded in the next. WESXXD/73Y3 0.55 Upile in the next. Bust spreaded in the next. WESXXD/73Y3 0.55 Upile in the next. Bust spreaded in the next. WESXXD/73Y3 0.55 Upile in the next. Bust spreaded in the next. WESXXD/73Y3 0.55 Upile in the next. Bust spreaded in the next. WESXD/73Y3 0.55 Upile in the next. Bust spreaded in the next. WESXD/73Y3 0.55 Upile in the next. Bust spreaded in the next. WESXD/73Y3 0.55 Upile in the next. Bust spreaded in the next. WESXD/73Y3 0.55 Upile in the next. Bust spreaded in the next. WESXD/73Y3 0.55 Upile in the next. Bust spreaded in the next. WESXD/75 0.50 Cord model in the next. Bust spread
VEHAFICI3 0.46 0.38 be very mop draft in the next, so underselve pools, their access to quality services. It trust that with the next, in the next, intersecting of the next, intersecting the next, is next, is next, intersecting the next, is next, is next, it is next
 WENDERICA OAG OAG<
 of calling lim by first mane, and loow that that and loow that the moves passions about social justice and partuality about improving for vulnerable people, their access to quality services. It that the next years , but especially moving for vulnerable people, their access to quality services. It that the next years and that the next have unwhen it was invited, they asked me to make a more an overview. This means that in when it was invited, they asked me to make a more an overview. This means that in when it was invited, they asked me to make a more an overview. This means that in when it was invited, they asked me to make a more an overview. This means that in when it was invited, they asked me to make a more an overview. This means that in this first, 1 d like to walcome valere is graveling or the sport and excites science dage at the equire barry of sess. 2.03 2.04 1.10 peokle. 2.03 2.04 1.10 peokle. 2.03 2.04 1.10 peokle. 2.03 2.04 1.10 peokle. 2.04 1.10 peokle. 2.03 2.24 of crime and sit safe to walk controller physicial activity of sess. 2.03 2.25 of crime and sit safe to walk controller physicial activity device the age of 65 than in the day 2 boes your neighborhood have access to great spaces and to paris and to state activity and activity of the sport and evence activities and a soluties physicial activity device the age of 65 than in the day 2 boes your neighborhood thave access to great spaces and to paris and to state activities and a soluties physicial activity device the age of 65 than in the day 2 boes your neighborhood thave access to great spaces and to paris and to state activities a soluties physicial activity device the age of 65 than in the day 2 boes your neighborhood thave access to great spaces and to paris and activities activities a soluties and activities activities activities activities activities activities activities activities activities activitities activities activities activities activities activitities ac
 quali, Lid not not have the privilege of conving Peter II may take the liberty or caling it mby first name. and Linow or caling it moving for vulnerable people, their access to quality services. Turns what the next is myost name and under a bout section in the next. MSPXXDJ73Y3 0.46 0.268 be very important in the next. MSPXXDJ73Y3 0.55 1.100 people, their access to quality value the next. MSPXXDJ73Y3 0.55 1.100 people, their access to quality value the next. MSPXXDJ73Y3 0.55 1.100 people, the next. MSPXXDJ73Y3 0.55 1.100 people, the next. MSPXXDJ73Y3 0.55 1.100 people the context we have un when university of essex. Her and account accelerate the second more gladowell good moring, value to a context we have university of essex. Her and account accelerate the second more or active active
 Itin coston, J'ul ke to pay my respects to the family of the lake Retrent of parts of calling him by first name, and know that the excession of the lake Peter fin name, and know that the excession of the lake Peter fin name, and know that the excession of the lake Peter fin name, and hour Peter fin name, and hour Peter fin name, and part unitrable people, that and thour Peter he was passionate about social justice and part unitrable people, that and thour Peter he was passionate about social justice and part unitrable people, that and thour Peter he was passionate about social justice and part unitrable people, that and thour Peter he was passionate about social justice and part unitrable people, that and thou hour each of quality services. This means that i m use motion in the mean context we have unit when i vas invited, they asked me to make a more an overview. This means that i m university of esset. 2.03 1.10 speakers 2.03 1.10 speakers 2.03 1.10 speakers and the transmolutie provide the evolution of the speakers include to a set of the natural environment. The sport and evolution of the speakers include to a set of the natural environment. The sport and evolution of the provide property the clears and the and to a set of a speaker side. If the provide provide the evolution of the provide the evolution of the provide the evolution of the provide to a set of the provide the evolution of the provide evolutis and the provide evolutis and the provide evolution of the
Kind Scheduler And the plane the privilege of the late freet in may take the latery of calling him by first name, and throw the intervent from what i vee expensions that is my take the latery of calling him by first name, and throw that that is my take the latery of calling him by first name, and throw that that is my take the latery of calling him by first name, and throw that that is my take the latery of calling him by first name, and throw that that is my take the latery of calling him by first name, and throw that that is my take the latery of calling him by first name, and throw that that is my take the duality services. It there exceeds an unit per exceed method with the next. MSPAXD/33Y3 0.46 1.058 be very mportant in the next. MSPAXD/33Y3 0.45 1.010 peters in the next reases to quality services. It there access to quality services. It the access to quality services in the next movel when it was invited, they asked me to make a more interest in the next. MSPAXD/33Y3 0.55 1.10 goteskers And the natural environment. The university of essets. 1.01 goteskers And the natural environment. The university of essets. 2.03 and the natural environment. The private access to green passes to walk tow income? And the natural environment. The private the university of essets. 2.03 and the natural environment. The private access to green passes to walk tow income? And the natural environment. The private activity device the age of 5 than university device. 2.03 and the natural environment. The private activity device to age of 5 than univers in the next is more acti
 The land on which we are meeting and on which we are meeting and on the langer of the l

		To Ik -Semantic Tagping Algorith			
PhysicalHealth PhysicalHealth	Environment	Environment	PhysicalHealth PhysicalHealth PhysicalHealth PhysicalHealth PhysicalHealth PhysicalHealth PhysicalHealth	Physical Health Psychological Health	Psychological Health Level Of Independence
PhysicalSymptoms			Physical Symptoms		
RespritoryIllness RespiratoryIllnessSymptoms			Physicalsymptoms Physicalsymptoms Respritorylliness Physicalsymptoms Pain Physicalsymptoms	PhysicalSymptoms CognitiveFunction	CognitiveFunction

<u>.</u>

disorders like bronchitis, asthma cemphysemo or lung cancer. Cardiovascular problems like artery, blockage heart attacks, and it can be linked to allergies like whereing, rashes and allergy reactions. But what i found where you looked at the professionals and their roles responsibilities. This is kind of a parallel to that, but it 's more about the place rather than the professional.	so when it chuices to beatin and social care in England and Worthern Ireland . Scotland was there are four key branches of the NHS. So the NHS is whu of the What are your strengths , your stressors , and what supports do you currently have available to you ? Are there any changes or adjustments that need to be made in your file or	we revironment? We then work out with you and the medical or mutidiscipinary team medical or mutidiscipinary team new we can support you for excample, do experience of about 20 or 25 years of doing that, that i want to bring to you the message this morning that pan is a supease. Now, most of the time you think of pain is a symptom of a disease and that 's	True, most on the time it: resymptom of a tumor of an infection or an inflammation or an operation but about 10 of the time after the pattert has recovered from One of those vents, pain persists. It So right at the top You know what is chronic pain . what do those wow works annyeah , and how common is that yeah , well vereyone 's experienced pain and we need pain is 'is a protector	pain Is something that it turns out. Our You ask people to call up the memory of what you ask back then, what you velte back then. And then what you do follow your fingers is that you ask people to follow your fingers as you move your fingers from side to side. "Man, stop this crazy nonsense, this is a crazy method." And then I got to see some of the results, I golie. "Wow, this is interesting." Wigging your fingers and full of people's vess Wigging your fingers and full of people's vess	as they think about the trauma made the traum away. Ihaven tracht of prepared with them beforehand , they may have a waiting list or soon The Department of Work and Pensions does have its own visiting service, so it's always worth thinking of contacting them as to whether there 's ast ownether they can come and ill the	form out foryou anyway. We ful just get responsible of judgment. You have to use your Judgment. For example, if a ditentis depending on you for so namy activities of daily living. You have to follow their the best choice. You have to follow their own selection: There were you have to follow : if s not we give them our choice but it is there twice we have to follow.
0.35	1.27	0.59	0.37	2.53	2.03	4.56
0.20	1.15	0.36	0.19	2.19	1.46	4.14
VGPACN2P_'42	89bvFq4XtW06	CZ0gliwjk_02	kWf7utae7.cg2	2oZT8Hq1646	12Wt0D0021g4	DmteRv5sVd9

PhysicalHealth	PhysicalHealth	Physical Health	SocialRelationship Environment Environment Environment	PsychologicalHealth PsychologicalHealth	SocialRelationship SocialRelationship	PsychologicalHealth	PsychologicaHealth
Behaviour	Behaviour	Behaviour	PersonalRelationship EconomicStatus EconomicStatus EconomicStatus	NegativeFeeling NegativeFeeling	PersonalRelatonship PersonalRelatonship	NegativeFeeling	NegativeFeeling

Alcohol Alcohol	Child Child Incertion Coupertion	Fear Anger	Parents Parents Fear	Fear
and for that we still have to make the judgment. What could be better for them ? do about it ? Well, the who's global strategy endorsed well, the who's global strategy endorsed parts to reduce the harmful use of alobol. Unfortunately , jimplementation of the strategy is uneven , implementation of the strategy is uneven , mean index a shot of voldsa, When you drink a shot of voldsa, When you drink a shot of voldsa, When you drink a shot of voldsa. 20% of the alcohol is abonded by the stomach, where it goes into your boodstream. then to your liver to be metabolized. Ethanol gest into eals by passive diffusion. The buzz, the slight headache and the hangover the next day are all pretty common.	Howeer, regularly dinking alcohol can health effecting your body. These health effects show up quickly for hose who binge dink but can show up acqually for proper bwd or firk regularly geven if they do n't drink too much. Here 'S For example, in 2016 the life expectancy years, while in australia twa 83 years , but even within australia. The life years, which in australia in the id JOL even within australia. The life to years lower than that of nonindigenous people	and even within a specife group of people, health can change based on a person's income level of education or occupation . The powers of both positivity and negativity are stronger than you may know. Lots of studies have shown that people with a positive out look on life tand to live longer, more fulfilling lives	than their mean and grumpy neighbors. Fear, anger, and other more difficult emotions related to this thing called stress. The stress l put a bit of stress on well for me, and year ways watered to do well for me, and years where a parents, so and the stress of earns, and it was also kind of hard because my parents were going through a fitte bit. A troubled time of hard because my parents were more and it was also kind of hard because my parents were more and it was also kind of hard because my parents were more and it was also kind of hard because my parents were mer provide the liet st March last April. Yeah take so around That means you offen suffer quite a lot, but that's help you survie. Feelings also tend to be more unpleasant if you flight them fryou allow here to be three, and the more and them fryou allow them to be three.	express many different feelings. Like feeling hopeless, irritated, anxious, exited, lonely, embarrased, suspicious, content and so on. These feeling suspicious, springs from fear. They include care, he capitalized these. So you see a re care be caused prief and panic
0.37	0.50	0.57	4.04	1.29
0.24 0.18 0.18 0.17	0.25	0.43	3.45	1.11
skýpmoaGIZ KWOpV9_kUUM2 WOZFIÞKXX02	zorgupodkjzvoz	4tbStXPOwik4	7LD8IC4NqXM10 SIOJpprbfeE4	Yh1y37:5044

NegativeFeeling	PsychologicalHealth
NegativeFeeling	PsychologicalHealth
NegativeFeeling NegativeFeeling	PsychologicalHealth PsychologicalHealth
0	LevelOfIndependence
NegativeFeeling	PsychologicalHealth
NegativeFeeling	PsychologicalHealth
NegativeFeeling	Psychological Health
0	0
NegativeFeeling	PsychologicalHealth
NegativeFeeling	PsychologicalHealth
NegativeFeeling	PsychologicalHealth
CognitiveFunction	PsychologicalHealth
-	
Ivegativereeiing	Psychologicalmealth
NegativeFeeling	PsychologicalHealth
Rest	Phy sical Health
PersonalRelationship	SocialRelationship
RespritoryIllness	Physical Health
DarconalBalationchin	SocialBalationshin
NegativeFeeling	PsychologicalHealth
NegativeFeeling	PsychologicalHealth
NegativeFeeling	Psy chological Health
Rest	Physical Health
NegativeFeeling	PsychologicalHealth
0	and a second sec

NegativeFeeling NegativeFeeling NegativeFeeling	PsychologicalHealth PsychologicalHealth PsychologicalHealth
sonalife lat briship sonalife lat briship sonalife lat briship SocialActivity VegativeFeeling VegativeFeeling	So dalRelationship So dalRelationship So dalRelationship So dalRelationship Pay chological Health Psy chological Health Psy chological Health
VegativeFeeling VegativeFeeling	Psychological Heal th Psychological Heal th Environment
ysicalSymptoms	Physical Health
sicalEnvironment	Environment
sonal Relationship sonal Relationship	SocialRelationship SocialRelationship

ere Descention	iors.	s of	Hopelessness oved.		Mom Mom	Family Community		Anxiety	Anxiety Anxiety				Danracsion	Depression				Pain			Safety		re Family	Husband
interfen	al factor	selings c	ce eniov	ficulty		8						E				_				_	a		av, ve	ы В

And given Emily's age and observations or in have to conclude that we can't allow to keep making tea for herself every mornin	5.07	4.41	2qUclQvnBC07
, we're presenting on home safety modifications . But we have a couple goals			
um , i 'm marlena and i 'm also presenting			
to be here today , i 'm just gon na so um as lori said			
over time and short distances . Therefore Welcome marlena hi everyone um , it 's nice	0.54	0.40	BOKCzrUBUE3
comes to moving aroung que to significant hip pain , which increases			
example , Mary scores Phil points when it			
Because of this people with the sium condition . May get different outcomes , for			
them on a daily basis and us in Labour the condition or disability itself .			
health condition or disability affects			I
possible . אס you statiu שא צבוונוץ נמוא As the assessment measures , high their	3.02	2.43	XOxW_4kJqa89
loved one as much independence as			
create a safe environment . Allow your			
Some things to remember are : go easy and			
I nererore , approacn your loved one with respect , patience and in a gentle way .	/ C'N	0.40	IVUTJY 3 DCE 2
and over time the normal day . Today			
recognize			
, the worst thing about depression is that when you to in it it is band to			
talk a little bit about his experience with depression	2.09	1.54	r0iMPxjRA2U7
and probiotics , are patented to deliver results . You ca n't get anywhere else			
, and Our magnesium , as well as our calciur			
, relieving herbs , vitamins and minerals prohiotics and enzymes			
deliver results , and it contains 100 stress and anxiety			
ensures nutrients absorb fast and			
, and you will feel results quickly because our onti bland delivery system			
, because we use whole food herbs and nutrients			
It relieves stress and anxiety , naturally			
, and this is why you need our stress and anxiety relief pack .			
then you are jeopardizing your health	1.04	0.27	RxSsPSNGnkw2
part of my job , as the social worker was			
started seeing a trierapist which is great			
, the family took our suggestion and they			
waiting in the waiting room with baited breath			
laying next to her . In a cot . when this patient was in the or mom . was			
So , while this patient was spending a month or more in the hospital , mom was	2.14	1.56	Y_M3elAw5
concentrating or making decisions.			
have trouble sleeping, experience			
that persists for weeks or months. They may lose interest in activities they onc			
sadness, hopelessness, and helplessness			
Doold with depending office office of the			
Depression can be caused by a variety of psychological, biological, and environmenta			
with daily life, work, and relationships.			
It's a common, but serious illness that can ir			

Psychological tea th	Psychologicaliteatth	PsychologicalHealth PsychologicalHealth	Physical Health Psychological Health Psychological Health	Physical Health Physical Health Physical Health	PhysicalHealth
			PhysicalSymptom	Physical Symptom Physical Symptom	
Cognitive Function	Negativefeeling	NegativeFeeling NegativeFeeling	RespiratoryulinessSymptoms MegativeFeeling NegativeFeeling	RepiratoryIlhessSymptoms RespiratoryIlhessSymptoms RespiratoryIlhessSymptoms	Respritory/liness

Memory	Anxiety	Anviety Anviety Breathlessness Anviety Fear	Breathless ness Breathless ness Breathless ness Breathless ness	lungDisease
Well. I think there is another way. I rang Emily's niece what? Consulting family is standard procedure too, jan't it Fank? O. Her niece told me that this teat ping has become a bit of a ritual apparently all started when her husband Charlie was alve. She'd get up every moning, make aro of tea then apparently all started when her husband Charlie was alve. She'd get up every moning, make a pool tea then those pottures of entire lows. Of the choices that people make and how those choices who thou frorthem. Those pottures are almost impossible to get. West of what we know about human life we know about human life	and as we know, mindsight is anything but Zuy Zu. We forget vast amounts of what happens to us in life, and sometimes memory is downight creative. 0.37 1'm nervous before an exam. You know, or 'infeling know, before, and with the video, but anakety is so much more than just beng in the head. it's also very physical, vit's metal and physical because it's focused on one thing, which is survival now. What this means is	0.30 final if mon the direct sides, so and share these with you and basically relate with you and basically relate with you and basically relate with you guys. So, if you 're eaady to kitk anekery hou burk, snash the like button down below and beave me a comment. Today's question of the day, what type of music do you like to listen to when the like the viologing of the day, you want 10.2 you breath feater or typ increase the flow of oxygenrich all into the lungs breathlesness, is n't just a phycal sympton. It's also a feeling that affects the way we think and act. If breathlesnes, you can feel very and act. If breathless, you can feel very and act.	paricky or overwhelmed, or that your chest is tight or like you 're sufficiaring , and you reed to take deep breaths, but you can from ike breathing is just such hard work and enhausing getting out or breath can be frightening and make you feel - rightening and make you feel - interase live breathless you feel - nircease live breathless you feel - nircease live you feel - nircease fur her with enhouse of the are of breathlessness, and thesa re optionsity our lungs and that certainly could cause breathlessness and that 's more of a concern in people who are a heavy smokers or people	who have a history of ashma people. Who'Ve been posed to ablestos and 1 suppose, if you are getting breathless, then it is very important. You What amazes me about a bit of patients I meet with 1.11 advanced lung disease is how accustomed many of them become to the limitations of their breathing.
106	0.23	0.30	0 8	0.38
8 Kiku TCFvu3	5tDGBBy c8tg2	VCSutY7IG382 4XqHVIZPYb82	Vs6DW155Y502	2 aulobkUzSQ2

		ignorent was picked for the s		
Physical Health Physical Health Physical Healt h	PhysicalH ealth PhysicalHealth	PersonalValuesAndBeliefs PersonalValuesAndBeliefs	Environment	Ervironment LevelOfindependence
	PhysicalSymptoms			
Respritory(liness Respritory(liness Respritory(liness	Respiratorylliness	Spirituality religion	Acquiringinformation	EconomicStatus

Copd Copd Copd PulmonaryHypertension	Cough	Belief budism		Training	Education DailyLuving
Some of them become very used to walking ten feet and it they at least tell me sometimes that it doesn't bother them and some people it does bother. So really it's taking a very personal, very patientfocused approach to how does it affect means you're not able to breathe to a means you're not able to breathe to a Another name for coold is emphysema . Some of the symptoms - there's quite a few. The most common ones are shortness of	Coughsten . Coughsten services the problem with copd is it's sa progressive condition that eventually leads to pulmonary hypertension, so yoou're getting blood pressure i'That's going up in your distressing or all chest pain , respiratory symptoms are common and I think that's to ned it the reasons we ve the because kowar this respiratory receit ado go down this respiratory relatively or up or un's creater respiratory symptoms on our screens creatively or up or our screens sympthe or not you're reading about somehody having A could n'Twitter or courte of the our of wells	someoner when by a worker or that we are sub- someone who 's worthed about wheezes that they might be experimenting is that it would a so we want to make sure that every book's understanding, the vocabulary and what we prove want to make sure that every understanding, the vocabulary and what we understanding, the vocabulary and what we understanding, the vocabulary and what we understanding, the vocabulary and what we even before this prough. Buddhism has roots in a watch is now known as laining, which i may eventually what is now known as laining, which i may eventually are the time, india put its smail focus on tradition, based on texts called the vedas, which were andernt Sanskrittexts which informed ther belief's and	made up a foundation for the Hinduism faith. There was a bit of division, as a man named Sudhartha Gauenna began to teach a shift from this normal way of being, based on discipline and abstaining from mindugences. See, he was a bit shettered, and as he discovered more about the world, he began to teach on	the lessons he's learned. Julogical care, Huntingdon's motor neuron bug as y mutting tectorosis. It puts a huge amount of stain on the company both in terms of fainting again the rate staff, but also in terms of training , which has a huge cost involved in that as wel. Wel. the girds of have a haf an	them into the Q.& Abox on the bottom question site of your screen. If your question site Barbara will be covering them at the end of the covering them at the end of the be Barbara West with capital nursing dedication
4 0.37	6 2.40	6		3.10	1 1.07
2 0.1	2.0	e. O		2.5	0.2.0
bPNdAGBMph0.	FDS80bF6085	j[tVtakws]w2		ITV53IhSwKk5	uZdWafHAXd2 j11CQGyloXo4

PersonalValuesAndBellefs	Phy sicalH ealth	Environment PhysicalHealth PhysicalHealth	Environment Environment	Environment	Physical Health	Environment Environment Environment Environment LevelOfIndependence LevelOfIndependence
	Physica ISymptoms	PhysicalSymptoms				
	Pain	Acquiring Information Physical Symptoms Pain		PhysicalEnvironment		

Environment	Physical Health Psychological Health Physical Health	PhysicalHealth PhysicalHealth PhysicalHealth PhysicalHealth	Environment Envisionment PhysicalHealth PhysicalHealth PhysicalHealth PhysicalHealth PhysicalHealth PhysicalHealth	Environment
		hhysical Symptom Physical Symptom Physical Symptom	Physical Symptom:	
Rehabilitation	Behavlour NegativeFeeling Rest	RespiratorylilinessSymptoms RespiratorylilinessSymptoms RespiratorylilinessSymptoms RespiratorylilinessSymptoms	Respritorylliness Respritorylliness Respritorylliness Respritorylliness Respritorylliness Respritorylliness Respritorylliness Respritoryliness Respritoryliness	Economicŝtatus

	_
	_
	_
	_
	_
	_
	-
Exercise	
	_
	_
Diet	_
Depression	_
	_
	_
Sleep	-
Breathless ness	
Cough	
Dyspnea	
Breathlessness	
Do kokiliteetion	
Cond	
Asthma	
LungDisease	
LU1801COC	
Covid	
Covid	
Cough	
Cough	_
	_
	_
	_
Education	
	_

medical legal process we 'lib be covering the rehabilitation Code and also the serious injury code and also the serious injury code payments and so the importance of interim legal advice legal advice covering the arresh in terms of access to realid a treatment and major trauma . Trealid a treatment and major trauma . Trealid a treatment and major trauma . Trealid a treatment where you start from it's not like. If you have no activity at all there thruss a week immediate v. If you	get those effects or if you already have to increase that even more to get the maximum benefits. The other key question maximum benefits. The other key question draft det. The hother with that is skty percent of the solid weight of your brain is fat and low fat diets can actually trigget depression, and sol like healtty' fat fats, although	and scen fish and sco swordfish, is out and never would have that. It's just loaded with Mercury And while omegas are good fast for your brain, longtert consumption of other fast, like trans and staurated fast.	may compromise brain health. Meanwhile, proteins and amino adds, the building block nutrients of growth and development, Amino adds contain the precursors to Amino adds contain the precursors to neurotransmittes, neurotransmittes, affecting things like mood,	attentivenes. and weight. and weight. history. you may find that patients present with multiple respiratory present with multiple respiratory breathiestness and a cough.	top of the list of spreador in the new point of the list of spreador in the list of spreador in the list of spreador in the list of the li	chronic respiratory disease majory COPD chronic respiratory disease majory COPD and today i'm going to tak about. Did we video episode and it was recorded before covid considerations, to it does not include covid considerations, in that episode. But if you want to think about the other if you want to think about the other definition have they a priced coult fuell	In the contract protect on a multiple contract protect on a multiple contract protect on a multiple contract protect on the contract protect on the contract on the contract protect on the contract protect on the contract protect on the contract protect on the contract o
0.57	1.22	1.47		2.03	1.32	0.54	1.44
0.43	1.01	1.16		1.45	1.13	0.36	1.17
v6n9snBQTic3	etW5Ycgdjyo4	xyQY8ang6g4		ONPVDBXnl2A7	uT4zD8knsE02	RQhfVbYaCws3	HP21va21LM4

CognitiveFunction	PsychologicalHealth
	LevelOfIndependence
Behaviour Behaviour	Physical Health Physical Health
Behaviour	Dhyveical Health
Deligning	ruy sicam calu
Rehaviour	Phy sical Health
Behaviour	Physical Health
RespritoryI liness RespritoryI liness	Physical Health Physical Health
	and the second se
PersonalRelationship	SocialRelationship
RespritoryIllness	Phy sical Health
	Environment
	Environment
	Environment
PhysicalEnvironment	Environment

Thinking Teatment Drug Alcohol	Diet	Diet	Tumor Tumor Tumor Relative Tumor		Circuit
high art because they have cultural capital, which means the professional capital, which means the professional capital and social sitils necessary to attain the property, power and presige to get ahead socially low culture or popular culture. The one low culture or popular culture. The one we stay cuto con megative throughts and limited mindsets ? So much yeah, that 's a big question. So um stories we've been told about, what's going on in our skull interesting, Joky. So the first thing is going on in our skull interesting, Joky. So the first thing is Bloodletting, purging, fasting Bloodletting, purging, fasting solidy, the welliness industry.	They've taken the word "detox," the medical treatment for people with drug and alcohol addiction. and used it to apply to market cleanses. and used it to apply to market cleanses. With their electrolyte needs with their electrolyte needs and is perfectly suited for people who are following a ketogenic, low carbohydrate, or peleo diet.	LIMNI contrains a sciencebacked electrolyte ratio 201 1000 milligrams of sodium, and 60 milligrams of prassium. experience on the running track for over five years jate only uncooked fruits, vegetables, nuts seeds and spouts an experience that inspired my first book. The price diet connector sells to your soul	The struct struct and on and the seeses tested my restlence, juid of the feel as good, physically or emotionally as when i first began a vegan dist. The defining Wow. Separate from the gut microbiome, they have their own microbiome and they took this further. They were taking a look at pancreatic cancer, which is one of our most deadly cancers. And they analyzed the microbiome of the tumor,	not the person's gut microbiome, of the turnor, and discovered that a person who has a low diversity, within the turnor microbiome has a worse prognosis or a lower likelihood of survival relates to the person whose turnor has a high diversity. Survive what happens to the health of person what happens to the aeth of fish. Environmental health is not just shows: comes back to you and whether the environment, you are part of is helping.	You say healthy or making you sick. What here the terms : physical , chemical , biological , social and psychosodia factors in the environment. These specific terms are recurring and you will see them in various definitions that we "I go over today
0.47 1.00	5.19	2.29 2.4	6.12 6.3	0.35	0.54 1.4
MalxBucEyg4 DESCCJSQSK13	15K2pMqU20k18	ydyaBfY6aY5	9JDsQUIUEA020	V5Pk_Qwejao2	NFHL_Zb2xo2

PersonalRelationship PersonalRelationship		SocialRelationship SocialRelationship
		Environment
Rest		Physical Health
		Physical Health
		Phy sical Health
		PhysicalHealth
		Alasses (1) and a late
Pain	Physicalsymptoms	PhysicalHealth
NegativeFeeling		Psychological Health
CognitiveFunction		PsychologicalHealth
NegativeFeeling		PsychologicalHealth
		PhysicalHealth
		- 10 - 10 -
Rest		PhysicalHealth

				Friend Family	Rehabilitation		Sleep	Physical Symptoms Physical Symptoms Physical Symptoms	Disconfort	Amilety Thinking Anviety Anviety PhysicalSymptoms	Sleep
: clean air, stable climate, adequate water, sanitation and hygiene, safe use of chemicals protection from adation. Iheatiny and safe protection from agricultural practices heath's supporting ches and built environments are all Prerequisite for good health	, healthy environments could prevent almost one quarter of the global burden of disease according	to the whole cov19 pandemic is a further reminder of the delicate relationship between people and	our planet . Another very important agency in the united states is atsdr , which stands for agency for	out by capita . The assessment may take around an hour , but you 'll get as much mar aroun each o bring photographic ID with You need to bring photographic ID with you , such as your passport or driving cares supont worker friend or driving	member why you to support you and they member why you to support you and they Contact your state's department of vocational or rehabilitation services for an about the steps it takes to receive a driver's license.	But they also have programs where the lessons and modifications are at little or no cost to you. Once you have figured out what general You will need to do some research on the senerific of You will need to do some research on the senerific of	your modifications. worse and worse as the evening went on . and then the evening went on . and then it was particularly bad in bed and could rit get a wink of sleep all night. Othary .can you tell me a bit	anything that we we teach if you 're dealing with physical symptoms. If you 're dealing with something that you know u've getosmost errouts physical symptoms teac errouts physical consulting with a health professional .	addition to seeing checking sur- addition to seeing checking making sure have, you know you want to be checking in with a doctor on ight. And i also had this discomfort in my back. Just didn't feel like doing there several weeks of this.	Major depressive disorder, or MDD, everything is fine the many different ways. everything is fine they do is they label you with and what they do is they label you with anxery, now you ve grown up thinking. Anxiety is all in your head. In fact, you we can anxer you have any not even be symptoms 2 in fact. you way not even be symptoms 2 in fact. you way not even be	sy mptoms is in fact, you may not even be them
				1.08	1.42		0.37	1.15	0.48	0.27	0.29
				0.49	1.20		0.23	0.45	0.27	71.0	0.17
				vmZVbXCv6wM3	A066kt0LZA4		elRIm6B0zP42	qSVVWE_2Bnc2	MNp9bmNI602	r8qsGPwgdAw2	Cg_GW7yhq202

PhysicalHealth	PhysicalHealth PhysicalHealth PhysicalHealth PhysicalHealth	Physical Health Physical Health	Physical Headth Bsychological Headth Psychological Headth Psychological Headth	
Behaviour	Behaviour Behaviour Behaviour Behaviour	Behaviour Behaviour	Respriterytliness Negativefeeling Negativefeeling Negativefeeling	
				New concepts suggested by Annotator 4
Aconol drugs??	Drug Drug Drug Drug	Smoking Smoking	Preumonia Anxiety Fear Depresion	294 To fix - Domain taxonomy
: physical activity releases, feelgood hormores alled enotophins which hep us steep and feel better. It also improves our physical fitness which tends to make people feel better in general. Even small a mounts of regular 0.47 substance in amounts or with methods that are harmfut for themselves or others. Some of the most commonly used drugs are eatonhol, costine , opoids, bencodarzepines cannabis, barbiturates, amphetamine , failuchogens, inhalants clud drugs and	other substances. 1.02 The slowest method is to take a drug orally because it must be absorbed by our digestive system before it takes effect. Inhaling a drug gets it into the bloodstream faster. And injecting a drug intravenously works quickly too because it pumps the chemicals directly into the blood.	Once there, the drug quickly reaches the gates of its destination, the brain. That's one of the reasons smoking can lead to oxygen 1.43 deprivation and shortness of breath. Within about 10 seconds, Within about 10 seconds, Within about 10 seconds, the brain. The brain. The brain. The brain. The brain second second and other neuroframmitters that create the pleations.	Amonga may neurophysical 0.45 dipsnia can result from physical lillnesses. for example preumonia, chronic obstructive pulmonary disease or organ failures. It can also be caused by psychological issues or organ failures ankety fear, depression .	
b6Dte96WdqM2 0.26	8qk0hkuXOC83 0.44	1.26 1.26	k8AuX2qQNa02 0.30	

To fix - Semantic Tagging Algorithm gnore - it was picked for the segment Total

Appendix N

Google Form for VIN-L Evaluation (Second Domain)

The Google Form designed and shared with the domain experts to evaluate VIN-L output in the second domain.

Hea Pati (Res	Health Related Quality of Life Needs of Patients With Chronic Illnesses (Respiratory Illness) - Expert Feedback					
videonar	videonarrativesinadvance@gmail.com Switch accounts					
Next	Page 1 of 17	Clear form				

Introduction

Thank you for helping with evaluating the video narratives that have been generated from Abrar Mohammed's PhD project conducted at the University of Leeds, UK.

Your evaluation is essential to assess the potential benefits of the generated video narratives

to help build awareness of health-related quality of life needs of patients living with chronic respiratory illnesses.

In this evaluation, you will **spend approximately 90 min** watching **video narratives with patient stories about living with a chronic respiratory Illness**. Video narratives are combinations of video segments which have been automatically characterised and linked together by computer algorithms. You will be given a scoring sheet for each video narrative and will be asked for your feedback on the usability of the narratives.

You will assess 16 video narratives (totalling 64 minutes video content) which are aimed at building awareness of health-related quality of life needs of patients living with chronic respiratory Illness. This will include: Physical Health, Environment, Psychological Health, Level of Independence, Personal Values and Beliefs, and Social Relationships.

You will have to log on with your **Google account**. When you fill out a Google Form in your Google account, **your progress is automatically saved**. This means if you can't complete the form or need to switch devices, your earlier responses will be saved, if you do not press SUBMIT. When you come back to the form, you will start from where you stopped last time. If you press SUBMIT and would like to come back to the form, you will start from the beginning, just be careful, please!

The study is **fully anonymous** - although you will logon with your Google account, the responses **do not save the email of the responders**.

Thank you very much for your help.

If you have any questions please contact:

- Abrar Mohammad <a.mohammed1@leeds.ac.uk>
- Vania Dimitrova <v.g.dimitrova@leeds.ac.uk>

Back	Next	Page 2 of 17	Clear form

Your involvement in this study will include the following steps:

- 1. Read and accept the consent form.
- 2. Complete a short pre-study section about your profile.
- 3. Watch several suggested video narratives.
- 4. Give your feedback on the video narratives you have watched and their description.
- 5. Give a short video summary.

 Answer questions providing feedback on the usability and usefulness of the recommended video narratives for learning about Health Related Quality of Life-Respiratory Illness areas.

7. Repeat steps 3-6 with all four types video narratives.

CONSENT

I have read and understood the information about the evaluation of research on using video narratives for learning based on user interaction and video content, conducted at the School of Computing, University of Leeds, UK. On this basis I agree to participate in this online survey, and I consent to publication of the results of the evaluation with the understanding that confidentiality will be preserved. I also understand that I may at any time withdraw from the online survey, including withdrawal of any information I have provided. I note that the survey is fully complaint with General Data Protection Regulation (GDPR). Also, this survey has been reviewed and approved by the Research Ethics Committee of the Faculty of Engineering and Physical Sciences , University of Leeds, UK.

Agre	e								
Back	Next	 Page 3 of 17	Clear form						
Your Prof	file								
------------------------	-------------------------------------	----------------------	--------------------------	------------------------	---	--	------------	--	--
Please d	Please describe briefly what you do								
Your answ	ver								
What is y a chronic	our awa: illness?	reness of he	ealth-relate Personal	d quality of Some	life needs of Some	patients liv Experience as	ing with		
	Not at all	General knowledge	experience as patient	experience as carer	experience as health professional	researcher in patient quality of life	Other		
Answer									
Back	Nex	t 💼			Page 4 of 17		Clear form		

To help you understand the structure of the generated Video Narratives, in the Image below, you will see pointers to the main parts of the Video Narratives: Introduction, Video segment description, Video segment (part of a video to watch), Conclusion statement.

	meaning of the local state and the second state and	Long-term Repiratory Illnesses	
	This is a collection of video segments from YouTube-	videos veiles policies stories.	Narrative Introduction
	They have been automatically segmented, characterised	and linked in domain concepts.	
	The videos in this page have been finited via "Combined	ional Linking" which means:	
	You will explace videos which link concepts from	teo topico (Server analt).	
	Ti Ti		
	Link the topics PERSONAL VALUES AND BELIEFS and	PSYCHOLOGICAL BEALTH	
ch segment will start from a time point in the video a	al will stay when the convergencing concepts are oversel.		,
u can re-match the segment or match other yors of th	e videa, if you wish us.		Video segment descriptio
es will now work a video segment that mentions Hill	IEF (edated to PERSONAL VALUELAND DELIEFS) and HOPE (edated to PETCH)	LOGICAL HEALTH). The segment is iden from a YosTate vide	with D+ and USAR V-He , and warm at 0.12 with dominan of 36 serve
	Video segment		
ou watched video segments that li	nk PERSONAL VALUES AND BELIEFS and PSYCHO	DLOGICAL HEALTH	ive conclusion
evideo arguments none nationatically extracted from	ppular Yin/Tube videos, using algorithms for segmentation, theoreticiation and liaking	based on a theory for concept learning	
	white the second s		

In this section you will find the definition of the Derivative linking and 3 links to Derivative Video Narratives. Please, pay attention to the main areas of patient quality of life mentioned in the videos: Physical Health, Environment, Psychological Health, Level Of Independence, Personal Values and Beliefs, and Social Relationships.

Derivative linking: To become aware of a concept (Green oval) from a topic *T*, you will explore videos about its parent concept which is more generic (Orange oval).



	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

Video Narrative: Exploring Strength by linking it to its parent concept Energy *

<u>Video Narrative: Exploring Exercise by linking it to its parent concept</u> <u>Rehabilitation</u>

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

*

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

Video Narrative: Exploring Sleep by linking it to its parent concept Rest *

You watched 3 video narratives that were created using Derivative Linking (learning a concept by linking it to a more generic concept). The questions below ask how you perceive the usefulness of the these narratives.

	Extremely likely	Quite likely	Slightly likely	Neutral	Slightly unlikely	Quite unlikely	Extremely unlikely
Identify key points related to patient quality of life.	0	0	0	0	0	0	0
Focus on key points at a time	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	0
Identify key topics/concepts to be aware of	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Link main points mentioned in the videos	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

The video narratives with the short description are useful to *

What do you find **positive** about the generated Derivative Video Narratives with regard to raising awareness of quality of life needs of patients living with respiratory illnesses?

Your answer

What do you find **negative** about the generated Derivative Video Narratives with regard to raising awareness of quality of life needs of patients living with respiratory illnesses?

Your answer

Back

Next

Page 7 of 17

Clear form

The following questions gather further feedback on the usability of the Derivative Video Narratives you watched:

MENTAL DEMAND (1-low 20-high)

How mentally demanding was it to watch recommended Derivative Video Narratives and notice quality of life needs of patients living with respiratory illnesses? For example, how much mental and perceptual activity was required thinking, deciding, remembering, looking, searching?

Your score should be within the range of low(1) to high(20).

Your score on MENTAL DEMAND (1-low 20-high) *

Your answer

Please explain your score

EFFORT (1-low 20-high) How hard did you have to work (mentally and physically) to watch recommended Derivative Video Narratives and notice quality of life needs of patients living with respiratory illnesses?

Your score should be within the range of low(1) to high(20).

Your score on EFFORT (1-low 20-high) *

Your answer

Please explain your score

Your answer

FRUSTRATION (1-low 20-high)

How discouraged, irritated, stressed and annoyed did you feel while watching the recommended Derivative Video Narratives to notice quality of life needs of patients living with respiratory illnesses?

Your score should be within the range of low(1) to high(20).

Your score on FRUSTRATION (1-low 20-high) *

Your answer

Please explain your score

PERFORMANCE	(1-low 20-high)
-------------	-----------------

How successful do you think you were to identify new useful terms from watching the recommended Derivative Video Narratives to notice quality of life needs of patients living with respiratory illnesses?

Your score should be within the range of low(1) to high(20).

Your score on PERFORMANCE (1-low 20-high) *	
Your answer	
Please explain your score	
Your answer	
Back Next Page 8 of 17	Clear form

I In this section you will find the definition of the Super Ordinate linking, and watch 3 Super Ordinate Video Narratives. Please, pay attention to the main areas of patient quality of life mentioned in the videos: Physical Health, Environment, Psychological Health, Level Of Independence, Personal Values and Beliefs, and Social Relationships.

Super Ordinate linking: To learn a concept (Green oval) from a topic, we link to its children concepts (Orange ovals).



Video Narrative: Exploring Rehabilitation by linking it to its specific concept Exercise

*

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

Video Narrative: Exploring Rest by linking it to its specific concept Sleep *

Video Narrative: Exploring Energy by linking it to its specific concept Strength *

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

You watched 3 video narratives that were created using Super Ordinate Linking (learning a concept by linking it to a more specific concept). The questions below ask how you perceive the usefulness of the these narratives.

	Extremely likely	Quite likely	Slightly likely	Neutral	Slightly unlikely	Quite unlikely	Extremely unlikely
Identify key points related to patient quality of life.	0	0	0	0	0	0	0
Focus on key points at a time	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Identify key topics/concepts to be aware of	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Link main points mentioned in the videos	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

The Super Ordinate Video Narratives with the short description are useful to *

What do you find **positive** about the generated Super Ordinate Video Narratives with regard to raising awareness of quality of life needs of patients living with respiratory illnesses?

What d with reg respirat	o you find ne gard to raisin tory illnesses	e gative about ig awareness s?	the generate of quality of l	d Super Ordinate Video ife needs of patients liv	Narratives ving with
	SWAL				

The following questions gather further feedback on the usability of the Super Ordinate Video Narratives you watched:

MENTAL DEMAND (1-low 20-high)

How mentally demanding was it to watch recommended Super Ordinate Video Narratives and notice quality of life needs of patients living with respiratory illnesses? For example, how much mental and perceptual activity was required thinking, deciding, remembering, looking, searching?

Your score should be within the range of low(1) to high(20).

Your score on MENTAL DEMAND (1-low 20-high) *

Your answer

Please explain your score

EFFORT (1-low 20-high)

How hard did you have to work (mentally and physically) to watch recommended Super Ordinate Video Narratives and notice quality of life needs of patients living with respiratory illnesses?

Your score should be within the range of low(1) to high(20).

Your score on EFFORT (1-low 20-high) *

Your answer

Please explain your score

Your answer

FRUSTRATION (1-low 20-high)

How discouraged, irritated, stressed and annoyed did you feel while watching the recommended Super Ordinate Video Narratives to notice quality of life needs of patients living with respiratory illnesses?

Your score should be within the range of low(1) to high(20).

Your score on FRUSTRATION (1-low 20-high) *

Your answer

Please explain your score

PERFORMANCE	(1-low	20-high)
-------------	--------	----------

How successful do you think you were to identify new useful terms from watching the recommended Super Ordinate Video Narratives to notice quality of life needs of patients living with respiratory illnesses?

Your score should be within the range of low(1) to high(20).

Your score on PERFORMANCE (1-low 20-high) *	
Your answer		
Please explain your score		
Your answer		
Back Next	Page 11 of 17	Clear form

In this section you will find the definition of the Combinational linking, and watch 4 Combinational Video Narratives. Please, pay attention to the main areas of the patient quality of life mentioned in the videos: Physical Health, Environment, Psychological Health, Level Of Independence, Personal Values and Beliefs, and Social Relationships.

Combinational linking: Learn about two topics (Green ovals) by linking concepts from both topics mentioned in the same video segment.



	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

Video Narrative: Linking Personal Values and Beliefs to Psychological Health *

Video Narrative: Linking Level Of Independence to Psychological Health *

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

Video Narrative: Linking Social Relationship to Psychological Health *

Video Narrative: Linking Level Of Independence to Social Relationship *

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

You watched 4 video narratives that were created using Combinational Linking (linking concepts from two topics mentioned in the same video segment). The questions below ask how you perceive the usefulness of the these narratives.

The Combinational Video Narratives with the short description are useful to *

	Extremely likely	Quite likely	Slightly likely	Neutral	Slightly unlikely	Quite unlikely	Extremely unlikely
Identify key points related to patient quality of life.	0	0	0	0	0	0	0
Focus on key points at a time	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Identify key topics/concepts to be aware of	\bigcirc	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Link main points mentioned in the videos	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
What do you find positive about the generated Combinational Video Narratives with regard to raising awareness of quality of life needs of patients living with respiratory illnesses? Your answer							
What do you find negative about the generated Combinational Video Narratives with regard to raising awareness of quality of life needs of patients living with respiratory illnesses?							

Your answer

Back

Clear form

The following questions gather further feedback on the usability of the Combinational Video Narratives you watched:

MENTAL DEMAND (1-low 20-high)

How mentally demanding was it to watch recommended Combinational Video Narratives and notice quality of life needs of patients living with respiratory illnesses? For example, how much mental and perceptual activity was required thinking, deciding, remembering, looking, searching?

Your score should be within the range of low(1) to high(20).

Your score on MENTAL DEMAND (1-low 20-high) *

Your answer

Please explain your score

EFFORT (1-low 20-high) How hard did you have to work (mentally and physically) to watch recommended Combinational Video Narratives and notice quality of life needs of patients living with respiratory illnesses?

Your score should be within the range of low(1) to high(20).

Your score on EFFORT (1-low 20-high) *

Your answer

Please explain your score

Your answer

FRUSTRATION (1-low 20-high) How discouraged, irritated, stressed and annoyed did you feel while watching the recommended Combinational Video Narratives to notice quality of life needs of patients living with respiratory illnesses?

Your score should be within the range of low(1) to high(20).

Your score on FRUSTRATION (1-low 20-high) *

Your answer

Please explain your score

PERFORMANCE (1-low 20-high)
How successful do you think you were to identify new useful terms from watching
the recommended Combinational Video Narratives to notice quality of life needs of
patients living with respiratory illnesses?
Your score should be within the range of low(1) to high(20).

Your score on PERFORMANCE	(1-low 20-high) *	
Your answer		
Please explain your score		
Your answer		
Back Next	Page 14 of 17	Clear form

In this section you will find the definition of the Correlative linking, and watch 6 Correlative Video Narratives. Please, pay attention to the main areas of the Respiratory Illness mentioned in the videos: Physical Health, Environment, Psychological Health, Level Of Independence, Personal Values and Beliefs and Social Relationship and their concepts.

Correlative linking: To learn a concept (Green oval) from a topic, we link it to similar concepts (Orange ovals).



<u>Video Narrative: become aware of Caregiver by linking it to similar concepts from</u> * <u>the topic Social Relationship</u>

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

Video Narrative: become aware of the concept	<u>t Fear by linking it to similar</u>
concepts from the topic Psychological Health	

*

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

<u>Video Narrative: become aware of the concept Shortness Of Breath by linking it</u> * to similar concepts from the topic Physical Health

Video Narrative: become aware of the concept Exercise by linking it to similar * concepts from the topic Environment									
	Strongly Agree Neutral Disagree disagree								
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0				
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0				
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0				
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0				
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0				
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0				

<u>Video Narrative: become aware of the concept Meditation by linking it to similar</u> * <u>concepts from the topic Personal Values and Beliefs</u>

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

concepts from the topic Physical Health								
	Strongly agree	Agree	Neutral	Disagree	Strongly disagree			
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0			
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0			
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0			
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0			
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0			
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0			

Video Narrative: become aware of the concept Drug by linking it to similar

You watched 6 video narratives that were created using Correlative Linking (learning a concept by linking it to similar concepts). The questions below ask how you perceive the usefulness of the these narratives.

	Extremely likely	Quite likely	Slightly likely	Neutral	Slightly unlikely	Quite unlikely	Extremely unlikely
Identify key points related to patient quality of life.	0	0	0	0	0	0	0
Focus on key points at a time	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	0
Identify key topics/concepts to be aware of	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Link main points mentioned in the videos	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	0

The Correlative Video Narratives with the short description are useful to *

What do you find **positive** about the generated Correlative Video Narratives with regard to raising awareness of quality of life needs of patients living with respiratory illnesses?

~	-		-	5	~	1.4.	10	P
			-		-	v		
	~	-	-		~		~	

What do you find **negative** about the generated Correlative Video Narratives with regard to raising awareness of quality of life needs of patients living with respiratory illnesses?

Your answer

Back

Next

Page 16 of 17

Clear form

The following questions gather further feedback on the usability of the Correlative Video Narratives you watched:

MENTAL DEMAND (1-low 20-high)

How mentally demanding was it to watch recommended Correlative Video Narratives and notice quality of life needs of patients living with respiratory illnesses? For example, how much mental and perceptual activity was required thinking, deciding, remembering, looking, searching?

Your score should be within the range of low(1) to high(20).

Your score on MENTAL DEMAND (1-low 20-high) *

Your answer

Please explain your score.
EFFORT (1-low 20-high) How hard did you have to work (mentally and physically) to watch recommended Correlative Video Narratives and notice quality of life needs of patients living with respiratory illnesses?

Your score should be within the range of low(1) to high(20).

Your score on EFFORT (1-low 20-high) *

Your answer

Please explain your score.

Your answer

FRUSTRATION (1-low 20-high) How discouraged, irritated, stressed and annoyed did you feel while watching the recommended Correlative Video Narratives to notice quality of life needs of patients living with respiratory illnesses?

Your score should be within the range of low(1) to high(20).

Your score on FRUSTRATION (1-low 20-high) *

Your answer

Please explain your score.

PERFORMANCE (1-low 20-high)

How successful do you think you were to identify new useful terms from watching the recommended Correlative Video Narratives to notice quality of life needs of patients living with respiratory illnesses?

Your score should be within the range of low(1) to high(20).

Your score on PERFORMANCE (1-low 20-high) *

Your answer

Please explain your score.

Your answer

Thank you very much for completing the study.

Please press SUBMIT to complete the study.

Back

Submit

Page 17 of 17

Clear form

Second Form used with the PhD researchers from the University of Leeds

Health Related Quality of Life Needs of Patients With Chronic Illnesses (Respiratory Illness) - Expert Feedback

videonarrativesinadv	ance@gmail.com Switch accoun	ts	Ø
Next		Page 1 of 11	Clear form

Introduction

Thank you for helping with evaluating the video narratives that have been generated from Abrar Mohammed's PhD project conducted at the University of Leeds, UK.

Your evaluation is essential to assess the potential benefits of the generated video narratives

to help build awareness of health-related quality of life needs of patients living with chronic respiratory illnesses.

In this evaluation, you will **spend approximately 60 min** watching **video narratives with patient stories about living with a chronic respiratory Illness**. Video narratives are combinations of video segments which have been automatically characterised and linked together by computer algorithms. You will be given a scoring sheet for each video narrative and will be asked for your feedback on the usability of the narratives.

You will assess 12 video narratives (totalling 46 minutes video content) which are aimed at building awareness of health-related quality of life needs of patients living with chronic respiratory Illness. This will include: Physical Health, Environment, Psychological Health, Level of Independence, Personal Values and Beliefs, and Social Relationships.

You will have to log on with your **Google account**. When you fill out a Google Form in your Google account, **your progress is automatically saved**. This means if you can't complete the form or need to switch devices, your earlier responses will be saved, if you do not press SUBMIT. When you come back to the form, you will start from where you stopped last time. If you press SUBMIT and would like to come back to the form, you will start from the beginning, just be careful, please!

The study is **fully anonymous** - although you will logon with your Google account, the responses **do not save the email of the responders**.

Thank you very much for your help.

If you have any questions please contact:

- Abrar Mohammad <a.mohammed1@leeds.ac.uk>
- Vania Dimitrova <v.g.dimitrova@leeds.ac.uk>

Back	Next	Page 2 of 11	Clear

form

Your involvement in this study will include the following steps:

- 1. Read and accept the consent form.
- 2. Complete a short pre-study section about your profile.
- 3. Watch several suggested video narratives.
- 4. Give your feedback on the video narratives you have watched and their description.
- 5. Give a short video summary.

 Answer questions providing feedback on the usability and usefulness of the recommended video narratives for learning about Health Related Quality of Life-Respiratory Illness areas.

7. Repeat steps 3-6 with all four types video narratives.

CONSENT

I have read and understood the information about the evaluation of research on using video narratives for learning based on user interaction and video content, conducted at the School of Computing, University of Leeds, UK. On this basis I agree to participate in this online survey, and I consent to publication of the results of the evaluation with the understanding that confidentiality will be preserved. I also understand that I may at any time withdraw from the online survey, including withdrawal of any information I have provided. I note that the survey is fully complaint with General Data Protection Regulation (GDPR). Also, this survey has been reviewed and approved by the Research Ethics Committee of the Faculty of Engineering and Physical Sciences , University of Leeds, UK.

Agre	e		
Back	Next	 Page 3 of 11	Clear form

Your Pro	file						
Please d	lescribe l	briefly what	you do				
Your ans	wer						
What is a chroni	your awa c illness? Not at all	General Knowledge	Personal experience as patient	d quality of Some experience as carer	Some experience as health professional	patients liv Experience as researcher in patient quality of life	ing with Other
Answer							
Back	Nex	t			Page 4 of 11		Clear form

To help you understand the structure of the generated Video Narratives, in the Image below, you will see pointers to the main parts of the Video Narratives: Introduction, Video segment description, Video segment (part of a video to watch), Conclusion statement.

	Health-related Quality of Life Needs of Patients with Long-term Respiratory II	Inesses
	This is a solice tion of video segments from Yaullabe video, with patient stories,	Narrative Introduction
	They have been automatically segmented, characterised and linked is domain concepts.	
	The videos in this page have been finded via "Combinational Linking" which means:	
	You will explore videos which link one opts from two inplies (Green avails).	
	To and the set of the	
	Link the topics PERSONAL VALUES AND BELIEFS and PSYCHOLOGICAL BEALTH .	
ich argunent voll start from a time point in the video and inc can re-ranch the segment or watch other parts of the v	nell step when the conceptuating concepts are oversed.	Video segment descriptio
		video segment descripto
	Video segment	
iou watched xideo segments that lin	k PERSONAL VALUES AND BELIEFS and PSYCHOLOGICAL HEALTH .	Narrative conclusion
he video arguments were nationalically entracted from po- in hope you found the suffection of sideo segments help	plar YncYsle video, uwg signebens for organization, characterisation and listing based on a theory for concept humang M	

In this section you will find the definition of the Combinational linking, and watch 5 Combinational Video Narratives. Please, pay attention to the main areas of the patient quality of life mentioned in the videos: Physical Health, Environment, Psychological Health, Level Of Independence, Personal Values and Beliefs, and Social Relationships.

Combinational linking: Learn about two topics (Green ovals) by linking concepts from both topics mentioned in the same video segment.



Video Narrative:	Physical Heal	th to Environ	ment *		
	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

Video Narrative: Linking Social Relationship to Physical Health *

Strongly Strongly Agree Neutral Disagree agree disagree The introduction text in the video narrative \bigcirc 0 0 0 0 gives a clear idea of the aim of the narrative. All video segments clearly link to the health 0 0 \bigcirc related quality \bigcirc \bigcirc of life areas mentioned in the introduction. The descriptions which introduce each 0 0 0 \bigcirc \bigcirc video segment provide a useful summary. All video segments provide relevant 0 0 0 0 Ο content for health related quality of life areas. The concluding text at the end of the video 0 0 0 \bigcirc \bigcirc narrative provides appropriate summary. This video narrative is useful for raising awareness of 0 0 \bigcirc quality of life \bigcirc \bigcirc needs of patients living with chronic respiratory illnesses.

Video Narrative: Linking Physical Health to Psychological Health *

Video Narrative: Linking Physical Health to Level of Independence*

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

Video Narrative: Linking Level of Independence to Environment *

You watched 5 video narratives that were created using Combinational Linking (linking concepts from two topics mentioned in the same video segment). The questions below ask how you perceive the usefulness of the these narratives.

The Combinational Video Narratives with the short description are useful to *

	Extremely likely	Quite likely	Slightly likely	Neutral	Slightly unlikely	Quite unlikely	Extremely unlikely
Identify key points related to patient quality of life.	0	0	0	0	0	0	0
Focus on key points at a time	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Identify key topics/concepts to be aware of	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Link main points mentioned in the videos	\bigcirc	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

What do you find **positive** about the generated Combinational Video Narratives with regard to raising awareness of quality of life needs of patients living with respiratory illnesses?

Your answer

What do you find **negative** about the generated Combinational Video Narratives with regard to raising awareness of quality of life needs of patients living with respiratory illnesses?

Your answer

Back

Next

Clear form

The following questions gather further feedback on the usability of the Combinational Video Narratives you watched:

MENTAL DEMAND (1-low 20-high)

How mentally demanding was it to watch recommended Combinational Video Narratives and notice quality of life needs of patients living with respiratory illnesses? For example, how much mental and perceptual activity was required thinking, deciding, remembering, looking, searching?

Your score should be within the range of low(1) to high(20).

Your score on MENTAL DEMAND (1-low 20-high) *

Your answer

Please explain your score

EFFORT (1-low 20-high) How hard did you have to work (mentally and physically) to watch recommended Combinational Video Narratives and notice quality of life needs of patients living with respiratory illnesses?

Your score should be within the range of low(1) to high(20).

Your score on EFFORT (1-low 20-high) *

Your answer

Please explain your score

Your answer

FRUSTRATION (1-low 20-high)

How discouraged, irritated, stressed and annoyed did you feel while watching the recommended Combinational Video Narratives to notice quality of life needs of patients living with respiratory illnesses?

Your score should be within the range of low(1) to high(20).

Your score on FRUSTRATION (1-low 20-high) *

Your answer

Please explain your score

How successful do you think you were to identify new useful terms from watching the recommended Combinational Video Narratives to notice quality of life needs of patients living with respiratory illnesses?

Your score should be within the range of low(1) to high(20).

Your score on PERFORMANCE ((1-low 20-high) *
-----------------------------	-------------------

Your answer

Please explain your score

Back	Next	Page 8 of 11	Clear form
Dack	Next	Page 8 01 11	Clear form

In this section you will find the definition of the Correlative linking, and watch 7 Correlative Video Narratives. Please, pay attention to the main areas of the Respiratory Illness mentioned in the videos: Physical Health, Environment, Psychological Health, Level Of Independence, Personal Values and Beliefs and Social Relationship and their concepts.

Correlative linking: To learn a concept (Green oval) from a topic, we link it to similar concepts (Orange ovals).



the topic Psychological Health						
	Strongly agree	Agree	Neutral	Disagree	Strongly disagree	
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0	
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0	
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0	
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0	
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0	
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0	

Video Narrative: become aware of Thinking, by linking it to similar concepts from *

<u>Video Narrative: become aware of the concept Treatment by linking it to similar</u> * <u>concepts from the topic Level of Independence</u>

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

<u>Video Narrative: become aware of the concept Lung Cancer by linking it to similar</u> * <u>concepts from the topic Physical Health</u>

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

Video Narrative: become aware of the concept Energy by linki	<u>ng it to similar</u>
concepts from the topic Physical Health	

*

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0

<u>Video Narrative: become aware of the concept Insurance by linking it to similar</u> <u>concepts from the topic Level of Independence</u>

concepts from the topic Physical Health						
	Strongly agree	Agree	Neutral	Disagree	Strongly disagree	
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0	
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0	
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0	
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0	
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0	
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0	

Video Narrative: become aware of the concept Pneumonia by linking it to similar *

concepts from the topic Physical Health						
	Strongly agree	Agree	Neutral	Disagree	Strongly disagree	
The introduction text in the video narrative gives a clear idea of the aim of the narrative.	0	0	0	0	0	
All video segments clearly link to the health related quality of life areas mentioned in the introduction.	0	0	0	0	0	
The descriptions which introduce each video segment provide a useful summary.	0	0	0	0	0	
All video segments provide relevant content for health related quality of life areas.	0	0	0	0	0	
The concluding text at the end of the video narrative provides appropriate summary.	0	0	0	0	0	
This video narrative is useful for raising awareness of quality of life needs of patients living with chronic respiratory illnesses.	0	0	0	0	0	

Video Narrative: become aware of the concept Cough by linking it to similar

*

You watched 7 video narratives that were created using Correlative Linking (learning a concept by linking it to similar concepts). The questions below ask how you perceive the usefulness of the these narratives.

Extremely Quite Slightly Slightly Quite Extremely Neutral likely unlikely unlikely likely likely unlikely Identify key points related to patient quality of life. Focus on key points at a time Identify key topics/concepts to be aware of Link main points mentioned in the videos

The Correlative Video Narratives with the short description are useful to *

What do you find **positive** about the generated Correlative Video Narratives with regard to raising awareness of quality of life needs of patients living with respiratory illnesses?

Your answer

What do you find **negative** about the generated Correlative Video Narratives with regard to raising awareness of quality of life needs of patients living with respiratory illnesses?

Your answer

Back

Next

Page 10 of 11

Clear form

The following questions gather further feedback on the usability of the Correlative Video Narratives you watched:

MENTAL DEMAND (1-low 20-high)

How mentally demanding was it to watch recommended Correlative Video Narratives and notice quality of life needs of patients living with respiratory illnesses? For example, how much mental and perceptual activity was required thinking, deciding, remembering, looking, searching?

Your score should be within the range of low(1) to high(20).

Your score on MENTAL DEMAND (1-low 20-high) *

Your answer

Please explain your score.

EFFORT (1-low 20-high) How hard did you have to work (mentally and physically) to watch recommended Correlative Video Narratives and notice quality of life needs of patients living with respiratory illnesses?

Your score should be within the range of low(1) to high(20).

Your score on EFFORT (1-low 20-high) *

Your answer

Please explain your score.

Your answer

FRUSTRATION (1-low 20-high) How discouraged, irritated, stressed and annoyed did you feel while watching the recommended Correlative Video Narratives to notice quality of life needs of patients living with respiratory illnesses?

Your score should be within the range of low(1) to high(20).

Your score on FRUSTRATION (1-low 20-high) *

Your answer

Please explain your score.

PERFORMANCE (1-low 20-high)

How successful do you think you were to identify new useful terms from watching the recommended Correlative Video Narratives to notice quality of life needs of patients living with respiratory illnesses?

Your score should be within the range of low(1) to high(20).

Your score on	PERFORMANCE	(1-low	20-high) *
1001 30010 011		(11011	20 mgm

Your answer

Please explain your score.

Your answer

Thank you very much for completing the study.

Please press SUBMIT to complete the study.

Back	Submit	Page	11 of 11 Clear form
Васк	Submit	Page	Clear form

Appendix O

Evaluation Results of VIN-L (Second Domain)

These are the evaluation results, from experts, of VIN-L in the second domain which includes the: Quality and the Perceived Usefulness results.

Quality of the video narratives as evaluated by the experts.

Narrative type	Response type	Q1	Q2	Q3	Q4	Q5	Usefulness
СО	Strongly Agree	10	12	6	13	6	18
	Agree	11	6	10	6	13	3
	Neutral	3	7	8	2	5	4
	Disagree	3	2	3	6	3	1
	Strongly Disagree	0	0	0	0	0	1
CR	Strongly Agree	19	16	23	22	15	24
	Agree	13	14	9	11	17	14
	Neutral	5	5	7	4	5	1
	Disagree	2	4	0	2	2	0
	Strongly Disagree	0	0	0	0	0	0
D	Strongly Agree	3	3	0	5	0	7
	Agree	3	4	5	3	7	2
	Neutral	2	1	3	1	1	0
	Disagree	1	1	1	0	1	0
	Strongly Disagree	0	0	0	0	0	0
S	Strongly Agree	0	3	0	4	1	6
	Agree	6	6	7	5	6	3
	Neutral	2	0	1	0	2	0
	Disagree	1	0	1	0	0	0
	Strongly Disagree	0	0	0	0	0	0

The numbers are the aggregation of the responses from the experts where each expert had evaluated more than one video narrative:

Perceived usefulness of the video narratives as evaluated by experts.

The numbers are the aggregation of the responses from the experts where each expert had evaluated more than one video narrative:

Narrative	Response type				
Туре		PU1	PU2	PU3	PU4
	Extremely Likely	2	1	2	3
со	Quiet Likely	2	3	2	1
	Slightly Likely	1	2	2	1
	Neutral	1	0	0	1
	Slightly Unlikely	0	0	0	0
	Quiet Unlikely	0	0	0	0
	Extremely Unlikely	0	0	0	0
	Extremely Likely	3	0	3	2
CR	Quiet Likely	2	5	3	3
	Slightly Likely	1	1	0	0
	Neutral	0	0	0	0
	Slightly Unlikely	0	0	0	1
	Quiet Unlikely	0	0	0	0
	Extremely Unlikely	0	0	0	0
	Extremely Likely	2	0	2	0
D	Quiet Likely	1	2	1	2
	Slightly Likely	0	0	0	1
	Neutral	0	1	0	0
	Slightly Unlikely	0	0	0	0
	Quiet Unlikely	0	0	0	0
	Extremely Unlikely	0	0	0	0
	Extremely Likely	2	0	2	0
S	Quiet Likely	1	3	1	2
	Slightly Likely	0	0	0	0
	Neutral	0	0	0	1
	Slightly Unlikely	0	0	0	0
	Quiet Unlikely	0	0	0	0
	Extremely Unlikely	0	0	0	0